Nutrition Assessment in Medical and Surgical Oncology

by

Lisa Martin

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

in

Nutrition and Metabolism

Department of Agricultural, Food and Nutritional Science University of Alberta

© Lisa Martin, 2018

ABTRACT

Introduction:

Cancer-associated cachexia is defined as a multifactorial syndrome characterized by loss of body weight with specific losses of skeletal muscle and adipose tissue. Cachexia is driven by a variable combination of reduced food intake (FI) and distinct tumor-associated metabolic changes, including elevated energy expenditure, excess catabolism, and inflammation. Patients with cachexia experience weakness, fatigue, loss of independence, poor treatment tolerance, and death. Cachexia may be present at the time of cancer diagnosis or develop and worsen with cancer treatment and disease progression. There are no widely agreed upon diagnostic criteria, which have hindered diagnosis, clinical management and development of effective treatments. An International Consensus Framework for the Definition and Classification of Cancer Cachexia made recommendations for development of diagnostic criteria, suggesting 'definitive cutoffs for variables (e.g. weight loss, low muscle mass) could be determined from large contemporary datasets by determining the values that relate optimally to meaningful patient-centered outcomes, such as loss of function or decreased survival' The objective of my thesis was to advance development of diagnostic criteria for cancer cachexia in at risk patient populations from two cancer treatment settings, medical and surgical oncology.

Methods:

Large contemporary data sets were lacking to study patients with cancer cachexia. International research partners contributed data from clinical cachexia research studies to create aggregated data sets for secondary data analysis. Data included candidate diagnostic criteria (e.g. body mass index (BMI), weight loss (WL), computed tomography (CT)-defined body composition, FI, and C-reactive protein (CRP, as a marker of inflammation), covariates (e.g. cancer type, stage, surgical approach, age, sex, performance status), and patient-centered

ii

outcomes (e.g. overall survival (OS), length of hospital stay (LOS)). In medical oncology, the prognostic impact of candidate diagnostic criteria (WL, BMI, FI, and CRP) on OS was evaluated with Kaplan Meier and multivariable Cox proportional hazard models; FI and CRP were evaluated as etiological criteria for WL with multivariable multinomial logistic regression (MLR) models. In the surgical setting, the prognostic impact of candidate diagnostic criteria (WL, FI, and CT-defined body composition) on LOS, postoperative complications, and 30-day hospital readmission were evaluated with multivariable logistic (LR) and negative binomial (NMR) regression models.

Results:

Large aggregated data sets were created to study patients from medical (N=18,173 patients) and surgical (N=5,739) oncology settings. Three studies were conducted with data from medical oncology. Study 1 demonstrated that increased %WL and decreased BMI independently predicted OS. A grading system combining %WL and BMI was developed based on OS (Grade 0 (longest OS) to 4 (shortest OS)), and was subsequently validated. In studies 2 & 3, FI and CRP were evaluated as etiological criteria for WL. Common values, based on relative risk of weight loss, were determined across the 3 most frequently used clinical measurement scales for FI, which were combined (normal, moderately or severely reduced) and evaluated. Reduced FI significantly associated with increasingly severe WL. The relationship between CRP and WL was also evaluated; CRP values (<10, 10-43, and \geq 43 mg/L) associated with distinct degrees of WL were defined. In multivariable MLR both CRP and FI significantly associated with increasingly severe WL, and the combination of WL, FI, and CRP identified patients with significantly different OS.

Two studies were conducted with data from surgical oncology. In the first study, nutrition risk (based on WL and reduced FI) was identified as an independent predictor of low compliance to a standardized multi-modal care pathway, which had a significant negative impact on postoperative complications and LOS. In the second study, age- and sex-specific preoperative body composition profiles were defined based on combinations of CT-defined low skeletal muscle, low skeletal muscle radiation attenuation, and high visceral fat. Multidimensional body composition profiles were at significant risk of longer LOS and increased 30-day hospital readmission.

Conclusions:

Candidate diagnostic available for aggregation were heterogeneous, and represented with variable frequency. Alterations in metabolism were only represented by a single criterion, CRP. Identifying common values between different measurement scales for FI facilitated data aggregation. Evaluation of aggregated data identified WL, BMI, FI, CRP, and CT-defined body composition as candidate diagnostic criteria for cancer cachexia that identified patients at risk for poor outcomes in different care settings. These criteria are a key first step toward development of definitive diagnostic criteria, and require prospective validation.

PREFACE

This doctoral thesis is original work by Lisa Martin. The research presented is based on data collected from human subjects. A portion of the data presented in Chapters 2, 5, 6, 7 and 8 were acquired through national and international collaborations with research partners, who agreed to share their data for secondary analysis as part of an international data aggregation effort. Research partners are listed in the table and all data were collected under the auspices of human ethics approvals from their respective institutions:

Principle Investigator	Institution
Pierre Senesse	Institut Régional du Cancer de Montpellier (ICM), France
Maurizio Muscaritoli	Sapienza University of Rome, Italy
Barry Laird	University of Edinburgh, Edinburgh, UK; European Palliative Care Research Centre
Bruno Gagnon	Université Laval, Quebec, Canada
Martin Chasen	University of Ottawa, Canada
Chris Deans	University of Edinburgh, Royal Infirmary, Edinburgh, United Kingdom
Ioannis Gioulbasanis	Larissa General Clinic "E Patsidis", Larissa, Thessaly, Greece
Ola Wallengren	Sahlgrenska University Hospital, Gothenburg, Sweden
Ingvar Bosaeus	Sahlgrenska University Hospital, Gothenburg, Sweden
Kent Lundholm	Gothenburg University, Gothenburg, Sweden
Anne Voss	Abbott Nutrition, Columbus, Ohio, USA
Francois Goldwasser	Paris Descartes University, Paris, France
R. Thomas Jagoe	Jewish General Hospital, Montreal, Quebec, Canada
Federico Bozzetti	University of Milan, Italy
Florian Strasser	Cantonal Hospital, St. Gallen, Switzerland
Lene Thoresen	St. Olavs University Hospital, Trondheim, Norway
Sami Antoun	Institut Gustave Roussy, Villejuif, France
Kenneth H. Fearon*	University of Edinburgh, Royal Infirmary, Edinburgh, United Kingdom
Ian Jenkins	Department of Surgery, St. Mark's Hospital, Harrow, London, United Kingdom;
	Department of Surgery and Cancer, Imperial College, London, United Kingdom
Georgios Malietzis	Department of Surgery, St. Mark's Hospital, Harrow, London, United Kingdom;
	Department of Surgery and Cancer, Imperial College, London, United Kingdom

*deceased September 3, 2016

Data collected from human subjects in Alberta, Canada, received research ethics approval from the University of Alberta Health Research Ethics Biomedical Panel and the University of Calgary Health Research Ethics Board. Data included in Chapters 2, 5, 6, 7 and 8 were collected with the following ethical approvals: HREBA.CC-16-0308 "Pro00055767 Chart review of nutritional status in colorectal cancer patients undergoing surgery; HREBA.CC-17-0433 "23198 Chart review of nutritional status and involuntary weight loss in advanced cancer patients"; Pro00046864 "Enhancing Patients' Recovery After Surgery: Strategy to Transform Care and Maximize Value".

The contributions made by the PhD candidate, Lisa Martin, and the co-authors to the completion of work included in this thesis are described here.

Chapter 2 of this thesis has been published as Martin L, Senesse P, Gioulbasanis I, Antoun A, Bozzetti F, Deans C, Strasser F, Thoresen L, Jagoe RT, Chasen M, Lundholm K, Bosaeus I, Fearon K, Baracos VE. Diagnostic criteria for cancer-associated weight loss. J Clin Oncol, 2015; 33:90-99. *Conception and design*: Lisa Martin, Kenneth H. Fearon, Vickie E. Baracos, *Provision of study materials or patients*: Pierre Senesse, Ioannis Gioulbasanis, Sami Antoun, Federico Bozzetti, Chris Deans, Florian Strasser, Lene Thoresen, R. Thomas Jagoe, Martin Chasen, Kent Lundholm, Ingvar Bosaeus, Kenneth H. Fearon, *Collection and assembly of data*: Lisa Martin, Pierre Senesse, Ioannis Gioulbasanis, Sami Antoun, Federico Bozzetti, Chris Deans, Florian Strasser, Lene Thoresen, R. Thomas Jagoe, Martin Chasen, Kent Lundholm, Ingvar Bosaeus, Kenneth H. Fearon, Vickie E. Baracos *Data analysis and interpretation*: Lisa Martin, Pierre Senesse, Vickie E. Baracos *Manuscript writing*: Lisa Martin and Vickie E. Baracos with contribution from other authors *Revision and final approval of manuscript*: All authors.

Chapter 3 of this thesis has been published as Martin L. Diagnostic criteria for cancer cachexia: Data versus dogma. Curr Opin Nutr Metab Care, 2016; 19:188-198. This article was an invited review, and was completed in its entirety (conception, design, collection and assembly of data, interpretation, manuscript writing and approval) by Lisa Martin

Chapter 4 of this thesis has been published online ahead of print as Martin L, Kubrak C. How much does reduced food intake contribute to cancer-associated weight loss? Curr Opin

vi

Support Palliat, 2018, Aug. 16; doi:10.1097/SPC.000000000000379. Lisa Martin was the invited author for this review, and responsible for the content as lead author. *Conception and design*: Lisa Martin, Catherine Kubrak, *Collection and assembly of data*: Lisa Martin, Catherine Kubrak, *Data analysis and interpretation*: Lisa Martin, Catherine Kubrak, *Manuscript writing*: Lisa Martin, Catherine Kubrak, *Revision and final approval of manuscript*: Lisa Martin, Catherine Kubrak.

Chapter 7 of this thesis has been published online ahead of print as Martin L, Gillis C, Gillam M, Sheppard C, Buhler S, Basualdo-Hammond C, Nelson G, Gramich L. Implementation of an Enhanced Recovery After Surgery Program can change nutrition care practice: A multicenter experience in elective colorectal surgery. JPEN, J Parenter Enter Nutr, 2018, Jul 23; doi.org/10.1002/jpen.1417. *Conception*: Lisa Martin, Leah Gramlich, Gregg Nelson Design: Lisa Martin, Leah Gramlich, Chelsia Gillis, Carlotta-Basualdo-Hammond, Gregg Nelson, *Provision of study materials or patients:* Leah Gramlich, Gregg Nelson, *Collection and assembly of data*: Lisa Martin, Caroline Sheppard, Carlotta Basualdo-Hammond, Gregg Nelson, Leah Gramlich, *Data analysis:* Lisa Martin, *Interpretation of results*: Lisa Martin, Chelsia Gillis, Melani Gillam, Marlis Atkins, Carlotta Basualdo-Hammond, Sue Bhuler, Gregg Nelson, Leah Gramlich *Manuscript writing:* Lisa Martin, Leah Gramlich, *Revision and final approval of manuscript:* All authors.

Chapter 8 of this thesis has been published as Martin L, Hopkins J, Malietzis G, Jenkins JT, Sawyer MB, Brisebois R, MacLean A, Nelson G, Gramlich L, Baracos VE. Assessment of CT-defined muscle and adipose tissue features in relation to short term outcomes after elective surgery for colorectal cancer: a multicenter approach. Ann Surg Oncol, 2018; 25:2669-2680. *Conception and design:* Lisa Martin, Vickie E. Baracos, *Provision of study materials or patients:*

Leah Gramlich, Gregg Nelson, Jessica Hopkins, Georgios Malietzis, Michael Sawyer, Ian Jenkins, Ronald Brisebois, Anthony MacLean, *Collection and assembly of data*: Leah Gramlich, Gregg Nelson, Jessica Hopkins, Georgios Malietzis, Michael Sawyer, Ian Jenkins, Ronald Brisebois, Anthony MacLean, Vickie E Baracos, *Data analysis:* Lisa Martin, *Interpretation of results*: Lisa Martin, Georgios Malietzis, Ian Jenkins, Vickie E. Baracos, Leah Gramlich, Gregg Nelson *Manuscript writing:* Lisa Martin, Vickie E. Baracos, *Revision and final approval of manuscript:* All authors. I would like to dedicate my thesis to Emlyn and Henry.

You are my greatest adventure.

"When in doubt, choose the kids. There will be plenty of time later to choose work."

- Anna Quindlen

ACKNOWLEDGMENTS

I would like to acknowledge all of the financial support I have received throughout my graduate studies. I am fortunate to have received funding from the Department of Agricultural, Food and Nutritional Science, the University of Alberta, Alberta Innovates Health Solutions, the Killam Trusts, and the Rhoads Research Foundation of the American Society of Parenteral and Enteral Nutrition. I am grateful to have had continuous financial support which made it possible to pursue my academic interests.

I would like to thank my supervisor, Dr. Vickie Baracos. We have known each other for many years, and you have been nothing but gracious with your time, patience, and support. I have been the recipient of your generosity and many valuable life lessons about, science, horses, and how keep calm at all times. You have taught me the importance of collaboration, and your enthusiasm for science has inspired me and many others to keep moving forward. "*The secret of genius is to carry the spirit of the child into old age, which means never losing your enthusiasm*."-Aldous Huxley. Thank you for everything.

I would like thank the members of my supervisory committee. Dr. Leah Gramlich, thank you for always providing me with opportunities and believing in my ability as a researcher. You have been kind, inclusive, and have provided excellent advice over the years. Everyone needs someone in their corner, thanks for being in mine. Dr. Gregg Nelson, thank you for sharing your ideas and experience, and for your kind and positive mentorship.

I would like to thank all of my collaborators: Pierre Senesse, Barry Laird, Bruno Gagnon, Martin Chasen, Chris Deans, Maurizio Muscaritoli, , Ioannis Gioulbasanis, Ola Wallengren, Anne C. Voss, Francois Goldwasser, Catherine Kubrak, R. Thomas Jagoe, Federico Bozzetti, Florian Strasser, Lene Thoresen, Sami Antoun, Seyyed Mohammad Reza Kazemi-Bajestani, Jessica Hopkins, Michael Sawyer, Georgios Malietzis, Ian Jenkins, and Ken Fearon. Thank you for sharing your research data, and for your willingness to entrust it to me. It has been a truly wonderful experience, and my work would not be possible without your contributions.

I would like to thank Dr. Vera Mazurak, and Dr. Marina Mourtzakis. I am grateful for your friendship and wonderful advice over the years.

I have had the great pleasure to work alongside many talented individuals. We have travelled far together and I hope we will continue our discussions and collaborations for many years to come. A special thank you to Mohammad, Cathy, Ana, Amrit, Jingjie, Ashok, Cynthia, David, and Lorena, your friendship and support has been tremendous.

Thank you to the people I love most in the world. To Andy, thank you for your unconditional love and support, and for always listening and encouraging me. You are the love of my life and the best father to our children. To my parents, Yvonne & Vern Whaley and Neil Sutcliffe & Peggy Pierce, thank you for your support of our family and for always saying yes to your grandkids. To Emlyn and Henry thank you for being the best distractions in the world. My accomplishments and successes are a reflection of all of you.

TABLE OF CONTENTS

CHAPTER 1: Intr	oduction and research plan	
1.1 Introduc	ction	1
1.2 Researcl	h approach	7
1.3 Research	h plan	8
1.4 Chanter	Format	
1.5 Reference	res	13
CHAPTER 2: Diag	nostic criteria for the classification of cancer-associated weight loss	
2.1 Introduc	ction	16
2.2 Methods	5	17
2.2.1	Data collection and Handling	17
2.2.2	Statistics	19
2.3 Results		20
231	Training Sample	20
2.5.1	RMI adjusted WI grading system	20
2.5.2	Subaroun analysis	·····21 24
2.3.3	Subgroup unalyses	24 24
2.3.4		
2.4 Discussion	on	
2.5 Kelerend	res	
CHADTED 2 D'	····	
CHAPTER 5: Diag	nostic criteria for cancer cachexia: data versus dogma	
3.1 Key Poir	nts	43
3.2 Cachexia	a has been known since ancient times but lacks agreed upon diag	gnostic
	······	
3.3 Progress	toward consensus diagnostic criteria for cancer cachexia	
3.4 Methods	and principles for the development of diagnostic criteria	
3.5 Re-defin	ing weight loss as a diagnostic criterion for cancer cachexia	47
3.6 Skeletal	muscle is an emerging diagnostic criterion for cancer cachexia	48
3.7 Reduced	l food intake is a candidate diagnostic criterion in development	52
3.8 Biologic	Aberrations (i.e. inflammation) in Cachexia - How do we factor	
them in to	cachexia classifications?	53
3.9 Conclusi	ion	54
3.10 Referen	nces	56
CHAPTER 4: How	much does reduced food intake contribute to cancer-associated	weight
loss?		
4.1 Introduc	ction	66
4.2 Identific	ation of Reduced Food Intake	68
4.3 Food Int	take and Cancer-Associated Weight Loss	70
4.3.1	Comparing food intake between patients with and without cancer-	
assoc	ciated weight loss	70
4.3.2	<i>The association between food intake and cancer-associated weight</i>	
loss.	,	71
433	Nutrition impact symptoms and Cancer-associated weight loss	
,.5.5		

4.4 Discussion	73
4.5 Conclusion	
4.6 References	77

CHAPTER 5: Reduced food intake as an explanatory variable for cancer-associated weight loss

5.1 Introduction	88
5.2 Methods	89
5.2.1 Study population	89
5.2.2 Assessments of food intake	90
5.2.3 Statistics	91
5.3 Results	92
5.3.1 Patient characteristics	92
5.3.2 The association between food intake and cancer-associated weight	
Loss	92
5.4 Discussion	95
5.5 References	98

CHAPTER 6: The variable contribution of reduced food intake and systemic inflammation to cancer-associated weight loss

6.1 Introduction	110
6.2 Methods	
6.2.1 Study population	
6.2.2 Assessments of food intake and C-reactive protein	
6.2.3 Statistics	
6.3 Results	116
6.3.1 Patient characteristics	115
6.3.2 CRP and cancer-associated weight loss	
6.3.3 Risk stratification for overall survival	
6.4 Discussion	118
6.5 References	

CHAPTER 7: Implementation of an Enhanced Recovery After Surgery program car abange putrition governmentations. A multi-context surgery program car	1
7.1 Clinical Relevancy Statement	ry 132
7.2 Introduction	132
7.3 Methods	134
7.3.1 ERAS Implementation Program	135
7.3.2 Study Design	135
7.3.3 Statistical Analysis	137
7.4 Results	138
7.4.1 The influence of ERAS implementation on the adoption of perioperation	ive
nutrition care practices	138
7.4.1.1 Patient characteristics	138
7.4.1.2 Changes to perioperative nutrition care practices	138
7.4.2 Impact of nutrition risk on compliance to AHS EIP and short-term	
postoperative outcomes	139

7.4.2.1 Patient characteristics	139
7.4.2.2 Nutrition risk and compliance to AHS EIP	140
7.4.2.3 Nutrition risk and short-term postoperative outcomes	140
7.5 Discussion	141
7.6 References	146

CHAPTER 8: Assessment of computed tomography (CT)-defined muscle and adipose tissue features in relation to short-term outcomes after elective surgery for colorectal cancer: a multicenter approach

8.1 Introduction	161
8.2 Methods	162
8.2.1 Study design & participants	162
8.2.2 Clinical data	163
8.2.3 Computed tomography	163
8.2.4 Statistics	164
8.3 Results	165
8.3.1 Patient characteristics	165
8.3.2 Individual body composition features associate with LOS	166
8.3.3 Multidimensional body composition profiles associate with LOS	168
8.3.4 Post-surgical complications	168
8.4 Discussion	169
8.5 References	172

CHAPTER 9: Discussion and future directions

9.1 The status of diagnostic criteria	185
9.1.1 Diagnostic criteria for cancer-associated weight loss	
9.1.2 Diagnostic criteria for CT-defined skeletal muscle	
9.1.3 Diagnostic criteria for food intake	
9.1.4 Diagnostic criteria for systemic inflammation	189
9.2 Strengths and limitations of a data driven approach	191
9.3 Future development and implementation	195
9.3.1 Future development: acquisition of data relevant to cancer cach	exia195
9.3.2 Future development: how can diagnostic criteria be used to adv	ance
nutrition care?	196
9.4 Conclusion	199
9.5 References	
BIBLIOGRAPHY	

APPENDIX

Appendix V: Chapter 6 Supplementary Table 1. Univariable multinomial logistic regression (MLR) analysis for the association of selected variables to cancer-associated weight loss....250

Appendix IX: Chapter 8. Supplementary Table 3. Factors predictive of major complications (Clavien Dindo Grade \geq 3) in colorectal cancer patients undergoing elective surgery.......259

LIST OF TABLES

Table 2-1 : Descriptions of data contributed to the training and validation samples
Table 2-2 : Characteristics of training and validation samples
Table 2-3 : Median survival and hazard ratios with 95% confidence intervals for proportional hazard model assessing the effect of variables associated with survival
Table 2-4 : Sub-group analysis: median survival by grade, and overall p-values from coxproportional hazards models assessing the prognostic significant of the BMI adjusted WLgrading system
Table 3-1 : A summary of narrative definitions for cachexia and diagnostic criteria61
Table 3-2 : A summary of candidate diagnostic criteria and the heterogeneity that exists within each criterion
Table 3-3 : A Summary of results from 19 articles (n=5,124 patients) published between Jan.2015 to Nov. 2015 evaluating skeletal muscle using CT L3 image analysis
Table 4-1: Assessments used to measure food intake in the oncology setting
Table 4-2 : A summary of studies that have examined the contribution or food intake to cancer- associated weight loss
Table 5-1 : A description of study cohorts that were selected from the International Cancer Cachexia Database (ICCDR) for this study
Table 5-2 : Patient characteristics for the three study cohorts102
Table 5-3 : Mean percent (%) weight loss by category of food intake for each measurement tool
Table 5-4 : Data from three multivariable multinomial logistic regression (MLR) models highlighting the association between assessments of food intake and cancer-associated weight loss (WL)
Table 5-5 : The crosstabulation for categories of weight loss (WL) by categories of food intake(FI) for each of the food intake assessment tools: PG-SGA, <i>I</i> -VVAS, MNA108
Table 6-1: Patient characteristics 125

Table 6-2 : A multivariable multinomial logistic regression model to evaluate the association of CRP and food intake to weight loss
Table 6-3 : Median overall survival (OS) for patients stratified by categories of weight loss(WL), food intake (FI), and CRP categories
Table 7-1 : Enhanced recovery after surgery (ERAS) care elements for people undergoing major colorectal surgeries in Alberta, Canada
Table 7-2 : Criteria for compliance to ERAS nutrition care elements
Table 7-3 : Characteristics of patients who underwent elective colorectal surgery within an ERAS implementation program
Table 7-4: The impact of ERAS implementation and nutrition risk on nutrition care elements
Table 7-5 : Logistic regression model of the factors associated with low compliance (<70%) toERAS care elements for patients in the ERAS implementation group (N=2628)159
Table 8-1: Characteristics for the pooled colorectal surgical cohort
Table 8-2 : Pre-surgical L3 CT-defined body composition features by sex and age for the pooled colorectal cancer (CRC) cohort (N=2100)
Table 8-3 : Regression models for variables associated with extended length of hospital stay (LOS) and 30-hospital readmission in patients with colorectal cancer undergoing elective surgery. 181
Table 8-4 : Estimated marginal mean LOS derived from the final adjusted NBM for each body composition profile demonstrating the independent effect of the three body composition profiles on LOS with or without major post-operative complications
Table 9-1: A summary of proposed diagnostic criteria for cancer cachexia, cachexia, and malnutrition

LIST OF FIGURES

Figure 1-1: Classification of cancer cachexia stage
Figure 2-1 : Line graphs representing the relationships between deciles of BMI (1A) and percent weight loss (%WL, 1B) to overall survival
Figure 2-2: Risk of reduced survival is a function of body mass index and percent weight loss (%WL)
Figure 2-3: Cumulative survival curves from the subgroup analysis of the training sample for (A) garteoesophageal and (B) head and neck cancers
Figure 3-1: The relationship between percent weight loss and overall survival47
Figure 3-2 : Distributions for skeletal muscle index (cm ² /m ²), evaluated by CT image analysis, for Caucasian and Japanese cancer patients
Figure 4-1: A conceptual diagram representing the pathophysiology of cancer cachexia66
Figure 4-2: A diagram representing the energy balance framework
Figure 4-3 : Heterogeneous body weight responses to a nutritional intervention
Figure 5-1: The relationship between percent (%) weight loss (WL) and three food intake measurement scales
Figure 6-1: A conceptual diagram representing the pathophysiology of cancer cachexia111
Figure 6-2: Mean %WL by decile of serum CRP (mg/L) values116
Figure 8-1 : Description of body composition features related to length of hospital stay167

LIST OF ABBREVIATIONS

ADL	Activities of daily living
AHS	Alberta Health Services
AJCC	American Joint Committee on Cancer
ASA	American Society of Anesthesiologists
BMI	Body mass index
CI	Confidence interval
CNST	Canadian nutrition risk screening tool
CRC	Colorectal cancer
CRP	C-reactive protein
СТ	Computed tomography
CTCAE	Common Terminology Criteria for Adverse Events
ECOG PS	Easter Cooperative Oncology Group performance status
EORTC	European Organization for Research and Treatment of Cancer
ERAS	Enhanced Recovery After Surgery
EIP	ERAS Implementation Program
EIAS	ERAS Interactive Audit System
FI	Food intake
GI	Gastrointestinal
HR	Hazard ratio
HU	Hounsfield Unit
ICD-10	International Statistical Classification of Diseases and Related Health Problems
	10 th revision
<i>I</i> -VVAS	Ingesta- verbal/visual analogue scale
KPS	Karnofsky Performance Scale
L3	Third lumbar vertebrae
LOS	Length of hospital stay
MLR	Multinomial logistic regression
MNA	Mini nutrition assessment
MST	Malnutrition screening tool
NBM	Negative binomial regression
NS	Non-significant
ONS	Oral nutritional supplements
OS	Overall survival
OR	Odds ratio
PG-SGA	Patient generated-subjective global assessment
POD	Postoperative day
PONV	Postoperative nausea and vomiting
PS	Performance status
QOL	Quality of life
SD	Standard deviation
SAT	Subcutaneous adipose tissue
SE	Standard error
SMA	Skeletal muscle area
SMI	Skeletal muscle index
SMR	Skeletal muscle radiation attenuation

TAT	Total adipose tissue
UICC	Union for International Cancer Control
VAS	Visual analogue scale
VAT	Visceral adipose tissue
VO	Visceral obesity
WL	Weight loss

CHAPTER 1: Introduction and research plan

1.1 Introduction

Cachexia was first described by Hippocrates (460-377 BC) as the gradual emaciation of an individual in response to chronic illness that ultimately leads to death, *'The flesh is consumed... the shoulders, clavicles, chest, and thighs melt away... this illness is fatal*'.¹ Such narrative portraits have continued to appear throughout the literature, whereby a cluster of clinical symptoms including weight loss (WL), wasting of fat and muscle, anorexia, and gastrointestinal disturbances have consistently been used to describe cachexia.¹ The etiology of cachexia has been difficult to characterize due to its multifactorial nature, which is thought to be driven by a variable combination of reduced food intake (FI) and metabolic changes, including elevated energy expenditure, excess catabolism, and inflammation.²⁻⁴ Owing to the prominence of reduced FI as well as the presence of inflammation; cachexia is also classified as a form of disease-associated malnutrition.⁴⁻⁶ Cachexia is distinct from starvation and nutritional deficiencies in otherwise healthy individuals and because of the underlying disease it can only be partly reversed by nutritional interventions.²⁻⁴

Patients with cancer-associated cachexia have distinctive tumor-related metabolic changes that are believed to drive wasting of skeletal muscle and adipose tissue. The variable contribution of tumor, inflammation, and FI leads to notable heterogeneity in the presentation, clinical course, and outcomes of patients with cancer cachexia, and was best summarized by Professor Kenneth Fearon '*When considering how a tumor might influence the host, one has to account for the heterogeneity of the response to cancer. Some patients will remain weight-stable, and some will become profoundly cachectic. For those who lose weight this may be via a series of metabolic changes, through the development of anorexia, or via a combination of the two. Furthermore, there is tremendous heterogeneity in the metabolic response which a host may* demonstrate and a whole variety of causes of reduced food intake. In fact, one might almost despair at trying to find one common final pathway to account for all these changes. Nevertheless, I think such heterogeneity gives us a clue. It suggests that for different tumors and even for patients with the same tumor we may be looking at entirely different mechanisms of weight loss "⁷

It seems highly unlikely that a single clinical, laboratory, or radiological feature would support the diagnosis of cancer cachexia.⁸ However, researchers and clinicians alike have used multiple definitions and diagnostic criteria to identify patients with cancer cachexia which have ranged from a single overall assessment (i.e. involuntary WL) to various combinations of assessments (i.e. FI, skeletal muscle depletion, and systemic inflammation).^{9,10} These inconsistencies were highlighted in a review by Dechaphunkul et al.¹⁰; 24 different definitions for cancer cachexia were identified in head and neck cancers alone, and the number of criteria used ranged from a single criterion up to six different criteria. Lack of a uniformly accepted definition and diagnostic criteria have been described as impediments to the identification and treatment of cancer cachexia, and for the development of new therapeutic agents.^{2,11,12} Inclusion criteria to clinical trials have favored the inclusion of patients with WL of any etiology and near the end of life. Treatment response is difficult to evaluate in patients whose WL was not due to the etiology targeted by the therapy, and who were too close to death to receive a benefit.¹² The importance of diagnosing patients with cancer cachexia cannot be understated; it has a negative impact on patients' physical functioning and quality of life, decreases their ability to tolerate cancer treatments (e.g. surgery and anti-cancer therapies), and reduces survival.^{2,13,14 15} Early identification is key to ensure appropriate clinical, nutritional, and metabolic care.¹⁶⁻¹⁸

The increasing prevalence of overweight and obesity add an additional layer of complexity to the assessment and diagnosis of cancer cachexia. Overweight and obese patients are not typically recognized as being at nutrition risk, however excess body fat can mask skeletal muscle wasting associated with cancer cachexia placing patients at increased risk of chemotherapy toxicity, surgical complications, and death.^{12,19,20} The prevalence of overweight plus obesity combined (i.e. $BMI > 25 \text{ kg/m}^2$) in Canada and the US is 60 and 70% respectively, and is a risk factor for the development of many cancers.²¹⁻²⁴ It is therefore not surprising that today's cancer patient is more likely to be overweight or obese at diagnosis than underweight (e.g. 40% vs. 10%).^{19,25} Computed tomography (CT) images completed as standard of care in oncology are routinely available, and offer an unprecedented window to noninvasively and precisely quantify skeletal muscle, and adipose tissues. CT-defined body composition has emerged as an important prognostic indicator in oncology, and has been suggested as a biomarker for consideration in all cancer patients.²⁶⁻³¹ The emerging clinical utility of CT images to assess body composition greatly expands the clinicians' toolbox beyond body weight, and contributes to the identification of patients at risk of cancer cachexia.

A unifying set of concepts to define and diagnose cancer cachexia was needed, which could serve to guide clinical management, and to help create more homogeneous cohorts for clinical research studies.^{2,8} This need was recognized and addressed within the International Consensus Framework for the Definition and Classification of Cancer Cachexia.² In 2010, a group of international experts in cancer cachexia came together with the purpose to develop a definition and diagnostic criteria specific to cancer cachexia. Making the diagnosis of cancer cachexia is the goal, because it allows for prognostication of outcomes and optimal clinical managment.¹⁶⁻¹⁸ To be able to diagnose cancer cachexia consensus was required with regard to several key elements:

1) A definition: a clear definition should describe what cancer cachexia is, and identifies key clinical and etiological characteristics.

2) **Diagnostic criteria:** a set of signs, symptoms, and tests for use in routine clinical care that when applied identify patients with the syndrome, and can guide the care of individual patients.⁸ General requirements for diagnostic criteria are that they are broad and reflect the different features of the syndrome (e.g. the heterogeneity).⁸

3) **Classification criteria:** are standardized definitions intended to capture the majority of patients with key features of the syndrome, and enables the creation of well-defined, homogeneous cohorts for clinical research.⁸

To gain consensus for these main elements, an international consortium was formed which included 19 experts in clinical cancer cachexia research such as medical and surgical oncologists, palliative medicine specialists, and nutritionists.² A formal Delphi consensus process was adopted that included focus groups, discussion, and two Delphi scoring and voting rounds.² Consensus was reached for a definition: cancer cachexia is defined as *'a multifactorial syndrome defined by ongoing skeletal muscle loss (with or without fat loss)...characterized by negative protein and energy balance driven by a variable combination of reduced food intake and altered metabolism* ^{v2} Novel to this definition is the statement that skeletal muscle loss can occur independently from fat mass; therefore the quantification of skeletal muscle mass is critical to the diagnosis of cancer cachexia. This definition acknowledges the specific etiology of cancer cachexia including reduced FI and tumor-driven components that alter metabolism (e.g. tumorderived catabolic factors, inflammation, and hypermetabolism) and contribute to excess catabolism of skeletal muscle and adipose tissue as well as to alterations in nutrient intake.^{2,32}

Consensus was achieved for the central concepts that will underlie the eventual classification of cancer cachexia according to stages. It was agreed that cancer cachexia is a syndrome that evolves across a continuum characterized by three stages of increasing severity: pre-cachexia, cachexia, and refractory cachexia, which may be variously characterized by severity of WL, reduced FI, altered metabolism, poor functional status, and decreased life expectancy (Figure 1-1).² However, there was no consensus with regard to the actual criteria to identify patients for each cachexia stage, and this remains an area for development.



Figure 1-1. Classification of cancer cachexia stage

A schematic for the stages of cachexia, re-drawn from Fearon et al.² This figure highlights the concepts that underlie the eventual classification of cancer cachexia, which is represented as a continuum that includes three stages of increasing severity, and presents provisional criteria that may identify patients by cachexia stage. Pre-cachexia is suggested to be characterized by early nutritional (e.g. anorexia) and metabolic changes that precede substantial involuntary loss of body weight and depletion of skeletal muscle (e.g. sarcopenia). Cachexia is characterized by varying degrees of involuntary weight and skeletal muscle depletion that is due to a variable combination of reduced food intake and systemic inflammation. Refractory cachexia is suggested to be characterized by patients with advanced cancers that are no longer responsive to anti-cancer therapies, who have low performance status (e.g. $ECOG \ge 3$) and life expectancy of <3 months.

Given the complexity and multifactorial nature of cancer cachexia the development and

validation of diagnostic criteria is expected to be challenging, and consensus was not achieved

for diagnostic criteria.^{2,7,8} Involuntary WL and skeletal muscle depletion had some degree of agreement regarding their use as diagnostic criteria, and provisional diagnostic criteria were presented.² Involuntary WL as a diagnostic criterion was agreed upon by all participants. However, the severity scale for WL was not agreed upon for the reasons that multiple severity scales are used in clinical practice (e.g. Common Terminology Criteria for Adverse Events $(CTCAE)^{33} > 5\%, >10\%, >20\%$), which were arbitrarily defined, and do not consider the severity of WL in combination with initial body weight.² A re-evaluation of WL was recommended, to define the impact of the rate of WL in combination with initial body weight in relation to clinical outcomes. There was consensus supporting the routine assessment of skeletal muscle, but no consensus was achieved regarding specific methodology, although preference was given to the quantification of skeletal muscle using CT.²

Consensus was not achieved for any diagnostic criteria pertaining to the etiological factors of cancer cachexia. However, an agreed upon list of candidate diagnostic criteria was provided for assessments of FI and altered metabolism.² Assessments of FI are recognized to be heterogeneous, and several types of assessments were suggested as possible diagnostic criteria including qualitative (e.g. patient-reported food intake) and quantitative (e.g. calculation of energy and protein intakes) assessments. There are few clinical measures of altered metabolism; however C-reactive protein (CRP, a marker of systemic inflammation) and the assessment of energy expenditure (to determine if patients were hypermetabolic) are recognized as indicators of metabolic change and were suggested as possible diagnostic criteria.²

To advance efforts to develop diagnostic criteria for cancer cachexia, authors of the consensus framework recommended a data-driven approach to develop 'definitive cutoffs for variables (e.g. weight loss, low muscle mass) which could be determined from large

6

contemporary datasets by determining the values that relate optimally to meaningful patientcentered outcomes, such as loss of function or decreased survival" This statement is the foundation for my thesis.

1.2 Research approach

My research approach has been to aggregate data from clinical research studies, and to conduct secondary analysis to evaluate body weight, WL, FI, CT-defined body composition, and systemic inflammation as diagnostic criteria for cancer cachexia. My goal was to create an aggregated data set large enough for adequately powered statistical and subset analyses. Data was eligible for aggregation based on the following criteria: i) original data were prospectively collected under the auspices of human ethics approvals from respective institutions, ii) data were anonymized, iii) data included a nutrition risk assessment at the point of referral for nutrition care (i.e. prior to nutrition intervention) which could include the following scenarios: at entry to investigational cachexia clinical trials, at cancer diagnosis as of standard of care, prior to undergoing elective surgery, and at entry to supportive/palliative care programs, iii) clinical data including: patient demographics, cancer type and stage, surgical type and approach, performance status, iv) patient centered outcomes (overall survival (OS), length of hospital stay (LOS), postoperative complications, 30-day hospital readmission). Clinical research studies of cachexia include patients most at risk for developing cachexia (i.e. specific cancers and those with locally advanced or metastatic tumors) therefore not all cancer types and stages are represented. Additionally, data on ethnicity is not collected. Details about my research approach are presented in Chapter 3. Cachexia-related data is often already part of standard of care (e.g. CT images; blood work; cancer diagnosis and stage), but is also enriched with data collected from additional nutrition (e.g. WL, FI, appetite) and functional assessments that are not typically found within the medical chart. A main limitation of aggregating data sets for retrospective secondary analysis is that the data collected can be heterogeneous and difficult to aggregate as different parameters and assessments are used in different studies. I address these limitations throughout my thesis in Chapters 2, 5, 6, and 9.

1.3 Research plan

My overall objective was to advance the development of diagnostic criteria for cancer cachexia. Three specific objectives, sub-objectives and respective hypotheses are outlined:

Objective 1. To evaluate WL as a diagnostic criterion based on the consensus statement that WL 'should be graded according to degree of weight loss and concurrent BMI, and the severity classification should be developed around the predictive value for outcomes such as treatment toxicity, quality of life, hospitalization, and survival'² The aim was to define the prognostic significance of WL and body mass index (BMI) in a large international data set of contemporary cancer patients, using mortality as the outcome.

Sub-Objective 1.1 was to form the International Cancer Cachexia Data Repository (ICCDR) by aggregating data from clinical cancer cachexia research studies made available from international collaborators.

Sub-Objective 1.2 was to test the prognostic significance of WL and BMI in a multivariable Cox proportional survival model, and develop an updated grading system incorporating both WL and BMI. I hypothesized that increased WL and low BMI would be independently predict decreased OS, and that WL would confer a worse prognosis in patients with lower initial BMIs. **Sub-Objective 1.3** was to validate the grading system for cancer-associated WL in an independent sample. **Objective 2** was to evaluate two candidate diagnostic criteria, reduced FI and CRP (a marker of systemic inflammation), as etiological factors for cancer-associated WL.

Sub-objective 2.1 was to describe the relationship between FI, as assessed by three different tools (the Patient-Generated Subjective Global Assessment (PG-SGA)^{34,35}, the *Ingesta*-Verbal/Visual Analogue Scale (I-VVAS)^{36,37}, and the Mini-Nutrition Assessment (MNA) ^{38,39}), and WL to determine if there was equivalency between tools using multinomial logistic regression analysis. I hypothesized the different measurements scales for FI would demonstrate similar associations to WL, and that FI data could be combined. I also hypothesized that increasingly severe reductions in FI would be associated with increasingly severe WL i.e. that FI would explain, in part, the etiology of WL.

Sub-objective 2.2 was to describe the relationship between CRP and WL, and to determine optimal values of CRP associated with increased WL. I hypothesized that higher CRP values would be associated with increased WL.

Sub-objective 2.3 was to evaluate the contribution of CRP *and* FI as etiological criteria for WL using multinomial logistic regression analysis. I hypothesized that both CRP and reduced FI would be independently associated with increased severity of WL.

Sub-objective 2.4 was to determine if the combination of WL, FI, and CRP could be used to stratify patients according to overall survival using the Kaplan Meier method. I hypothesized that increased WL would identify patients with poor OS and that high CRP values and reduced FI would contribute to additional risk stratification.

Objective 3 was to determine if features of cancer cachexia, including WL, reduced FI (as assessed by a nutrition risk screen), and CT-defined body composition (i.e. skeletal muscle depletion, low skeletal muscle radiation attenuation, and high visceral adipose tissue) could

identify surgical patients' poor postoperative outcomes. Outcomes including compliance to standardized surgical care pathways, LOS, postoperative complications, and 30-day hospital readmission were evaluated using logistic and negative binomial regression analysis.

Sub-objective 3.1 was to evaluate surgical data collected from a standardized evidence-based perioperative surgical care pathway (Enhanced Recovery After Surgery (ERAS)) to determine if implementation of ERAS across colorectal surgical programs in Alberta, Canada improved the rate of nutrition risk screening. I hypothesized that nutrition risk screening would be improved, and that patients with nutrition risk (i.e. WL and reduced FI) would be identified.

Sub-objective 3.2 was to determine if patients identified with nutrition risk had worse surgical outcomes, including compliance to surgical pathways, LOS, and postoperative complications using logistic regression. I hypothesized that patients with nutrition risk would have lower compliance to ERAS care pathways, longer LOS, and increased postoperative complications. **Sub-objective 3.3** was to acquire and aggregate clinical data from research studies on cancer cachexia that evaluated preoperative CT-defined body composition in colorectal cancer (CRC) patients eligible for elective surgery.

Sub-objective 3.4 was to establish a sex and age-specific reference group for preoperative CTdefined body composition features in elective CRC patients.

Sub-objective 3.5 was to define sex- and age-specific thresholds for CT-defined low skeletal muscle, low skeletal muscle radiation attenuation, and high visceral adipose tissue that were significantly associated with longer LOS using negative binomial regression. I also wanted to determine if the combination of one, two, or three body composition features conferred increased risk of longer LOS, major complications, and a higher rate of 30-day hospital readmission. I hypothesized that low skeletal muscle, low skeletal muscle radiation attenuation, and high

10

visceral adipose tissue, would each be associated with longer LOS and that multidimensional profiles would confer additional risk for poor postoperative outcomes.

1.4 Chapter Format

The objectives, sub-objectives, and hypotheses stated above were tested in a series of studies, which have been organized into thesis chapters. Chapters 2 to 4 and 7 & 8 have been published. The thesis chapters will be presented in the order in which the research was completed, and are organized by oncology practice setting (e.g. medical or surgical oncology). Chapters 2 through 6 focus on research conducted in the medical oncology setting, followed by research conducted in the surgical setting (Chapters 7 & 8).

Chapter 2 is the first publication to use data from the ICCDR. This work presents an updated classification system for cancer-associated weight loss based on the prognostic significance of both WL and BMI. Objective 1 and sub-objectives 1.1 to 1.3 are addressed in this chapter.

Chapters 3 & 4 are literature reviews describing recent efforts to develop diagnostic criteria for cancer cachexia according to the principles outlined in the International Consensus Framework for Cancer Cachexia.² Chapter 3 provides an in-depth explanation of my research approach and additional context for the work completed in Chapters 2, 5 & 6. Chapter 4 provides discussion about the factors contributing to the etiology of cancer cachexia, as well as context for the research presented in Chapters 5 & 6.

Chapter 5 describes the relationship between FI and WL, and demonstrates how different assessments of FI were aligned based on their associations to WL. FI as an etiological criterion

for WL was also evaluated in multivariate regression. Objective 2 and sub-objective 2.1 are addressed in this chapter.

Chapter 6 describes the association between CRP and WL, and defines optimal values of CRP based on this relationship. CRP as an etiological criterion for WL was evaluated when adjusted for other covariates including reductions in FI. Lastly, the combination of WL, CRP, and FI was evaluated in relation to OS. Objective 2 and sub-objectives 2.2 to 2.4 are addressed in this chapter.

Chapter 7 describes the implementation of a standardized evidence-based perioperative surgical care pathway, ERAS, in colorectal surgeries. This study evaluates changes to nutrition risk screening across surgical programs and evaluates whether nutrition risk screening identifies patients with poor surgical outcomes. Objective 3 and sub-objectives 3.1 to 3.2 are addressed in this chapter.

Chapter 8 describes how the aggregation of clinical data was use to establish age and sex norms for features of body composition specific to preoperative CRC patients, and evaluates features of body composition in relation to postoperative outcomes. Objective 3 and sub-objectives 3.3 to 3.5 are addressed in this chapter.

Chapter 9 provides an overall discussion and directions for future research.

1.5 References

- 1. Bennani-Baiti NW, D. What is cancer anorexia-cachexia syndrome? A historical perspective. R Coll Physicians Edinb. 2009; 39:257-262.
- 2. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.
- 3. Evans WJ, Morley JE, Argiles J, et al. Cachexia: a new definition. Clin Nutr. 2008; 27:793-799.
- 4. Jensen GL, Mirtallo J, Compher C, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. JPEN J Parenter Enteral Nutr. 2010; 34:156-159.
- 5. Cederholm T, Bosaeus I, Barazzoni R, et al. Diagnostic criteria for malnutrition An ESPEN Consensus Statement. Clin Nutr. 2015; 34:335-340.
- 6. Cederholm T, Jensen GL. To Create a Consensus on Malnutrition Diagnostic Criteria. JPEN J Parenter Enteral Nutr. 2017; 41:311-314.
- 7. Fearon K. The Sir David Cuthbertson Medal Lecture 1991: The mechanisms and treatment of weight loss in cancer. Proc Nutr Soc. 1992; 51:251-265.
- 8. Aggarwal R, Ringold S, Khanna D, et al. Distinctions between diagnostic and classification criteria? Arthritis Care Res. 2015; 67:891-897.
- 9. Blum D, Omlin A, Baracos VE, et al. Cancer cachexia: a systematic literature review of items and domains associated with involuntary weight loss in cancer. Crit Rev Oncol Hematol. 2011; 80:114-144.
- 10. Dechaphunkul T, Martin L, Alberda C, Olson K, Baracos V, Gramlich L. Malnutrition assessment in patients with cancers of the head and neck: a call to action and consensus. Crit Rev Oncol Hematol. 2013; 88:459-476.
- 11. Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.
- 12. Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol. 2013; 10:90-99.
- 13. Farkas J, von HS, Kalantar-Zadeh K, Morley JE, Anker SD, Lainscak M. Cachexia as a major public health problem: frequent, costly, and deadly. J Cachexia Sarcopenia Muscle. 2013; 4:173-178.

- Thompson KL, Elliott L, Fuchs-Tarlovsky V, Levin RM, Voss AC, Piemonte T. Oncology Evidence-Based Nutrition Practice Guideline for Adults. J Acad Nutr Diet. 2016.
- 15. Chan LN, Compher C, DiBaise JK, et al. American Society for Parenteral and Enteral Nutrition research agenda. JPEN J Parenter Enteral Nutr. 2014; 38:13-18.
- 16. Arends J. Struggling with nutrition in patients with advanced cancer: nutrition and nourishment-focusing on metabolism and supportive care. Ann Oncol. 2018; 29:ii27-ii34.
- 17. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 18. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 19. Gioulbasanis I, Martin L, Baracos VE, Thezenas S, Koinis F, Senesse P. Nutritional assessment in overweight and obese patients with metastatic cancer: does it make sense? Ann Oncol. 2015; 26:217-221.
- 20. Baracos VE, Arribas L. Sarcopenic obesity: hidden muscle wasting and its impact for survival and complications of cancer therapy. Ann Oncol. 2018; 29:ii1-ii9.
- 21. De PG, Silvestris F. Obesity as a major risk factor for cancer. J Obes. 2013; 2013:291546.
- 22. Center for Disease Control and Prevention. Obesity and Overweight. 2016; http://www.cdc.gov/nchs/fastats/obesity-overweight.htm.Accessed: October 11, 2018.
- 23. Canada S. Body mass index, overweight or obese, self-reported, adult, by age group and sex. 2016; http://www.statcan.gc.ca/tables-tableaux/sum-som/l01/cst01/health81b-eng.htm.Accessed: October 11, 2018.
- 24. Wolin KY, Carson K, Colditz GA. Obesity and cancer. Oncologist. 2010; 15:556-565.
- 25. Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancer-associated weight loss. J Clin Oncol. 2015; 33:90-99.
- 26. Hubbard JM, Cohen HJ, Muss HB. Incorporating biomarkers into cancer and aging research. J Clin Oncol. 2014; 32:2611-2616.
- 27. Kazemi-Bajestani SM, Mazurak VC, Baracos V. Computed tomography-defined muscle and fat wasting are associated with cancer clinical outcomes. Semin Cell Dev Biol. 2016; 54:2-10.

- 28. Levolger S, van Vugt JL, de Bruin RW, JN IJ. Systematic review of sarcopenia in patients operated on for gastrointestinal and hepatopancreatobiliary malignancies. Br J Surg. 2015; 102:1448-1458.
- 29. Malietzis G, Aziz O, Bagnall NM, Johns N, Fearon KC, Jenkins JT. The role of body composition evaluation by computerized tomography in determining colorectal cancer treatment outcomes: a systematic review. Eur J Surg Oncol. 2015; 41:186-196.
- 30. van Vugt JL, Levolger S, de Bruin RW, van RJ, Metselaar HJ, Ijzermans JN. Systematic review and meta-analysis of the impact of computed tomography assessed skeletal muscle mass on outcome in patients awaiting or undergoing liver transplantation. Am J Transplant. 2016; 16:2277-2292.
- 31. Yu H, Joh YG, Son GM, Kim HS, Jo HJ, Kim HY. Distribution and Impact of the Visceral Fat Area in Patients With Colorectal Cancer. Ann Coloproctol. 2016; 32:20-26.
- 32. Burfeind KG, Michaelis KA, Marks DL. The central role of hypothalamic inflammation in the acute illness response and cachexia. Semin Cell Dev Biol. 2016; 54:42-52.
- 33. National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE) v.5.0. 2017.
 https://ctep.cancer.gov/protocolDevelopment/electronic_applications/docs/CTCAE_v5_Q uick_Reference_8.5x11.pdf. Accesssed: October 11, 2018.
- 34. Jager-Wittenaar H, Ottery FD. Assessing nutritional status in cancer: role of the Patient-Generated Subjective Global Assessment. Curr Opin Clin Nutr Metab Care. 2017; 20:322-329.
- 35. Ottery FD. Definition of standardized nutritional assessment and interventional pathways in oncology. Nutrition. 1996; 12:S15-19.
- Thibault R, Goujon N, Le Gallic E, et al. Use of 10-point analogue scales to estimate dietary intake: a prospective study in patients nutritionally at-risk. Clin Nutr. 2009; 28:134-140.
- 37. Guerdoux-Ninot E, Flori N, Janiszewski C, et al. Assessing dietary intake in accordance with guidelines: Useful correlations with an Ingesta Verbal/Visual Analogue Scale in medical oncology patients. Clinical Nutrition. 2018;DOI: 10.1016/j.clnu.2018.06.974.
- 38. Vellas B, Villars H, Abellan G, et al. Overview of the MNA Its history and challenges. The journal of nutrition, health & aging. 2006; 10:456-635.
- 39. Rubenstein LZ, Harker JO, Salva A, Guigoz Y, Vellas B. Screening for undernutrition in geriatric practice: developing the short-form mini-nutritional assessment (MNA-SF). J Gerontol A Biol Sci Med Sci. 2001; 56:M366-372.

CHAPTER 2: Diagnostic criteria for the classification of cancer-associated weight loss¹

2.1 Introduction

Involuntary weight loss (WL) is the cardinal diagnostic criterion of cancer cachexia.¹ WL at the start of chemotherapy is associated with reduced response rates and increased toxicity, and is included as one of the key Common Toxicity Criteria of Adverse Events (CTCAE).² Percent WL (%WL) is an index of severity but there are marked inconsistencies as to what %WL oncologists consider clinically important (varying from 5 to >20%).^{3, 4} Likewise, levels of body mass index (BMI) used to define clinically underweight and cachexia cited by various authorities and authors are inconsistent (<18.5, <21, <23 kg/m²).⁵

Individual patients do not start their cancer journey with identical body habitus. The demographics of body weight in cancer patients have changed since most existing grading schemes for WL were established. Overweight and obesity are now prevalent worldwide⁶, and the upward shift in body weight renders definitions of clinically significant WL in cancer patients increasingly unclear.^{7,8} Cancer cachexia contributes to poor prognosis through progressive depletion of the body's energy and protein reserves, therefore it is relevant to determine the impact of WL on survival as a function of initial body reserves. Obesity is usually considered a disadvantage based on studies of all-cause mortality.⁹ However, obesity confers a survival advantage in patients with several diseases associated with WL (e.g. heart and renal failure).¹⁰ The larger energy reserve of obese persons may confer this advantage.¹¹

Diagnostic criteria for cancer cachexia will inevitably include additional information beyond merely assessing WL, such as the presence of skeletal muscle wasting, anorexia and

¹A version of this chapter has been published: Martin L, Senesse P, Gioulbasanis I, Antoun A, Bozzetti F, Deans C, Strasser F, Thoresen L, Jagoe RT, Chasen M, Lundholm K, Bosaeus I, Fearon K, Baracos VE. Diagnostic criteria for cancer-associated weight loss. J Clin Oncol, 2015; 33:90-99.
inflammation.¹² However, we propose that in the first instance a robust classification of WL is needed, that it should be based on contemporary data and that it should evaluate the prognostic significance of WL in patients who are initially of low, intermediate, and high BMI. Current WL grading schemes do not take into account potential benefit of higher initial body weight in risk assessment of patients with cancer- or treatment-associated WL. The European School of Oncology recommends early recognition of cachexia¹³, and an international consensus group¹² recommended the collection of contemporary data to classify the severity of cancer cachexia in relation to outcomes such as treatment toxicity and survival. To that end, we undertook to define the prognostic significance of BMI and WL in a large international data set.

2.2 Methods

2.2.1 Data collection and Handling

Two samples of prospectively collected data were assembled (Table 2-1): a training sample for development of a BMI adjusted WL grading system and a validation sample to assess the performance of the grading system. Data were collected at presentation under the auspices of research ethics approvals at the contributing institutions and were anonymized. Data included: age (>18 years), sex, confirmed cancer site and stage, performance status (PS), WL history, BMI, and time to death or censoring. Cachexia is defined as a continuum with increasing cumulative losses over time to death.¹² To represent this continuum we included patients at most risk for developing cachexia (i.e. specific cancers and those with locally advanced or metastatic tumors). A main point of contact with the health care system at which the data were collected was medical oncology (i.e. consecutively referred patients). The training sample included several previously published population-based cohorts¹⁴⁻²⁴ and participants in randomized clinical trials.²⁵⁻²⁷ Other

data were from screening programs for nutritional risk as part of standard care in medical oncology and integrated supportive/palliative care programs at participating cancer centres. The validation sample included nutritional risk screening data (consecutively referred patients) from a cancer center in Montpellier, France. Compiled data were evaluated against inclusion criteria, and patients were excluded if they were <18 years of age, had a diagnosis of cancer *in situ*, or were missing WL and BMI data.

Weight loss history

Patient-reported weight history was collected over various time frames: previous 1, 2, 3, and/or 6 months, and/or from usual body weight (UBW; Table 2-1). Patient-reported height, weight, and WL history are reliable^{28, 29} and patient report is part of the standard medical approach in completing a weight history.³⁰

Percent WL was calculated:

[(current weight (kg) - previous weight (kg))/previous weight (kg)] x 100

A majority of cases (8396/10768) of all patients (training and validation samples) had WL over the preceding 6 months so we elected to make the initial analysis using this as the primary WL time frame. If 6 month %WL was missing, the next most frequently reported WL time frame was used (UBW, 3, 1 and 2 months). Different WL time frames were evaluated for their ability to predict survival (see results). BMI is reported as current weight (kg) / height² (expressed as meters²).

Performance status and Cancer Stage

Performance status was recorded as Eastern Cooperative Oncology Group Performance Status (ECOG PS). Karnofsky Performance Status (KPS) data were converted to ECOG PS using categories (³¹): KPS 100 (ECOG 0), 90-80 (ECOG 1), 70-60 (ECOG 2), 50-40 (ECOG 3), 0-30 (ECOG 4). Cancer stage was based on the American Joint Committee for Cancer stage groupings.

2.2.2 Statistics

The primary outcome was overall survival, defined as the number of months a patient survived between the date of WL and BMI assessment and the date of death. All patients were followed until death or were censored at their last confirmed contact with the healthcare system. Survival analysis included: Kaplan-Meier method (comparisons with Mantel-Cox log-rank tests) and Cox proportional hazards model (estimated hazard ratios (HR) and 95% confidence intervals (CI)). Concordance (c-)statistics assessed discrimination of the BMI adjusted WL grading system in predicting survival³² as previously described¹⁵. C-statistics are applicable to all regression models, including survival models³³, a value of 0.5 indicates a prediction no better than chance and 1.0 is perfect prediction.³² Analyses were completed using IBM SPSS Statistics for Windows (v22.0, Armonk, NY: IBM Corp.) and C-statistics (95%CI) were estimated using a macro in SAS (v9.1.3, SAS Institute Inc., Cary, NC, U.S.A.). Results were considered significant at the p<0.05 level.

Part I: Survival Prediction

The training sample was used to determine the prognostic significance of %WL and BMI (as continuous variables) in a multivariate model controlled for age, sex, cancer site, stage and PS.

19

Part II: Categorical Assignment of %WL and BMI

Categories were defined for %WL and BMI that related to overall survival from the training sample. Each variable was divided into deciles (i.e. divides the distribution of a continuous variable into ten equal groups) to explore the impact of increasing %WL and decreasing BMI on overall survival. Deciles were compared based on differences in median survival (Kaplan Meier) and prognostic impact on survival (estimated HR from Cox proportional hazard model).

Part III: Grading system

The grading system was created using the training sample based on the combination of %WL and BMI categories. This analysis was laid out in a 5x5 matrix representing 5 different %WL categories within each of the 5 different BMI categories (25 possible combinations of WL and BMI). Grade 0 was assigned to the least risk subgroup in the matrix (longest survival) and Grades 1, 2, 3, 4 were assigned to the subgroups according to decreasing survival and increasing HR.

Part IV: Validation

The %WL and BMI of patients in the validation sample were graded according to BMI adjusted WL grading system, and survival discrimination was evaluated according to median survival and c-statistics, and compared to the training sample.

2.3 Results

2.3.1 Training Sample

Cancer stage for the sample (n=8675) was predominantly (89%) locally advanced or metastatic; the majority (73%) were weight losers (mean WL -9.7%). A few patients (6%) experienced weight gain (defined as >2.4%; mean +8.3%) and 21% were weight stable (within

 $\pm 2.4\%$ to reflect diurnal variation in body weight; mean -0.2%). Despite prevalent WL, only 10% were underweight (BMI <18.5), 48% normal weight (18.5-24.9), 29% overweight (25.0-29.9) and 13% were obese (\geq 30.0).

Previous work¹⁴ demonstrated weight gain was prognostic of reduced survival in advanced cancer patients in the last months of life. Several forms of weight gain signal disease progression: edema, ascites, increased organ volume and tumor mass. Given that we could not discriminate the specific nature of weight gain, the main analysis focused specifically on weight stable and weight losing individuals (N=8160). Patient characteristics for the training sample are presented (Table 2-2). There were 6294 deaths (out of 8160 patients), an overall median survival of 8.2 months (95% CI 7.9-8.6), and a median follow-up of 41.3 months (95% CI 39.8-42.8).

%WL and BMI (continuous variables) were modeled along with conventional covariates known to impact survival (Table 2-3). Predictors of survival in the univariate analysis included %WL, BMI, sex, age (continuous), cancer site, cancer stage, and PS. All variables, except sex, predicted survival at the multivariate level, both increasing %WL (P<0.001) and decreasing BMI (P=0.010) independently predicted survival for the training sample overall. %WL over all time frames was predictive of survival (data not shown): 1 month (P<0.001), 2 months (P<0.001), 3 months (P=0.003), 6 months (P<0.001), and UBW (P<0.001). A multivariate model including advanced stages only (III+IV) controlled for diagnosis, sex, age, BMI, and %WL (not shown) was not different from the overall multivariate model presented in Table 2-3.

2.3.2 BMI adjusted WL grading system

Figure 1 depicts the relationship between deciles of %WL and BMI to overall survival. For BMI deciles (n=816/decile), there were no differences in median survival between deciles 1+2, 3+4, 5+6, 7+8, 9+10 (Figure 2-1A). Therefore, 5 categories of BMI (<20.0, 20.0-21.9, 22.0-24.9, 25.0-27.9, \geq 28.0 kg/m²) were considered that differed in overall survival (P<0.001). Weight stable (± 2.4%) patients were grouped into one category (N=1847) and weight losing patients (N=6290) were split into deciles (n=629/decile). There were no differences in median survival between deciles 1+2+3, 4+5, 6+7+8, 9+10, and 11 (Figure 2-1B). Thus 5 categories of WL were considered that differed in overall survival: weight stable (± 2.4%) and %WL (2.5 to - 5.9%, -6.0 to -10.9%, -11.0 to -14.9%, and \geq -15.0%, P<0.001).

Figure 2-1: Line graphs representing the relationships between deciles of BMI (1A) and percent weight loss (%WL, 1B) to overall survival.



Decile 1 represents the lowest BMI (1A) and the highest %WL (1B), decile 10 represents the highest BMI (1A) and lowest %WL (1A). Blue lines represent unadjusted estimated hazard ratios associated with reduced overall survival. Reference categories are BMI decile 10 (BMI >30.9, HR=1.0, 1A) and weight stable ($\pm 2.4\%$, HR=1.0, 1B). The risk of reduced survival increases with decreasing BMI (1A) and increasing %WL (1B). Yellow lines represent the estimated median overall survival in months. Median survival decreases with decreasing BMI (1A) and increasing %WL (1B). Different shades of blue in the line graphs and legends indicate significant differences (P<0.05) in median survival between deciles.

The 5x5 matrix analysis representing 25 possible combinations for %WL and BMI are presented (Figure 2-2). Sample size (n), median survival and unadjusted HRs (see below for adjusted HRs) are shown in Figures 2-2A-C, respectively. Combining groups with similar HRs yielded 5 distinct grades with significantly different survival (Figure 2-2D). A gradient of decreasing survival was observed with increasing %WL and decreasing BMI; the highest risk in the lower right hand corner (Grade 4, median survival 4.3 months) and the least risk in the upper left corner (Grade 0, 20.9 months).

Figure 2-2: Risk of reduced survival is a function of body mass index *and* percent weight loss (%WL).



Panels A-C represent a 5x5 matrix analysis of the 5 categories of BMI and 5 categories of %WL for a total 25 possible combinations. The sample size (A), median overall survival (months; B), and unadjusted estimated hazard

ratios (*reference category BMI \geq 28.0/weight stable \pm 2.4%, HR=1.0; 2C) are presented for each cell. Different colors (A-C) represent significant differences (P<0.05) in median overall survival and HRs within and between cells of the matrix. Figure 3-2D represents the BMI adjusted WL grading system (grades 0 to 4). Grade 0 (20.9 months (95%17.9-23.9), unadjusted HR=1.0), Grade 1 (14.6 months (12.9-16.2), HR=1.3), Grade 2 (10.8 months (9.7-11.9), HR=1.5), Grade 3 (7.6 months (7.0-8.2), HR=2.0), Grade 4 (4.3 months (4.1-4.6), HR=3.1, P<0.001). Figures 3-2E,F represent cumulative survival curves from the sub-group analysis of the training sample for gastro-esophageal (E) and head & neck (F) cancers by grade.

The grading system showed good discrimination of survival (c-statistic 0.88 (95% CI 0.86-0.89)). The BMI adjusted WL grades were entered into a multivariate analysis controlled for age, sex, cancer site, cancer stage, and performance status and they were independent predictors of survival: Grade 0 (adjusted HR=1.0, reference category), Grade 1 (HR=1.1 (95% CI 1.0-1.3), P=0.02), Grade 2 (HR=1.2 (1.1-1.3), P<0.001), Grade 3 (HR=1.4 (1.2-1.5), P<0.001), Grade 4 (HR=1.7 (1.5-1.9), P<0.001). These results are similar to the model presented in Table 2 with %WL and BMI as continuous variables.

2.3.3 Subgroup analyses

The training sample included a range of cancer sites, stages and PS and each of these were independent predictors of survival (Table 2-3). Clinically, we evaluate prognosis within specific cancer sites and stages. Therefore the grades were assessed for survival discrimination within major sub-groups of the training sample for which there was adequate sample size and number of events (Table 2-4). Overall, the grades showed good survival discrimination (Figure 2-3 A,B) and were independent predictors of survival when the multivariate analysis was stratified according to cancer site, cancer stage, age, PS, and health care setting (Table 2-4). WL grades were independently predictive of survival in patients with both good (0-2) and poor (3-4) performance status when controlled for age, sex, cancer diagnosis and stage (Table 2-4).

Figure 2-3: Cumulative survival curves from the subgroup analysis of the training sample for (A) garteoesophageal and (B) head and neck cancers.



2.3.4 Validation Sample

A total of 2093 patients were included (Table 2-2). Median overall survival was 12.3 months (95% CI 11.4-13.3) with a median follow-up of 25.7 months (95% CI 24.7-26.8). BMI and %WL (continuous variables) were independently predictive of survival in multivariate analysis (Table 2-3). Patients in the validation sample had a different case mix, however the BMI adjusted WL grades when applied to this sample gave a c-statistic of 0.89 (95% CI 0.87-0.92), concordant with the training sample. The grades gave good survival discrimination overall and was a significant independent predictor of survival in a multivariate analysis (Table 2-4).

2.4 Discussion

Ours is the first systematically developed cancer WL grading system that incorporates the 2 dimensions of WL and BMI and links them to survival. We assembled data representing the spectrum of these features in contemporary cancer patients, and demonstrated that both %WL and BMI predict survival independently of conventional prognostic factors including cancer site, stage, and PS. This large study validates the concept proposed within the international cachexia

classification framework¹² that the severity of WL should be evaluated based on the rate of loss and the level of depletion of body reserves. The proposed grading system takes into account the impact of high versus low initial BMI in the risk assessment of patients with WL. On average there was a 4.9 fold difference in median survival between grades 0 and 4 (20.9 versus 4.3 months, Table 2-4), and these differences were even higher in patient subgroups with the greatest long term nutritional risk, such as cancers of the head and neck (12.8 fold difference) which are notorious for impairment of food intake. Energy deficits of $\sim 120,000$ total calories during chemoradiation have been reported for patients with head and neck cancer³⁴, thus substantial energy reserves may confer an advantage. In patients nearing the end of their cancer trajectory (e.g. supportive/palliative care setting) it might be speculated that the quantity of energy reserves are less important in predicting survival, as death may ensue from tumor invasion of vital organs and processes. While indeed the magnitude of difference in median survival between grades 0 and 4 was less in poor prognosis subgroups, they still gave good survival discrimination at later stages of the disease trajectory. Survival prediction at the end of life is important for clinicians to make decisions regarding patient care i.e. placement into palliative care.

The power of this exercise lies in the assembly of a large sample of patients to clarify interactions between simple markers of nutritional status. BMI and %WL can be combined to provide a severity grade related to the risk of shortened survival. This is most clearly understood and presented in a matrix of combinations of BMI and %WL. The least risk category (longest survivors) comprises individuals with a high BMI who are weight stable or have minimal WL. Likewise, the highest risk category (lower right corner of the matrix) has low initial BMI and high %WL. It has been conventional to use one or more %WL cutoffs in grading systems applied to cancer patients, including CTCAE, cachexia scores, and screening tools for malnutrition.

Using a single cutoff has the pitfall of subgrouping patients with disparate degrees of risk. For example, in Figure 3-2B within all patients with WL <10.9%, there are significantly different subsets of patients with characteristic median survivals as long as 21.5 months (weight stable with high BMI) and as short as 4.7 months (initial BMI <20 kg/m²).

Our results reinforce the interest of documenting WL and BMI, which when assembled into a meaningful classification scheme, can reclaim their clinical utility. Loss of ability to maintain body weight, even subtle WL >2.4% is significantly related to decreased survival, and is consistent with the recently proposed notion of "pre-cachexia"¹². This could be a useful time to start preventative nutrition and metabolic interventions, rather than delaying intervention for WL until some arbitrarily defined high level. Clinical management of obese patients with cancer is a developing sphere of cancer research.⁷ Here we note that for individuals with the highest BMI (i.e. largest energy reserves) the risk of mortality associated with any degree of WL is less than for individuals with lower initial BMI. We caution that this does not mean their WL has no clinical consequence. Separate work clearly demonstrates that skeletal muscle loss is particularly detrimental in obese and overweight cancer patients.^{15, 35} In obese patients with advanced solid tumors, sarcopenia (severe muscle depletion)¹⁴ or sarcopenia and concurrent WL³⁴, associated with poor prognosis compared with obese patients without these features.

We suggest that our BMI adjusted WL grading system is a useful tool in efforts to predict survival as it is independent of cancer site, stage, and PS, and strongly discriminates survival differences. Grade 4 carries a particularly poor prognosis (Table 2-4). Grade 4 is thus of value in identifying individuals whose expected survival is too short to be consistent with certain specific treatment plans or interventions. In current CTCAE criteria, grade 4 (life-limiting toxicity) is undefined for WL. The consistently short median survival for grade 4 BMI adjusted WL across a broad range of cancer sites, stages, and settings, is a potentially useful and consistent criterion for grade 4 toxicity. The grades will assist in more effective stratification of patients to clinical trials involving interventions for cachexia as well as investigations of cancer therapy, allowing inclusion of more homogeneous populations with respect to expected survival.

There are some limitations to our work. Although our data sets were all collected prospectively there are differences in the time frame of WL and types of PS measures. Concerns about these differences are partly addressed by the subgroup analysis within our large training (n=8160) and validation (n=2693) samples. It would seem that as long as %WL is recorded over a consistent time frame within a given setting, WL during 1, 3, or 6 months should have clinical utility with our grading scheme. Given the broad array of cancer sites and stages, we are unable to exactly account for previous and present cancer treatments in the current survival analysis. It is not clear to what extent WL is specifically cancer associated, as it may also be associated with acute illness, comorbid conditions, or side effect of treatment. We speculate that this simply may not matter, if the WL is analogous to the expenditure of capital (body reserves of energy and protein) and it culminates in metabolic bankruptcy regardless of why it was spent.

2.5 References

- Dewys WD, Begg C, Lavin PT, et al. Prognostic effect of weight loss prior to chemotherapy in cancer patients. Eastern Cooperative Oncology Group. Am J Med, 1980; 69:491-497.
- National Cancer Institute: Common Terminology Criteria for Adverse Events (CTCAE) version 4.0, 2009; https://www.eortc.be/services/doc/ctc/CTCAE_4.03_2010-06-14_QuickReference_5x7.pdf
- 3. Blum D, Omlin A, Baracos VE, et al. Cancer cachexia: A systematic literature review of items and domains associated with involuntary weight loss in cancer. Crit Rev Oncol Hematol, 2011; 80:114-144.
- 4. Spiro A, Baldwin C, Patterson A, et al. The views and practice of oncologists towards nutritional support in patients receiving chemotherapy. Br J Cancer, 2006; 95:431-434.
- 5. Jensen GL, Hsiao PY, Wheeler D. Adult nutrition assessment tutorial. JPEN J Parenter Enteral Nutr, 2012; 36:267-274.
- Nguyen D, El-Serag H. The Epidemiology of Obesity. Gastroenterol Clin North Am, 2010; 39:1-7.
- 7. Tao W, Lagergren J. Clinical management of obese patients with cancer. Nat Rev Clin Oncol, 2013; 10:519-533.
- 8. Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol, 2013; 10:90-99.
- 9. Flegal KM, Kit BK, Orpana H, et al. Association of all-cause mortality with overweight and obesity using standard body mass index categories: a systematic review and meta-analysis. JAMA, 2013; 309:71-82.
- 10. Kalantar-Zadeh K, Horwich TB, Oreopoulos A, et al. Risk factor paradox in wasting diseases. Curr Opin Clin Nutr Metab Care, 2007; 10:433-442.
- 11. Li L, Kalantar-Zadeh K. Obesity that makes kidney cancer more likely but helps fight it more strongly. J Natl Cancer Inst, 2013; 105:1848-1849.
- 12. Fearon K, Strasser F, Anker SD, et al.: Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol, 2010; 12:489-495.

- Aapro M, Arends J, Bozzetti F, et al. Early recognition of malnutrition and cachexia in the cancer patient: a position paper of a European School of Oncology Task Force. Ann Oncol, 2014; 25:1492-1499.
- 14. Martin L, Watanabe S, Fainsinger R, et al. Prognostic factors in patients with advanced cancer: use of the patient-generated subjective global assessment in survival prediction. J Clin Oncol, 2010; 28:4376-4383.
- 15. Martin L, Birdsell L, MacDonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol, 2013; 31:1539-1547.
- Parmar MP, Swanson T, Jagoe RT. Weight changes correlate with alterations in subjective physical function in advanced cancer patients referred to a specialized nutrition and rehabilitation team. Support Care Cancer, 2013; 21:2049-2057
- 17. Thoresen L, Fjeldstad I, Krogstad K, et al. Nutritional status of patients with advanced cancer: the value of using the subjective global assessment of nutritional status as a screening tool. Palliat Med, 2002; 16:33-42.
- 18. Thoresen L, Frykholm G, Lydersen S, et al. The association of nutritional assessment criteria with health-related quality of life in patients with advanced colorectal carcinoma. Eur J Cancer Care (Engl), 2012; 21:505-516.
- 19. Gagnon B, Agulnik JS, Gioulbasanis I, et al. Montreal prognostic score: estimating survival of patients with non-small cell lung cancer using clinical biomarkers. Br J Cancer, 2013; 109:2066-2071.
- 20. Gioulbasanis I, Georgoulias P, Vlachostergios PJ, et al. Mini Nutritional Assessment (MNA) and biochemical markers of cachexia in metastatic lung cancer patients: interrelations and associations with prognosis. Lung Cancer, 2011; 74:516-520.
- 21. Bozzetti F, Mariani L, Lo VS, et al. The nutritional risk in oncology: a study of 1,453 cancer outpatients. Support Care Cancer, 2012; 20:1919-1928.
- 22. Chasen MR, Feldstain A, Gravelle D, et al. An interprofessional palliative care oncology rehabilitation program: effects on function and predictors of program completion. Curr Oncol, 2013; 20:301-309.
- 23. Deans DAC, Tan BH, Wigmore SJ, et al. The influence of systemic inflammation, dietary intake and stage of disease on rate of weight loss in patients with gastro-oesophageal cancer. Br J Cancer, 2009; 100:63-69.

- 24. Strasser F, Lutz TA, Maeder MT, et al. Safety, tolerability and pharmacokinetics of intravenous ghrelin for cancer-related anorexia/cachexia: a randomised, placebo-controlled, double-blind, double-crossover study. Br J Cancer, 2008; 98:300-308.
- Fearon KC, Von Meyenfeldt MF, Moses AG, et al. Effect of a protein and energy dense N-3 fatty acid enriched oral supplement on loss of weight and lean tissue in cancer cachexia: a randomised double blind trial. Gut, 2003; 52:1479-1486.
- 26. Fearon KC, Barber MD, Moses AG, et al. Double-blind, placebo-controlled, randomized study of eicosapentaenoic acid diester in patients with cancer cachexia. J Clin Oncol, 2006; 24:3401-3407.
- 27. Wallengren O, Bosaeus I, Lundholm K. Dietary energy density, inflammation and energy balance in palliative care cancer patients. Clin Nutr, 2013; 32:88-92.
- 28. Haverkort EB, de Haan RJ, Binnekade JM, et al. Self-reporting of height and weight: valid and reliable identification of malnutrition in preoperative patients. Am J Surg, 2012; 203:700-707.
- 29. Perry GS, Byers TE, Mokdad AH, et al. The validity of self-reports of past body weights by U.S. adults. Epidemiology, 1995; 6:61-66.
- Blainey S: Consultation and clinical history-taking skills, in Ranson M, Abbott H, Braithwaite W (eds): Clinical examination skills for health professionals, M&K Update Ltd, 2014, pp 1-19
- 31. Ma C, Bandukwala S, Burman D, et al. Interconversion of three measures of performance status: an empirical analysis. Eur J Cancer, 2010; 46:3175-3183.
- Harrell FE, Jr., Lee KL, Mark DB. Multivariable prognostic models: issues in developing models, evaluating assumptions and adequacy, and measuring and reducing errors. Stat Med, 1996; 15:361-387.
- Pencina MJ, D'Agostino RB. Overall C as a measure of discrimination in survival analysis: model specific population value and confidence interval estimation. Stat Med, 2004; 23:2109-2123.
- 34. Kubrak C, Olson K, Jha N, et al. Clinical determinants of weight loss in patients receiving radiation and chemoirradiation for head and neck cancer: a prospective longitudinal view. Head Neck, 2013; 35:695-703.

35. Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumors of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol, 2008; 9:629-635.

	Training Samples											Validation Sample	
Ref	13	14, UP	15	21	18,19	24,25	22	26	20	23	16,17	UP	UP
Country	Canada ^{1a}	Canada ^{1b}	Canada ²	Canada ³	Greece	Scotland	UK	Sweden	Italy	Switzer- land	Norway	France	France
Data Source	Nutrition risk screening	Nutrition risk screenin g	Base-line data, clinical study	Nutrition risk screening	Base- line data, clinical study	Base-line data, RCT	Base- line data, clinical study	Base- line data, RCT	Base- line data, clinical study	Base-line data, clinical study	Base-line data, clinical study	Base-line data, clinical study	Nutrition risk screening
Health Care Setting*	Support/ Palliat care	Med oncol	Med oncol	Support/ Palliat care	Med oncol	Support/ Palliat care; Med oncol	Surgica l oncol	Support / Palliat care	Support / Palliat care	Support/ Palliat care	Support/ Palliat care	Med oncol	Med oncol
Total N	1631	3815	529	151	738	706	215	486	180	112	92	82	3075
Exclude	133	274	56	18	39	4	0	32	3	7	0	11	382

Table 2-1. Descriptions of data contributed to the training and validation samples.

N [†]													
Include N	1498	3541	473	133	699	702	215	454	177	105	92	71	2693
Weight History	Prior 1,6 months	Prior 1,6 months	Prior 2,6 months	Prior 6 months	Prior 3 months	UBW	UBW	UBW	Prior 6 months	Prior 6 months	Prior 3 months	Prior 6 months	Prior 1, 6 months & UBW

Med Oncol, medical oncology; Palliat, palliative; RCT, Randomized Clinical trial; Support, supportive; UK, United Kingdom; UBW usual body weight; UP, unpublished; *Description of the treatments delivered in the health care setting to which patients were referred: **Supportive/Palliative Care:** these samples represent patients referred for multidisciplinary pain and symptom management in routine clinical practice. This setting includes patients with metastatic disease who were receiving treatment with palliative intent and/or no longer receiving active treatment. **Medical Oncology:** these samples represent patients referred for chemotherapy treatment in routine clinical practice, chemotherapy was administered according to cancer site, stage, and performance status. This setting includes patients newly diagnosed or with recurrent disease. Samples included patients who were treatment naïve and who had received prior surgery, chemotherapy, and/or radiation.

[†]Exclusions: <18 years of age, cancer in situ, missing weight loss and BMI, missing survival information, weight gain (>2.4%)

	Tr	aining Sam (N=8160)	ple	Validation Sample			
		(10 0100)			(N=2693)		
Continuous Variables	Ν	Mean	SD	N	Mean	SD	
Age (years)	8160	65.3	11.8	2693	61.3	12.7	
Weight (kg)	7848	69.6	16.9	2693	65.9	14.6	
Height (m)	7532	1.69	0.1	2690	1.67	0.09	
BMI (kg/m ²)	8160	24.4	5.1	2690	23.4	4.6	
% weight loss*	8138	-9.7	8.4	2693	-7.0	6.7	
Categorical Variables	Ν	Perc	ent	Ν	Per	cent	
Sex							
male	4949	60.	6	1367	50	0.7	
female	3211	39.4		1326	49.2		
Cancer Site							
colorectal	1395	17.	1	300	1	1.1	
breast	227	2.8	3	453	10	16.8	
gastro-esophageal	947	11.6		222	8.2		
genitourinary	300	3.7	7	544	20.2		
head & neck	997	12.	2	308	1	1.4	
other cancers	285	3.5	5	339	12	2.6	
other gastrointestinal	207	2.5	5	27	1	.0	
pancreas	831	10.	2	162	6	0.0	
respiratory	2561	31.	4	234	8	.7	
unknown primary	121	1.5	5	1	0	0.0	
hematological	148	1.8	3	54	2	.0	
liver & intrahepatic bile ducts	141	1.7	7	49	1	.8	
Cancer Stage							
stage I	279	3.4	ŀ	77		3	

Table 2-2. Characteristics of training and validation samples

stage II	555	6.8	127	4.9
stage III	1274	15.7	221	8.5
stage IV	6010	74.0	2173	83.6
ECOG Performance Status				
score 0	1234	17.6	571	21.2
score 1	2560	36.5	899	33.4
score 2	1551	22.1	767	28.5
score 3	1494	21.3	434	16.1
score 4	176	2.5	18	0.7
WHO BMI Categories				
<18.5	817	10.0	320	11.9
18.5-24.9	3974	48.7	1504	55.8
25.0-29.9	2325	28.5	656	24.4
≥30.0	1044	12.8	210	7.8
Weight change				
weight stable (±2.4%)	1847	22.6	808	30.0
weight loss (>-2.4%)	6290	77.1	1885	70.0

BMI, body mass index; ECOG, Eastern Cooperative Oncology Group; N, number; SD, standard deviation; WHO, World Health Organization

*% weight loss based on weight reported in previous 6 months, if missing, the next the longest time frame for reported %WL was substituted where available i.e. UBW, 3, 2, or 1 months respectively: %WL = [(current weight-previous weight (kg))/previous weight (kg)]x100%

 Table 2-3. Median survival and hazard ratios with 95% confidence intervals for proportional hazard model assessing the effect of variables associated with survival.

	N. 1	Median Survival	Univariate	e	Multivaria	te	Multivariate		
Variables	Number of Deaths	(months, 95% CI)	HR (95% CI)	P- value	HR (95% CI)	P-value	HR (95% CI)	P- Value	
% weight loss (continuous)*	6279/8137	8.2 (7.9-8.6)	0.96 (0.96-0.96)	< 0.001	0.98 (0.98-0.99)	< 0.001	0.98 (0.97-0.99)	< 0.001	
BMI	6294/8160	8.2 (7.9-8.6	0.95 (0.95-0.96)	< 0.001	0.99 (0.99-1.00)	.010	0.97 (0.96-0.99)	< 0.001	
(kg/m ² , continuous)									
Age	6294/8160	8.2 (7.9-8.6)	1.01 (1.01-1.01)	< 0.001	1.01 (1.01-1.01)	< 0.001	1.01 (1.01-1.02)	< 0.001	
(years, continuous)									
Sex									
Male	3768/4949	8.5 (8.1-9.0)	1.00		1.00		1.00		
Female	2526/3211	7.7 (7.2-8.2)	1.08 (1.02-1.13)	.004	0.96 (0.90-1.01)	.133	0.86 (0.77-0.96)	0.008	
Cancer Site				< 0.001		< 0.001		< 0.001	
colorectal	903/1395	16.3 (14.6-18.1)	1.00		1.00		1.00		
breast	194/227	7.5 (5.8-9.1)	1.85 (1.59-2.17)	< 0.001	1.24 (1.04-1.47)	.014	1.18 (0.96-1.46)	0.123	
gastro-esophageal	805/948	6.8 (6.1-7.5)	1.82 (1.66-2.00)	< 0.001	1.88 (1.70-2.10)	< 0.001	1.62 (1.29-2.05)	< 0.001	
genitourinary	275/300	4.0 (3.3-4.7)	2.83 (2.47-3.24)	< 0.001	1.52 (1.30-1.76)	< 0.001	1.13 (0.94-1.37)	0.199	
head & neck	455/997	64.3 (48.1-80.5)	0.45 (0.40-0.51)	< 0.001	0.58 (0.51-0.66)	< 0.001	0.89 (0.71-1.12)	0.321	

other cancers	240/276	6.0 (4.5-7.4)	1.96 (1.70-2.26)	< 0.001	1.28 (1.10-1.49)	.001	1.32 (1.07-1.64)	0.01
other	192/215	4.2 (3.3-5.1)	2.49 (2.13-2.91)	< 0.001	1.81 (1.54-2.13)	< 0.001	1.09 (0.65-1.82)	0.75
gastrointestinal								
pancreas	731/831	4.3 (3.8-4.8)	2.79 (2.53-3.09)	< 0.001	1.98 (1.75-2.25)	< 0.001	1.91 (1.51-2.41)	< 0.001
respiratory	2144/2561	6.9 (6.4-7.3)	1.96 (1.81-2.12)	< 0.001	1.80 (1.65-1.96)	< 0.001	1.47 (1.18-1.83)	0.001
unknown primary	114/121	4.0 (3.1-4.9)	2.67 (2.20-3.25)	< 0.001	1.67 (1.31-2.13)	< 0.001	7.29 (1.01-52.4)	0.048
hematological	107/148	6.1 (3.8-8.4)	1.56 (1.28-1.91)	< 0.001	0.95 (0.77-1.18)	.647	0.46 (0.28-0.74)	0.001
liver &	134/141	4.8 (3.5-6.1)	3.01 (2.51-3.61)	< 0.001	2.40 (1.85-3.11)	< 0.001	1.75 (1.20-2.56)	0.004
intrahepatic bile								
ducts								
Cancer Stage				< 0.001		< 0.001		<0.001
Ι	84/279	72.0 [†] (66.3-77.7)	1.00	< 0.001	1.00		1.00	
II	231/555	45.6 (39.1-52.1)	1.73 (1.35-2.22)	< 0.001	1.16 (0.89-1.51)	.277	1.30 (0.601-2.82)	0.504
III	860/1274	16.2 (14.4-18.0)	3.29 (2.63-4.11)	< 0.001	1.76 (1.40-2.22)	< 0.001	2.43 (1.21-4.86)	0.012
IV	5093/6010	5.9 (5.6-6.2)	6.70 (5.39-8.32)	< 0.001	3.95 (3.17-4.93)	< 0.001	6.98 (3.61-13.49)	<0.001
ECOG PS				< 0.001		< 0.001		<0.001
score 0	636/1234	33.3 (29.6-37.0)	1.00	< 0.001	1.00		1.00	
score 1	1790/2560	13.2 (12.4-14.1)	1.86 (1.70-2.04)	< 0.001	1.29 (1.17-1.41)	< 0.001	1.66 (1.41-1.97)	< 0.001
score 2	1276/1551	5.4 (4.9-5.9)	3.50 (3.17-3.85)	< 0.001	1.92 (1.73-2.13)	< 0.001	2.44 (2.06-2.90)	< 0.001
score 3	1369/1494	3.2 (2.9-3.4)	5.17 (4.70-5.69)	< 0.001	2.92 (2.63-3.24)	< 0.001	4.24 (3.51-5.14)	< 0.001

score 4	167/176	1.3 (0.9-1.7)	8.44 (7.11-10.02)	< 0.001	5.05 (4.23-6.05)	< 0.001	6.48 (3.91-10.74)	< 0.001

BMI, body mass index; CI, confidence interval; ECOG PS, Eastern Cooperative Oncology Group Performance Status; HR, hazard ratio

Table 2-4. Sub-group analysis: median survival by grade, and overall p-values from cox proportional hazards models assessing the prognostic significant of the BMI adjusted WL grading system.

	BMI adjuste	Prognostic significance of BMI adjusted WL grades stratified by covariate					
	Grade 0	Grade 1	Grade 2	Grade 3	Grade 4	Overall Uni- variate P- Value	Overall Multi- variate P-Value‡
Training Sample							
Overall							
N deaths	696/1132	678/973	858/1189	1853/2326	2195/2518		
Median Survival (months)	20.9	14.6	10.8	7.6	4.3	< 0.001	< 0.001
95% CI	(17.9-23.9)	(12.9-16.2)	(9.7-11.9)	(7.0-8.2)	(4.1-4.6)		

Subgroup analysis stratified by covariate

Cancer Site							
Colorectal							
N deaths	95/170	97/167	146/251	297/472	268/335		
Median Survival (months)	28.3	22.8	21.6	17.3	7.3	< 0.001	0.002
95% CI	(18.9-37.6)	(17.7-27.9)	(17.2-26.0)	(13.8-20.8)	(5.5-9.2)		
Gastro- esophageal							
N deaths	63/78	63/81	67/83	243/288	368/417		
Median Survival (months)	18.4	12.8	10.3	7.6	4.4	< 0.001	0.003
95% CI	(9.8-27.0)	(10.7-14.9)	(7.8-12.9)	(6.1-9.0)	(3.8-5.0)		
Head & Neck							
N deaths	85/304	73/195	66/155	94/184	137/155		
Median Survival (months)	77.9*	66.4*	67.0	36.2	6.1	< 0.001	< 0.001
95% CI	(73.0-82.8)	(59.9-72-9)	(45.6-88.4)	(6.5-66.0)	(4.7-7.6)		
Respiratory tract							
N deaths	296/389	273/330	350/434	620/709	592/684		
Median Survival (months)	11.3	9.9	8.2	5.6	4.2	< 0.001	< 0.001
95% CI	(9.7-12.9)	(7.9-11.9)	(7.1-9.3)	(4.9-6.3)	(3.5-4.9)		
Cancer Stage							
$I + II_{\downarrow}$							
N deaths	70/217	52/165	36/123	76/200	81/127		

Median Survival (months)	70.2*	67.3*	57.6*	45.9	13.0	< 0.001	0.047
95% Cl	(63.9-76.6)	(60.2-74.5)	(50.6-64.6)	(23.2-68.6)	(8.8-17.3)		
N deaths	99/185	105/176	114/199	248/373	293/340		
Median Survival	30.4	26.9	24.5	17.1	7.3	< 0.001	< 0.001
95% CI	(19.8-41.0)	(20.8-33.1)	(17.4-31.7)	(13.4-20.7)	(6.1-8.5)		
IV							
N deaths	525/722	514/620	702/860	1522/1743	1817/2047		
Median Survival (months)	12.2	9.1	7.6	5.6	3.8	< 0.001	< 0.001
95% CI	(10.6-13.7)	(8.0-10.2)	(6.7-8.4)	(5.1-6.0)	(3.6-4.1)		
Age							
<65							
N deaths	268/526	262/431	346/523	766/999	996/1144		
Median Survival	30.9	19.5	12.6	8.7	4.7	< 0.001	< 0.001
(months) 95% CI	(24.5-37.4)	(16.2-22.9)	(10.5-14.6)	(7.8-9.7)	(4.2-5.1)		
≥65							
N deaths	428/606	416/542	512/666	1087/1327	1199/1374		
Median Survival	15.2	12.0	9.3	6.6	4.1	< 0.001	< 0.001
(months) 95% CI	(12.7-17.8)	(10.1-13.9)	(8.1-10.6)	(5.9-7.3)	(3.8-4.5)		
Performance							
Status							
2)							
N deaths	510/912	461-732	562/866	1073/1481	1085/1336		
Median Survival	27.8	19.7	15.0	10.7	6.0	< 0.001	< 0.001
(months) 95% CI	(24.0-31.7)	(17.3-22.0)	(13.1-17.0)	(9.7-11.7)	(5.5-6.5)		
poor (ECOG 3-4)							
N deaths	116/133	109/124	175/188	465/513	670/710		
Median Survival	6.1	4.4	3.5	2.9	2.6	< 0.001	< 0.001
(months) 95% CI	(3.6-8.6)	(3.2-5.7)	(2.8-4.1)	(2.4-3.4)	(2.3-2.9)		
Health Care Setting							
Medical							
Oncology N deaths	529/949	475/756	592/902	1115/1544	1116/1384		

Median Survival (months)	28.3	19.8	14.9	11.0	6.4	< 0.001	< 0.001
95% CI	(24.2-	(17.5-	(13.1-	(10.0-	(5.8-6.9)		
	32.3)	22.2)	16.8)	12.0)			
Supportive/Palliative Care							
N deaths	167/183	203/217	266/287	738/782	1079/113		
					4		
Median Survival (months)	5.8	5.0	3.9	3.7	3.2	< 0.001	0.002
95% CI	(4.0-7.6)	(4.1-5.9)	(3.2-4.6)	(3.3-4.1)	(2.9-3.4)		
Validation Sample							
Overall							
N deaths	166/353	245/430	260/450	568/842	474/615		
Median Survival (months)	25.1	15.7	17.6	11.5	6.9	< 0.001	< 0.001
95% CI	(19.0-	(12.5-	(13.7-	(9.9-13.2)	(5.9-8.0)		
	31.3)	18.9)	21.4)				

BMI, body mass index; CI, confidence interval; ECOG, Eastern Cooperative Oncology Group; WL, weight loss *median survival not reached, estimated mean overall survival is reported

[†]Stage I and II were combined as they represent patients earlier in the cancer trajectory

‡Multivariate P-values for the prognostic significance of the BMI adjusted WL grades are adjusted for covariates including: age, sex, cancer site,

cancer stage, and ECOG performance status

CHAPTER 3: Diagnostic Criteria for Cancer Cachexia: Data versus Dogma¹

3.1 Key Points

- There are notable disparities in diagnostic criteria for cancer cachexia, and there are no agreed upon diagnostic criteria.
- Through international consensus, a cancer cachexia framework was proposed to address disparities in diagnostic criteria. Candidate diagnostic criteria for cancer cachexia include: weight loss, muscle mass, reductions in food intake, and alterations in metabolism.
- Definitive diagnostic criteria for cancer cachexia can be developed with large contemporary data sets that include variables related to cancer cachexia. Clearly defined statistical methods will identify candidate diagnostic criteria with the best utility.
- A new grading system for cancer-associated weight loss has been proposed confirming the concept that severity of weight loss should be classified according to the degree of weight loss and current body weight.
- Further levels of refinement for diagnostic criteria include advancements related to skeletal muscle depletion (and other features of body composition), and causes of cancer cachexia including reduced food intake, and alterations in metabolism. Development in these areas will enrich cancer cachexia classification systems and improve their utility.

3.2 Cachexia has been known since ancient times but lacks agreed upon diagnostic criteria

Historical descriptions of cachexia as a wasting syndrome date back to Hippocrates, and are full of rich observations about dramatic weight loss, the experience of anorexia, weakness and fatigue that eventually lead to death. Unintended weight loss (WL) has always been a key clinical sign, and often triggers investigations for malignancy as a possible explanation. Often cachexia is defined by a single criterion based on its cardinal clinical feature, WL, which is typically classified as being above a certain threshold (>5%, >10%).¹⁻⁴ There has been a

¹A version of this chapter has been published: Martin L. Diagnostic criteria for cancer cachexia: Data versus dogma. Curr Opin Nutr Metab Care, 2016; 19:188-198.

renaissance in cancer cachexia research over the last decade, with support from a Society on Cachexia and Wasting Disorders, a series of international conferences, and an expanding research community. There have been advancements in basic science, clinical research, and in the identification of therapeutic targets for cachexia therapies. Advancements have improved our understanding of the pathophysiology of cachexia, also termed disease-associated malnutrition, now invariably described as a multifactorial syndrome of ongoing loss of muscle (with or without fat), driven by variable impairments of food intake and metabolic alterations that elicit excess catabolism (e.g. inflammation) (Table 3-1).¹⁻⁵ These metabolic changes explain why cachexia cannot be reversed by conventional nutritional support alone, and are the key difference between cachexia and starvation-related malnutrition.^{1,4,5}

Criteria to define cachexia, in addition to WL, now include variable combinations of low body mass index (BMI), skeletal muscle mass, reduced dietary intake, and biological indicators of inflammation as markers of metabolic change (Table 3-1). Measurement of these dimensions is increasingly specific, precise, clinically available, and can be related to diverse clinically relevant outcomes including risk of chemotherapy toxicity, surgical complications, functional impairment, patient-family psychological distress, attrition from clinical trials, and mortality.^{1,4}

This new knowledge has been variably understood and translated producing an abundance of clinical tests and assessments. As a result, there is a heterogeneous mix of available diagnostic criteria for cancer cachexia which have been applied in a dogmatic fashion.⁶ Further to the possible combinations of diagnostic criteria, is an added layer of heterogeneity within each individual criterion (Table 3-2). For example, WL can be collected over various time frames and defined according to different thresholds. Overall, there are not widely agreed upon diagnostic criteria for cancer cachexia, and disparities among the existing ones are notable in

published research, clinical trials, clinical assessments, and practice guidelines.^{3,4,7-11} A main challenge is to identify the most robust diagnostic criteria among a sea of credible candidates.

3.3 Progress toward consensus diagnostic criteria for cancer cachexia

Several authors and consensus groups flagged disparities among cachexia diagnostic criteria as a detriment to the identification and treatment of cachectic patients, and for the development of therapeutic agents.^{1,4,9,10} International consensus groups^{1,4} have addressed these disparities, and proposed frameworks for the general classification of cachexia associated with chronic disease¹, and specifically for cancer cachexia.⁴ This review focuses on the cancer cachexia framework⁴ where agreement was reached on several main themes: an etiology-based definition for cancer cachexia, the concept of cachexia as a continuum with three stages of clinical relevance, the presentation of provisional diagnostic criteria, and recommendations for assessments to describe cancer cachexia.

The cancer cachexia framework⁴ proposes a series of next steps, notably a call for data resulting in a data-driven approach for the development of a robust set of diagnostic criteria, *"definitive cutoffs for variables (e.g. weight loss, low muscle mass) could be determined from large contemporary datasets by determining the values that relate optimally to meaningful patient-centered outcomes, such as loss of function or decreased survival… the criteria suggested in the present consensus remain arbitrary".*

With this task is underway, several authors^{7,9-12} adopted the provisional diagnostic criteria⁴ and compared classification of cachectic patients to other definitions for cachexia.^{1,3} Not surprisingly different classifications yield different results, stratification of patients according to cachexia stage improved with combinations of criteria (e.g. WL + reduced food intake, WL + inflammation), but development of more clearly defined diagnostic criteria is necessary.

Deployment of the right diagnostic criteria and their combinations may improve identification of cachexia patient subsets to guide clinical decisions and therapeutic research. For example, identification of subsets of cachectic patients with poor prognosis unlikely to live long enough to receive benefit from cancer therapies or withdraw from clinical trials, but who may be candidates for supportive therapies such as nutrition support.¹³

3.4 Methods and principles for the development of diagnostic criteria

The cancer cachexia framework⁴ provides an agreed upon list of potential candidates to classify cancer cachexia (e.g. WL, skeletal muscle (SM) depletion, reduced food intake, metabolic alterations). The first set of candidate diagnostic criteria includes descriptions of the clinical manifestation of cachexia including WL and SM depletion. The next set includes etiologically based criteria that explain WL and SM depletion including reductions in food intake and alterations in metabolism. Candidate diagnostic criteria must be clinically relevant, practical, and readily available. To define definitive diagnostic criteria, data sets of similarly collected variables are necessary and should be large enough to capture representative distributions of candidate diagnostic criteria, important covariates, and outcomes for adequately powered statistical analyses, including subset analyses. Inclusion of contemporary cancer patients ensures representation from a variety of cancer populations, body weight demographics, and treatment plans. Survival is a key outcome, clinically relevant in oncology and is unambiguous in its collection and interpretation. However, other outcomes must be considered including treatment complications, functional impairment, quality of life, and attrition from clinical trials.

A typical analytical approach applies clearly defined statistical methods (i.e. regression analysis) to examine the distribution of candidate diagnostic criterion and the relationship to relevant outcomes (Figure 3-1A). This approach encourages examination of the shape of the

46

distribution and changes in risk across a variable (e.g. linear vs. non-linear) to generate risk stratifications and optimal values linked to an outcome, which can then be adjusted for other covariates. New values for diagnostic criteria and/or their combinations require independent validation for proof of concept and applicability in different populations. This review will discuss recent work on data driven approaches to define diagnostic criteria for WL, SM depletion, reduced food intake, and altered metabolism.

3.5 Re-defining weight loss as a diagnostic criterion for cancer cachexia

The cancer cachexia framework⁴ suggested that "severity should be graded according to degree of weight loss and concurrent BMI, and the severity classification should be developed around the predictive value for outcomes such as treatment toxicity, quality of life, hospitalization, and survival"



Figure 3-1. The relationship between percent weight loss and overall survival

a) A line graph illustrating the shape of the relationship between % weight loss, divided in to deciles, and overall survival in a sample of cancer patients (N=8138).¹⁴ The black arrow represents the cutoff for a bivariate classification for weight loss above 5%⁴, and the gray dotted arrows represent the data-derived cutoffs for five categories of weight loss wherein risk of reduced survival significantly (P<0.001) increases for each category. (b) Cumulative survival curve (Kaplan-Meier Method, log rank tests) representing the bivariate classification for weight loss.¹⁴ (c) Cumulative survival curves for weight loss grades (0–4) defined by combining five categories of weight loss (Fig. 1a) and five categories of BMI (not shown).¹⁴ The weight loss grades add several levels of stratification when compared with a bivariate classification for weight loss.

The conventional, but arbitrary provisional diagnostic criterion⁴ for cancer cachexia is unintentional WL of >5% in the previous 6 months. This bivariate classification is recognized to be an oversimplification, it assumes an equal risk for all degrees of WL >5% that disregards all notions of severity (Figure 3-1A, B). WL is actually understood with a greater level of sophistication; increased WL equals increased risk of poor outcome (Figure 3-1A).¹⁴ A concept first proposed by De Wys et al.¹⁵ and represented in the Common Terminology Criteria for Adverse Events $(CTCAE v4)^2$, a system to grade the severity of an adverse event in oncology, including three levels of clinically relevant WL (Table 3-1). Additionally, 7 classes of BMI with clinical relevance are accepted. The importance of WL is not the same across the distribution of BMI; people with little energy and protein reserves to begin with, experience a greater impact of WL. These concepts were confirmed by Martin et al.¹⁴, who compiled a large data set of similarly collected variables on >11,000 cancer patients from Europe and Canada, which was used to develop and validate a new grading system for cancer-associated WL based on a risk stratification with survival as the outcome. WL and BMI independently predicted overall survival, and when considered together provide excellent stratification of patients with disparate survivals, confirming that severity of WL should be evaluated based on the rate of WL and current BMI (Figure 3-1C). The WL grading system remains to be validated according to cancer type, treatment setting, and for other outcomes such as treatment complications or attrition from clinical trials. Next levels of refinement include diagnostic criteria for SM, food intake, and altered metabolism.

3.6 Skeletal muscle is an emerging diagnostic criterion for cancer cachexia

The cancer cachexia framework⁴ specifies that "cancer cachexia is characterized by an ongoing loss of skeletal muscle mass (with or without loss of fat mass)..." and includes SM

depletion, in addition to WL, as a provisional diagnostic criterion.⁴ There is no consensus regarding methodology for assessment of muscle mass, however preference was given to computed tomography cross-sectional images taken at the L3 region (CT L3).⁴ CT images are completed as standard of care in oncology, are routinely available, and offer an unprecedented window to non-invasively and precisely quantify SM. When CT images are unavailable another modality should be considered (e.g. DEXA).

The rapidity with which CT analysis has been adopted for the quantification of SM in oncology made it the topic of several reviews in 2015.¹⁶⁻¹⁹ The most extensive review by Kazemi-Bejestani et al.¹⁷ included 53 studies (from 2000 to Jan.2015) on a total of 9,138 patients. The authors concluded there is a consistent association between SM depletion and outcomes including chemotherapy response and toxicity, increased post-surgical complications and length of hospital stay, and mortality. Quantification of other tissues (e.g. fat) is an evolving area of research, but not sufficiently refined to be included as diagnostic criterion at this time. From Jan. 2015-Nov. 2015, an additional 19 publications on 5,124 patients were evaluated for SM using CT analysis (Table 3-3). These studies²⁰⁻³⁷ corroborate the concepts highlighted by Kazemi-Bejestani et al.¹⁷ including associations between SM depletion and increased chemotherapy-related toxicity²⁰ and post-surgical complications²¹⁻²⁴, and decreased survival²⁵⁻²⁸. Treatment induced changes in SM is an emerging theme, and recent work highlights significant SM loss during the course of chemotherapy^{29.34}, which associated with reduced survial in four studies.³⁰⁻³³

Up for debate is the application of pre-defined cutoffs for SM depletion in patient populations with different demographics (e.g. race, obesity rates, muscularity, tumor types). The provisional diagnostic criteria⁴ includes sex-specific cutoffs for CT L3 defined SM depletion

49

(skeletal muscle index (SMI) Men: $<52.4 \text{ cm}^2/\text{m}^2$; SMI Women: $<38.5 \text{ cm}^2/\text{m}^2$) associated with reduced survival.^{4,38} These cut offs were derived from a Caucasian sample of obese (BMI \ge 30.0, N=250) cancer patients, and may have limited relevance when applied to other populations. Development of appropriate reference populations and data derived cutoffs specific to sex, age, and race are clearly needed to advance the concepts of SM depletion.^{4,17}

Martin et al.³⁹ completed an analysis including all BMI classes in a large sample (N=1473) of Caucasian cancer patients. BMI and sex specific cutoffs for SM depletion (SMI Men: <43 cm2/m2 (BMI<25.0), <53 cm2/m2 (BMI \geq 25.0); Women: <41 cm2/m2) were defined in relation to survival, demonstrating SM differs by sex and body weight. More recent studies include analyses to define cutoffs for SM depletion specific to their patient populations (Table 3-3).^{21,25,30,32} Fujiwara et al.²⁵ offer an excellent example in a large (N=1257) sample of Japanese hepatocellular carcinoma patients. Sex-specific cutoffs for skeletal muscle depletion (SMI Men: \leq 36 cm²/m²; SMI Women: \leq 30 cm²/m²) were defined based on survival. Notably, these cutoffs are lower than those derived from a Caucasian population, confirming careful consideration should be given to the choice of cutoffs applied to a given population (Figure 3-2).

Figure 3-2: Distributions for skeletal muscle index (cm²/m²), evaluated by CT image analysis, for Caucasian and Japanese cancer patients



Normal distributions for skeletal muscle index (cm²/m²), evaluated by CT image analysis, for Caucasian (gray dotted lines)³⁹ and Japanese (black solid lines) cancer patients.²⁵ Also depicted are the sex-specific cutoff values derived from Caucasian (vertical grey dotted lines) and Japanese (vertical black solid lines) patient populations.^{4,25}

CT analysis provides a plethora of prognostic information, and these findings are of such importance they should be accepted as a standard oncology biomarker.⁴⁰ It is acknowledged that SM, which is the focus of this section, is not the only information of importance. CT analyses reveal the highly dimensional nature of body composition, including different tissue amounts and distributions (e.g. total, subcutaneous, and visceral fat), and tissue specific radiation attenuation values. For SM in particular, low attenuation values indicate fatty infiltration to the muscle. All of these dimensions can potentially relate to clinical outcomes as well as functionality of muscle.^{22,24,25,27,28,30,33-35,38} Other areas for development include evaluating the interplay between

SM (and other tissues) with candidate diagnostic criteria including food intake, and altered metabolism.

3.7 Reduced food intake is a candidate diagnostic criterion in development

The cancer cachexia framework⁴ specifies two factors responsible for the development of cancer cachexia, which *"is driven by a variable combination of reduced food intake and abnormal metabolism."* Characterization of these features is necessary to identify patients with cachexia. The assessment of food intake or anorexia is recommended, but there is no consensus regarding measurements of food intake, and no provisional diagnostic criteria were presented.⁴ However, reduced food intake or anorexia have been included as diagnostic criteria in other definitions of cachexia^{1,3} (Table 3-1), and when combined with other diagnostic criteria (e.g. WL, inflammation) demonstrate improved identification of cachexic patient subsets.^{3,7,9,11}

A notable limitation in defining diagnostic criteria for food intake and anorexia is the heterogeneity of available assessments (Table 3-2). It is unknown which assessments are most associated with the features defining cachexia such as WL, SM depletion, and altered metabolism, or how they relate optimally to clinical outcomes. Recent work highlights the clinical utility of food intake measures within existing assessment tools commonly collected in cachexia research⁴¹⁻⁴³, and Solheim et al.⁴³ point out that assessments of anorexia and food intake are both necessary to fully describe the food intake of cachectic patients due to conscious control over appetite loss. Linkages between actual food intake and clinical assessments remain to be evaluated and may offer validation to various classifications of food intake such energy intake <1500 kcal³, and commonly used categorical estimations of food intake (e.g. normal, less than normal, etc). As well, capturing actual food intake inclusive of macronutrients (energy, protein, fat) and micronutrients (vitamins, minerals) may allow for the identification of specific nutrient
deficiencies associated with cancer cachexia. Refining food intake as a diagnostic criterion will be enabled by the emergence of rich data sets to identify assessments with the best utility.

3.8 Biologic Aberrations (i.e. inflammation) in Cachexia - How do we factor them in to cachexia classifications?

To date classification work has centered on the effects of cachexia (e.g. WL, SM depletion) and not causation. Current oncology research, is marrying genetic and other biologic information with traditional clinical findings to determine patient subsets which may respond to tailored cancer therapies. A similar approach is needed to enable us to establish cachexia patient subsets to guide therapeutic research.

The cancer cachexia framework⁴ describes altered metabolism as a variable combination of factors (e.g. inflammation, tumor metabolism, tumor mediated effects) leading to a hypercatabolic state resulting in cachexia. There is no consensus regarding measurements for altered metabolism. However, inflammation an important driver of tumor growth and metastasis has also emerged as an important driver of energy imbalance and muscle wasting in cancer cachexia.⁴ C-reactive protein (CRP) is the most widely accepted index of systemic inflammation.⁴ Inclusion of other routinely available clinical markers of inflammation (e.g. neutrophils and lymphocytes) also predict the effect of inflammation on cancer progress and survival. The most accepted scale to characterize systemic inflammation is the Glasgow Prognostic Score (GPS). McMillan⁴⁴ together with other investigators^{45,46} have clearly shown the devastating effect of systemic inflammation on survival and development of cancer-related symptoms.⁴⁷ Adding systemic inflammation to traditional clinical findings offers improved identification of patient subsets with worse prognosis and unlikely to benefit from treatment.^{7,9,13,46} Efforts to understand systemic inflammation as a cause of SM depletion in cancer cachexia are emerging (Table 3-3). Two studies^{36,48} showed systemic inflammation to be

an independent predictor of SM depletion, as well as a contributor to accelerated muscle loss during treatment.⁴⁸

Measures of systemic inflammation are clinically available, have been collected on a large number of patients, and it is possible to compile this data to explore their interplay with other features of cachexia. Subsequent biologic information such as evidence of altered autonomic function, immune and hormonal activity, microbiota, and genetic background may enrich cachexia classification, albeit with increasing complexity. For example, preliminary evidence determined a series of *single nucleotide polymorphisms* (SNPs) either confer susceptibility or resistance to cancer cachexia⁴⁹ Cancer cachexia is complex but it is anticipated that our advancing knowledge of cachexia causation will enrich our classification systems and improve their utility.

To summarize, there are limitations to what is presented here. Cancer cachexia research has many facets, however the focus of this review highlights where we have data to address disparities in diagnostic criteria as outlined in the cancer cachexia framework. Other areas of assessment such as physical function and quality of life are outcomes or consequences of cancer cachexia and continuing areas for development. Data collection efforts towards a large international data repository with an expanded number of clinical variables are ongoing with support from the cancer cachexia consensus group. Of particular interest is the development of prospective treatment related outcomes for cancer therapies and surgery. Ultimately, new values for diagnostic criteria will require validation and adoption into practice.

3.9 Conclusion

There is a consensus definition and framework outlining assessments necessary to diagnose cancer cachexia⁴, and extensive collaboration produced an international data set to enable development and refinement of diagnostic criteria for cancer cachexia, with potential to

influence other spheres of cachexia research. Identification of cachectic patient subsets may lead to better inclusion criteria for clinical trials, decreased attrition rates, improved outcomes, and approval of therapeutics. As well, better identification of patients who are likely to benefit from treatment (aggressive vs. supportive) and earlier deployment of beneficial therapies. Developing diagnostic criteria for cancer cachexia using methodologies similar to those that have been used in other chronic conditions⁵⁰ indicates the potential of a data driven approach to define diagnostic criteria with clinical utility.

3.10 References

- 1. Evans WJ, Morley JE, Argiles J, et al. Cachexia: a new definition. Clin Nutr 2008; 27:793-9.
- National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE) v.4.03, 2009; https://www.eortc.be/services/doc/ctc/CTCAE_4.03_2010-06-14_QuickReference_5x7.pdf
- Fearon KC, Voss AC, Hustead DS. Definition of cancer cachexia: effect of weight loss, reduced food intake, and systemic inflammation on functional status and prognosis. Am J Clin Nutr 2006; 83:1345-50.
- 4. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol 2011; 12:489-95.
- Jensen GL, Mirtallo J, Compher C, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the international consensus guideline committee. JPEN J Parenter Enteral Nutr 2010; 34:156-9.
- 6. Dechaphunkul T, Martin L, Alberda C, Olson K, Baracos V, Gramlich L. Malnutrition assessment in patients with cancers of the head and neck: a call to action and consensus. Crit Rev Oncol Hematol 2013; 88:459-76.
- 7. Blum D, Stene GB, Solheim TS, et al. Validation of the consensus-definition for cancer cachexia and evaluation of a classification model--a study based on data from an international multicentre project (EPCRC-CSA). Ann Oncol 2014; 25:1635-42.
- 8. Garcia JM, Boccia RV, Graham CD, et al. Anamorelin for patients with cancer cachexia: an integrated analysis of two phase 2, randomised, placebo-controlled, double-blind trials. Lancet Oncol 2015; 16:108-16.
- 9. Thoresen L, Frykholm G, Lydersen S, et al. Nutritional status, cachexia and survival in patients with advanced colorectal carcinoma. Different assessment criteria for nutritional status provide unequal results. Clin Nutr 2013; 32:65-72.
- 10. Wallengren O, Lundholm K, Bosaeus I. Diagnostic criteria of cancer cachexia: relation to quality of life, exercise capacity and survival in unselected palliative care patients. Support Care Cancer 2013; 21:1569-77.
- 11. Wesseltoft-Rao N, Hjermstad MJ, Ikdahl T, et al. Comparing two classifications of cancer cachexia and their association with survival in patients with unresected pancreatic cancer. Nutr Cancer 2015; 67:472-80.
- 12. LeBlanc TW, Nipp RD, Rushing CN, et al. Correlation between the international consensus definition of the Cancer Anorexia-Cachexia Syndrome (CACS) and patient-

centered outcomes in advanced non-small cell lung cancer. J Pain Symptom Manage 2015; 49:680-9.

- 13. Bozzetti F, Cotogni P, Lo VS, Pironi L, Giardiello D, Mariani L. Development and validation of a nomogram to predict survival in incurable cachectic cancer patients on home parenteral nutrition. Ann Oncol 2015; 26:2335-40.
- 14. Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancer-associated weight loss. J Clin Oncol 2015; 33:90-9.
- Dewys WD, Begg C, Lavin PT, et al. Prognostic effect of weight loss prior to chemotherapy in cancer patients. Eastern Cooperative Oncology Group. Am J Med 1980; 69:491-7.
- 16. Gibson DJ, Burden ST, Strauss BJ, Todd C, Lal S. The role of computed tomography in evaluating body composition and the influence of reduced muscle mass on clinical outcome in abdominal malignancy: a systematic review. Eur J Clin Nutr 2015; 69:1079-86.
- 17. Kazemi-Bajestani SM, Mazurak VC, Baracos V. Computed tomography-defined muscle and fat wasting are associated with cancer clinical outcomes. Semin Cell Dev Biol 2016; 54:2-10.
- 18. Levolger S, van Vugt JL, de Bruin RW, IJzermans JN. Systematic review of sarcopenia in patients operated on for gastrointestinal and hepatopancreatobiliary malignancies. Br J Surg 2015; 102:1448-58.
- 19. Malietzis G, Aziz O, Bagnall NM, Johns N, Fearon KC, Jenkins JT. The role of body composition evaluation by computerized tomography in determining colorectal cancer treatment outcomes: a systematic review. Eur J Surg Oncol 2015; 41:186-96.
- 20. Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapy-induced haematological toxicity in advanced non-small cell lung cancer. Lung Cancer 2015; 90:85-91.
- 21. Harada K, Ida S, Baba Y, et al. Prognostic and clinical impact of sarcopenia in esophageal squamous cell carcinoma. Dis Esophagus 2016; 29:627-633.
- 22. Huang DD, Wang SL, Zhuang CL, et al. Sarcopenia, as defined by low muscle mass, strength and physical performance, predicts complications after surgery for colorectal cancer. Colorectal Dis 2015; 17:O256-O264.
- 23. Tegels JJ, van Vugt JL, Reisinger KW, et al. Sarcopenia is highly prevalent in patients undergoing surgery for gastric cancer but not associated with worse outcomes. J Surg Oncol 2015; 112:403-7.
- 24. van Vugt JL, Braam HJ, van Oudheusden TR, et al. Skeletal muscle depletion is associated with severe postoperative complications in patients undergoing cytoreductive

surgery with hyperthermic intraperitoneal chemotherapy for peritoneal carcinomatosis of colorectal cancer. Ann Surg Oncol 2015; 22:3625-31.

- 25. Fujiwara N, Nakagawa H, Kudo Y, et al. Sarcopenia, intramuscular fat deposition, and visceral adiposity independently predict the outcomes of hepatocellular carcinoma. J Hepatol 2015; 63:131-40.
- 26. Fukushima H, Nakanishi Y, Kataoka M, Tobisu K, Koga F. Prognostic significance of sarcopenia in patients with metastatic renal cell carcinoma. J Urol 2016; 195:26-32.
- 27. Psutka SP, Boorjian SA, Moynagh MR, Schmit GD, Costello BA, Thompson RH, et al. Decreased skeletal muscle mass is associated with an increased risk of mortality after radical nephrectomy for localized renal cell cancer. J Urol 2016; 195:270-6.
- 28. Tamandl D, Paireder M, Asari R, Baltzer PA, Schoppmann SF, Ba-Ssalamah A. Markers of sarcopenia quantified by computed tomography predict adverse long-term outcome in patients with resected oesophageal or gastro-oesophageal junction cancer. Eur Radiol 2016; 26:1359-67.
- 29. Antoun S, Bayar A, Ileana E, et al. High subcutaneous adipose tissue predicts the prognosis in metastatic castration-resistant prostate cancer patients in post chemotherapy setting. Eur J Cancer 2015; 51:2570-7.
- 30. Choi Y, Oh DY, Kim TY, et al. Skeletal muscle depletion predicts the prognosis of patients with advanced pancreatic cancer undergoing palliative chemotherapy, independent of body mass index. PLoS One 2015; 10:e0139749.
- 31. Cooper AB, Slack R, Fogelman D, et al. Characterization of anthropometric changes that occur during neoadjuvant therapy for potentially resectable pancreatic cancer. Ann Surg Oncol 2015; 22:2416-23.
- 32. Kimura M, Naito T, Kenmotsu H, et al. Prognostic impact of cancer cachexia in patients with advanced non-small cell lung cancer. Support Care Cancer 2015; 23:1699-708.
- 33. Miyamoto Y, Baba Y, Sakamoto Y, et al. Negative impact of skeletal muscle loss after systemic chemotherapy in patients with unresectable colorectal cancer. PLoS One 2015; 10:e0129742.
- 34. Rollins KE, Tewari N, Ackner A, et al. The impact of sarcopenia and myosteatosis on outcomes of unresectable pancreatic cancer or distal cholangiocarcinoma. Clin Nutr 2016;35:1103-9.
- 35. Aust S, Knogler T, Pils D, et al. Skeletal muscle depletion and markers for cancer cachexia are strong prognostic factors in epithelial ovarian cancer. PLoS One 2015; 10:e0140403.

- 36. Malietzis G, Johns N, Al-Hassi HO, et al. Low muscularity and myosteatosis is related to the host systemic inflammatory response in patients undergoing surgery for colorectal cancer. Ann Surg 2016; 263:320-5.
- 37. Reisinger KW, Derikx JP, van Vugt JL, et al. Sarcopenia is associated with an increased inflammatory response to surgery in colorectal cancer. Clin Nutr 2016; 35:924-7.
- 38. Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumors of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol 2008; 9:629-35.
- 39. Martin L, Birdsell L, MacDonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol 2013; 31:1539-47.
- 40. Hubbard JM, Cohen HJ, Muss HB. Incorporating biomarkers into cancer and aging research. J Clin Oncol 2014; 32:2611-6.
- 41. LeBlanc TW, Samsa GP, Wolf SP, Locke SC, Cella DF, Abernethy AP. Validation and real-world assessment of the Functional Assessment of Anorexia-Cachexia Therapy (FAACT) scale in patients with advanced non-small cell lung cancer and the cancer anorexia-cachexia syndrome (CACS). Support Care Cancer 2015; 23:2341-7.
- 42. Salsman JM, Beaumont JL, Wortman K, Yan Y, Friend J, Cella D. Brief versions of the FACIT-fatigue and FAACT subscales for patients with non-small cell lung cancer cachexia. Support Care Cancer 2015; 23:1355-64.
- 43. Solheim TS, Blum D, Fayers PM, et al. Weight loss, appetite loss and food intake in cancer patients with cancer cachexia: three peas in a pod? analysis from a multicenter cross sectional study. Acta Oncol 2014; 53:539-46.
- 44. McMillan DC. The systemic inflammation-based Glasgow Prognostic Score: a decade of experience in patients with cancer. Cancer Treat Rev 2013; 39:534-40.
- 45. Guthrie GJ, Charles KA, Roxburgh CS, Horgan PG, McMillan DC, Clarke SJ. The systemic inflammation-based neutrophil-lymphocyte ratio: experience in patients with cancer. Crit Rev Oncol Hematol 2013; 88:218-30.
- Simmons CP, Koinis F, Fallon MT, et al. Prognosis in advanced lung cancer--A prospective study examining key clinicopathological factors. Lung Cancer 2015; 88:304-9.
- 47. Laird BJ, McMillan DC, Fayers P, et al. The systemic inflammatory response and its relationship to pain and other symptoms in advanced cancer. Oncologist 2013; 18:1050-5.
- 48. Wallengren O, Iresjo BM, Lundholm K, Bosaeus I. Loss of muscle mass in the end of life in patients with advanced cancer. Support Care Cancer 2015; 23:79-86.

- 49. Johns N, Tan BH, MacMillan M, et al. Genetic basis of interindividual susceptibility to cancer cachexia: selection of potential candidate gene polymorphisms for association studies. J Genet 2014; 93:893-916.
- 50. International Expert Committee report on the role of the A1C assay in the diagnosis of diabetes. Diabetes Care 2009; 32:1327-34.

Narrative Definitions		Diagnostic Criteria
Cachexia Associated with	th Chronic Disease	
ICD-10 cachexia (R64)*	A wasting syndrome, associated with an underlying condition e.g. HIV, cancer	None specified
Consensus definition disease-associated cachexia ¹	"A complex metabolic syndrome associated with underlying illness characterized by loss of muscle with or without loss of fat mass prominent clinical feature is weight loss Anorexia, inflammation, insulin resistance and increased muscle protein breakdown frequently associated with wasting disease."	 Weight loss >5 % <u>AND</u> 3 of the following: decreased muscle strength, fatigue, anorexia, low fat-free mass index, abnormal biochemistry (inflammation, low albumin, anemia)
Cancer Cachexia NCI CTCAE v4.0 ²	A decrease in overall body weight	Grade 1: 5 to 10% from baseline; intervention not indicated; Grade 2: 10 to <20% from baseline; nutritional support indicated; Grade 3: ≥20% from baseline; tube feeding or TPN indicated; Grade 4: not defined, life threatening; Grade 5: not defined, fatal
3-factor classification for cancer cachexia ³	"A multifactorial syndrome characterized by ongoing weight loss, low food intake, and presence of systemic inflammation"	At least 2 of the following factors: • weight loss ≥10%, • food intake <1500 kcal/day, • CRP ≥10 mg/L
Consensus Definition Cancer cachexia ⁴	"A multifactorial syndrome defined by an ongoing loss of skeletal muscle mass (with or without loss of fat mass) not fully reversed by conventional nutritional support leading to progressive functional impairment. Its pathophysiology is characterized by a negative protein and energy balance driven by a variable combination of reduced food intake and abnormal metabolism."	Weight loss >5% over past 6 months OR BMI <20 <u>AND</u> any degree of weight loss >2% OR Muscle depletion <u>AND</u> any degree of weight loss >2%

Table 3-1. A summary of narrative definitions for cachexia and diagnostic criteria

*ICD-10, International Classification of Disease, http://apps.who.int/classifications/icd10/browse/2016/en#/R6; HIV, human immunodeficiency virus; NCI CTCAE v4.0, National Cancer Institute Common Terminology Criteria for Adverse Events; CRP, C-reactive protein; BMI, body mass index.

Diagnostic Criteria	Examples of heterogeneity in data collection and in the reporting of diagnostic criteria
Weight Loss	Data collection:
Weight Loss	 clinical tools e.g. SGA, PG-SGA, MNA, MST, MUST, SNAQ, NRS 2002, NCI CTCAE v.4.0 different time frames e.g. WL (actual or %) over previous 1, 2, 3, 6 months, or from
	pre-illness weight
	Reported as:
	• continuous variable e.g. % or amount (kg., lbs.)
	• bivariate categorization e.g. WL present (yes or no), WL > than a defined threshold (>5%, >10%, >3kg, >5kg)
	• multiple categories of increasing severity e.g. Grade 1 (5-10%), Grade 2 (10-20%), Grade 3 (>20%)
Reduced food	Data Collection:
intake	 clinical tools e.g. SGA, PG-SGA, MNA, MUST, SNAQ, NRS 2002, FAACT, VAS (estimated food intake), and food records, food frequency questionnaires (recorded food intake)
	 different time frames e.g. retrospectively (previous day, 1 week, 1 or 3 months) or prospectively (over 1 to 7 days)
	Reported as:
	• continuous variable e.g. kcal energy, g protein, g fat from recorded food intake, or VAS (0-100) for estimated intake
	• bivariate classification e.g. < or > 1500kcal/d from recorded food intake, or estimated
	food intake less than usual (yes or no)
	• multiple categories of increasing severity e.g. estimated food intake (normal, reduced, severely reduced)
Appetite Loss /	Data Collection:
Anorexia	• clinical tools e.g. SGA, PG-SGA, MST, FAACT, ESAS, EORTC QLQ-C30, NCI CTCAE v.4.0
	• different time frames e.g. same day, previous 1-2 weeks or month
	Reported as :
	• continuous variable e.g. VAS (0-100)
	• bivariate categorization e.g. appetite loss/anorexia present (yes or no)
	• multiple categories of increasing severity e.g. Likert scales (0, no loss; 4, severe loss)
Inflammation	Data Collection:
	• biological indicators for systemic inflammation: CRP, albumin, neutrophils, lymphocytes, cytokines
	Reported as:
	• continuous individual parameters e.g. CRP, albumin, neutrophils, lymphocytes, cytokines
	• continuous composite parameters e.g. NLR, PINI
	• bivariate categorization e.g. CRP (>5 or > 10 mg/L), NLR (>3 or >5)
	• multiple categories of increasing severity e.g. GPS, mGPS, PINI

Table 3-2. A summary of candidate diagnostic criteria⁴ and the heterogeneity that exists within each criterion.

SGA, Subjective Global Assessment; PG-SGA, Patient Generated-Subjective Global Assessment; MNA, Mini-Nutritional Assessment; MST, Malnutrition Screening Tool; MUST, Malnutrition Universal Screening Tool; SNAQ, Short Nutritional Assessment Questionnaire; NRS 2002, Nutrition Risk Screening; NCI CTCAE, National Cancer Institute Common Terminology Criteria for Adverse Events; FAACT, Functional Assessment of Anorexia Cachexia Therapy; VAS, visual analogue scale; ESAS, Edmonton Symptom Assessment Scale; EORTC QLQ-C30, European Organization for Research and Treatment of Cancer Quality of Life Questionnaire Core-30; CRP, C-reactive protein; PINI, prognostic inflammatory and nutritional index; NLR, neutrophil to lymphocyte ratio; GPS, Glasgow Prognostic Score; mGPS, modified Glasgow Prognostic Score. *This is not intended to be an exhaustive list of all possible assessments and biological indicators.

Table 3-3. A Summary of results from 19 articles (n=5,124 patients) published between Jan. 2015 to Nov. 2015 evaluating skeletal muscle using CT L3 image analysis.

Authors	(n) Patients/cancer type/therapy	Definition of Sarcopenia* (SMI cm ² /m ²)	% sarcopenic	Major findings related to skeletal muscle
Skeletal mus	cle and chemotherapy toxicity			
Sjoblom ²⁰	153/non-small cell lung/ chemotherapy	_	_	Chemotherapy dosed as mg/kg LBM independently associated with grade $3-4$ haematological toxicity (OR = 1.15 , P= 0.018).
Skeletal mus	cle pre-surgery and post-surgical out	comes		
Harada ²¹	325/esophageal/surgery or chemoradiation	Men: <44.5 Women: <36.5	33	Anastomotic leak higher in sarcopenic vs. non-sarcopenic (25% vs. 14%, $P = 0.032$); no difference for other complications.
Huang ²²	142/colorectal/surgery	25	12	Sarcopenia (OR 4.5, $P = 0.007$) independent risk factor for post-surgical complications.
Tegels ²³	152/gastric/surgery	39	57	Sarcopenia did not predict in-hospital mortality (P=0.52), severe complications (P=1.00) or 6-month mortality (P=0.69).
van Vught ²⁴	206/colorectal /surgery + chemotherapy	38	44	SMI lower in patients with severe postoperative complications (86.6 vs. $110.2 \text{ cm}^2/\text{m}^2$; p = 0.008). SMI (cm ² /m ²) independently associated with severe postoperative complications (OR 0.9, P=0.018), higher SMI less complications
Skeletal mus	cle and survival prediction			
Fujiwara ²⁵	1257/liver/treatment	Men: ≤36.2 Women: ≤29.6	11	Sarcopenia independent predictor of reduced OS (HR 1.5, P=0.001).
Fukishima ²⁶	63/renal/prior treatment	39	68	Sarcopenia independent predictor of OS (HR 2.6, p=0.015). Three-year OS for sarcopenic vs. non-sarcopenic were 31% and 73% (P <0.001).
Psutka ²⁷	387/renal/surgery	38	47	Sarcopenia independently associated with increased cancer specific mortality (HR 1.70, p=0.047) and all-cause mortality (HR 1.5, p=0.039).
Tamandl ²⁸	200/oesophageal + gastro- oesophageal junction/surgery	38	65	Reduced OS (HR 1.9, P=0.011) for sarcopenic vs. non-sarcopenic post- surgery.
Treatment ir	nduced changes in skeletal muscle			
Antoun ²⁹	120/prostate/treatment	39	70	Significant SMI loss from baseline to 3 months (-2.5 cm^2/m^2 , P<0.001), and baseline to 6 months (-2.8 cm^2/m^2 , P<0.001). Sarcopenia not associated with OS, PFS.

Choi ³⁰	484/pancreas/chemotherapy	Men: <42.2 Women: <33.9	21	Significant SMI loss (-2.8 cm^2/m^2 , P<0.001) from baseline to progression. Baseline sarcopenia (HR 1.7, P<0.001), loss of SMI (HR 1.4, P=0.004) independent predictors of OS.
Cooper ³¹	89/pancreas/chemotherapy or chemoradiation	39	52	Significant SMI loss (-1.2 cm ² /m ² , P<0.01) from baseline to post-treatment. Post-treatment SMI independently predicted DFS (HR=0.9, P=0.04), higher SMI for improved DFS.
Kimura ³²	134/non-small cell lung / chemotherapy	Men: <41 Women: <38	38	Significant SMI loss from baseline to 3 months (-1.6 cm ² /m ² , p<0.01), 6 months (-1.3 cm ² /m ² , p<0.01), 12 months (-1.5 cm ² /m ² , p<0.01). Increase SMI independently predicted better OS after 6 months (0.92, P<0.01), and 12 months (0.90, P<0.01) of treatmen.
Miyamoto ³³	182/colorectal/chemotherapy	-	29	Change in SMI was +4.2% and not associated with OS. When SMI loss examined as quartiles, the highest quartile of SMI loss (>5%) independently predicted reduced OS (HR 2.1, P=0.010).
Rollins ³⁴	228/pancreas + cholangiocarcinoma/ chemotherapy	39	61	Significant loss SMI (-2.4 cm^2/m^2 , P<0.001) from baseline to follow-up. Sarcopenia did not predict OS (p=0.739) or chemotherapy completion or toxicity.
Skeletal mus	cle and inflammatory markers			
Aust ³⁵	140/ovary/surgery + chemotherapy	39	29	Sarcopenia (HR 1.23, p=0.565) did not predict OS. There were no correlation between inflammatory markers and SMI.
Malietzis ³⁶	763/colorectal/surgery	39	65	Patients with NLR>3 had lower SMI (P=0.002) than those with NLR <3. NLR >3 (OR 1.8, p<0.001) an independent predictor of low SMI.
Reisinger ³⁷	87/colorectal/surgery	-	-	Low pre-operative SMI significantly predicted high concentrations for calprotectin (a biological indicator of inflammation) on PO days 2-5.
Rollins ³⁴	228/pancreas + cholangiocarcinoma/ chemotherapy	39	60	No correlation between inflammatory markers and SMI.

SMI, skeletal muscle index (cm^2/m^2) ; LBM, lean body mass; OR, odds ratio; HR, hazard ratio; OS, overall survival; PFS, progression-free survival; C-index, concordance index; DFS, disease-free survival; NLR, neutrophil to lymphocyte ratio; PO, post-operative. *Sarcopenia = skeletal muscle depletion (measured by CT L3 image analysis) defined by cutoffs for skeletal muscle index (cm^2/m^2) associated with a clinical outcome (e.g. overall survival). CHAPTER 4: How much does reduced food intake contribute to cancer-associated weight loss?¹

4.1 Introduction

In 2011, an international consensus group defined cancer cachexia as a syndrome of involuntary weight loss (WL), characterized by the loss of skeletal muscle (with or without fat loss), which is driven by a variable combination of reduced food intake (FI) and altered metabolism.¹ Many factors conspire to impair FI and to alter metabolism, resulting in WL (Figure 4-1).

Figure 4-1. A conceptual diagram representing the pathophysiology of cancer cachexia



A conceptual diagram representing the pathophysiology of cancer cachexia, which is driven by a variable combination of reduced food intake (FI) and altered metabolism.¹ The focus of this review, identified by the dotted black box, is the contribution of reduced FI to cancer-associated weight loss (WL). There are many factors that conspire to reduce FI resulting in involuntary WL in patients with cancer cachexia (represented by dark grey boxes). Many of these factors are directly or indirectly related to the presence of the tumor e.g. nutrition impact symptoms due to location of the tumor in the gastrointestinal tract or development of symptoms in response to cancer therapies; decreased central drive to eat in response to pro-inflammatory tumor-mediated factors, and/or cancer-related pain or

¹ A version of this chapter will be published: Martin L, Kubrak C. How much does reduced food intake contribute to cancer-associated weight loss? Curr Opin Support Palliat Care, 2018; 12, doi:10.1097/SPC.00000000000379

nausea. Non-cancer related psychosocial factors can also act as barriers to FI e.g. mood disorders, food insecurity, lack of social support for assistance with meal preparation, poor dietary habits, etc. Metabolic alterations (represented by light grey boxes) are also key drivers of cancer-associated WL, and include tumor-mediated catabolic factors that act directly on skeletal muscle and adipose tissue, or via pro-inflammatory factors arising from interactions with the immune system. The specific nature of these metabolic alterations continues to evolve and requires further elucidation. Ultimately our understanding of the factors contributing to cancer-associated WL will only be fully described when both reductions in food intake and alterations in metabolism are accounted for.

Reduced FI may be instigated by pain, nausea, anxiety, depression, or inflammation as examples, whereby the central drive to eat within the brain (appetite center) is decreased or abolished.²⁻⁴ Additionally, other clinically distinct causes of reduced FI (referred to as nutrition impact symptoms) including decreased upper gastrointestinal motility (early satiety and nausea), dysphagia, stomatitis, bowel obstruction, dental issues, and psychosocial factors may also contribute to a diminished FI.^{1,2,4-6} Altered metabolism is characterized by tumor-derived catabolic factors and inflammation, which promote skeletal muscle and adipose tissue catabolism, and increased energy expenditure, resulting in WL (Figure 5-1).^{2,4} The extent to which FI is reduced and metabolism is altered is not currently known.^{1,2,4,5} However, early identification of reduced FI is the dominate theme within clinical practice guidelines for nutrition care in patients with cancer, since it's treatment has the potential to stem continued WL, improve patient outcomes (i.e. quality of life (QOL)), and reduce patients' and family's distress^{1,5,7-11}

There is a paucity of research studies evaluating the contribution of reduced FI to cancerassociated WL. Lack of emphasis in the current research on reduced FI is disproportionate to its' importance to clinicians, patients, and families.¹¹ In this review, using cancer-associated WL as the model (Figure 4-1), we will present recent studies focusing on the assessment of FI and its association with cancer-associated WL.

4.2 Identification of Reduced Food Intake

An individual's energy intake must align with their energy expenditure (i.e. sum of resting energy expenditure, physical activity, and diet-induced thermogenesis) to maintain energy balance and weight stability (Figure 4-2).^{2,5} Cancer can alter the homeostatic control of energy balance²; tumor derived catabolic factors, inflammation, and cancer treatments can increase energy expenditure and reduce energy intake resulting in negative energy balance and WL (Figure 4-2).



Figure 4-2. A diagram representing the energy balance framework.

A diagram representing the energy balance framework. Energy intake represents the sum total calories ingested and metabolized from the diet; energy expenditure represents the sum total calories of resting energy expenditure (REE), physical activity, and diet-induced thermogenesis. Total energy expenditure is difficult to measure; therefore REE is often used to represent energy expenditure. There are several scenarios within the energy balance framework that will impact on body weight. **i.** Energy balance refers to a net balance between energy intake and energy expenditure, resulting in overall weight stability, **ii.** Positive energy balance occurs with an increase in energy intake relative to energy expenditure resulting in weight gain. **iii.** Negative energy balance occurs when there is an imbalance between energy intake and energy expenditure resulting in weight loss (WL). Four scenarios highlight how the variable contributions of food intake and altered metabolism (represented by REE) can result in WL: a) decreased energy intake, no change in REE, b) no change in energy intake, increased REE, c) disproportionate increase in both energy intake and REE, d) decreased energy intake, increased REE.

To understand the factors contributing to the continuum of cancer-associated WL, information about energy intake and energy expenditure over time is essential. A variety of methods exist to quantify both energy intake and expenditure, each with inherent advantages and limitations.¹²⁻¹⁴ However, as Chow & Hall¹² eloquently point out, a main barrier to our understanding of energy balance is the lack of precise and accurate information about daily fluctuations in both energy intake and expenditure that can capture persistent changes over time. Quantification of energy expenditure is complex and challenging, and at present we have an incomplete understanding about the role of altered energy expenditure in weight losing cancer patients.^{5,13,15-17} In absence of routine, clinically expedient assessments of energy expenditure, current recommendations are to evaluate and manage energy intakes according to clinical effects on body weight and muscle mass.⁵

There are a variety of clinical tools used to assess FI, which can be divided into three main types: 1) Screening for reduced FI using clinical questionnaires: patients compare their current or recent FI to what is normal (i.e. adequate) for them using various categorical or numerical responses (Table 4-1), 2) Assessment of food and fluid intake using recall or diaries: patients record their current food and fluid intake; macro- and micro-nutrient intakes can be calculated, and information about diet quality, food patterns, and preferences can be obtained (Table 4-1). A main limitation for these types of tools is the short time frame (e.g. 1 day to 1 month) over which FI is captured relative to the period of WL (e.g. 1 to 6 months, from pre-illness), which cannot capture the day-to-day variations in FI. 3) A third approach is to use appetite as a surrogate for FI, for which there are also a variety of clinical assessments.¹⁸ However, appetite is unlikely to be adequate surrogate of FI. Measures of appetite and FI have been demonstrated to be only moderately correlated^{19,20}, likely because appetite and FI represent

69

different aspects of FI behavior; appetite is a dimension of ingestive behavior that also includes hunger and satiety, which together influence the outcome which is FI.²¹ There is currently no agreement for how to measure or define reduced FI; however, given the variability in available assessments, reduced FI has been variously classified as patient reported reductions in FI, or energy intakes that are either below a measured energy expenditure or below the recommended energy and protein intakes outlined in clinical practice guidelines.⁵

4.3 Food Intake and Cancer-Associated Weight Loss

The specific contribution of FI to cancer-associated WL has not been extensively studied. Therefore, it is not possible to draw definitive conclusions regarding the magnitude of the contribution of reduced FI to cancer-associated WL; results were difficult to aggregate due to differences in study design (cross-sectional vs. longitudinal; observational vs. intervention), definitions for WL as an outcome (e.g. continuous, >5%), FI measures (questionnaire vs. food/fluid records), and limited presentation of statistical information. However, common themes were identified and will be discussed.

4.3.1 Comparing food intake between patients with and without cancer-associated weight loss

Four studies compared the FIs of patients with or without cancer cachexia (i.e. patients with or without WL).²²⁻²⁵ Similar to what Blum et al.¹⁵ had previously reported, the use of prospectively collected FI records to estimate energy intakes in cancer patients was limited.^{22,23,25} Overall, mean energy (and protein) intakes were not different between cachectic or non-cachectic patients, although a higher proportion of cachectic patients reported reductions in FI (measured by the PG-SGA)²⁴ and energy intakes of <20kcal/kg/d.²² Small sample sizes limit the ability to detect differences due to large inter-individual variations in FI, and energy expenditure

was either not quantified^{22,25} or quantified inappropriately²³ so the gap between energy intake and energy expenditure could not be adequately assessed as an explanatory factor for WL. 4.3.2 The association between food intake and cancer-associated weight loss

Four studies evaluated the association between FI and WL (Table 4-2).^{19,20,26,27} Vagnildhaug et al.²⁷ observed increasingly severe reductions in FI (measured by the PG-SGA) with increased severity of WL. Two studies^{19,20} sought to evaluate the association of FI to cancer-associated WL using multivariable linear regression analysis. These studies demonstrated that reduced FI (measured by the PG-SGA as normal vs. reduced) or low energy and protein intakes (continuous as kcal/kg/d; g/kg/d) had significant, independent associations with WL. Notably, both studies also sought to evaluate the association of altered metabolism to WL by including a measure of systemic inflammation, C-reactive protein (CRP), but did not find an association.^{19,20}

Two studies^{19,26} undertook specific interventions to improve FI in cancer patients (Table 4-2). Kapoor et al.²⁶ randomized 63 female cachectic patients into two groups, both groups received nutritional and physical activity counseling and the intervention group (N= 30) received 100 g/d of fortified flour for 6 months. Patients were assessed 3 and 6 months after initial visit. The intervention group significantly increased total energy (kcal/d) and protein (g/d) intakes at each visit, resulting in a trend towards body weight gain. In the study by Nasrah et al.¹⁹, cachexic patients (N=320) attending a multi-disciplinary cancer nutrition rehabilitation program received, pain, symptom, and psychosocial management, exercise training, and individualized nutrition counseling to achieve target dietary intakes of at least 30 kcal/kg and 1.3 g protein/kg. Patients were seen approximately 6 and 12 weeks after initial visit. Overall, researchers found as mean energy (from 25 to 32 kcal/kg/d) and protein (from 1.0 to 1.4 g/kg/d) intakes increased, weight

stabilized (weight change from -2 to 0.4 kg) over time. Interestingly, researchers observed an unpredictable weight change response with a given change in energy intake suggesting that unaccounted for alterations in metabolism (e.g. increased energy expenditure) might be contributing to WL. Taken together these two studies suggest that reduced FI is an important contributor to cancer-associated WL, which can be mediated with efforts to improve FI in subsets of patients. Identifying patients likely to respond to nutritional interventions will require further characterization.

A single study²⁸ assessed resting energy expenditure (REE kcal/d; portable indirect calorimetry) and energy intake (24hr recall, kcal/d) in a heterogeneous group of cancer patients (Table 4-2). Interestingly, when patients were classified as hypermetabolic (defined as an REE >110% of predicted REE), hypometabolic, or normometabolic, the mean energy intakes (kcal/d) were not different between groups, and thus not surprisingly hypermetabolic patients had the greatest WL. This observation suggests patients with hypermetabolism fail to augment their energy intakes in relation to their increased energy expenditure resulting in negative energy balance and WL. In a multivariable linear regression analysis hypermetabolism, low energy intake (kcal/d), and elevated CRP had significant, independent associations with WL >5% (Table 4-2). Although the magnitude of the association for each of these factors was not reported, this study suggests that both reduced FI and alterations in metabolism associate with WL.

4.3.3 Nutrition impact symptoms and Cancer-associated weight loss

In the current literature, researchers recorded results of self-perceived symptoms that associated with reduce FI in weight losing patients with cancer. Appetite loss (measured with a variety tools) related to reduced FI (measured by food/fluid records, VAS FI scores) in a majority of weight losing patients.^{19,20,24,26,29-31} Patients with cachexia often described dysgeusia,

72

nausea, vomiting, pain, fatigue, depression, weakness, and meat aversion, as reasons for reduced FI.^{19,26,29-33} These symptoms are consistent with our current understanding of the factors likely to interfere with FI in patients with cancer (Figure 4-1).^{1,2}

4.4 Discussion

A key feature of cancer cachexia is reduced FI that leads to progressive WL, which is the subject of this review. The current literature does not dispute this concept, but lack of consensus regarding the assessment of FI is a principal factor confounding this area of research and limits our ability to explain the magnitude of the association with WL. It is improbable that reductions in FI alone will explain the progression of WL in all patients with cachexia, and without adequate consideration of the factors driving alterations metabolism (e.g. increased energy expenditure, systemic inflammation, and tumor-derived catabolic factors), we have an incomplete understanding about the magnitude by which these alterations also contribute to cancer-associated WL (Figure 4-1).

Diminishing FI is recognized to be a dominant driver of WL, and efforts to improve FI with the intention to promote maintenance or gain of weight are regarded as the first line of treatment for cancer cachexia.^{2,4-6,34} It is our understanding that cancer cachexia cannot be fully reversed by nutrition support alone due to metabolic alterations that independently contribute to WL, and because of this variable combination of reduced FI and altered metabolism attempts to correct FI alone will result in variable body weight responses (Figure 4-3).^{1,2,10,19,35-37}



Figure 4-3. Heterogeneous body weight responses to a nutritional intervention

Body weight response to a nutritional intervention. The line graph, from Wigmore et al.³⁵, shows the variable progression (rate) of WL before and after introducing a nutritional intervention and highlights the variability in body weight responses, which include weight stability, weight gain, WL, and fluctuation between all three of body weight responses. Monitoring body weight responses in conjunction food intake provides information about energy balance; which essentially, quantifies the gap between energy intake and expenditure. For example patients in negative energy balance scenario iii. c) from Fig 2A) increase their energy intake but continue to lose weight because energy intake remains below REE.

A recent meta-analysis did find that nutrition interventions promoted weight gain in subsets of patients undergoing chemotherapy or chemoradiation. This positive effect on body weight was attributed to interventions with energy dense high protein oral nutritional supplements enriched with n-3 fatty acids, which aim to improve both energy and protein intakes as well as modulate alterations in metabolism.¹⁰ Understanding which patients are likely to respond to specific nutrition interventions requires further investigation, and many additional factors (e.g. patient compliance, high symptom burden, increased tissue catabolism, proximity to death) can also affect individual response to a given intervention^{10,19} de van der Schueren et al.¹⁰ made a series of recommendations to improve the design of future clinical trials studying nutrition interventions in cancer cachexia, including the study of interventions in the context of multimodal care plans, which is also the recommend treatment approach for cancer cachexia.^{6,34,38-40} Core components of multimodal care include the medical management of pain and symptoms, and combined interventions to improve FI, increase exercise to promote anabolism, modulate inflammation, and establish adequate psychosocial supports.^{4,10,34,39,40} Indeed specialized multimodal cachexia clinics have demonstrated positive impacts on body weight, improved FI, and other important patient outcomes such as quality of life, physical function, pain and distress.^{19,41-45} Although not every center will have access to specialist cachexia clinics, Maddocks et al.⁴⁰ suggest effective multimodal care can be practical and incorporated into the clinical practice of health care professionals. Key components for the successful adoption of multimodal care plans include patient engagement, communication between health care professionals, consistent messaging regarding the treatment plan, and practical and individualized interventions.^{40,41}

In order to provide the right nutrition interventions at the right time, identification of patients at risk of nutrition problems is an essential first step.^{6,38} Clinical practice guidelines recommend regularly screening cancer patients for nutrition risk (i.e. weight loss, reductions in FI, nutrition impact symptoms) and if identified, followed-up with a comprehensive nutrition assessment, and appropriate interventions defined in the context of individual prognosis and treatment goals⁵ This review, albeit limited in scope, demonstrates current research is aligned with international consensus and clinical guidelines to assess FI in patients with cancer, but there is no agreement on how to best measure FI in clinical practice, which has been a longstanding issue.^{1,15,36,37} Future studies to evaluate equivalency between different measures of FI based on

75

their association to cancer-associated WL are needed to enhance our ability to aggregate and compare research results, reduce heterogeneity, and improve our ability to draw conclusions about the impact of FI on features of cancer.

4.5 Conclusion

The etiology of cancer cachexia is driven by a variable combination of reduced FI and altered metabolism. In the clinical setting, identifying reductions in FI is the predominant theme, but few studies have examined the magnitude of the association to WL. Additionally, research studies have adopted a variety of different tools to evaluate FI in patients with cancer, rendering the comparison of available studies challenging. The literature reviewed here points to reductions in FI as an important driver of cancer-associated WL, which is independent of alterations in metabolism. Until there is a more complete understanding about the alterations in metabolism that occur in patients with cancer and how they relate to WL, multimodal interventions that remove barriers to FI should remain a focus of care.

4.6 References

- 1. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.
- 2. Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.
- 3. Burfeind KG, Michaelis KA, Marks DL. The central role of hypothalamic inflammation in the acute illness response and cachexia. Semin Cell Dev Biol. 2016; 54:42-52.
- 4. Arends J. Struggling with nutrition in patients with advanced cancer: nutrition and nourishment-focusing on metabolism and supportive care. Ann Oncol. 2018; 29:ii27-ii34.
- 5. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 6. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 7. Senesse P, Isambert A, Janiszewski C, et al. Management of cancer cachexia and guidelines implementation in a comprehensive cancer center: a physician-led cancer nutrition program adapted to the practices of a country. Journal of pain and symptom management. 2017; doi:10.1016/j.jpainsymman.2017.01.010.
- 8. Amano K, Maeda I, Morita T, et al. Eating-related distress and need for nutritional support of families of advanced cancer patients: a nationwide survey of bereaved family members. J Cachexia Sarcopenia Muscle. 2016; 7:527-534.
- 9. Vagnildhaug OM, Balstad TR, Almberg SS, et al. A cross-sectional study examining the prevalence of cachexia and areas of unmet need in patients with cancer. Support Care Cancer. 2018; 26:1871-1880.
- de van der Schueren MAE, Laviano A, Blanchard H, Jourdan M, Arends J, Baracos VE. Systematic review and meta-analysis of the evidence for oral nutritional intervention on nutritional and clinical outcomes during chemo(radio)therapy: current evidence and guidance for design of future trials. Ann Oncol. 2018; 29:1141-1153.
- 11. Cooper C, Burden ST, Cheng H, Molassiotis A. Understanding and managing cancerrelated weight loss and anorexia: insights from a systematic review of qualitative research. J Cachexia Sarcopenia Muscle. 2015; 6:99-111.
- 12. Chow CC, Hall KD. Short and long-term energy intake patterns and their implications for human body weight regulation. Physiol Behav. 2014; 134:60-65.

- Hills AP, Mokhtar N, Byrne NM. Assessment of physical activity and energy expenditure: an overview of objective measures. Front Nutr. 2014; doi: 10.3389/fnut.2014.00005.
- 14. Lam YY, Ravussin E. Analysis of energy metabolism in humans: A review of methodologies. Mol Metab. 2016; 5:1057-1071.
- 15. Blum D, Omlin A, Baracos VE, et al. Cancer cachexia: a systematic literature review of items and domains associated with involuntary weight loss in cancer. Crit Rev Oncol Hematol. 2011; 80:114-144.
- Purcell SA, Elliott SA, Ryan AM, Sawyer MB, Prado CM. Accuracy of a portable indirect calorimeter for measuring resting energy expenditure in individuals with cancer. JPEN J Parenter Enteral Nutr. 2018; doi.10.1002/jpen.1310.
- Nguyen TY, Batterham MJ, Edwards C. Comparison of resting energy expenditure between cancer subjects and healthy controls: a meta-analysis. Nutr Cancer. 2016; 68:374-387.
- 18. Martin L. Diagnostic criteria for cancer cachexia: data versus dogma. Curr Opin Clin Nutr Metab Care. 2016; 19:188-198.
- 19. Nasrah R, Kanbalian M, Van Der Borch C, Swinton N, Wing S, Jagoe RT. Defining the role of dietary intake in determining weight change in patients with cancer cachexia. Clin Nutr. 2018; 37:235-241.
- 20. Solheim TS, Blum D, Fayers PM, et al. Weight loss, appetite loss and food intake in cancer patients with cancer cachexia: three peas in a pod? analysis from a multicenter cross sectional study. Acta Oncol. 2014; 53:539-546.
- 21. French SA, Epstein LH, Jeffery RW, Blundell JE, Wardle J. Eating behavior dimensions: associations with energy intake and body weight: a review. Appetite. 2012; 59:541-549.
- 22. Jager-Wittenaar H, Dijkstra PU, Dijkstra G, et al. High prevalence of cachexia in newly diagnosed head and neck cancer patients: An exploratory study. Nutrition. 2017; 35:114-118.
- 23. Mohan A, Poulose R, Kulshreshtha I, et al. High prevalence of malnutrition and deranged relationship between energy demands and food intake in advanced non-small cell lung cancer. Eur J Cancer Care (Engl). 2017; doi: 10.1111/ecc.12503.
- Vigano AAL, Morais JA, Ciutto L, et al. Use of routinely available clinical, nutritional, and functional criteria to classify cachexia in advanced cancer patients. Clin Nutr. 2017; 36:1378-1390.

- 25. Bye A, Wesseltoft-Rao N, Iversen PO, et al. Alterations in inflammatory biomarkers and energy intake in cancer cachexia: a prospective study in patients with inoperable pancreatic cancer. Med Oncol. 2016; 33:54.
- 26. Kapoor N, Naufahu J, Tewfik S, Bhatnagar S, Garg R, Tewfik I. A Prospective Randomized Controlled Trial to Study the Impact of a Nutrition-Sensitive Intervention on Adult Women With Cancer Cachexia Undergoing Palliative Care in India. Integr Cancer Ther. 2017; 16:74-84.
- Vagnildhaug OM, Blum D, Wilcock A, et al. The applicability of a weight loss grading system in cancer cachexia: a longitudinal analysis. J Cachexia Sarcopenia Muscle. 2017; 8:789-797.
- 28. Vazeille C, Jouinot A, Durand JP, et al. Relation between hypermetabolism, cachexia, and survival in cancer patients: a prospective study in 390 cancer patients before initiation of anticancer therapy. Am J Clin Nutr. 2017; 105:1139-1147.
- Barajas Galindo DE, Vidal-Casariego A, Calleja-Fernandez A, et al. Appetite disorders in cancer patients: Impact on nutritional status and quality of life. Appetite. 2017; 114:23-27.
- 30. Muscaritoli M, Lucia S, Farcomeni A, et al. Prevalence of malnutrition in patients at first medical oncology visit: the PreMiO study. Oncotarget. 2017; 8:79884-79896.
- 31. van der Werf A, Arthey K, Hiesmayr M, et al. The determinants of reduced dietary intake in hospitalised colorectal cancer patients. Support Care Cancer. 2018; 26:2039-2047.
- 32. Blauwhoff-Buskermolen S, Langius JAE, Becker A, Verheul HMW, de van der Schueren MAE. The influence of different muscle mass measurements on the diagnosis of cancer cachexia. J Cachexia Sarcopenia Muscle. 2017; 8:615-622.
- 33. Guerdoux-Ninot E, Kilgour RD, Janiszewski C, et al. Meal context and food preferences in cancer patients: results from a French self-report survey. Springer Plus. 2016; 5:810.
- 34. Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol. 2013; 10:90-99.
- 35. Wigmore SJ, Barber MD, Ross JA, Tisdale MJ, Fearon KC. Effect of oral eicosapentaenoic acid on weight loss in patients with pancreatic cancer. Nutr Cancer. 2000; 36:177-184.
- Baldwin C, Spiro A, Ahern R, Emery PW. Oral nutritional interventions in malnourished patients with cancer: a systematic review and meta-analysis. J Natl Cancer Inst. 2012; 104:371-385.

- 37. Balstad TR, Solheim TS, Strasser F, Kaasa S, Bye A. Dietary treatment of weight loss in patients with advanced cancer and cachexia: a systematic literature review. Crit Rev Oncol Hematol. 2014; 91:210-221.
- 38. Aapro M, Arends J, Bozzetti F, et al. Early recognition of malnutrition and cachexia in the cancer patient: a position paper of a European School of Oncology Task Force. Ann Oncol. 2014; 25:1492-1499.
- 39. Balstad TR, Kaasa S, Solheim TS. Multimodal nutrition/anabolic therapy for wasting conditions. Curr Opin Clin Nutr Metab Care. 2014; 17:226-235.
- 40. Maddocks M, Hopkinson J, Conibear J, Reeves A, Shaw C, Fearon KC. Practical multimodal care for cancer cachexia. Curr Opin Support Palliat Care. 2016; 10:298-305.
- 41. Gagnon B, Murphy J, Eades M, et al. A prospective evaluation of an interdisciplinary nutrition-rehabilitation program for patients with advanced cancer. Curr Oncol. 2013; 20:310-318.
- 42. Glare P, Jongs W, Zafiropoulos B. Establishing a cancer nutrition rehabilitation program (CNRP) for ambulatory patients attending an Australian cancer center. Support Care Cancer. 2011; 19:445-454.
- 43. Parmar MP, Swanson T, Jagoe RT. Weight changes correlate with alterations in subjective physical function in advanced cancer patients referred to a specialized nutrition and rehabilitation team. Support Care Cancer. 2013; 21:2049-2057.
- 44. Parmar MP, Vanderbyl BL, Kanbalian M, Windholz TY, Tran A-T, Jagoe RT. A multidisciplinary rehabilitation programme for cancer cachexia improves quality of life. BMJ Support Palliat Care. 2017; 7:441-449.
- 45. Uster A, Ruehlin M, Mey S, et al. Effects of nutrition and physical exercise intervention in palliative cancer patients: A randomized controlled trial. Clin Nutr. 2018; 37:1202-1209.

Measures of Food Intake in Clinical Practice	Example Tools	Example Data	Advantages	Limitations
Food intake as a component of clinical questionnaires	PG-SGA MNA Food intake VAS NRS-2002	Food intake in the past month: Normal / less than usual/ more than usual Current food intake: Normal food, in normal amount/ normal food, reduced amount/ little solid food/ only oral nutritional supplements/ only liquids/ very little anything Food intake over the past 3 months: Normal vs. moderately or severely reduced Current food intake VAS (0- none at all to 10-as normal) Food intake in the preceding week: 50-75% of normal requirement/ 25-50% of normal requirement/	Completed by patient or health care providers Screens for nutritional problems Can be used to track changes over time Clinically expedient Current food intake is benchmarked to past intake	No estimate of macro- and micronutrient intakes None or limited information about patient food patterns and preferences Food intake assessed over different time frames (e.g. intake now, past week, or over one or 3 months)

Table 4-1. Assessments used to measure food intake in the oncology setti
--

		0-25% of normal requirement		
Food and fluid records	24-hour recall (food and fluids consumed the previous day) Food and fluid intake diaries (food and fluids consumed over 1-7 days)	Macronutrients: Energy intake (kcal/d) Protein intake (g/d) Fat intake (g/d) Micronutrients	Completed by patient or care giver and health care provider Information about food type, patterns, and preferences Useful for in-depth nutritional assessment and for guiding individualized nutrition interventions	Requires specialized software and country-specific nutrient database Time consuming and burdensome to patient/caregiver Accuracy declines with increased recording days Resource intensive (health care provider must collect, verify, and enter data) Potential for mis-quantification of food items, and underestimation of nutrients

MNA, Mini-Nutrition Assessment; NRS 2002, Nutrition Risk Screening; PG-SGA, Patient Generated-Subjective Global Assessment; VAS, visual analogue scale.

Study	N/Setting/Cancer Type	Variables	Measure of WL	Correlation with Weight Change	Outcome(s)
Vagnildhaug / Prospective observational cross-sectional multi-center	1406/Palliative Care Program/Heterogeneous, incurable, 60% receiving treatment	Food Intake (current): PG-SGA; 5 sub-categories Appetite loss (previous week): EORTC QLQ-C15 PAL Function: EORTC-QLQ- C15 PAL, KPS Fatigue: EORTC-QLQ- C15-PAL Emotional Functioning: EORTC-QLQ-C15-PAL	Weight loss grades (WLG) (0-4)	Not reported	One way ANOVA: compared differences between WLG 0-4 Worsening of parameters with increasing WLG: Food intake (P<0.001) Appetite loss (P<0.001) KPS (P<0.001) Fatigue (P<0.001) Physical function (P<0.001) Emotional functioning (P<0.001)
Solheim/ Prospective, observational, cross-sectional multi-center	885/Palliative Care Program/Heterogeneous, incurable, 75% receiving treatment	Food Intake (past month): PG-SGA; normal vs. reduced Appetite loss (previous week): EORTC QLQ-C30 Function: EORTC- QLQ-C30, KPS Inflammation: CRP _{log} (mg/L)	% weight loss past 6 months	Reduced food intake R _s =0.34, P<0.01 Appetite loss R _s =0.34, P<0.01	Multivariable linear regression: %WL (dependent variable):Reduced food intake (β =0.15, P<0.01)
Nasrah/ Retrospective,	320/Cancer Nutrition Rehabilitation Program/	Energy intake (kcal/kg/d) : 24-hour recall	Visit 1(Baseline): weight change (kg)	Energy intake (kcal/kg/d) (R=0.25, P<0.001)	Multivariable linear regression: weight change in kg (dependent

Table 4-2. A summary of studies that have examined the contribution or food intake to cancer-associated weight loss.

intervention, longitudinal, moncentric	Heterogeneous, advanced (Stage III-IV), 35% not receiving treatment *Intervention = individualized diet counselling and exercise training	Protein intake (g/kg/d): 24-hour recall Appetite loss: VAS 0-10 Performance Status: ECOG Inflammation: CRP _{log} (mg/L), Albumin (g/L)	past 6 weeks	Protein intake $(g/kg/d)$ (R=0.28, P<0.001) Appetite loss (R _s =-0.24, P<0.001) PS (R _s =-0.24, P<0.001) CRP _{log} (R=-0.11, P<0.05)	variable): Significant independent predictors of weight change: Energy intake† (P=0.002) Performance status (P=0.002) Appetite loss score (P<0.05) (adjusted model R ² = 0.12) age, sex, tumor type, chemotherapy treatment, and CRP _{log} did not associate with weight change
			<i>Visit 2:</i> weight change (kg) past 6 weeks (V2-V1)	Energy intake (R=0.26, P<0.001) Protein intake (R=0.19, P<0.01) Appetite loss (R _s =-0.31, P<0.001), PS (R _s =-0.18, P<0.01) CRP _{log} (R=-0.09, P=NS)	Significant independent predictors of weight change: Energy intake ⁺ (P=0.004) Appetite loss score (P=0.005) only predictors at V2 (adjusted model R ² = 0.10) age, sex, tumor type, chemotherapy treatment, and CRP _{log} , PS, did not associate with weight change
			<i>Visit 3:</i> weight change (kg) past 6 weeks (V2-V3)	Energy intake (R=0.35, P<0.001) Protein intake (R=0.23, P<0.01) Appetite loss $(R_s=-0.38, P<0.001)$ PS $(R_s=-0.28, P<0.001)$ CRP _{log} (R=-0.06, P=NS)	Significant independent predictors of weight change: Energy intake (P=0.0005) Appetite loss score (P=0.001) only predictors at V3 (adjusted model R ² = 0.17) age, sex, tumor type, chemotherapy treatment, and CRP _{log} . PS, did not associate with

					weight change
Kapoor/ prospective, intervention, longitudinal, moncentric	N=33/Heterogeneous tumor group, all female, palliative care clinic Treatment group N=17; Control Group, N=15 * Both groups individualized diet counselling; Intervention group received 14 packets of 100 g of Enhanced Atta (flour), providing approximately 400 kcal/d	Energy intake (kcal/d): 24-hour recall Protein intake (g/kg/d): 24-hour recall Food Frequency Questionnaire (FFQ) by Indian Migrant FFQ EORTC-QLQ-C30 measured (fatigue, nausea and vomiting, pain, appetite loss, constipation, diarrhea) Physical Activity Questionnaire (Indian Migrant study – PAQ)	Visit 1 (Baseline): weight (kg)	Not reported	Student's t-test compared differences between intervention and control groups for each time point, and repeated measures ANOVA determined changes within groups overtime:
			<i>Visit 2:</i> weight change (kg) past 3 months (V2- V1)	Not reported	Compared to the control group, the intervention groups significantly increased (t-test): Energy intake (P=0.003) Protein intake (P<0.001) Physical activity (P=0.007) Appetite loss (P<0.001) and Reduced pain (P<0.001) Reduced fatigue (P<0.001)
			<i>Visit 3:</i> weight change (kg) past 3 months (V3- V2)	Not reported	Compared to the control group, the intervention groups significantly increased (t-test): Energy intake (P=0.001) Protein intake (P<0.001) Physical activity (P<0.007) Appetite loss (P=0.001) and Reduced fatigue (P<0.001) At the end of 6 months

					(repeated measures ANOVA): Intervention group increased: Body weight (P=0.081) Energy intake (P<0.001) Macronutrient intake (P<0.001) Appetite loss (P=0.006) and Reduced pain (P=0.012) Reduced fatigue (P=0.002) No change in performance status (NS)
					Control group decreased: Body weight (P=0.003), PS (P=0.004) and showed no improvement in energy or macronutrient intakes, and appetite loss.
Vazeille/ Cross- sectional, observational, monocentric	390/Multi-disciplinary risk assessment program/Heterogeneous, solid tumors, any stage, prior to systemic therapy	Energy intake (kcal/d): 24-hour recall Measured REE (kcal/d): indirect calorimetry (portable) Brudiated REE (kcal/d):	% weight loss from UBW	Energy balance R=-0.23, P<0.001) NRI score R=0.27, P<0.001 Albumin (R=0.22,	Multivariable linear regression: WL >5% (dependent variable): Significant independent predictors of WL >5%:
		Predicted REE (kcal/d): Harris-Benedict Equation Energy Balance (kcal/d): energy intake– measured REE Inflammation: CRP (mg/L), Albumin (g/L), Nutrition Status: NRI		P<0.001) CRP R=-0.20, P<0.001	Energy intake (OR 0.9998 [95% CI 0.9997-0.9999), P=0.013) CRP (OR 1.0019 [1.0006, 1.0032], P=0.004), Hypermetabolism† (OR 1.1742 [95% CI 1.0428, 1.3223], P=0.008)

Abbreviations: ANOVA, analysis of variance; BW, body weight; CI, confidence interval; CRP, C-reactive protein; DI, dietary intake; ECOG, Eastern Cooperative Oncology Group; EORTC, European Organization for Research and Treatment of Cancer; EORTC-QLQ-C15 PAL/C30, European Organization for Research and Treatment of Cancer Quality of Life Questionnaire; FFQ, Food Frequency Questionnaire; Ingesta-VAS, Ingesta visual analogue scale; KPS, Karnofsky Performance Scale; NS, non-significant; NRI, nutrition risk index; OR, odds ratio; PG-SGA, Patient-Generated Subjective Global Assessment; PS, Performance Status; PAQ, Physical Activity Questionnaire; QOL, Quality of life; R, pearson correlation; Rs spearman correlation; REE, resting energy expenditure; UBW, usual body weight; VAS, visual analogue scale; WL, weight loss, WLG, weight loss grade. *dietary intake in this model was not specified as energy, protein or diet category; †Hypermetabolism measured REE >110% of predicted REE.

CHAPTER 5: Reduced food intake as an explanatory variable for cancer-associated weight loss

5.1 Introduction

Cancer cachexia is defined, as a multifactorial syndrome of involuntary weight loss (WL) characterized by skeletal muscle loss (with or with fat loss), which is driven by "a variable combination of reduced food intake (FI) and altered metabolism".¹ Reduced FI is an important and, in some cases the dominant driver of cancer-associated WL, and quantifying the severity of FI impairment is recommended as an essential first step to guide nutrition care.¹⁻¹⁰ Impairments to FI are multifactorial, and can include decreased central drive to eat, uncontrolled pain and symptoms, physical obstruction or gastrointestinal tract dysfunction, reduced physical function, and/or psychosocial factors ². Some of these limitations may be reversible, and their identification can assist care teams to determine appropriate feeding strategies.^{2,11,12}

Despite the importance placed on the early identification of reduced FI, its specific contribution to WL has not been fully elucidated.¹³ There is a lack of consensus with regard to how reduced FI ought to be measured due to significant heterogeneity of available assessments.¹³ FI data acquired from the clinical setting is predominantly acquired from validated patient-reported nutrition risk assessment questionnaires, *e.g.* the Patient-Generated Subjective Global Assessment (PG-SGA) ^{14,15}, the *Ingesta*-Verbal/Visual Analogue Scale (I-VVAS)^{16,17}, and the Mini-Nutrition Assessment (MNA) ^{18,19}, as opposed to the more labor intensive food and fluid records.²⁰ A concept inherent to all of these tools is to capture the quantity of FI anchored to an individual's perceived normal food intake; however the ratings of severity are numerical or descriptive and have a 3-, 7- or 10-levels to describe intake.

Our international research groups have been involved in the aggregation of an International Cancer Cachexia Data Repository (ICCDR) of patients with cancers of advanced
stage who are at risk for cancer cachexia. Our goal is to generate samples of sufficient size to capture representative distributions of key patient features, including FI and WL history, and clinical outcomes for adequately powered statistical and subset analyses. The ICCDR includes FI data for 11,704 cases (evaluated by PG-SGA (n=6161), *I*-VVAS (n=3186) and MNA (n=2357)), and is therefore suited to answer our research questions: what is the association between FI as assessed by three different tools and WL?

5.2 Methods

5.2.1 Study population

The ICCDR includes prospective clinical and nutrition risk assessment data collected from 1999 to 2016. Nutrition risk assessment occurred at several points of referral for nutrition care (i.e. prior to nutrition intervention): at entry to investigational cachexia clinical trials, at cancer diagnosis as of standard of care, and at entry to supportive/palliative care programs. Martin et al.²¹ provide a description of the study cohorts initially included in the ICCDR, to which additional data have subsequently been added. The ICCDR has specific inclusion criteria intended to facilitate data pooling: i) Original study data were prospectively collected under the auspices of human ethics approvals from respective institutions, ii) data are anonymized and iii) clinical data include age (\geq 18 years), sex, confirmed cancer site (based on International Classification of Diseases (ICD)-10 classifications) and stage (based on the American Joint Committee on Cancer (AJCC) stage groupings versions 6 to 7), performance status (PS; Eastern Cooperative Oncology Group (ECOG), Karnofsky Performance Status (KPS), or other measures of PS), WL history, measures of FI, body mass index (BMI), and if available time to death/censoring.

For this study, patients were selected from the ICCDR with complete data for the following: demographic and clinical data (age, sex, cancer diagnosis and stage, PS, BMI, WL history), and nutrition risk assessments that included a measure of FI. Data for these patients was collected between 2004 and 2014. Patients with data from 2004 to 2010 used AJCC staging version 6.0, from 2011 onwards version 7.0 was used. We did not select patients with weight gain (e.g. $\geq 2.5\%$) for the reasons that our primary outcome is cancer-associated WL, and we previously showed that weight gaining patients in this population had different prognoses compared to weight stable patients.²²

5.2.2 Assessments of Food Intake

Three data sets were identified each with a different assessment for measuring reductions in FI: the PG-SGA, the 10-point *I*-VVAS, and the MNA.

The PG-SGA data includes data pooled from 5 study cohorts (Table 5-1). The PG-SGA has 7 categories of FI, and patients choose the category that best represents their current FI compared to what is normal for them. The categories of FI capture the severity of FI impairment, represented by both quantity (e.g. less than normal, very little of anything) and quality indicators (normal food, little solid food, only liquids/ONS). Patients who reported receiving "only tube feedings or nutrition by vein" were excluded (N=69) as they were no longer relying on volitional FI.

The *I*-VVAS data was collected from a nutrition risk screening program at the Cancer Institute of Montpellier, France (Table 5-1).²¹ FI was assessed at patients' first visit using the patient reported 10-point *I*-VVAS^{16,17}, and patients could respond to the question verbally or visually. The 10-point scale conveys the severity of current FI impairment based on quantity (normal to nothing at all). Patients rating their FI as 0 were excluded (N=47) as they were

receiving artificial nutrition support and no longer relying on volitional FI; an *I*-VVAS of 1 was therefore defined as the lowest level of volitional FI.

MNA data includes data pooled from two study cohorts (Table 5-1). ^{21,23,24} The MNA includes a question about FI in the screening portion of the tool, which assesses reductions in the quantity of FI over the previous 3 months. Patients relying on artificial nutrition support were excluded (N=47) as they were no longer relying on volitional food intake.

5.2.3 Statistics

Summary statistics were used to describe patients. The primary outcome was WL history, recorded as the percentage of WL over the previous 6 months or from usual body weight and calculated as:

% weight loss (%WL) =

[(current weight in kg -previous weight in kg)/previous weight in kg]*100

Statistical analysis was carried out in two parts:

1) The relationship between assessments of FI (PG-SGA, I-VVAS, MNA) and %WL were evaluated. Mean %WL was calculated for each of the three FI measurement scales; statistical differences in mean %WL were evaluated using one-way ANOVA (post-hoc tests: Games-Howell), a difference of >2% was recognized to be clinically meaningful WL (i.e. account for measurement error).

2) Multinomial logistic regression (MLR) was used to evaluate the association between reductions in food intake and %WL; MLR is an extension of logistic regression and is a robust statistical approach when the dependent variable has more than two categories. The dependent variable in this analysis is WL, which has 5 categories of severity ($\pm 2.4\%$, 2.5-5.9%, 6.0-10.9%, 11.0-14.9%, and $\geq 15.0\%$), as outlined in the diagnostic criteria for cancer-associated WL.²¹ The

reference group for the dependent variable was weight stable ($\pm 2.4\%$), to which all other categories of WL were compared. MLR models were adjusted for age (continuous), sex, cancer site (ICD-10), cancer stage (AJCC), PS (ECOG or MNA mobility score), and BMI (<20.0, 20.0-21.9, 22.0-27.9, ≥ 28.0).²¹ Results are reported as odds ratios (OR) and 95% confidence intervals (CI).

All analyses were completed using IBM SPSS Statistics for Windows version 23.0 (SPSS, Chicago, IL) and were considered statistically significant at the P < 0.05 level. Figures were drawn with GraphPad Prism version 7.04 for Windows (GraphPad Software, La Jolla California USA).

5.3 Results

5.3.1 Patient Characteristics

Characteristics for the PG-SGA, *I*-VVAS, and MNA data sets are presented in Table 5-2. Each data set represents patients with cancer cachexia: patients with predominantly locally advanced or metastatic disease and losing weight.

5.3.2 The association between food intake and cancer-associated weight loss

Figure 5-1 depicts the mean %WL for each of the FI measurement scales from the MNA, PG-SGA, and *I*-VVAS respectively. Within each tool, measurement scales represent clinical and statistical differences in %WL (P<0.001).





Three graphs depict the relationship between percent weight loss (%WL) and food intake (FI) measurement scales collected from different nutrition risk assessment tools: (A) MNA, (B) PG-SGA, and (C) *Ingesta*-VVAS. For each graph (A,B,C), different shades of gray represent significant differences (P<0.001) in mean %WL within the respective measurement scales. The 3 categories of FI measured by the (A) MNA associated with significantly different degrees of WL (p<0.001). Based on differences in %WL, three levels of FI were defined for the (B) PG-SGA, and (C) *Ingesta*-VVAS, which aligned with the categories of FI from the (A) MNA. Each measurement scale has a category representative of normal FI (black box). Moderately reduced FI (blue diamond) is represented by the following categories: (A) MNA "moderately decreased food intake"; (B) PG-SGA "normal food in reduced amount"; and (C) *Ingesta*-VVAS numeric scores 5 to 9. Severely reduced FI (red circle) is represented by the following categories: (A) MNA "severely decreased food intake"; (B) PG-SGA "little solid food", "only oral nutritional supplements", "only liquids", and "very little of anything"; and (C) *Ingesta*-VVAS numeric scores 1 to 4. White lines highlight regions of similar mean %WL values for normal, moderately reduced, and severely reduced FI descriptors, which align the three tools.

The MNA (Figure 5-1A) has 3 FI categories associated with different degrees of WL (P<0.001; Table 5-1). The PG-SGA and *I*-VVAS also had three distinct homogeneous subsets within their measurement scales that associated with different degrees of WL (P<0.001; Figure 5-1 B,C). Therefore, the measurement scales of the PG-SGA and *I*-VVAS were each combined into three categories of FI, similar to the MNA: normal, moderately reduced, and severely reduced, which associated with significantly different degrees of WL (P<0.001; Table 5-3). Each category of FI (normal, moderately reduced, severely reduced) was compared across tools. There

were some significant statistical differences between similar categories of FI (P<0.05), however the mean differences in %WL were <2%, and therefore not clinically meaningful (Table 5-3).

Three MLR models, one for each of the PG-SGA, I-VVAS and MNA data sets, evaluated the association between reduced FI and WL. The univariable analysis can be found in Appendix III, Suppl Table 1. In multivariable MLR analysis, all models were adjusted for the same variables: age, sex, cancer diagnosis, cancer stage, PS, and BMI, and the full models can be found in Appendix IV, Suppl Table 2. Table 5-4 depicts the adjusted odds ratios (OR) of experiencing weight loss with different levels of severity, in patients with moderately and severely reduced FI, as compared to those with normal intake. In each of the adjusted MLR models, reduced FI had the strongest association to WL, and each of the models demonstrated similar associations (i.e. overlapping OR with 95% CI; no differences between beta coefficients) between FI categories and WL severity (Table 5-4; Appendix IV Suppl Table 2). The odds of belonging to the most severe WL category ($\geq 15.0\%$) for patients reporting severely reduced FI (vs. normal) were overlapping for each of the 3 tools: the PG-SGA (OR 15.3 (95% CI 11.3-20.7), P<0.001), the I-VVAS (OR 14.4 (95% CI 9.3-22.2), P<0.001), and the MNA: OR 26.9 (95% CI 14.3-50.5), P<0.001; Table 5-4). Likewise the odds of WL $\geq 15.0\%$ for patients reporting moderately reduced FI (vs. normal) also overlapped for the PGSGA (OR 5.02 (95% CI 3.95-6.38), P<0.001), the *I*-VVAS (OR 5.62 (95% CI 3.83-8.26, P<0.001)), and the MNA (OR 7.1 (95% CI 4.3-11.6), P<0.001; Table 5-4).

Reductions in FI did not account for WL in all patients with cancer (Table 5-5). For example, the majority of patients with weight stability reported a normal FI (76% MNA; 61% PG-SGA; 63% *I*-VVAS), however patients with weight stability also reported moderately or severely reduced FI (24% MNA; 39% PG-SGA; 37% *I*-VVAS; Table 5-5). Similarly, a small

proportion of patients with \geq 15% WL reported a normal FI (13% MNA; 12% PG-SGA; 11% *I*-VVAS; Table 5-5). These results are not unexpected given that the etiology of cancer-associated WL is driven by a variable combination of reduced FI and altered metabolism.

5.4 Discussion

The pathophysiology of cancer cachexia is said to be "....characterized by a negative protein and energy balance driven by a variable combination of 2 major etiologies, reduced FI and abnormal metabolism".¹ The identification of reduced FI is suggested to be a diagnostic criterion for cancer cachexia, yet there has been no consensus regarding how reductions in FI ought to be measured.¹ We evaluated the quantitative significance of reduced FI as a driver of cancer-associated WL. This is the first study to demonstrate equivalency between three different assessments of FI, the MNA, PG-SGA, and I-VVAS, and their association to cancer-associated WL. We approached this using MLR analyses to determine the probability of weight loss (categorical dependent variable) in patients with varying levels of FI. Our findings suggest that reduced FI had the strongest association to WL, in models adjusted for age, sex, cancer site, cancer stage, performance status, and BMI. These findings were consistent across three independent, international samples totaling 11,704 patients. Patient-reported "severely reduced food intake" captured subsets of patients with a high likelihood (OR>14) of having already lost \geq 15% of their body weight at first assessment. This is not surprising, as mean daily energy intakes for patients with the lowest *I*-VVAS scores (≤ 4) are estimated to between 5-13 kcal/kg/day, which is grossly insifficient.¹⁷

The MNA, PG-SGA, and *I*-VVAS are major clinical assessments recommended in oncology nutrition clinical practice ², and despite their diversity of concept and scales, the questions used to identify low levels of FI associated with WL. These tools assess FI based on

patients' own estimates of current or recent FI in relation to what is normal for them, which allows patients to directly convey the impact cancer has on them.²⁵ There are however, limitations to this analysis. The data used in this study was collected at a single point in time (i.e. to determine the need for nutrition care), and as a result there are discrepancies between the time frames for which reductions in FI (e.g. current or past 3 months) were recorded in relation to the time frame over which WL occurred (e.g. previous 6 months or from usual body weight). We do not have information regarding the actual onset of FI problems or about fluctuations in intake and body weight that may have occurred up until the point of assessment. Additionally, we did not account for alterations in metabolism (e.g. tumor mediated catabolic factors, systemic inflammation, and/or increased energy expenditure), which are also important in the etiology of cancer-associated WL.

In cancer cachexia, the respective contributions of FI and altered metabolism are painted as a spectrum; most patients experience both to some degree, and in others it is reduced FI or hypermetabolism that have the predominant effect on weight loss.²⁶ Our results indirectly support this concept; a subset of patients (N=2089, 18%; Table 5-5) reported normal FI but also had WL of varying degrees. WL in these patients may be attributable to various alterations in metabolism that have not been accounted for which has also been observed by others.²⁷⁻²⁹ However, at present we have an incomplete understanding about how to clinically identify or evaluate alterations in metabolism and there are currently no viable treatment options, therefore treatment for cancer cachexia has predominantly focused on improving FI to support weight maintenance and weight gain, albeit with limited sucess.³⁰⁻³²

There are many factors that conspire to reduce FI in patients with cancer including a decreased central drive to eat (anorexia), various symptoms (stomatitis, early satiety,

constipation, nausea, vomiting, pain), depression, anxiety, functional impairment, social isolation, lack of finances, poor dentition and dietary habits, all of which should be recognized early. ^{26,30} Identification of specific issues affecting FI should be prioritized, as they may be readily managed and reversed by appropriate treatments. Optimization of nutritional intake can be achieved when cachexia is managed within a multimodal context.^{27,33} This includes medical management of pain and symptoms, interventions to increase physical activity to promote anabolism, modulation of inflammation, and psychosocial support.³⁴ When barriers to FI are removed, the escalation of nutrition care (e.g. from oral to enteral feeding) should be viewed as an iterative process based on changes to FI and body weight response.²⁷

In order to provide the right nutrition interventions at the right time, identification of patients at risk of nutrition problems (WL and reductions in FI) is an essential first step ^{3,35}, and if identified, followed-up with a comprehensive nutrition assessment, and appropriate interventions defined in the context of individual prognosis and treatment goals.² Our findings underscore the clinical importance of assessing patients' FI using clinically expedient assessments as patients with reduced food intake have a high probability of presenting with severe WL.

5.5 References

- 1. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.
- 2. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 3. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 4. Cederholm T, Jensen GL. To Create a Consensus on Malnutrition Diagnostic Criteria. JPEN J Parenter Enteral Nutr. 2017; 41:311-314.
- French Speaking Society of Clinical Nutrition, Metabolism. Clinical nutrition guidelines of the French Speaking Society of Clinical Nutrition and Metabolism (SFNEP): Summary of recommendations for adults undergoing non-surgical anticancer treatment. Dig Liver Dis. 2014; 46:667-674.
- Thompson KL, Elliott L, Fuchs-Tarlovsky V, Levin RM, Voss AC, Piemonte T. Oncology Evidence-Based Nutrition Practice Guideline for Adults. J Acad Nutr Diet. 2017; 117:297-310 e247.
- 7. Mislang AR, Di Donato S, Hubbard J, et al. Nutritional management of older adults with gastrointestinal cancers: An International Society of Geriatric Oncology (SIOG) review paper. J Geriatr Oncol. 2018; 9:382-392.
- 8. Clinical Oncology Society of Australia (COSA). Head and Neck Guideline Steering Committee. Evidence-based practice guidelines for the nutritional management of adult patients with head and neck cancer. Sydney: Cancer Council Australia. 2018; https://wiki.cancer.org.au/australiawiki/index.php?oldid=116710. Accessed July 19, 2018.
- 9. Caccialanza R, Pedrazzoli P, Cereda E, et al. Nutritional Support in Cancer Patients: A Position Paper from the Italian Society of Medical Oncology (AIOM) and the Italian Society of Artificial Nutrition and Metabolism (SINPE). J Cancer. 2016; 7:131-135.
- 10. Isenring E, Zabel R, Bannister M, et al. Updated evidence-based practice guidelines for the nutritional management of patients receiving radiation therapy and/or chemotherapy. Nutrition & Dietetics. 2013; 70:312-324.
- 11. MacDonald N. Terminology in cancer cachexia: importance and status. Curr Opin Clin Nutr Metab Care. 2012; 15:220-225.
- 12. Senesse P, Isambert A, Janiszewski C, et al. Management of cancer cachexia and guidelines implementation in a comprehensive cancer center: a physician-led cancer

nutrition program adapted to the practices of a country. Journal of pain and symptom management. 2017; doi:10.1016/j.jpainsymman.2017.01.010.

- Martin L, Kubrak C. How much does reduced food intake contribute to cancer-associated weight loss? Curr Opin Support Palliat Care. 2018; doi:10.1097/SPC.0000000000379.
- 14. Jager-Wittenaar H, Ottery FD. Assessing nutritional status in cancer: role of the Patient-Generated Subjective Global Assessment. Curr Opin Clin Nutr Metab Care. 2017; 20:322-329.
- 15. Ottery FD. Definition of standardized nutritional assessment and interventional pathways in oncology. Nutrition. 1996; 12:S15-19.
- Thibault R, Goujon N, Le Gallic E, et al. Use of 10-point analogue scales to estimate dietary intake: a prospective study in patients nutritionally at-risk. Clin Nutr. 2009; 28:134-140.
- 17. Guerdoux-Ninot E, Flori N, Janiszewski C, et al. Assessing dietary intake in accordance with guidelines: Useful correlations with an Ingesta Verbal/Visual Analogue Scale in medical oncology patients. Clinical Nutrition. 2018; doi:10.1016/j.clnu.2018.06.974.
- 18. Vellas B, Villars H, Abellan G, et al. Overview of the MNA Its history and challenges. J Nutr Health Aging. 2006; 10:456-635.
- 19. Rubenstein LZ, Harker JO, Salva A, Guigoz Y, Vellas B. Screening for undernutrition in geriatric practice: developing the short-form mini-nutritional assessment (MNA-SF). J Gerontol A Bio Sci Med Sci. 2001; 56:M366-372.
- 20. Blum D, Omlin A, Baracos VE, et al. Cancer cachexia: a systematic literature review of items and domains associated with involuntary weight loss in cancer. Crit Rev Oncol Hematol. 2011; 80:114-144.
- 21. Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancer-associated weight loss. J Clin Oncol. 2015; 33:90-99.
- 22. Martin L, Watanabe S, Fainsinger R, et al. Prognostic factors in patients with advanced cancer: use of the patient-generated subjective global assessment in survival prediction. J Clin Oncol. 2010; 28:4376-4383.
- 23. Gioulbasanis I, Martin L, Baracos VE, Thezenas S, Koinis F, Senesse P. Nutritional assessment in overweight and obese patients with metastatic cancer: does it make sense? Ann Oncol. 2015; 26:217-221.
- 24. Muscaritoli M, Lucia S, Farcomeni A, et al. Prevalence of malnutrition in patients at first medical oncology visit: the PreMiO study. Oncotarget. 2017; 8:79884-79896.

- 25. LeBlanc TW, Abernethy AP. Patient-reported outcomes in cancer care hearing the patient voice at greater volume. Nat Rev Clin Oncol. 2017; 14:763-772.
- 26. Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol. 2013; 10:90-99.
- 27. Nasrah R, Kanbalian M, Van Der Borch C, Swinton N, Wing S, Jagoe RT. Defining the role of dietary intake in determining weight change in patients with cancer cachexia. Clin Nutr. 2018; 37:235-241.
- 28. Vazeille C, Jouinot A, Durand JP, et al. Relation between hypermetabolism, cachexia, and survival in cancer patients: a prospective study in 390 cancer patients before initiation of anticancer therapy. Am J Clin Nutr. 2017; 105:1139-1147.
- 29. Lee JY, Hong N, Kim HR, et al. Effects of Serum Albumin, Calcium Levels, Cancer Stage and Performance Status on Weight Loss in Parathyroid Hormone-Related Peptide Positive or Negative Patients with Cancer. Endocrinolo Metab (Seoul, Korea). 2018; 33:97-104.
- 30. Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.
- 31. Baldwin C, Spiro A, Ahern R, Emery PW. Oral nutritional interventions in malnourished patients with cancer: a systematic review and meta-analysis. J Natl Cancer Inst. 2012; 104:371-385.
- 32. de van der Schueren MAE, Laviano A, Blanchard H, Jourdan M, Arends J, Baracos VE. Systematic review and meta-analysis of the evidence for oral nutritional intervention on nutritional and clinical outcomes during chemo(radio)therapy: current evidence and guidance for design of future trials. Ann Oncol. 2018; 29:1141-1153.
- 33. Gagnon B, Murphy J, Eades M, et al. A prospective evaluation of an interdisciplinary nutrition-rehabilitation program for patients with advanced cancer. Curr Oncol. 2013; 20:310-318.
- 34. Maddocks M, Hopkinson J, Conibear J, Reeves A, Shaw C, Fearon KC. Practical multimodal care for cancer cachexia. Curr Opin Support Palliat Care. 2016; 10:298-305.
- 35. Aapro M, Arends J, Bozzetti F, et al. Early recognition of malnutrition and cachexia in the cancer patient: a position paper of a European School of Oncology Task Force. Ann Oncol. 2014; 25:1492-1499.

		PG	S-SGA Data Set N=6161	<i>I-</i> VVAS Data Set N=3186	MNA Data Set N=2357								
Contributor	Martin ²¹	Kubrak ^{21, UP}	Gagnon ³⁶	Chasen ³³	Laird ³⁷	Senesse ³⁸	Senesse ^{UP}	Muscaritoli ³⁹					
Data Collection		Nutrition Risk Screening in Prospective Observational Cohorts											
Country		Can	ada		Europe	France	France	Italy					
Health Care Setting	Medical Oncology / Palliative Care	Medical Oncology	Medical Oncology	Palliative Care	Palliative Care	Medical Oncology	Medical Oncology	Medical Oncology					
Total N	4647	494	382	150	488	3186	605	1752					

Table 5-1. A description of study cohorts that were selected from the International Cachexia Database (ICDR) for this study.

MNA, Mini-Nutrition Assessment; N, number; PGSGA, Patient-generated subjective global assessment; *I*-VVAS, *Ingesta*-verbal/visual analogue scale; WL, weight loss; UP, unpublished

Variables	Current Food Measured v PGSGA Food Categoric (N=6161	Intake vith Intake es)	Current Food I measured with point Visual An Scale (N=3186)	ntake a 10- alogue	Past Food Intake Measured with MNA Food Intake Categories (N=2357)		
	Ν	%	Ν	%	Ν	%	
Region							
Canada	5373	92.1	-	-			
Europe	488	7.9	-	-			
France	-	-	3186	100	605	26	
Italy	-	-	-	-	1752	74	
Sex							
Male	3603	58.5	1633	51.3	1150	48.8	
Female	2558	41.5	1553	48.7	1207	51.2	
Age, years (mean, SD) Status	64.6 (12.1)		62.1 (12.4)		66.2 (12.9)		
Censored	1522	24.7	728	22.8	-	-	
Dead	4639	75.3	2458	77.2	-	-	
Overall Survival, months (median, 95% CI)	9.6 (9.2-10.1)		10.9 (10.1-11.7)		-	-	
Cancer Diagnosis* Head & neck	1191	19.3	346	10.9	99	4.2	
Breast	258	4.2	533	16.7	481	20.4	
Upper gastrointestinal	950	15.4	433	13.6	384	16.3	
Lower gastrointestinal	1009	16.4	519	16.3	403	17.1	
organs Respiratory	1913	31.1	363	11.4	364	15.4	
Other	454	7.4	514	16.1	193	8.2	
Cancer Stage ⁺		,					
1	280	4.5	85	2.7	240	10.0	
2	442	7.2	166	5.2	274	11.6	
3	859	13.9	299	9.4	423	17.9	
4	4580	74.3	2636	82.7	1420	60.2	
ECOG PS							
PS 0	1387	22.5	554	17.4	-	-	
PS 1	2110	34.2	1055	33.1	-	-	
PS 2	1181	19.2	994	31.2	-	-	
PS 3	1327	21.5	567	17.8	-	-	
PS 4	156	2.5	16	0.5	-	-	

Table 5-2. Patient characteristics for the three study cohorts

MNA Section C: Mobility bed or chair bound					225	9.5
able to get out of bed / chair but does not go out					417	17.7
goes out					1715	72.8
% WL (mean, SD)	-8.0 (7.5)		-7.2 (6.7)		-5.8 (6.1)	
WL Categories						
$\pm 2.4\%$ (weight	1777	28.8	928	29.1	864	36.7
2.5-5.9%	1139	18.5	655	20.6	534	22.7
6.0-10.9%	1347	21.9	818	25.7	537	22.8
11.0-14.9%	816	13.2	376	11.8	211	9.0
≥15.0%	1082	17.6	409	12.8	211	9.0
BMI, kg/m2 (mean, SD) BMI Categories	25.0 (5.3)		23.5 (4.6)		24.6 (4.4)	
	935	15.2	707	22.2	278	11.8
20.0-21.9	863	14.0	551	17.3	337	14.3
22 0-24 9	1540	25.0	890	27.9	792	33.6
25.0-27.9	1289	20.9	571	17.9	528	22.4
≥28.0	1534	24.9	467	14.7	422	17.9
WL Grade						
Grade 0	1066	17.3	417	13.1	453	19.2
Grade 1	900	14.6	510	16.0	462	19.6
Grade 2	992	16.1	538	16.9	396	16.8
Grade 3	1766	28.7	973	30.5	697	29.6
Grade 4	1437	23.3	748	23.5	349	14.8
PGSGA Box 2: current food intake						
normal food	2112	34.3	-	-	-	-
normal food, less amount	2765	44.9	-	-	-	-
little solid food	588	9.5	-	-	-	-
only liquids or ONS	289	4.7	-	-	-	-
very little anything	407	6.6	-	-	-	-
I-VVAS						
1	-	-	113	3.5	-	-
2	-	-	171	5.4	-	-
3	-	-	233	7.3	-	-
4	-	-	220	6.9	-	-
5	-	-	481	15.1	-	-
6	-	-	212	6.7	-	-

7	-	-	281	8.8	-	-
8	-	-	229	7.2	-	-
9	-	-	53	1.7	-	-
10	-	-	1193	37.4	-	-
MNA Section A: food intake past 3 months						
severe decrease	-	-	-	-	267	11.3
moderate decrease	-	-	-	-	1000	42.4
no decrease	-	-	-	-	1090	46.2

BMI, body mass index; ECOG PS, Eastern Cooperative Oncology Group performance status; ; *I*-VVAS, *Ingesta*-Verbal/Visual Analogue Scale; MNA, mini-nutrition assessment; N, number; PG-SGA, Patient-Generated Subjective Global Assessment; SD, standard deviation; WL, weight loss

*Upper gastrointestinal (esophageal, stomach, pancreas, liver, biliary tract, small bowel); lower gastrointestinal (colon, rectum, anus); genitourinary (kidney, bladder, adrenal, prostate, testes, penis); Other (gynecological, hematological, peritoneum, unknown, thyroid)

†based on AJCC version 6 or 7

(-) indicates there is no data for these fields

Categorie reduced f	s for severity of ood intake			Comparison between categories of food intake within each assessment tool	Mean di betw assessm catego	ifferences veen food nents tools ory of food	Comparisons between food intake assessment tools for each category of food intake	
PG-SGA		Ν	Mean %WL (SD)	P-Value*	<i>I</i> -VVAS ^a	MNA ^b	PG-SGA ^c	P-Value*
•	severely reduced	1284	-12.2 (7.8)		1.3	0.0		^a P<0.05
٠	moderately reduced	2765	-8.9 (7.2)	■◆● <0.001	1.0	1.8		^{a,b,c} P<0.05
	normal	2112	-4.1 (5.8)		0.1	0.3		^b P<0.05
<i>I</i> -VVAS								
•	severely reduced	737	-10.9 (7.2)			-1.3	-1.3	^{a, b} P<0.05
•	moderately reduced	1256	-7.9 (6.3)	■◆● <0.001		0.8	-1.0	^{a,b,c} P<0.05
	normal	1193	-4.0 (5.1)			1.0	-0.1	^b P<0.05
MNA								
•	severely reduced	267	-12.2 (7.8)		1.3		0.0	^a P<0.05
•	moderately reduced	1000	-7.1 (5.4)	■◆● <0.001	-0.8		-1.8	^{a,b,c} P<0.05
	normal	1090	-3.0 (4.4)		-1.0		-1.1	^{a,c} P<0.05
Overall								
•	severely reduced	2288	-12.1 (11.5)					
•	moderately reduced	5021	-8.3 (8.1)	■◆● <0.001				
	normal	4395	-3.8 (5.3)					

Table 5-3. Mean percent (%) weight loss by category of food intake for each measurement tool.

MNA, Mini-Nutrition Assessment; PG-SGA, Patient-Generated Subjective Global Assessment; *I-VVAS*, *Ingesta* Verbal/Visual Analogue Scale; WL, weight loss *Comparison of means with One-way ANOVA

						8	<i>i j</i>			i				
PG-SGA ML model ¹	R	Ν	WL 2.5-5.9% N= 1139 (28.8%) ⁄0)	I	WL 6.0-10.9% N= 1347 (21.9%) ()		WL 11.0-14.9% N= 816 (13.2%	/o o)	I	WL ≥15.0% N= 1082 (17.6%))	
Current Food Intake	N (%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	
severely reduced*	1284 (20.8)	1.09 (0.14)	2.98 (2.26-3.92)	< 0.001	1.79 (0.13)	5.98 (4.59-7.78)	< 0.001	2.37 (0.16)	10.70 (7.85-14.58)	< 0.001	2.73 (0.16)	15.27 (11.25-20.73)	< 0.001	
normal food, reduced amount	2765 (44.9)	0.66 (0.09)	1.94 (1.63-2.30)	<0.001	1.08 (0.09)	2.93 (2.46-3.50)	<0.001	1.45 (0.12)	4.25 (3.36-5.38)	<0.001	1.61 (0.12)	5.02 (3.95-6.38)	<0.001	
normal food, normal amount	2112 (34.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)		
I-VVAS MLR model ²		WL 2.5-5.9% N= 655 (20.6%)			WL 6.0-10.9% N= 818 (25.7%)			WL 11.0-14.9% N= 376 (11.8%)			WL ≥15.0% N= 409 (12.8%)			
Current Food	N (%)	β	OR	P-	0 (CE)	OR	P-	β	OR	Р-	0 (077)	OR	P-	
Intake	()	(SE)	(95% CI)	value	р (SE)	(95% CI)	value	(SE)	(95% CI)	value	β (SE)	(95% CI)	value	
Intake severely reduced intake (scores 1-4)	736 (23.1)	(SE) 0.86 (0.18)	(95% CI) 2.37 (1.68-3.35)	value <0.001	1.53 (0.16)	(95% CI) 4.62 (3.36-6.37)	value <0.001	(SE) 2.13 (0.21)	(95% CI) 8.41 (5.63-12.58)	value <0.001	β (SE) 2.67 (0.22)	(95% CI) 14.36 (9.28-22.23)	value <0.001	
Intake severely reduced intake (scores 1-4) moderately reduced intake (scores 5-9)	736 (23.1) 1256 (39.4)	(SE) 0.86 (0.18) 0.82 (0.12)	(95% CI) 2.37 (1.68-3.35) 2.28 (1.79-2.90)	value <0.001 <0.001	1.53 (0.16) 1.13 (0.12)	(95% CI) 4.62 (3.36-6.37) 3.10 (2.45-3.92)	value <0.001 <0.001	(SE) 2.13 (0.21) 1.45 (0.17)	(95% CI) 8.41 (5.63-12.58) 4.25 (3.04-5.94)	value <0.001 <0.001	β (SE) 2.67 (0.22) 1.73 (0.20)	(95% CI) 14.36 (9.28-22.23) 5.62 (3.83-8.26)	value <0.001 <0.001	
Intake severely reduced intake (scores 1-4) moderately reduced intake (scores 5-9) normal (score 10)	736 (23.1) 1256 (39.4) 1193 (37.5)	(SE) 0.86 (0.18) 0.82 (0.12)	(95% CI) 2.37 (1.68-3.35) 2.28 (1.79-2.90) 1.0 (ref)	value <0.001 <0.001	1.53 (0.16) 1.13 (0.12)	(95% CI) 4.62 (3.36-6.37) 3.10 (2.45-3.92) 1.0 (ref)	value <0.001 <0.001	(SE) 2.13 (0.21) 1.45 (0.17)	(95% CI) 8.41 (5.63-12.58) 4.25 (3.04-5.94) 1.0 (ref)	value <0.001 <0.001	β (SE) 2.67 (0.22) 1.73 (0.20)	(95% CI) 14.36 (9.28-22.23) 5.62 (3.83-8.26) 1.0 (ref)	value <0.001 <0.001	
Intake severely reduced intake (scores 1-4) moderately reduced intake (scores 5-9) normal (score 10) MNA MLR model ³	736 (23.1) 1256 (39.4) 1193 (37.5)	(SE) 0.86 (0.18) 0.82 (0.12)	(95% CI) 2.37 (1.68-3.35) 2.28 (1.79-2.90) 1.0 (ref) WL 2.5-5.9% N=534 (22.7%	value <0.001 <0.001	1.53 (0.16) 1.13 (0.12)	(95% CI) 4.62 (3.36-6.37) 3.10 (2.45-3.92) 1.0 (ref) WL 6.0-10.9% N=537 (22.8%	value <0.001 <0.001	(SE) 2.13 (0.21) 1.45 (0.17)	(95% CI) 8.41 (5.63-12.58) 4.25 (3.04-5.94) 1.0 (ref) WL 11.0-14.9% N=211 (9.0%)	value <0.001 <0.001	β (SE) 2.67 (0.22) 1.73 (0.20)	(95% CI) 14.36 (9.28-22.23) 5.62 (3.83-8.26) 1.0 (ref) WL ≥15.0% N=211 (9.0%)	value <0.001 <0.001	
Intake severely reduced intake (scores 1-4) moderately reduced intake (scores 5-9) normal (score 10) MNA MLR model ³ Past Food Intake	736 (23.1) 1256 (39.4) 1193 (37.5) N (%)	(SE) 0.86 (0.18) 0.82 (0.12) β (SE)	(95% CI) 2.37 (1.68-3.35) 2.28 (1.79-2.90) 1.0 (ref) WL 2.5-5.9% N=534 (22.7% OR (95% CI)	value <0.001 <0.001	β (SE) 1.53 (0.16) 1.13 (0.12) β (SE)	(95% CI) 4.62 (3.36-6.37) 3.10 (2.45-3.92) 1.0 (ref) WL 6.0-10.9% N=537 (22.8% OR (95% CI)	value <0.001 <0.001 P- value	(SE) 2.13 (0.21) 1.45 (0.17) β (SE)	(95% CI) 8.41 (5.63-12.58) 4.25 (3.04-5.94) 1.0 (ref) WL 11.0-14.9% N=211 (9.0%) OR (95% CI)	value <0.001 <0.001 6 9 9 9 9 9 9 9 9 9 9	β (SE) 2.67 (0.22) 1.73 (0.20) β (SE)	(95% CI) 14.36 (9.28-22.23) 5.62 (3.83-8.26) 1.0 (ref) WL ≥15.0% N=211 (9.0%) OR (95% CI)	value <0.001 <0.001 P- value	

Table 5-4. Data from three multivariable multinomial logistic regression (MLR) models highlighting the association between assessments of food intake and cancer-associated weight loss (WL).

reduced	(11.3)	(0.29)	(1.24-3.80)		(0.26)	(4.12-11.46)		(0.31)	(5.33-17.75)		(0.32)	(14.29-50.49)	
moderately reduced	1000 (42.4)	1.32 (0.13)	3.74 (2.89-4.83)	<0001	2.00 (0.14)	7.38 (5.57-9.78)	< 0.001	1.78 (0.20)	5.94 (3.99-8.86)	< 0.001	1.95 (0.25)	7.05 (4.29-11.60)	< 0.001
not reduced	1090 (46.2)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	

β, beta coefficient; ; *I*-VVAS, *Ingesta*-verbal/visual analogue scale, MNA, Mini-Nutrition Assessment; MLR, multinomial logistic regression; N, number; OR, odds ratio; PGSGA, Patient-generated subjective global assessment; SE, standard error; WL, weight loss

*Severely reduced includes the following PG-SGA categories (little solid food, only liquids/oral nutritional supplements, very little of anything) ¹PG-SGA MLR Model - Reference weight stable (+/- 2.4%) N=1777 (28.8%); Intercept only model: -2 log likelihood (LL)=18682.909, AIC 18690.909; Final Model: -2LL=16403.448, AIC=16555.448, χ 2=2279.461 (df=72), P<0.001; Pseudo R2 (Nagelkerke) = 0.323; Pearson goodness-of-fit P=0.063 ²*I*-VVAS MLR Model - Reference weight stable (+/-2.4%) N=928 (21.9%); Intercept only model: -2 LL=9671.159, AIC=9679.159; Final Model: -2LL=8658.277, AIC=8818.277, χ 2=1012.883 (df=76), P<0.001; Pseudo r2 (Nagelkerke)=0.285; Pearson goodness-of-fit P=0.792 ³MNA MLR Model - Reference weight stable (±2.4%) N=864 (36.7%); Intercept only model: -2 log likelihood (LL)= 6755.457, AIC 6763.457; Final Model: -2LL=5533.028, AIC=5677.028, χ 2=1222.429 (df=68), P<0.001; Pseudo R2 (Nagelkerke)=0.427, Pearson goodness-of-fit P=0.198

WL Categories			PG-SGA FI	Categories			I-VVAS FI (Categories		MNA* FI Categories			
		Severe	Moderate	Normal	Total	Severe	Moderate	Normal	Total	Severe	Moderate	Normal	Total
≥-15.0%	N	426 _a	532 _b	124 _c	1082	195 _a	168 _b	46 _c	409	95 _a	89 _b	27 _c	211
	% within WL Categories	39.4%	49.2%	11.5%	100%	47.7%	41.1%	11.2%	100%	45.0%	42.2%	12.8%	100%
	% within FI Categories	33.2%	19.2%	5.9%	17.6%	26.5%	13.4%	3.9%	12.8%	35.6%	8.9%	2.5%	9.0%
	% of Total	6.9%	8.6%	2.0%	17.6%	6.1%	5.3%	1.4%	12.8%	4.0%	3.8%	1.1%	9.0%
-11.0 to - 14.9%	N	257 _a	431 _b	128 _c	816	138 _a	169 _b	69 _c	376	46 _a	116 _b	49 _c	211
	% within WL Categories	31.5%	52.8%	15.7%	100%	36.7%	44.9%	18.4%	100%	21.8%	55.0%	23.2%	100%
	% within FI Categories	20.0%	15.6%	6.1%	13.2%	18.7%	13.5%	5.8%	11.8%	17.2%	11.6%	4.5%	9.0%
	% of Total	4.2%	7.0%	2.1%	13.2%	4.3%	5.3%	2.2%	11.8%	2.0%	4.9%	2.1%	9.0%
-6.0 to - 10.9%	N	309 _a	704 _a	334 _b	1347	207	376 _a	235 _b	818	68 _a	340 _b	129 _c	537
100770	% within WL Categories	22.9%	52.3%	24.8%	100%	25.3%	46.0%	28.7%	100%	12.7%	63.3%	24.0%	100%
	% within FI Categories	24.1%	25.5%	15.8%	21.9%	28.1%	29.9%	19.7%	25.7%	25.5%	34.0%	11.8%	22.8%
	% of Total	5.0%	11.4%	5.4%	21.9%	6.5%	11.8%	7.4%	25.7%	2.9%	14.4%	5.5%	22.8%
-2.5 to - 5.9%	N	167 _a	530 _b	442 _b	1139	109 _a	285 _b	261 _b	655	30 _a	274 _b	230 _c	534
J. Y %	% within WL Categories	14.7%	46.5%	38.8%	100%	16.6%	43.5%	39.8%	100%	5.6%	51.3%	43.1%	100%
	% within FI Categories	13.0%	19.2%	20.9%	18.5%	14.8%	22.7%	21.9%	20.6%	11.2%	27.4%	21.1%	22.7%
	% of Total	2.7%	8.6%	7.2%	18.5%	3.4%	8.9%	8.2%	20.6%	1.3%	11.6%	9.8%	22.7%

Table 5-5. The crosstabulation for categories of weight loss (WL) by categories of food intake (FI) for each of the food intake assessment tools: PG-SGA, *I*-VVAS, MNA

±2.4% (weight stable)	Ν	125 _a	568 _b	1084 _c	1777	88 _a	258 _b	582 _c	928	28 _a	181 _b	655 _c	864
	% within WL Categories	7.0%	32.0%	61.0%	100%	9.5%	27.8%	62.7%	100%	3.2%	20.9%	75.8%	100%
	% within FI Categories	9.7%	20.5%	51.3%	28.8%	11.9%	20.5%	48.8%	29.1%	10.5%	18.1%	60.1%	36.7%
	% of Total	2.0%	9.2%	17.6%	28.8%	2.8%	8.1%	18.3%	29.1%	1.2%	7.7%	27.8%	36.7%
Total	N	1284	2765	2112	6161	737	1256	1193	3186	267	1000	1090	2357
	% within WL Categories	20.8%	44.9%	34.3%	100%	23.1%	39.4%	37.4%	100%	11.3%	42.4%	46.2%	100%
	% within FI Categories	100%	100%	100%	100%	100%	100%	100%	100%	100.0 %	100.0%	100.0%	100%
	% of Total	20.8%	44.9%	34.3%	100%	23.1%	39.4%	37.4%	100%	11.3%	42.4%	46.2%	100%

FI, food intake; MNA, Mini-Nutrition Assessment; N, number; PGSGA, Patient-generated subjective global assessment; *I*-VVAS, *Ingesta*-verbal/visual analogue scale; WL, weight loss

_{a,b,c} Subscript letters denote significant differences (P<0.05; Pearson Chi-Square, Bonferroni correction) in column proportions between categories of food intake within each weight loss category for each assessment tool.

Chapter 6: The variable contribution of reduced food intake and systemic inflammation to cancer-associated weight loss

6.1 Introduction

Cancer cachexia is defined as a syndrome of involuntary weight loss (WL), which is characterized by the progressive loss of skeletal muscle, which can occur with or without loss of fat mass.¹ The etiology of cancer-associated WL is suggested to be driven by a variable combination of reduced food intake (FI) and altered metabolism. We previously demonstrated that reductions in FI independently associated with increased probability of WL in a multivariable regression model adjusted for covariates including age, sex, cancer site, cancer stage, performance status, and body mass index (BMI).² However, this model did not explain all the variation in WL, and one explanation was that we did not account for alterations in metabolism. We hypothesized that the addition of covariates related to altered metabolism would contribute to explaining the etiology of WL in cachectic patients.

Alterations in metabolism are a defining characteristic of cancer cachexia, and their manifestations are thought to explain why cancer cachexia cannot be fully reversed by nutrition support alone (Figure 6-1).¹ Alterations in metabolism arise from a variable combination of factors including tumor metabolism, tumor-derived catabolic factors, and pro-inflammatory mediators. The majority of our understanding about metabolic alterations in response to malignancy is from animal studies, which are not always consistent with findings in human subjects³, and at present the identification and characterization of metabolic alterations in patients with cancer remains elusive. There is no consensus regarding a measure that best represents altered metabolism in patients with cancer. There is, however, clinical evidence suggesting an independent contribution of systemic inflammation (e.g. elevated C-reactive protein (CRP)), hypermetabolism (i.e. increased energy expenditure), and tumor-derived factors

such as parathyroid hormone-related peptide (PTHrP) to an increased risk of cancer-associated WL.⁴⁻⁶



Figure 6-1. A conceptual diagram representing the pathophysiology of cancer cachexia

A conceptual diagram representing the pathophysiology of cancer cachexia, which is driven by a variable combination of two major etiologies: reduced food intake (FI) and altered metabolism.¹ There are many factors that conspire to cause involuntary WL in patients with cancer cachexia. Reductions in FI can be directly or indirectly related to the presence of the tumor e.g. nutrition impact symptoms due to location of the tumor in the gastrointestinal tract or development of symptoms in response to cancer therapies; decreased central drive to eat in the central nervous system in response to pro-inflammatory tumor-mediated factors, and/or cancer-related pain or nausea. Non-cancer related psychosocial factors can also act as barriers to FI e.g. mood disorders, food insecurity, lack of social support for assistance with meal preparation, poor dietary habits, etc. Metabolic alterations (represented by light grey boxes) are also key drivers of cancer-associated WL, and include tumor-mediated catabolic factors that act directly on skeletal muscle and adipose tissue, or via pro-inflammatory factors arising from interactions with the immune system. Tumor metabolism and inflammation lead to hypermetabolism resulting in increased proteolysis and lipolysis to support the increased energy requirements. Inflammation in the CNS can elicit the production of glucocorticoids from the adrenal glands, which acts directly on skeletal muscle. However, the specific nature of these metabolic alterations continues to evolve and requires further elucidation. Ultimately our understanding of the factors contributing to cancer-associated WL will only be fully described when both reductions in FI and alterations in metabolism are accounted for. (figure adapted from Martin & Kubrak⁷, Baracos et al.³)

CRP is the most common clinically available indicator of systemic inflammation reported in the cachexia literature, and has been included as a diagnostic criterion in prior definitions of cachexia.^{8,9} CRP has also been demonstrated to be an important prognostic marker in oncology, and when combined with clinical characteristics such as WL, performance status (PS), and FI improves the identification of patient subsets with worse prognosis and who are unlikely to benefit from treatment.⁹⁻¹⁷ Threshold values for CRP (\geq 5, \geq 10 mg/L) commonly used in the oncology setting have been defined based on their association to overall survival, and while weight losing cancer patients often have higher CRP levels, values of CRP that relate optimally to WL as an outcome have not been extensively studied.^{6,8,9,18,19}

Our international research groups have aggregated clinical data of patients with advanced cancers who are at risk of cancer cachexia to create the International Cancer Cachexia Data Repository (ICCDR).^{2,20,21} Data from the ICCDR was used for this analysis, to address the following primary objectives: 1) evaluate the relationship between CRP and WL, and determine optimal CRP values for risk stratification based on WL, 2) evaluate the contribution of CRP *and* FI as explanatory variables for WL. A secondary objective was to determine if we could stratify patients according to overall survival based on the combination of CRP, FI, and WL.

6.2 Methods

6.2.1 Study population

Patients were selected from the ICCDR. A detailed description of the ICCDR has been provided elsewhere.^{2,20,21} Briefly, the ICCDR is an international effort to pool original study data that has been collected at several points of referral for nutrition care within the oncology setting: at entry to investigational cachexia clinical trials, at cancer diagnosis as standard of care, and at entry to supportive/palliative care programs. This analysis includes original study data that were

prospectively collected from 2002 to 2014, under the auspices of human ethics approvals from their respective institutions. To be included in the current analysis, patients required complete data for the following: demographic and clinical data (age, sex, cancer diagnosis (ICD-10) and AJCC stage (version 6.0 prior to 2011, version 7.0 thereafter), performance status (PS), body mass index (BMI), WL history), and a nutrition risk assessment that included a measure of FI and serum CRP (g/ml). For each original study cohort, high sensitivity serum CRP was collected at the same time as the nutrition risk assessment was completed, and analyzed in laboratories with automated analyzers that were standardized according to international certified reference material.²²

We did not select patients with weight gain (e.g. $\geq 2.5\%$) for the reasons that our primary outcome is cancer-associated WL, and we previously showed that weight gaining patients in this population had different prognosis from weight stable patients.²³

6.2.2 Assessments of food intake and C-reactive protein

Patient-reported FI was collected from one of two nutrition risk assessment tools: the Patient Generate Subjective Global Assessment (PG-SGA), and the *Ingesta*-Verbal/Visual Analogue Scale (*I*-VVAS). We previously demonstrated that each of these measurements scales could be combined into three distinct FI categories: normal, moderately, or severely reduced FI based on their association to %WL.²

6.2.3 Statistics

Summary statistics were used to describe patients. The primary outcome was WL history, recoded as the percentage of WL over the previous 6 months calculated as:

% weight loss (%WL) =

[(current weight in kg -previous weight in kg)/previous weight in kg]*100 Statistical analysis was carried out in three parts:

1) The relationship between serum CRP (mg/L) and %WL was evaluated, and optimal values were defined for CRP that related to %WL as follows: the distribution of CRP values was divided into deciles (i.e. 10 equal groups) and the mean %WL was calculated for each decile to explore the impact of increasing CRP on WL. A similar approach to define optimal values for WL and BMI has been previously reported.²¹ Deciles were compared based on differences in mean %WL using one-way ANOVA (post-hoc tests: Games-Howell). Deciles of CRP that did not have statistical differences for mean %WL were combined in to categories.

2) Multinomial logistic regression (MLR) evaluated the association between CRP and %WL adjusted for other variables likely to associate with WL including: FI, age (continuous), sex, cancer site (ICD-10), cancer stage (AJCC version 6.0 or 7.0), PS (ECOG), and BMI (<20.0, 20.0-21.9, 22.0-27.9, \geq 28.0)).²¹ MLR is an extension of logistic regression and is a robust statistical approach when the dependent variable has more than two categories. The dependent variable in this analysis is WL, which has 5 categories of severity (±2.4%, 2.5-5.9%, 6.0-10.9%, 11.0-14.9%, and \geq 15.0%), as outlined in the diagnostic criteria for cancer-associated WL.²¹ The reference group for the dependent variable was weight stable (±2.4%), to which all other categories of WL were compared. Results are reported as odds ratios (OR) and 95% confidence intervals (CI).

3) Survival analysis evaluated differences in overall survival (OS) for patients stratified by categories of CRP, FI, and %WL. OS was defined as the number of months a patient survived between the date of nutrition risk assessment and the date of death; patients were observed until

death or censored at their last confirmed contact with the health care system. Survival analysis was performed with the Kaplan-Meier method (comparisons with Cox-Mantel log-rank tests).

All analyses were completed using IBM SPSS Statistics for Windows version 23.0 (SPSS, Chicago, IL) and were considered statistically significant at the P < 0.05 level. Figures were drawn with GraphPad Prism version 7.04 for Windows (GraphPad Software, La Jolla California USA).

6.3 Results

6.3.1 Patient characteristics

A total of 3443 patients were selected and included in this analysis (Table 6-1). Patients represented patients with cancer cachexia: they had predominantly locally advanced or metastatic disease, with an average weight loss of $-7.3\% \pm 7.3$ over the previous 6 months.

6.3.2 CRP and cancer-associated weight loss

The relationship between CRP and % WL was examined by dividing serum CRP values (mg/L) into deciles (i.e. 10 equal groups), and calculating the mean %WL for each decile (Figure 6-1).





A graph depicting the mean %WL by decile of serum CRP (mg/L) values (mean CRP values are indicated for each decile). Different colored symbols ($\bullet \bullet \blacksquare$) represent significant differences in mean %WL (P<0.05) between deciles; deciles with similarly colored symbols were not significantly different, and were combined into three distinct CRP categories, highlighted by different shades of gray. The legend indicates the corresponding range of CRP values and mean % WL for each of the 3 CRP categories.

There were three distinct subsets with significant differences (P<0.001) in mean %WL (Figure 6-1). Deciles of CRP were therefore combined, and 3 categories of CRP were defined (<10.0 mg/L, 10.0-42.9 mg/L, and \geq 43.0 mg/L), representing increased CRP values that associated with increased WL. The values and mean % WL for each CRP category are listed in Figure 6-1.

The association between categories of CRP and increased severity of WL was evaluated with MLR analysis. The univariable MLR analysis can be found in Appendix V Suppl Table 1. In the multivariable analysis, the MLR model was adjusted for age, sex, cancer diagnosis, cancer stage, PS, BMI, and categories of food intake (Table 6-2). At the multivariable level, age and cancer stage were not associated with WL and were not included in the final MLR model. In the final MLR model, both reduced FI and CRP independently associated with increased probability of WL, demonstrating progressively increasing ORs as the severity of WL increased. Table 6-2 presents the adjusted odds ratio (OR) of experiencing WL with different levels of severity, for patients with moderate or severely reduced FI compared to those with normal intake, and with moderate (10.0-42.9 mg/L) or high (\geq 43.0 mg/L) CRP values compared to CRP values <10.0 mg/L. CRP did not associate with WL below 6%, whereas reductions in FI had strong associations within each category of WL (i.e. larger beta coefficients and ORs as severity of WL increased). Even moderately reduced FI had stronger associations to WL (ORs from 2.4 to 4.6, P<0.001) compared with moderate (ORs from 1.3 to 1.8, P<0.05) or high (ORs from 1.7 to 2.4, P<0.001) CRP values. Other covariates that were significantly associated with increased probability of WL included cancer diagnosis (lung, head & neck, upper and lower GI cancers), PS \geq 1, male sex, and BMI <25.0.

Our MLR model, adjusted for sex, cancer diagnosis, performance status, BMI, reductions in FI and CRP did not explain all the variation in WL. Patients experienced variable combinations of FI and CRP values across WL categories; overall, patients with elevated CRP values (>10 mg/L) and/or reduced FI (moderate or severe), had a greater % of patients with WL ≥6.0% compared to those with normal FI (37% vs. 12%, P<0.001, Table 6-3). For example, a subset of patients (N=384, 10%) reported normal FI and CRP values <10.0 mg/L but also had some degree of unexplained WL (i.e. WL >2.4%; Table 6-3). It is possible that other unaccounted factors (e.g. tumor mediated pro-catabolic factors, increased energy expenditure) may explain WL in these patients.

6.3.3 Risk stratification for overall survival

Median overall survival (OS) for this patient cohort was 9.5 months (95% CI 8.9-10.1; Table 6-1). The combined impact of increased CRP values, reduced FI, and WL on OS was evaluated. Patients were stratified by WL category (\pm 2.4%, 2.5-5.9%, 6.0-10.9%, 11.0-14.9%, \geq 15.0%), and for each WL category a 3x3 matrix was created including 3 categories of FI (normal, moderately, and severely reduced), and 3 categories of CRP (< 10 mg/L, 10-42.9 mg/L, \geq 43.0 mg/L). There was a total of 9 combinations per WL category for which median OS was calculated (Table 6-3). Patients with weight stability, normal FI, and CRP values <10 mg/L had the longest median OS of 25.9 months (95% CI 20.8-31.0). In each WL category, patients with severely reduced FI and CRP values >43 mg/L had the lowest median OS ranging from 1.1 to 4.1 months (Table 6-3).

6.4 Discussion

Cancer cachexia is driven by a variable combination of two major etiologies, reduced FI and altered metabolism.¹ We had previously demonstrated that patient-reported reductions in FI associated with increased probability of WL independent of age, sex, cancer diagnosis and stage, and performance status.² In this analysis, our regression model did not explain all the variation in WL, and a possible explanation was that we had not accounted for alterations in metabolism. In the ICCDR, the only marker of metabolic alteration available for evaluation as an explanatory factor for WL was CRP (i.e. systemic inflammation). We characterized the relationship between CRP and WL, and defined three categories for CRP (<10; 10-43; \geq 43 mg/L) based on values related to increased WL; patients with CRP <10 mg/L had the lowest WL (mean 5.7±6.2%). CRP values of \geq 10 are a common threshold used in oncology for prognostication of survival, and here we demonstrated they were also associated with increased WL.¹⁷

In an adjusted MLR analysis, both CRP and reduced FI independently associated with increased probability of WL, as did other covariates including cancer diagnoses of the lung, head & neck, and gastrointestinal tract, PS \geq 1, male sex, and BMI <25.0. These covariates contributed to the variation in WL explained by our model, and likely capture other reasons for WL beyond what we can measure with FI and CRP. Cancer diagnoses involving the GI tract including head & neck and upper and lower GI tract, as well as their treatments (e.g. GI toxicity) can impair ingestion, digestion, and absorption of nutrients, contributing to WL and not captured by our FI measures. Similarly, CRP may not capture the full scope of the underlying inflammatory processes contributing to WL in different tumor types, treatments, or in some cases behaviors (e.g. smoking).^{24,25} As well, tumor invasion of metabolically active organ tissue (e.g. liver) may lead to increased energy expenditure (i.e. hypermetabolism), which represents an additional contribution to WL beyond measures of FI and CRP.^{26,27} Reduced performance status may contribute to WL and muscle atrophy as a result of disuse and fatigue in response to cancer treatment and inflammation in the central nervous system.^{3,27} Sex differences in immune response may, in part, contribute to higher WL in men.²⁸ For example, estrogen can modulate inflammatory responses (e.g. inhibit production of interleukin-6, NfkB and tumor necrosis factor- α), which have been linked to WL and muscle wasting in cancer patients, and are not captured by CRP.^{28,29}

Some of the unexplained variation in WL may also be due to methodological limitations. There are discrepancies between the time frames for which reduced FI (e.g. current) was recorded in relation to the time frame over which WL occurred (e.g. previous 6 months). We do not know how FI and CRP values fluctuated over the period of time period during which WL occurred. These patterns may be revealed with longitudinal data collected using repeated

assessments. Some of the unexplained variation in WL may be the result of an incomplete set of relevant covariates in our statistical models. In cancer cachexia, the contribution of FI and altered metabolism to WL are painted as a spectrum; most patients experience both to some degree, and in others it is reduced FI or hypermetabolism or tumor-derived catabolic factors that have the predominant effect on WL.³⁰ We do not currently have a full accounting of factors that contribute to altered metabolism including assessments of energy expenditure (overall and of the tumor), measures of tumor-derived catabolic factors, or information about the tumor or cancer treatment. In our analysis, cancer stage did not associated with WL but information about tumor mass and metabolism may improve our understanding about the factors contributing to WL.^{26,31} At present, clinical data representing some of these factors is limited, although there is evidence to suggest they are important to the etiology of WL. A recent meta-analysis suggests there is an overall small increase (8-9%) in the resting energy expenditure (REE) of cancer patients compared to healthy controls, with larger elevations in cancers of the head & neck, esophageal, pancreas and liver.³² Two longitudinal studies demonstrated that patient's positive for PTHrP, which is thought to increase REE through adipose tissue browning, independently increased the risk of weight loss when adjusted for CRP, PS, cancer stage, albumin, and serum calcium.^{4,5} There is a long list of catabolic factors arising from animal studies, and more are currently being investigated but we do not have confirmatory studies or values for these factors in humans.³ A way forward would be to conduct prospective studies in a single tumor group with the aim to characterize factors that are likely to related to reductions in FI and metabolic alterations (REE, inflammation, tumor-mediated catabolic factors, cancer treatment) leading to cancer cachexia.

In oncology, patients are faced with a life limiting disease, and we therefore require information about their prognosis to determine the most appropriate type of clinical

interventions.³³⁻³⁶ CRP continues to be an important prognostic indicator in this patient cohort, and when combined with reduced FI, subsets of patients are identified with significantly reduced OS across different severities of WL (Table 6-3). This information is clinically available, and may facilitate identification of individuals entering end of life phase, which requires important consideration of treatments and care planning. These results suggest clinical assessments of patient-reported FI and a widely available clinical marker of systemic inflammation are associated with cancer-associated WL and provide useful prognostic information.

6.5 References

- 1. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.
- 2. Martin L, Kubrak C, Laird, B, et al. Reduced food intake as an explanatory variable for cancer-associated weight loss: Different clinical assessments of food intake provide equivalent results. University of Alberta; 2018, unpublished manuscript.
- 3. Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.
- 4. Hong N, Yoon HJ, Lee YH, et al. Serum PTHrP Predicts Weight Loss in Cancer Patients Independent of Hypercalcemia, Inflammation, and Tumor Burden. J Clin Endocrinol Metab. 2016; 101:1207-1214.
- 5. Lee JY, Hong N, Kim HR, et al. Effects of Serum Albumin, Calcium Levels, Cancer Stage and Performance Status on Weight Loss in Parathyroid Hormone-Related Peptide Positive or Negative Patients with Cancer. Endocrinol Metab (Seoul, Korea). 2018; 33:97-104.
- 6. Vazeille C, Jouinot A, Durand JP, et al. Relation between hypermetabolism, cachexia, and survival in cancer patients: a prospective study in 390 cancer patients before initiation of anticancer therapy. Am J Clin Nutr. 2017; 105:1139-1147.
- Martin L, Kubrak C. How much does reduced food intake contribute to cancer-associated weight loss? Curr Opin Support Palliat Care. 2018; doi:10.1097/SPC.0000000000379.
- 8. Evans WJ, Morley JE, Argiles J, et al. Cachexia: a new definition. Clin Nutr. 2008; 27:793-799.
- 9. Fearon KC, Voss AC, Hustead DS. Definition of cancer cachexia: effect of weight loss, reduced food intake, and systemic inflammation on functional status and prognosis. Am J Clin Nutr. 2006; 83:1345-1350.
- 10. Blum D, Stene GB, Solheim TS, et al. Validation of the Consensus-Definition for Cancer Cachexia and evaluation of a classification model--a study based on data from an international multicentre project (EPCRC-CSA). Ann Oncol. 2014; 25:1635-1642.
- 11. Thoresen L, Frykholm G, Lydersen S, et al. Nutritional status, cachexia and survival in patients with advanced colorectal carcinoma. Different assessment criteria for nutritional status provide unequal results. Clin Nutr. 2013; 32:65-72.

- 12. Wallengren O, Lundholm K, Bosaeus I. Diagnostic criteria of cancer cachexia: relation to quality of life, exercise capacity and survival in unselected palliative care patients. Support Care Cancer. 2013; 21:1569-1577.
- 13. Laird BJ, McMillan DC, Fayers P, et al. The systemic inflammatory response and its relationship to pain and other symptoms in advanced cancer. Oncologist. 2013; 18:1050-1055.
- 14. Wesseltoft-Rao N, Hjermstad MJ, Ikdahl T, et al. Comparing two classifications of cancer cachexia and their association with survival in patients with unresected pancreatic cancer. Nutr Cancer. 2015; 67:472-480.
- Simmons CP, Koinis F, Fallon MT, et al. Prognosis in advanced lung cancer--A prospective study examining key clinicopathological factors. Lung Cancer. 2015; 88:304-309.
- 16. Laird BJ, Kaasa S, McMillan DC, et al. Prognostic factors in patients with advanced cancer: a comparison of clinicopathological factors and the development of an inflammation-based prognostic system. Clin Cancer Res. 2013; 19:5456-5464.
- 17. McMillan DC. The systemic inflammation-based Glasgow Prognostic Score: a decade of experience in patients with cancer. Cancer Treat Rev. 2013; 39:534-540.
- 18. Nasrah R, Kanbalian M, Van Der Borch C, Swinton N, Wing S, Jagoe RT. Defining the role of dietary intake in determining weight change in patients with cancer cachexia. Clin Nutr. 2018; 37:235-241.
- 19. Gagnon B, Abrahamowicz M, Xiao Y, et al. Flexible modeling improves assessment of prognostic value of C-reactive protein in advanced non-small cell lung cancer. Br J Cancer. 2010; 102:1113-1122.
- 20. Martin L. Diagnostic criteria for cancer cachexia: data versus dogma. Curr Opin Clin Nutr Metab Care. 2016; 19:188-198.
- 21. Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancer-associated weight loss. J Clin Oncol. 2015; 33:90-99.
- 22. World Health Organization. Annex 2: Recommendations for the preparation, characterization and establishment of international and other biological reference standards (revised 2004). WHO Technical Report Series. 2006; 932:73-131.
- 23. Martin L, Watanabe S, Fainsinger R, et al. Prognostic factors in patients with advanced cancer: use of the patient-generated subjective global assessment in survival prediction. J Clin Oncol. 2010; 28:4376-4383.

- 24. Lee J, Taneja V, Vassallo R. Cigarette Smoking and Inflammation: Cellular and Molecular Mechanisms. Journal of Dental Research. 2012; 91:142-149.
- 25. Schiessel DL, Baracos VE. Barriers to cancer nutrition therapy: excess catabolism of muscle and adipose tissues induced by tumour products and chemotherapy. Proc Nutr Soc. 2018; doi:10.1017/s0029665118000186:1-9.
- 26. Lieffers JR, Mourtzakis M, Hall KD, McCargar LJ, Prado CM, Baracos VE. A viscerally driven cachexia syndrome in patients with advanced colorectal cancer: contributions of organ and tumor mass to whole-body energy demands. Am J Clin Nutr. 2009; 89:1173-1179.
- 27. Lawrence DP, Kupelnick B, Miller K, Devine D, Lau J. Evidence Report on the Occurrence, Assessment, and Treatment of Fatigue in Cancer Patients. JNCI Monographs. 2004; 2004:40-50.
- 28. Darnall BD, Suarez EC. Sex and gender in psychoneuroimmunology research: Past, present and future. Brain Behav Immun 2009; 23:595-604.
- 29. Hetzler KL, Hardee JP, Puppa MJ, et al. Sex Differences in the Relationship of IL-6 Signaling to Cancer Cachexia Progression. Biochim Biophys Acta 2015; 1852:816-825.
- 30. Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol. 2013; 10:90-99.
- 31. Friesen DE, Baracos VE, Tuszynski JA. Modeling the energetic cost of cancer as a result of altered energy metabolism: implications for cachexia. Theor Biol Med Model. 2015; 12:17.
- Nguyen TY, Batterham MJ, Edwards C. Comparison of Resting Energy Expenditure Between Cancer Subjects and Healthy Controls: A Meta-Analysis. Nutr Cancer. 2016; 68:374-387.
- 33. Arends J. Struggling with nutrition in patients with advanced cancer: nutrition and nourishment-focusing on metabolism and supportive care. Ann Oncol. 2018; 29:ii27-ii34.
- 34. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 35. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 36. Mislang AR, Di Donato S, Hubbard J, et al. Nutritional management of older adults with gastrointestinal cancers: An International Society of Geriatric Oncology (SIOG) review paper. J Geriatr Oncol. 2018; 9:382-392.
| Variables | N | % |
|---|----------------|------|
| Region | | |
| Canada | 513 | 14.9 |
| Europe | 2930 | 85.1 |
| Sex | | |
| Male | 1873 | 54.4 |
| Female | 1570 | 45.6 |
| Age, years (mean, SD) | 63.6 (12.3) | |
| Status | | |
| Censored | 754 | 21.9 |
| Dead | 2689 | 78.1 |
| Overall Survival, months (median, 95% CI) | 9.5 (8.9-10.1) | |
| Cancer Diagnosis* | | |
| Head & neck | 307 | 8.9 |
| Breast | 429 | 12.5 |
| Upper gastrointestinal | 685 | 19.9 |
| Lower gastrointestinal | 276 | 8.0 |
| Genitourinary organs | 607 | 17.6 |
| Respiratory | 786 | 22.8 |
| Other | 353 | 10.3 |
| Cancer Stage ⁺ | | |
| 1 | 86 | 2.4 |
| 2 | 177 | 5.1 |
| 3 | 482 | 14.0 |
| 4 | 2698 | 78.4 |
| ECOG PS | | |
| PS 0 | 734 | 21.3 |
| PS 1 | 1081 | 31.4 |
| PS 2 | 1058 | 30.7 |
| PS 3 | 546 | 15.9 |
| PS 4 | 24 | 0.7 |
| % WL (mean, SD) | -7.3 (7.0) | |
| WL Categories | | |
| ±2.4% (weight stable) | 1019 | 29.6 |
| 2.5-5.9% | 711 | 20.7 |
| 6.0-10.9% | 822 | 23.9 |
| 11.0-14.9% | 390 | 11.3 |
| ≥15.0% | 501 | 14.6 |
| BMI, kg/m2 (mean, SD) | 24.1 (4.9) | |

Table 6-1. Patient characteristics

BMI Categories		
<20.0	655	19.0
20.0-21.9	566	16.4
22.0-24.9	953	27.7
25.0-27.9	646	18.8
≥28.0	623	18.1
WL Grade [‡]		
Grade 0	512	14.9
Grade 1	542	15.7
Grade 2	590	17.1
Grade 3	1028	29.9
Grade 4	771	22.4
Reduced Food Intake		
normal	686	19.9
moderately reduced	1380	40.1
severely reduced	1377	40.0
Food Intake Tool Used		
PG-SGA	1001	29.1
<i>I</i> -VVAS	2247	70.9
CRP, mg/L (mean, SD)	43.6 (64.3)	
CRP _{log} (mean, SD)	2.7 (1.7)	
CRP <10 mg/L	1375	39.9
$CRP \ge 10 \text{ mg/L}$	2068	60.1

BMI, body mass index; ECOG PS, Eastern Cooperative Oncology Group performance status; *I*-VVAS, *Ingesta*-Verbal/Visual Analogue Scale; N, number; PG-SGA, Patient-Generated Subjective Global Assessment; SD, standard deviation; WL, weight loss

*Upper gastrointestinal (esophageal, stomach, pancreas, liver, biliary tract, small bowel); lower gastrointestinal (colon, rectum, anus); genitourinary (kidney, bladder, adrenal, prostate, testes, penis); Other cancers (gynecological, hematological, peritoneum, unknown, thyroid)

+based on AJCC version 6 or 7

[‡]based on WL grades by Martin et al.⁴

		V	VL 2.5-5.9%	%	V	VL 6.0-10.9	%	W	L 11.0-14.9	0%	l v	VL >=15.0%	6
X7 · 11 ·	NT (0/)	N=	= 711 (20.79	%)	N=	= 822 (23.9)%)	N=	= 390 (11.3)	%)	N	= 501 (14.6%	%)
variables in model	N (%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value
CRP Categories													
≥43.0 mg/L	1032 (30.0)	0.18 (0.14)	1.19 (0.91- 1.56)	0.193	0.51 (0.13)	1.67 (1.29- 2.15)	<0.001	0.56 (0.16)	1.75 (1.27- 2.42)	<0.001	0.87 (0.16)	2.39 (1.74- 3.29)	< 0.001
10.0-42.9 mg/L	1036 (30.1)	0.20 (0.12)	1.22 (0.97- 1.55)	0.088	0.26 (0.12)	1.30 (1.03- 1.65)	0.028	0.32 (0.16)	1.37 (1.01- 1.87)	0.045	0.59 (0.16)	1.80 (1.32- 2.45)	< 0.001
<10.0 mg/L	1375 (39.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Current Food Intake													
severely reduced intake	686 (19.9)	0.92 (0.17)	2.52 (1.80- 3.53)	< 0.001	1.37 (0.16)	3.94 (2.87- 5.40)	< 0.001	2.02 (0.20)	7.51 (5.12- 11.02)	<0.001	2.29 (0.19)	9.84 (6.76- 14.33)	<0.001
moderately reduced intake	1380 (40.1)	0.87 (0.11)	2.40 (1.92- 2.99)	< 0.001	1.12 (0.11)	3.07 (2.46- 3.84)	<0.001	1.36 (0.16)	3.90 (2.88- 5.29)	< 0.001	1.52 (0.16)	4.59 (3.37- 6.24)	<0.001
normal	1377 (40.0)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis													
respiratory	786 (22.8)	0.43 (0.19)	1.54 (1.06- 2.25)	0.025	0.27 (0.19)	1.32 (0.91- 1.90)	0.141	0.65 (0.26)	1.91 (1.15- 3.17)	0.012	0.64 (0.27)	1.90 (1.12- 3.22)	0.017
other	353 (10.3)	0.05 (0.22)	1.06 (0.69- 1.61)	0.800	-0.18 (0.21)	0.83 (0.55- 1.25)	0.379	0.17 (0.29)	1.19 (0.68- 2.09)	0.544	0.15 (0.30)	1.17 (0.65- 2.08)	0.603
genitourinary	607 (17.6)	0.34 (0.20)	1.41 (0.96- 2.07)	0.081	0.01 (0.19)	1.01 (0.70- 1.47)	0.947	0.24 (0.26)	1.27 (0.76- 2.12)	0.362	0.43 (0.26)	1.54 (0.92- 2.58)	0.099

Table 6-2. A multivariable multinomial logistic regression model to evaluate the association of CRP and food intake to weight loss.

upper GI	685 (19.9)	0.58 (0.20)	1.79 (1.20- 2.66)	0.004	0.45 (0.19)	1.56 (1.07- 2.28)	0.022	1.04 (0.26)	2.83 (1.70- 4.70	< 0.001	1.63 (0.26)	5.11 (3.07- 8.49)	<0.001
lower GI	276 (8.0)	0.24 (0.24)	1.27 (0.80- 2.04)	0.314	0.26 (0.23)	1.30 (0.83- 2.02)	0.251	0.72 (0.30)	2.06 (1.15- 3.70)	0.016	0.59 (0.31)	1.80 (0.97- 3.34)	0.062
head & neck	307 (8.9)	0.33 (0.25)	1.39 (0.85- 2.27)	0.183	0.53 (0.23)	1.70 (1.08- 2.69)	0.022	0.63 (0.32)	1.87 (1.00- 3.49)	0.050	1.25 (0.31)	3.50 (1.92- 6.37)	<0.001
breast	429 (12.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	570 (16.6)	0.12 (0.19)	1.12 (0.77- 1.63)	0.541	0.44 (0.19)	1.56 (1.07- 2.25)	0.019	0.54 (0.24)	1.72 (1.07- 2.77)	0.026	0.93 (0.25)	2.54 (1.57- 4.10)	<0.001
PS 2	1058 (30.7)	0.12 (0.15)	1.13 (0.83- 1.53)	0.432	0.67 (0.16)	1.95 (1.44- 2.64)	<0.001	0.75 (0.21)	2.13 (1.42- 3.18)	<0.001	0.99 (0.22)	2.69 (1.77- 4.11)	<0.001
PS 1	1081 (31.4)	0.30 (0.14)	1.35 (1.03- 1.77)	0.032	0.52 (0.15)	1.69 (1.27- 2.26)	<0.001	0.53 (0.20)	1.69 (1.14- 2.51)	0.009	0.68 (0.21)	1.97 (1.30- 2.99)	0.001
PS 0	734 (21.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
male	1873 (54.4)	0.20 (0.11)	1.23 (0.98- 1.53)	0.072	0.20 (0.11)	1.23 (0.98- 1.53)	0.069	0.45 (0.14)	1.57 (1.19- 2.08)	0.002	0.70 (0.14)	2.02 (1.53- 2.66)	<0.001
female	1570 (45.6)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
BMI Categories (kg/m2)													
<20.0	655 (19.0)	0.60 (0.18)	1.82 (1.28- 2.59)	0.001	0.96 (0.18)	2.62 (1.85- 3.70)	<0.001	1.86 (0.24)	6.42 (3.98- 10.36)	< 0.001	2.17 (0.21)	8.75 (5.76- 13.29)	<0.001
20.0 to 21.9	566 (16.4)	0.37 (0.17)	1.45 (1.04- 2.02)	0.029	0.92 (0.17)	2.52 (1.82- 3.49)	<0.001	1.48 (0.24)	4.40 (2.74- 7.06)	< 0.001	0.94 (0.23)	2.55 (1.64- 3.98)	<0.001

22.0 to 24.9	953 (27.7)	0.45 (0.14)	1.57 (1.19- 2.07)	0.002	0.59 (0.15)	1.80 (1.35- 2.40)	<0.001	1.18 (0.22)	3.24 (2.09- 5.02)	<0.001	0.75 (0.20)	2.12 (1.42- 3.15)	<0.001
25.0 to 27.9	646 (18.8)	0.11 (0.15)	1.12 (0.83- 1.51)	0.460	0.32 (0.16)	1.38 (1.01- 1.88)	0.040	0.73 (0.24)	2.08 (1.30- 3.33)	0.002	-0.05 (0.23)	0.95 (0.60- 1.51)	0.839
≥28.0	643 (18.1)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	

β, beta coefficient; BMI, body mass index; CRP, C-reactive protein; MLR, multinomial logistic regression; N, number; OR, odds ratio; PS, performance status; SE, standard error; WL, weight loss

MLR Model adjusted for age, sex, cancer diagnosis, cancer stage, performance status, and BMI - Reference weight stable ($\pm 2.4\%$) N=1019 (29.6%); Intercept only model: -2 log likelihood (LL)=10598.216, AIC 10606.21; age (P=0.259) and cancer stage (P=0.843) were not significant at the multivariable level and not included in the final model.

Final Model: -2LL=9592.50, AIC=9744.50, χ 2=1005.71 (df=72), P<0.001; Pseudo R²: Nagelkerke=0.265, Cox and Snell=0.253, McFadden's=0.094, Pearson goodness-of-fit P=0.694

0	WL Categories*														
	N=1	±2.4%	(0)	N	2.5-5.9% =711 (20.	7)	N	6.0-10.9% =822 (23.9	%)	1 N=	1.0-14.9% 390 (11.3	% %)	V N=	VL ≥15.0 =501 (14.0	% 5%)
	CRP (r	ng/L) categ	gories	CRP (1	ng/L) cat	egories	gories CRP (mg/L) categories			CRP (mg/L) categories			CRP (mg/L) categories		
Food Intake Categories	<10.0	10.0- 42.9	≥43.0	<10.0	10.0- 42.9	≥43.0	<10.0	10.0- 42.9	≥43.0	<10.0	10.0- 42.9	≥43.0	<10.0	10.0- 42.9	≥43.0
Normal															
N events/N	222/385	146/189	68/81	97/159	73/82	47/57	85/127	57/67	58/61	32/43	25/31	13/14	20/25	22/25	31/34
%N within WL category	37.8%	18.5%	7.9%	22.4%	11.5%	7.6%	15.5%	8.2%	7.4%	11.0%	7.9%	3.6%	5.0%	5.0%	6.8%
OS, months	25.9 ^a	12.2 ^b	5.6°	21.9 ^a	11.9 ^a	7.1ª	19.0 ^a	8.6 ^b	4.9 ^c	8.5 ^a	9.9 ^a	6.3b	9.2a	8.6a	5.0a
(95% CI)	(20.8- 31.0)	(9.5- 14.9)	(1.9- 9.3)	(16.3- 27.5)	(6.9- 16.9)	(5.9- 8.4)	(13.1- 24.9)	(5.9- 11.3)	(2.9- 6.9)	(3.7- 13.2)	(8.4- 11.4)	(1.8- 10.8)	(2.5- 15.9)	(3.0- 14.2)	(2.7- 7.3)
Moderately reduced															
N events/N	76/117	68/81	73/80	82/118	89/112	72/82	91/132	105/127	121/132	36/56	41/51	64/71	44/58	64/72	87/91
%N within WL category	11.5%	7.9%	7.9%	16.6%	15.8%	11.5%	16.1%	15.5%	16.1%	14.4%	13.1%	18.2%	11.6%	14.4%	18.2%
OS, months	22.2ª	10.7 ^b	3.8 ^c	19.7 ^a	8.8 ^b	5.0°	19.3ª	8.1 ^b	4.0 ^c	13.8 ^a	7.5 ^b	3.5°	13.0 ^a	6.4 ^b	3.1°
(95% CI)	(14.9- 29.5)	(8.6- 12.8)	(1.8- 5.8)	(16.2- 23.2)	(6.8- 10.9)	(3.6- 6.5)	(13.5- 25.1)	(5.9- 10.3)	(3.2- 4.8)	(4.6- 23.0)	(4.9- 10.1)	(2.6- 4.3)	(6.8- 19.2)	(5.5- 7.3)	(1.5- 4.7)
Severely reduced															
%N within WL category	2.2%	2.1%	4.2%	4.1%	4.4%	6.2%	4.3%	6.6%	10.6%	6.7%	8.7%	16.4%	8.6%	11.8%	18.8%
N events/N	13/22	16/21	41/43	24/29	28/31	36/44	23/35	46/54	78/87	17/26	29/34	59/64	34/43	53/59	85/94

Table 6-3. Median overall survival (OS) for patients stratified by categories of weight loss (WL), food intake (FI), and CRP categories

%N within WL category	2.2%	2.1%	4.2%	4.1%	4.4%	6.2%	4.3%	6.6%	10.6%	6.7%	8.7%	16.4%	8.6%	11.8%	18.8%
OS, months	13.2 ^a	12.1 ^a	1.1 ^b	8.7 ^a	5.3 ^b	2.2 ^b	14.5 ^ª	6.2 ^b	3.1 ^b	14.4 ^a	6.6 ^{a,b}	4.1 ^b	5.9 ^a	5.2 ^{a,b}	2.5 ^b
(95% CI)	(11.5- 14.9)	(0.0- 24.8)	(0.6- 1.6)	(6.1- 11.3)	(3.8- 6.8)	(0.0- 5.0)	(10.5- 18.5)	(5.1- 7.3)	(2.3- 3.9)	(0.0- 30.5)	(5.1- 8.1)	(2.5- 5.7)	(4.8- 7.0)	(3.8- 6.6)	(1.7- 3.3)
Overall															
OS, months	24.4	11.8	3.8	19.7	8.8	5.7	18.4	7.3	3.8	12.0	7.9	3.8	9.2	6	3.0
(95% CI)	(20.0- 28.8)	(9.8- 13.9)	(2.8- 4.8)	(16.6- 22.8)	(6.9- 10.8)	(4.3- 7.1)	(14.4- 22.3)	(6.0- 8.6)	(3.1- 4.4)	(9.3- 14.7)	(6.0- 9.8)	(2.9- 4.7)	(6.2- 12.2)	(5.1- 6.9)	(2.2- 3.8)

CI, confidence interval; CRP, C-reactive protein; N, number; OS, median overall survival; WL, weight loss

*WL categories as defined by Martin et al.(ref)

^{a,b,c} Superscript letters indicate significant differences (P<0.05) between median OS (Kaplan Meier with log rank (Mantel-Cox) tests) within a food intake category when stratified by CRP, for each WL category.

CHAPTER 7: Implementation of an Enhanced Recovery After Surgery program can change nutrition care practice: A multi-center experience in elective colorectal surgery¹

7.1 Clinical Relevancy Statement

Enhanced Recovery After Surgery (ERAS) programs are multimodal surgical care pathways that optimize patient care to promote recovery and improve postoperative outcomes. ERAS programs are emerging as the standard in perioperative management around the world. A key component is integration of evidence-based perioperative nutrition care into overall patient management. Implementation of an ERAS program for elective colorectal surgeries had a positive impact on the optimization of nutrition care and is a step towards greater quality of nutrition care throughout surgical programs in Alberta, Canada. ERAS implementation aligns surgical care with clinical practice guidelines for perioperative nutrition care.

7.2 Introduction

Suboptimal nutrition status has long been recognized as an independent predictor of poor surgical outcomes.¹ Unfortunately, surgical nutrition care practices have not been well characterized, especially with regard to nutrition risk screening, nutrition diagnoses, and in characterizing effective nutrition interventions.² Several nutrition and surgical societies have called for significant improvements to perioperative nutrition care practices^{1,3-6}, with a consistent take home message: implementation of evidence-based perioperative nutrition care pathways that are integrated into overall management of the surgical patient improve nutrition care practices and clinical outcomes.¹

¹A version of this chapter had been published: Martin L, Gillis C, Atkins M, Gillam M, Sheppard C, Buhler S, Basualdo Hammond C, Nelson G, Gramlich L. Implementation of an Enhanced Recovery After Surgery Program can change nutrition care practice: A multi-center experience in elective colorectal surgery. JPEN J Parenter Enteral Nutr, 2018; Jul 23; doi.org/10.1002/jpen.1417

Enhanced Recovery After Surgery (ERAS) programs are increasingly being adopted as the standard of perioperative care around the world.¹ ERAS programs are multimodal, evidencebased care improvement processes that bundle more than 20 care elements throughout the perioperative period, and consistently demonstrate improved postoperative outcomes in a variety of surgical settings.⁷⁻¹² The goals are to manage the stress response to surgery, maintain body stores, and improve physiologic function for early recovery.^{7,8} These goals are achieved by standardizing care elements related to surgical practices including analgesia, anesthesia, fluid and symptom management, and perioperative nutrition care.^{7,8} Central to ERAS programs is recognition that nutritional status affects outcomes of surgery and must be optimized for recovery; therefore, perioperative nutrition care is integrated into the overall management of the patient.^{1,13} ERAS programs are recognized as important drivers of standardized perioperative nutrition care, and have provided the foundation of key perioperative nutrition care elements, which have been incorporated in to clinical practice guidelines for surgical patients.^{1-4,6,14}

Despite the recognized benefits of ERAS programs^{1,3,4}, implementation of evidencedbased guidelines is a continual challenge.^{2,4,8} ERAS programs are complex, they must be adapted to local contexts and require extensive education, cooperation, and collaboration between the patient, health care provider, and health system.^{7,15} To facilitate education and training for successful adoption of ERAS protocols, the ERAS Society has a formalized ERAS Implementation Program (EIP).¹⁶ To be effective at improving postoperative outcomes, compliance to ERAS care elements is essential. This requires multidisciplinary teams to cooperate to carry out the care processes throughout the perioperative period, and for patients to take part in their recovery.⁷ Higher compliance (e.g. >70%) to ERAS care processes is an independent predictor of improved postoperative outcomes including reduced LOS and

133

complications.¹⁷⁻¹⁹ Within any ERAS program a minority of patients will not benefit from these strategies, and their identification is not well characterized.¹⁸ Identifying patients likely to deviate from an ERAS program, and therefore at risk of poor postoperative outcomes, is of interest as it may inform development of alternative care elements that would benefit these patients.¹⁸

In 2013, Alberta Health Services (AHS), a provincial health system, led an EIP¹⁶ in elective colorectal surgeries as a strategic initiative to enhance the quality of surgical care with a focus on efficiency, effectiveness, safety, accessibility, and cost.^{20 16} Initial results demonstrated reductions in length of hospital stay and cost, similar to other ERAS programs in colorectal surgery. 9-12,20-22 Prior to ERAS implementation, perioperative nutrition care practices for colorectal surgeries were not standardized and ill-defined. The AHS EIP provided opportunity to standardize and initiate several perioperative nutrition care processes.²³ The goal of this work is to generate knowledge regarding the impact of an EIP on the local adoption of perioperative nutrition care practices for elective colorectal surgeries, and identify patients who were more likely to deviate from the EIP. Outcomes of implementation studies focus on the rate of adoption of evidence-based practices,¹⁵Our primary focus is nutrition risk screening, and characterizing patients with nutrition risk in relation to adherence to ERAS care elements, and a secondary focus on postoperative outcomes. Our first objective describes the perioperative nutrition care elements implemented as part of the AHS EIP and the rate of compliance to these elements. The second objective was to evaluate whether nutrition risk affected compliance to the EIP and impacted postoperative outcomes.

7.3 Methods

7.3.1 ERAS Implementation Program

A description of the AHS EIP undertaken as a quality improvement project has been published.²² AHS is a provincial health system servicing ~4.4 million people in Alberta, Canada. In 2013, AHS implemented ERAS Society guidelines for colorectal surgery²³ in 6 acute care centers in the cities of Edmonton and Calgary.²² The AHS EIP was implemented according to a formalized process outlined by the ERAS Society¹⁶ and includes 22 evidence-based clinical care elements implemented by a multidisciplinary team throughout the perioperative period (Table 7-1). A core component of AHS EIP is data collection for the purposes of auditing and evaluating care processes, which in turn support and enhance practice change at both the provider and system level. Patient, surgery, compliance, and outcome data are prospectively collected by dedicated ERAS nurse coordinators (N=6) using the ERAS Interactive Audit System (EIAS, a secure online data capture tool) and data definitions developed by the ERAS Society.^{24,23} Compliance to each of the 22 care elements is recorded as compliant vs. non-compliant, according to pre-specified criteria, and overall compliance is calculated as the number of elements achieved (Table 7-2).²² Short-term postoperative outcomes include: 1) length of hospital stay (LOS) defined as the period between primary surgical date and discharge from hospital (readmissions were not included in the calculation), 2) complications recorded as any complication occurring during the index hospitalization (graded according to the Clavien-Dindo classification²⁵); serious complications defined as grade $\geq 3, 3$) 30-day readmission and mortality rates are also recorded. This quality improvement data captured within the EIAS was analyzed to assess the impact on nutrition care.

7.3.2 Study Design

The study population included consecutive, adult (≥18 years) patients (from September 2013-December 2016) who underwent elective colorectal surgeries as part of AHS EIP.

Multiple ERAS care elements contribute to patients' nutrition and metabolic status (e.g. fluid balance, anesthesia/analgesia), we chose to evaluate the care elements specific to nutrition care practices: nutrition risk screening, minimizing pre-operative fasting, use of pre-operative carbohydrate loading, prevention of postoperative nausea and vomiting, initiation of nutrition support when appropriate, early oral feeding, and mobilization (Tables 7-1, 7-2).^{1,7,8,13,23} Changes to perioperative nutrition care practices were evaluated using pre- and post-comparisons between 2 groups: 1) Pre-ERAS group: data collection prior to ERAS implementation to assess baseline compliance, 2) ERAS group: data collection with full ERAS implementation.

From a nutrition care perspective, the adoption of nutrition risk screening is a significant benefit of the AHS EIP. ERAS care elements, including nutrition risk screening tools, are adapted to meet local needs, hence centers selected nutrition risk screening tools best suited to their local context. For the ERAS Group, nutrition risk screening is completed by a health care provider at a preadmission visit with one of two validated tools: the Malnutrition Screening $Tool^{26}$ (MST; nutrition risk defined as an MST score ≥ 2 , which equates to eating poorly and/or recent weight loss) or the Canadian Nutrition Screening $Tool^{27}$ (CNST, nutrition risk defined based on responding "yes" to two questions, which equates to eating poorly and recent weight loss). Nutrition risk refers to a nutritional problem with potential to negatively impact nutritional status resulting in malnutrition. For the ERAS Group, patients' deemed to be at nutrition risk based on either tool were grouped together for analysis. We evaluated the impact of nutrition risk on compliance to ERAS care elements, and on short-term postoperative outcomes. Additional nutrition care elements included use of postoperative nutrition support prescribed by a Registered

136

Dietitian (e.g. enteral or parenteral nutrition), monitoring of patient's body weight, time taken for the patient to experience flatus, have a bowel movement, tolerate solid food (defined as consuming a solid diet in normal portions without vomiting, excludes use of nasogastric tube), and achieve activities of daily living (ADL; defined as the number of days after surgery when able to independently physically perform basic ADL to the same extent as before surgery). These data were evaluated as indicators of recovery.

The AHS EIP had ethical approval from institutional research ethics boards as a health system quality improvement initiative, individual consent from patients was not sought or required. Individual patient data were anonymized prior to inclusion in the EIAS.

7.3.3 Statistical Analysis

Frequencies and summary statistics describe the patient sample. Differences between groups were assessed with parametric (Independent T-tests, Chi-Square Tests and Fisher Exact Tests with post-hoc Bonferroni corrections) or non-parametric (Mann-Whitney) tests where appropriate. Binary logistic regression models evaluated the association between patient and surgical variables, and hospital site with dependent variables: low overall compliance to ERAS care elements (defined as $<70\%^{19}$), extended length of hospital stay (defined as >5 days), and postoperative complications (yes vs. no). Relationships between these outcomes and study variables were assessed at the univariable level, and variables significantly associated with an outcome (P<0.1) were entered into multivariable analyses (full model). Non-significant variables were removed in a backward step-wise manner, and the most parsimonious model was retained as the final model. Results are reported as odds ratios (OR with 95% CI) and considered

significant at the P<0.05 level (two-tailed). Data analysis was performed with IBM SPSS Statistics for Windows (v.23, Armonk, NY: IBM Corp).

7.4 Results

7.4.1 The influence of ERAS implementation on the adoption of perioperative nutrition care practices

7.4.1.1 Patient characteristics

4023 patients were included in the AHS EIP (pre-ERAS group, N=476; ERAS group, N=3536; Table 7-3). There were few differences between these groups. Overall, 3% of patients were underweight (BMI <18.5), 36% of patients were overweight (BMI 25.0-29.9), and 33% were obese (BMI \geq 30.0). The ERAS group had more laparoscopic surgeries compared to the pre-ERAS group (46% vs. 36%, P<0.001), due to changes in surgical practice as a result of AHS EIP. Mean overall compliance to the EIP was 71% in the ERAS group vs. 51% in the pre-ERAS group (Table 7-3). Pre- and intra-operative elements had the highest compliance while postoperative elements had the lowest compliance. More than half (67%) of patients in the ERAS group achieved good overall compliance (>70%), compared to essentially no patients (1%) in the pre-ERAS group.

7.4.1.2 Changes to perioperative nutrition care practices

AHS EIP had a significant impact on nutrition care (Table 7-4). In the pre-operative period, the ERAS group had significant improvements with regard to nutrition screening, carbohydrate loading, and postoperative nausea and vomiting prophylaxis (PONV). Nutrition risk screening increased from 8% (pre-ERAS group) to 74% (ERAS group, P<0.001). However,

a high proportion of patients in the ERAS group (N=908, 26%) were not screened for nutrition risk.

In the postoperative period, the ERAS group had improvements with regard to the adoption of strategies to stimulate gut motility, initiate early oral feeding, and obtaining patient weights (Table 7-4). Calories from oral nutritional supplements (ONS) on postoperative day (POD) 0-3 were increased (P<0.001). The ERAS group reduced the number of days to tolerate solid food, from 5.9 to 2.3 days (p<0.001). Use of artificial nutrition increased from 2% (Pre-ERAS Group) to 25% (ERAS Group, (P<0.001). More patients in the ERAS group met targets for mobilization on POD 0-3 (P<0.001), and time to return to ADL was reduced (3.4 days vs. 7.7 days, P<0.001).

The frequency of unknown/missing data nutrition care elements was highly variable. Elements missing <10% of cases included: carbohydrate loading, oral bowel prep, and PONV prophylaxis, mobilization on POD 0, time to tolerating solid food (Table 7-4). Elements missing >20% of cases included: nutrition risk screening, patient weights, energy intake from ONS on POD 2-3, use of artificial nutrition, and mobilization on POD 1-3 (Table 7-4). The ERAS group had fewer missing data compared to the pre-ERAS group.

7.4.2 Impact of nutrition risk on compliance to AHS EIP and short-term postoperative outcomes

7.4.2.1 Patient characteristics

A total of N=2628 (74%) patients from the ERAS group were screened for nutrition risk and included in this analysis. The majority (88%, N=2317) of patients were not at nutrition risk, and 12% (N=311; MST N=218, CNST N=93) were at nutrition risk. Characteristics of these two groups are presented in Table 7-3. Patients at nutrition risk had more smokers (P<0.001), cancer diagnoses (P=0.004), open surgeries (P=0.016), complex surgical procedures (P=0.034), and BMIs <25.0 (P<0.001). More than half (53%) of patients at nutrition risk were also overweight or obese.

7.4.2.2 Nutrition risk and compliance to AHS EIP

Patients at nutrition risk had lower overall compliance compared to those not at nutrition risk (68% vs. 74 %, P<0.001, Table 7-3). There was no difference in compliance for patients at nutrition risk as defined by the MST or CNST (P=0.623). For nutrition care elements, the nutrition risk group had less patients with morning weights (P<0.05), able to meet energy intake targets for early oral feeding (e.g. ONS consumption on POD 0-3, P<0.05), more patients requiring postoperative nutrition support (31% vs. 25%, P<0.001, Table 7-4), and a trend to take longer to return to solid food (P=0.079). Patients at nutrition risk had reduced mobility on POD 0-2, and took longer to recover ADL (4.5 vs. 3.1 days, P=0.046, Table 7-4). In multivariable logistic regression (Table 7-5), low compliance to AHS EIP (<70%) was predicted by nutrition risk (OR 2.3 95% CI 1.8-3.0, P<0.001), sex, age, surgical complexity, ASA \geq 3, postoperative complications, and hospital site. Patients with BMI \geq 30.0 were more likely to have good compliance (>70%) to ERAS care elements.

7.4.2.3 Nutrition risk and short-term postoperative outcomes

Short-term postoperative outcomes are listed in Table 7-3. Overall, AHS EIP significantly reduced LOS (median 5.0 days (IQR 4.0) vs 6.0 days (IQR 5.0), P<0.001), and the proportion of patients with LOS >5 days (58% vs. 49%, P<0.001), but had no effect on the rate of postoperative complications (44 vs. 48%, P=0.069), 30-day readmission (11% vs. 10%, P=0.364), or 30-day mortality (0.2% vs. 0.7%, P=0.388).

In contrast, patients at nutrition risk in the ERAS group had longer LOS (median 6.0 days (IQR 5.0) vs. 5.0 days (IQR 4.0), P<0.001, Table 7-3). LOS did not differ between patients with nutrition risk defined by MST or CNST (P=0.396). In multivariable logistic regression (Table 7-6), LOS >5 days was predicted by postoperative complications, surgical approach, surgical complexity, ASA class, sex, age, hospital site, and overall compliance to AHS EIP. Nutrition risk trended towards significance for longer LOS (OR 1.40 95% CI 1.00-1.97, P=0.052).

Patients at nutrition risk had more postoperative complications (56% vs. 49%, P=0.016), and more serious (Grade \geq 3) postoperative complications (14% vs. 8%, P=0.022). Postoperative complications differed (P<0.001) between patients with nutrition risk defined by MST (N=135 (62%)) or CNST (N=39 (42%)) but not for serious complications (P=0.955). In multivariable logistic regression (Appendix VI), postoperative complication were predicted by surgical approach (open or converted procedures), greater surgical complexity, ASA class \geq 3, age, male sex, low compliance to AHS EIP, and hospital site. ERAS care elements. Overall nutrition risk (and as defined by MST or CNST) did not predict of postoperative complications.

There was no difference in readmission rates for patients with or without nutrition risk (P=0.156, Table 7-3), nor when nutrition risk was defined by MST or CNST (P=0.447). Thirty-day mortality was low for this patient cohort (N=22, 0.8%), however 30-day mortality was higher for patients at nutrition risk (2.2% vs. 0.6%, P<0.001, Table 7-3). There was no difference in mortality for patients with nutrition risk defined by MST or CNST (P=0.805).

7.5 Discussion

AHS EIP for elective colorectal surgeries had a positive impact on the adoption of standardized perioperative care practices with an overall compliance rate of 72%. This is of

significance as enhanced adherence (>70%) to ERAS protocols are associated with improved surgical outcomes¹⁹. Adoption of an EIP aligned AHS more closely with clinical practice guidelines for perioperative nutrition care in surgical patients. Prior to ERAS implementation, the majority of perioperative nutrition care elements were either not practiced, or were highly variable. Key areas of improvement included nutrition risk screening, pre-operative carbohydrate loading, stimulation of gut motility, obtaining morning weights, early oral feeding, and mobilization. Nutrition risk screening, increased from 8% to 74%, and identified a subset of patients (12%) who were less compliant with the ERAS protocol, slower to recover oral intake and physical functioning, and had longer LOS and increased mortality. Collectively these results suggest patients at nutrition risk have difficulty recovering from elective surgery, which is independent of postoperative complications.

A limitation of our study was 26% of patients in the ERAS group were not screened for nutrition risk, potentially introducing a selection bias, which may explain, in part the lower rate of nutrition risk compared to other studies in similar patient populations (20-47%). However, the rate of nutrition risk is largely dependent on the case mix within the patient cohort and the tool selected for screening.^{13,28-30} There is heterogeneity in nutrition risk screening tools used across surgical programs, and for this reason results are not directly aggregable. Importantly, in programs where validated nutrition risk screening tools were used (e.g. Patient Subjective Global Assessment (PGSGA), Nutrition Risk Screen (NRS)-2002, Malnutrition Universal Screening Tool (MUST)), nutrition risk was associated with worse postoperative outcomes including longer LOS and increased postoperative complications, similar to our findings.^{13,28-30} In this study, nutrition risk was defined by two different screening tools, the MST and CNST,^{26,27} and it is possible the rate of nutrition risk detection differed between the tools. However, the MST and

CNST use similar criteria to identify nutrition risk (reduced food intake and weight loss), and have similar sensitivity and specificity versus the subjective global assessment (MST sensitivity = 39-93%, specificity = 55-93%; CNST sensitivity 67-73%, specificity 80-86%).³¹ Outcomes were not different whether patients were screened with the MST or CSNT. This is an important finding as both tools are designed to detect patients with pre-existing nutritional problems, (weight loss and reduced food intake), and when these problems are identified patients were less able to comply with AHS EIP, and importantly for nutritional professionals, had lower compliance to nutrition care elements. It seems plausible when nutritional problems pre-exist they continue to be problematic in the postoperative period and for patient recovery. We did observe an increase of 6% in the use of postoperative nutrition support for patients at nutrition risk, but it is not clear if this can be attributed to nutrition risk screening. A better understanding of care processes related to nutrition support in AHS is needed and is an opportunity for study to develop standardized nutrition care pathways. ERAS guidelines include recommendations for pre-operative nutrition interventions for patients at nutrition risk, these guidelines are not sufficiently developed in the AHS EIP, and this has been identified as a major gap in nutrition care and an area for development.

With any large scale quality improvement program there are limitations to consider. Nutrition risk was an important patient factor in identifying low compliance to AHS EIP; however other factors may also impact compliance to perioperative nutrition care elements. The AHS EIP achieved highest compliance in the preoperative period (86%) and lowest in the postoperative period (57%), similar to other colorectal ERAS programs.^{18,32} Possible explanations include development of postoperative complications, which can result in deviations from protocols, and the high degree of patient and multidisciplinary involvement in the

143

postoperative period, which may explain reported differences in the rates of adherence to ERAS care elements, and in data recording.^{7,8,33} Some ERAS elements are more easily tracked and recorded by health care professionals, for example ERAS elements found in the medical record or included on order sets e.g. PONV prophylaxis were missed less frequently by the multidisciplinary teams.³³ Data for elements not found in the medical record, and collected using other methodologies such as patient journals (e.g. mobility and nutrition) had lower rates of adherence.³³ Barriers to nutrition care within the AHS EIP have been identified and are related to care processes at individual hospital sites, and information and communication for providers and patients related to eating and drinking after surgery.^{33,34} Strategies to enhance nutrition care include process improvement for ensuring delivery of appropriate snack and ONS, and encouragement and education for patients and health care providers.³³ Improved adherence to other ERAS elements that influence oral intake and bowel function (e.g. optimal management of pain/nausea, fluid balance, non-opioid analgesia) will contribute to improved postoperative oral intake.

Strengths of this study include the large, representative patient cohort, with data collected over a whole health system. Our results are consistent with other EIPs, and studies examining the impact of nutrition risk on outcomes of surgery. ERAS care elements may need to be modified and expanded to account for the growing literature identifying the many factors that influence compliance and impact on postoperative outcomes.^{7,8} This study highlights nutrition risk as one of these factors, and development of alternate standardized care pathways may benefit these patients.^{35,36} For example, evidence-based multimodal prehabilitation programs have demonstrated some efficacy in contributing to improved patient outcomes within ERAS programs.^{37,38} Prehabilitation offers individualized nutrition (e.g. counselling and supplemental

144

protein, n-3 fatty acids), function (exercise program), and anxiety-reduction to promote metabolic preparation and optimization of health status beginning pre-operatively and extending into recovery.³⁷ AHS is working towards a standardized perioperative nutrition care pathway within the ERAS paradigm.

Ultimately, AHS EIP changed practice with the successful introduction of nutrition risk screening, and basic aspects of nutrition care that align with clinical guidelines for surgical patients. Areas for further development include reducing the number of missing nutrition risk screens, and development of processes that link nutrition screening to assessment, provision of nutrition support for at risk patients, and development of nutrition care elements tailored to patients likely to deviate from the standard ERAS care protocol.

7.6 References

- 1. Weimann A, Braga M, Carli F, et al. ESPEN guideline: Clinical nutrition in surgery. Clin Nutr. 2017; 36:623-650.
- 2. Williams JD, Wischmeyer PE. Assessment of perioperative nutrition practices and attitudes-A national survey of colorectal and GI surgical oncology programs. Am J Surg. 2017; 213:1010-1018.
- 3. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 4. Sandrucci S, Beets G, Braga M, Dejong K, Demartines N. Perioperative nutrition and enhanced recovery after surgery in gastrointestinal cancer patients. A position paper by the ESSO Task Force in collaboration with the ERAS Society (ERAS Coalition). European Journal of Surgical Oncology. 2018; doi:10.1016/j.ejso.2017.12.010.
- 5. Tappenden KA, Quatrara B, Parkhurst ML, Malone AM, Fanjiang G, Ziegler TR. Critical role of nutrition in improving quality of care: an interdisciplinary call to action to address adult hospital malnutrition. J Acad Nutr Diet. 2013; 113:1219-1237.
- 6. Wischmeyer PE, Carli F, Evans DC, et al. American Society for Enhanced Recovery and Perioperative Quality Initiative Joint Consensus Statement on Nutrition Screening and Therapy Within a Surgical Enhanced Recovery Pathway. Anesth Analg. 2018; 126:1883-1895.
- 7. Ljungqvist O, Scott M, Fearon KC. Enhanced Recovery After Surgery: A Review. JAMA Surg. 2017; 152:292-298.
- 8. Steenhagen E. Enhanced Recovery After Surgery: It's time to change practice! Nutr in Clin Pract. 2016; 31:18-29.
- 9. Grant MC, Yang D, Wu CL, Makary MA, Wick EC. Impact of Enhanced Recovery After Surgery and Fast Track Surgery Pathways on Healthcare-associated Infections: Results From a Systematic Review and Meta-analysis. Ann Surg. 2017; 265:68-79.
- Greco MC, G; Beretta,L; Gemma,M; Pecorelli,N; Braga M. Enhanced Recovery Program in Colorectal Surgery: A Meta-analysis of Randomized Controlled Trials. World J Surg. 2014; 38:1531-1541.
- 11. Visioni A, Shah R, Gabriel E, Attwood K, Kukar M, Nurkin S. Enhanced Recovery After Surgery for Noncolorectal Surgery?: A Systematic Review and Meta-analysis of Major Abdominal Surgery. Ann Surg. 2018; 267:57-65.
- 12. Lau CSM, Chamberlain RS. Enhanced Recovery After Surgery Programs Improve Patient Outcomes and Recovery: A Meta-analysis. World J Surg. 2017; 41:899-913.

- 13. Gustafsson UO, Ljungqvist O. Perioperative nutritional management in digestive tract surgery. Curr Opin Clin Nutr Metab Care. 2011; 14:504-509.
- 14. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 15. Bauer MS, Damschroder L, Hagedorn H, Smith J, Kilbourne AM. An introduction to implementation science for the non-specialist. BMC Psychol. 2015; 3:32.
- 16. ERAS Society. ERAS Implementation Program (EIP). 2018. https://www.encare.net/healthcare-professionals/products-andservices/ eras-implementation-program-eip. Accessed March 5, 2018.
- Group EC. The Impact of Enhanced Recovery Protocol Compliance on Elective Colorectal Cancer Resection: Results From an International Registry. Ann Surg. 2015; 261:1153-1159.
- 18. Messenger DE, Curtis NJ, Jones A, Jones EL, Smart NJ, Francis NK. Factors predicting outcome from enhanced recovery programmes in laparoscopic colorectal surgery: a systematic review. Surg Endosc. 2016; 31:2050-2071.
- 19. Gustafsson UO, Hausel J, Thorell A, et al. Adherence to the enhanced recovery after surgery protocol and outcomes after colorectal cancer surgery. Arch Surg. 2011; 146:571-577.
- 20. Nelson G, Kiyang LN, Chuck A, Thanh NX, Gramlich LM. Cost impact analysis of Enhanced Recovery After Surgery program implementation in Alberta colon cancer patients. Curr Oncol. 2016; 23:e221-227.
- 21. International Expert Committee report on the role of the A1C assay in the diagnosis of diabetes. Diabetes Care. 2009; 32:1327-1334.
- 22. Nelson G, Kiyang LN, Crumley ET, et al. Implementation of Enhanced Recovery After Surgery (ERAS) Across a Provincial Healthcare System: The ERAS Alberta Colorectal Surgery Experience. World J Surg. 2016; 40:1092-1103.
- 23. Gustafsson UO, Scott MJ, Schwenk W, et al. Guidelines for perioperative care in elective colonic surgery: Enhanced Recovery After Surgery (ERAS(R)) Society recommendations. Clin Nutr. 2012; 31:783-800.
- 24. Ljungqvist O. ERAS Enhanced Recovery After Surgery: moving evidence-based perioperative care to practice. JPEN J Parenter Enteral Nutr. 2014; 38:559-566.
- 25. Dindo D, Demartines N, Clavien PA. Classification of Surgical Complications. Annals of Surgery. 2004; 240:205-213.

- 26. Ferguson M, Capra S, Bauer J, Banks M. Development of a valid and reliable malnutrition screening tool for adult acute hospital patients. Nutrition. 1999; 15:458-464.
- 27. Laporte M, Keller HH, Payette H, et al. Validity and reliability of the new Canadian Nutrition Screening Tool in the 'real-world' hospital setting. Eur J Clin Nutr. 2015; 69:558-564.
- 28. Gillis C, Nguyen TH, Liberman AS, Carli F. Nutrition adequacy in enhanced recovery after surgery: a single academic center experience. Nutr Clin Pract. 2015; 30:414-419.
- 29. Jie B, Jiang ZM, Nolan MT, Zhu SN, Yu K, Kondrup J. Impact of preoperative nutritional support on clinical outcome in abdominal surgical patients at nutritional risk. Nutrition. 2012; 28:1022-1027.
- 30. Burden ST, Hill J, Shaffer JL, Todd C. Nutritional status of preoperative colorectal cancer patients. J Hum Nutr Diet. 2010; 23:402-407.
- 31. Laporte, M. Comparison of nutrition screening tools for hospital use. Canadian Malnutrition Task Force. 2013, http://nutritioncareincanada.ca/ sites/default/uploads/files/Comparison-of-nutrition-screening-toolsforhospital-use.pdf. Accessed June 27, 2018.
- 32. Aarts MA, Rotstein OD, Pearsall EA, et al. Postoperative ERAS Interventions Have the Greatest Impact on Optimal Recovery: Experience With Implementation of ERAS Across Multiple Hospitals. Ann Surg. 2018; 267: 992-997.
- 33. Gramlich LM, Sheppard CE, Wasylak T, et al. Implementation of Enhanced Recovery After Surgery: a strategy to transform surgical care across a health system. Implement Sci. 2017; 12:67.
- Gillis CG, M.; Marlett, N.; MacKean, G.; GermAnn, K.; Gilmour, L.; Nelson, G.; Wasylak, T.; Nguyen, S.; Araujo, E.; Zelinsky, S.; Gramlich, L. Patients as partners in Enhanced Recovery After Surgery: A qualitative patient-led study. BMJ Open. 2017; 7:10.
- 35. Correia MI, Hegazi RA, Higashiguchi T, et al. Evidence-based recommendations for addressing malnutrition in health care: an updated strategy from the feedM.E. Global Study Group. J Am Med Dir Assoc. 2014; 15:544-550.
- 36. Keller HH, McCullough J, Davidson B, et al. The Integrated Nutrition Pathway for Acute Care (INPAC): Building consensus with a modified Delphi. Nutr J. 2015; 14:63.
- 37. Carli F, Gillis C, Scheede-Bergdahl C. Promoting a culture of prehabilitation for the surgical cancer patient. Acta Oncol. 2017; 56:128-133.

38. Barberan-Garcia A, Ubre M, Roca J, et al. Personalised Prehabilitation in High-risk Patients Undergoing Elective Major Abdominal Surgery: A Randomized Blinded Controlled Trial. Ann Surg. 2018; 267:50-56.

Table 7-1. Enhanced recovery after surgery (ERAS) care elements for people undergoing major colorectal surgeries in Alberta, Canada

Preoperative Care Elements	Intraoperative Care Elements	Postoperative Care Elements
Preadmission counselling	Nausea and vomiting prophylaxis	Mid-thoracic epidural anesthesia/analgesia
Nutrition risk screening	Short acting anesthetic agents	No nasogastric tubes
Fluid and carbohydrate loading	Mid-thoracic epidural anesthesia/analgesia	Prevention of nausea and vomiting
No prolonged fasting	No drains	Avoidance of salt and water overload
No/selective bowel preparation	Avoidance of salt and water overload	Early removal of catheter
Antibiotic prophylaxis	Maintenance of normothermia (body warmer/warm intravenous fluids)	Early oral nutrition
Thromboprophylaxis		Non-opioid oral analgesia/NSAIDs
No premedication		Early mobilization
		Stimulation of gut motility
		Audit of compliance and outcomes

NSAIDs, non-steroidal anti-inflammatory drugs

*Nutrition care elements related to nutrition care practices are listed in bold

ERAS elements in italics are recognized to have an important impact on nutrition status and metabolism but do not fall under the purview of nutrition care practices

Nutrition Care Elements	ERAS Guideline	ERAS Guideline Compliant	ERAS Guideline Non-compliant
Nutrition risk* screening [‡]	Nutrition risk score recorded: Malnutrition Screening Tool OR Canadian Nutrition Screening Tool	yes	no
Fluid and carbohydrate loading [‡]	Patient treated with a preoperative carbohydrate rich drink: At least 50g of carbohydrates together with at least 400 ml fluid, be given up until 2 hours prior to anesthesia	yes	no
No prolonged fasting [‡]	Patient allowed solid food up to 6 hours before and clear fluids up to 2 hours before induction of anesthesia	yes	no
No/selective bowel preparation [‡]	Patient received oral bowel preparation /cleansing preoperatively	no	yes
Postoperative nausea and vomiting prophylaxis [‡]	PONV prophylaxis given before end of operation	yes	no
Stimulation of gut motility [‡]	Patient received gut motility stimulation (gum, laxative or both)	yes	no
Early oral nutrition [§]			
POD 0	Energy from ONS on POD 0 until early morning POD 1	≥300 kcal	< 300 kcal
POD 1	Energy from ONS on POD 1 until early morning POD 2	≥600 kcal	<600 kcal
POD 2	Energy from ONS on POD 2 until early morning POD 3	≥600 kcal	<600 kcal
POD 3	Energy from ONS on POD 2 until early morning POD 4	≥600 kcal	<600 kcal
Early mobilization [§]			
POD 0	Patient mobilized at all	yes	no
POD1	Number of hours, in total, patient mobilized	≥4 hours	< 4 hours
POD 2 to 3	Number of hours, in total, patient mobilized	≥6 hours	< 6 hours

Table 7-2. Criteria for compliance to ERAS nutrition care elements

ERAS, Enhanced Recovery After Surgery; g, grams; ml, milliliters; ONS, oral nutritional supplements; POD, postoperative day

*Nutrition risk defined as ≥ 2 on the Malnutrition Screening Tool (MST) or two "yes" answers on the Canadian Nutrition Screening Tool (CNST); completed by a health care professional

[†]Mobile refers to a patient who can rise from bed to walk, or to sit in chair

[‡]Data collected and recorded in medical chart by a health care professional; data abstracted and entered in to EIAS by ERAS nurse coordinator according to ERAS standardized data definitions

[§]Data collected from patient report: patients recorded in an ERAS journal the amount of ONS consumed/day and time spent mobilizing/day for a total of 3 days; data was entered into EIAS by ERAS nurse coordinator, calories from ONS were estimated based on reported volume of ONS consumed.

Patients Characteristics	Pre-ERAS	ERAS	P- Value	Not At	At Nutrition	P- Value
	(11-407)	(11-3530)	value †	Risk	Risk*	v alue †
				(N=2317)	(N=311)	
Sex (N, %)			0.372			0.306
female	222 (46)	1539 (43)		994 (43)	143 (46)	
male	264 (54)	1996 (57)		1322 (57)	168 (54)	
Age, years (mean, SD)	62.4 (13.7)	61.4 (14.3)	0.128	61.6 (14.1)	62.5 (14.9)	0.296
AgeCategory			0.105			0.091
18-25	3 (0.6)	49 (1)		27 (1)	6 (2)	
26-50	81 (17)	631 (18)		400 (17)	53 (17)	
51-75	311 (64)	2317 (66)		1539 (66)	190 (61)	
76-100	92 (19)	539 (15)		351 (15)	62 (20)	
Height, cm (mean, SD)	168.4 (10.5)	168.2 (10.2)	0.762	168.4 (10.1)	167.9 (9.7)	0.395
Weight, kg (mean, SD)	80.5 (21.0)	80.2 (19.8)	0.798	80.5 (19.6)	74.5 (20.2)	<0.001
BMI, kg/m ² (mean, SD)	28.3 (6.4)	28.2 (6.2)	0.923	28.3 (5.9)	26.3 (6.4)	<0.001
BMI class (N,%)			0.060			<0.001
<18.5 (underweight)	20 (4)	81 (2)		43 (2)	14 (5)	
18.5-24.9 (normal	138 (28)	1019 (29)		658 (28)	131 (42)	
weight) 25.0-29.9 (overweight)	159 (33)	1269 (36)		846 (37)	97 (31)	
≥30.0 (obese)	170 (35)	1167 (33)		770 (33)	69 (22)	
Alcohol use (N, %)			<0.001			0.470
no	393 (81)	3191 (90)		2101 (91)	277 (89)	
yes	78 (16)	299 (9)		190 (8)	29 (9)	
stopped due to surgery	2 (0.2)	8 (0.4)		5 (0.2)	0 (0)	
unknown	15 (3)	38 (1)		21 (1)	5 (2)	
Smoker (N ,%)			0.487			<0.001
no	402 (82)	2900 (82)		1929 (83)	231 (74)	
yes	84 (17)	613 (17)		379 (16)	76 (24)	
unknown	1 (1)	23 (1)		9 (0.4)	4 (1)	
Diabetes (N, %)			0.700			0.708
no	413 (85)	2964 (84)		1934 (83)	258 (83)	
yes	74 (15)	564 (16)		380 (16)	52 (17)	
unknown	0 (0)	8 (0.2)		3 (0.1)	1 (0.3)	
ASA Class (N, %)			0.523			0.067
1 to 2	326 (67)	2385 (67)		1582 (68)	192 (62)	
3 to 4	137 (28)	1074 (30)		68 (30)	111 (36)	
unknown	24 (5)	76 (2)		47 (2)	8 (3)	
Cancer Diagnosis (N, %)			0.204			0.004

 Table 7-3. Characteristics of patients who underwent elective colorectal surgery within an ERAS implementation program.

non-cancer diagnosis	244 (50)	1880 (53)		1105 (48)	175 (56)	
cancer diagnosis	243 (50)	1656 (47)		1212 (52)	136 (44)	
Diagnosis (N, %)			<0.001			<0.001
benign tumor including polyp(s)	54 (11)	531 (15)		402 (17)	23 (7)	
Crohn's disease	25 (5)	51 (1)		28 (1)	7 (2)	
diverticular disease	50 (10)	236 (7)		148 (6)	31 (10)	
functional disorder	23 (5)	152 (4)		86 (4)	9 (3)	
inflammatory bowel	22 (5)	266 (8)		168 (7)	25 (8)	
disease other benign disease or disorder	68 (14)	644 (18)		380 (16)	41 (13)	
other primary malignancy	9 (2)	48 (1)		29 (1)	6 (6)	
primary adenocarcinoma	222 (46)	1574 (45)		1053 (45)	164 (53)	
metastasis or recurrence	12 (3)	34 (1)		23 (1)	5 (2)	
malignancy Procedure Type (N, %)			0.627			0.143
rectal procedure	327 (67)	2335 (66)		818 (35)	123 (40)	
colon and small bowel	160 (33)	1201 (34)		1499 (65)	188 (60)	
procedure Surgical approach (N, %)			<0.001			0.016
laparoscopic	174 (36)	1628 (46)		1198 (52)	145 (47)	
open	232 (48)	1336 (38)		795 (34)	135 (43)	
stoma approach	48 (10)	410 (11)		213 (9)	19 (6)	
converted	26 (5)	134 (4)		101 (4)	12 (4)	
unknown	7 (1)	28 (1)		10 (0.4)	0 (0)	
Surgical Complexity [‡] (N, %)			0.863			0.034
less complex procedures	286 (59)	2062 (58)		1324 (57)	158 (51)	
more complex procedures	201 (41)	1474 (42)		993 (43)	153 (49)	
Acute Care Centre			<0.001			<0.001
Hospital Site #1	50 (13)	502 (14)		196 (9)	31 (10)	
Hospital Site #2	75 (19)	1182 (33)		1003 (43)	86 (28)	
Hospital Site #3	56 (14)	504 (14)		415 (18)	44 (14)	
Hospital Site #4	48 (12)	513 (15)		251 (11)	66 (21)	
Hospital Site #5	97 (25)	389 (11)		227 (10)	50 (16)	
Hospital Site #6	61 (16)	446 (13)		225 (10)	34 (11)	
Short-term post- operative outcomes Length of hospital stay (days)						
mean (SD)	8.8 (18.1)	7.6 (8.8)	<0.001	7.4 (7.6)	8.1 (7.6)	<0.001
median (IQR)	6.0 (5.0)	5.0 (4.0)		5.0 (4.0)	6.0 (5.0)	
Extended LOS (N, %)						

<5 days	204 (42)	1800 (51)	<0.001	1228 (53)	121 (39)	<0.001
>5 days	280 (58)	1706 (49)		1104 (47)	191 (61)	
Complication(s) during primary stay (N, %)			0.069			0.016
yes	212 (44)	1707 (48)		1127(49)	174 (56)	
no	273 (56)	1829 (52)		1190 (51)	137 (44)	
Complication severity [‡] (N, %)			0.764			0.022
non-serious (Grade I to II)	184 (90)	1537 (90)		1027 (92)	149 (86)	
serious (Grade III to IV)	21 (10)	163 (10)		95 (8)	24 (14)	
30-day readmission rate (N,%)	52 (11)	343 (10)	0.364	236 (10)	27 (9)	0.156
30-day mortality rate (N, %)	1 (0.2)	26 (0.7)	0.388	15 (0.6)	7 (2.2)	<0.001
Compliance to ERAS care % (Mean, SD)	e elements,					
pre-operative care elements	61.5 (13.3)	86.3 (13.8)	<0.001	86.7 (13.6)	87.6 (13.2)	0.274
intra-operative care elements	69.5 (15.2)	77.7 (16.5)	<0.001	79.5 (15.7)	75.4 (17.9)	<0.001
post-operative care elements	34.8 (10.8)	58.5 (16.7)	<0.001	59.1 (16.3)	55.0 (16.0)	<0.001
overall	51.3 (7.0)	71.8 (10.1)	<0.001	73.6 (9.5)	68.1 (9.3)	<0.001
Overall Compliance to all ERAS care elements						<0.001
>70% (good compliance)	4 (1)	2245 (63)	<0.001	1642 (71)	154 (50)	
<70% (low compliance)	483 (99)	1291 (37)		675 (29)	157 (50)	

ASA, American Society of Anesthesiologists physical status classification; BMI, body mass index; ERAS, Enhanced Recovery After Surgery; IQR, interquartile range; LOS, length of hospital stay; N, number; SD, standard deviation

*Nutrition risk defined as ≥ 2 on the Malnutrition Screening Tool (MST) or two "yes" answers on the Canadian Nutrition Screening Tool (CNST), patients in the ERAS implementation group whose nutrition screen status was unknown (N=908) were excluded from this analysis

[†]P-value based on comparison between groups (Pre-ERAS vs. ERAS; Not at Nutrition Risk vs. At Nutrition Risk) Chi-square, Fischer's exact test, or independent t-test were used where appropriate

[‡]Surgical complexity = more complex procedures: abdominoperineal resection, anterior resection of rectum, total/subtotal colectomy, reversal of Hartmann's procedure; less complex procedures: right hemicolectomy, left hemicolectomy, other large/small bowel resection, ileostomy reversal

[§] based on Clavien-Dindo complication classification

ERAS Protocol	Pre-ERAS	ERAS	Р-	Not At	At	Р-
Nutrition Care Elements	(N=487)	(N=3536)	Value	Nutrition	Nutrition	Value
				(N=2317)	KISK* (N=311)	
Pre-Surgery						
Nutrition screen (N, %)			<0.001	-	-	-
yes	43 (9)	2628 (74)		-	-	-
no	445 (91)	908 (26)		-	-	-
Carbohydrate treatment (N, %)			<0.001			0.377
yes	18 (4)	2142 (61)		1425 (62)	196 (63)	
no	467 (96)	1238 (35)		810 (35)	100 (32)	
unknown	2 (0)	156 (4)		82 (4)	15 (5%)	
Oral bowel preparation (N, %)			<0.001			0.310
no	270 (55)	2561 (72)		1691 (73)	236 (76)	
yes	210 (43)	951 (27)		619 (27)	73 (23)	
unknown	7 (1)	24 (1)		7 (0.3)	2 (1)	
PONV prophylaxis administered (N, %)			<0.001			0.193
yes	386 (79)	3133 (89)		2069 (89)	267 (86)	
no	94 (19)	369 (10)		231 (10)	41 (13)	
unknown	7 (1)	34 (1)		17 (1)	3 (1)	
Post-Surgery						
Stimulation of gut motility (N, %)			<0.001			<0.001
yes (laxatives or gum or both)	67 (14)	2559 (72)		1715 (74)	216 (69)	
no stimulation given	420 (86)	445 (13)		195 (8)	51 (16)	
unknown	0 (0)	532 (15)		407 (18)	44 (14)	
POD 0 ONS Intake ≥300 kcal (N, %)			<0.001			<0.001
yes	0 (0)	709 (20)		507 (22)	61 (20)	
no	287 (59)	2243 (63)		1477 (64)	178 (57)	
unknown	300 (41)	584 (17)		333 (14)	72 (23)	0.00 -
POD 0 Energy intake (kcal) from ONS (Mean, SD)	1.6 (18.3)	156.3 (148.1)		173.1 (147.1)	145.8 (153.7)	0.007
POD 0 Mobile [†] at all (N, %)			<0.001			<0.001
yes	209 (43)	2150 (61)		1528 (66)	162 (52)	
no	253 (52)	1300 (37)		739 (32)	145 (47)	
unknown/not applicable	25 (5)	86 (2)		50 (2)	4 (1)	
POD 1 Patient's Morning Weight Measured [‡]			<0.001			0.001
yes	40 (8)	2269 (64)		1626 (70)	189 (61)	

Table 7-4. The impact of ERAS implementation and nutrition risk on nutrition care elements.

no	447 (92)	1267 (36)		691 (30)	122 (39)	
POD 1 ONS Intake ≥600			<0.001			<0.001
kcal (N, %)	1 (0)	10(1(20)				
yes	1 (0)	1061 (30)		775 (33)	77 (25)	
no	286 (59)	1977 (56)		1266 (55)	174 (56)	
unknown	200 (41)	498 (14)		276 (12)	60 (19)	
POD 1 Energy intake (kcal) from ONS (Mean, SD)	8.3 (58.0)	359.3 (245.7)		389.9 (235.2)	343.5 (242.8)	0.003
POD 1 Hours mobile ≥4 hours (N, %)			<0.001			<0.001
yes	32 (7)	1650 (47)		995 (43)	134 (43)	
no	89 (18)	695 (18)		401 (17)	83 (27)	
unknown / not applicable	366 (75)	1191 (34)		921 (40)	94 (30)	
POD 1 Hours mobile	2.8 (2.7)	4.8 (2.5)	<0.001	4.9 (2.6)	4.1 (2.6)	<0.001
(mean, SD) POD 2 Patient's Morning Weight			<0.001			
Measured						
yes	53 (11)	2195 (62)		1545 (67)	187 (60)	0.026
no	434 (89)	1341 (38)		772 (33)	124 (40)	
POD 2 ONS intake ≥600 kcal (N, %)		427 (12)	<0.001	202 (12)	20 (0)	0.002
yes	0(0)	427 (12)		303 (13)	29 (9)	
no	288 (59)	2543 (72)		1686 (73)	216 (69)	
unknown	199 (41)	566 (14)		328 (14)	66 (21)	
POD 2 Energy intake (kcal) from ONS (Mean, SD)	10.0 (57.4)	299.6 (250.2)		319.2 (247.4)	305.1 (243.4)	0.400
POD 2 Hours mobile ≥ 6 hours (N, %)			<0.001			<0.001
yes	21 (4)	1074 (30)		661 (29)	84 (27)	
no	85 (18)	1075 (30)		604 (26)	122 (39)	
unknown / not applicable	381 (78)	1387 (39)		1052 (45)	105 (34)	
POD 2 Hours mobile	3.6 (2.7)	5.4 (2.7)	<0.001	5.4 (2.7)	4.9 (2.7)	0.001
POD 3 Patient's Morning Weight Measured			<0.001			0.060
yes	56 (11)	1657 (47)		1182 (51)	141 (45)	
no	431 (89)	1879 (53)		1135 (49)	170 (55)	
POD 3 ONS intake ≥600 kcal (N, %)			<0.001			0.066
yes	2 (0)	511 (15)		1419 (61)	183 (60)	
no	282 (58)	2177 (62)		521 (23)	187 (60)	
unknown	203 (42)	848 (24)		527 (23)	86 (28)	
POD 3 Energy intake (kcal) from ONS (Mean, SD)	14.37 (98.5)	224.9 (244.1)		244.9 (246.7)	211.4 (235.1)	0.046
POD 3 Hours mobile ≥6			<0.001			<0.001

hours (N, %)						
yes	20 (4)	855 (24)		494 (21)	75 (24)	
no	70 (14)	822 (23)		470 (20)	89 (29)	
unknown / not applicable	397 (82)	1859 (53)		1353 (58)	147 (47)	
POD 3 Hours mobile	3.9 (2.7)	5.4 (2.7)	<0.001	5.4 (2.7)	5.2 (2.8)	0.289
(mean, SD) Use of artificial nutrition (N, %)			<0.001			<0.001
none	140 (29)	1763 (50)		1233 (53)	118 (38)	
ONS only	7 (1)	659 (19)		418 (18)	68 (22)	
enteral nutrition only	0 (0)	9 (0)		4 (0.2)	2 (1)	
enteral and parenteral nutrition	0 (0)	47 (1)		21 (1)	8 (3)	
parenteral nutrition only	5 (1)	171 (5)		130 (6)	16 (5)	
unknown	335 (69)	887 (25)		511 (22)	99 (32)	
Days to recover ADL [§] (mean, SD)	7.7 (24.2)	3.4 (4.7)	<0.001	3.1 (10.0)	4.5 (6.1)	0.046
assessed (N, %)				1738 (75)	198 (64)	
unknown / not applicable (N, %)				579 (25)	113 (36)	
Days to first flatus (mean, SD)	3.4 (16.9)	2.1 (2.5)	0.078	2.1 (2.9)	2.1 (1.7)	0.938
assessed (N, %)	481 (99)	3426 (97)		2259 (97)	299 (96)	
unknown / not applicable (N, %)	6 (1)	110 (3)		58 (3)	12 (4)	
Days to first stool (mean, SD)	3.7 (3.6)	3.1 (3.3)	0.004	3.1 (3.3)	3.2 (3.1)	0.417
assessed (N, %)	413 (85)	2871 (81)		1820 (78)	266 (86)	
unknown / not applicable (N, %)	74 (15)	665 (19)		497 (22)	45 (15)	
Days to tolerating solid food ¹ (mean, SD)	5.9 (18.4)	2.3 (4.6)	0.001	2.8 (4.5)	3.3 (6.4)	0.079
assessed (N, %)	461 (95)	3377 (96)		2234 (96)	288 (93)	
unknown / not applicable (N, %)	26 (5)	159 (4)		83 (4)	23 (7)	

ADL, activities of daily living; ERAS, Enhanced Recovery After Surgery; kcal, calories; N, number; ONS, oral nutritional supplments; PONV, post-operative nausea and vomiting; POD, post-operative day; SD, standard deviation; VAS, visual analogue scale;

*Nutrition risk defined as ≥ 2 on the Malnutrition Screening Tool (MST) or two "yes" answers on the Canadian Nutrition Screening Tool (CNST), patients in the ERAS implementation group whose nutrition screen status was unknown (N=908) were excluded from this analysis

[†]Mobile refers to a patient who can rise from

bed to walk, or to sit in chair

[‡]Morning weights = patient's were weighed each morning using an electric scale located on the surgical wards (calibrated weekly), weight was recorded in medical chart by health care professional

[§]Days to achieve ADL = the number of days after surgery until patient was out of bed for more than 6 hours per day and at same level of independence with respect to daily living as before surgery.

Tolerating solid food = consuming solid food in normal portions without vomiting and excludes the use of a nasogastric tube

Table 7-5. Logistic regression model of the factors associated with low compliance (<70%) to ERAS care elements for patients in the ERAS implementation group (N=2628)

	 ⊺	Univariable		Final Multivariable			
Factors	Ν	OR (95% CI)	P-Value	Ν	OR (95% CI)	P-Value	
Nutrition Screen*							
not at nutritional risk	2317	1.0 (ref)		2316	1.0 (ref)		
at nutritional risk	311	2.48 (1.95-3.15)	<0.001	311	2.77 (2.11-3.64)	<0.001	
Sex							
female	1137	1.0 (ref)					
male	1490	0.95 (0.81-1.12)	0.559	-	-	-	
Age (years)	2627	1.0 (0.99-1.00)	0.227	-	-	-	
ASA Class							
1 to 2	1774	1.0 (ref)		1774	1.0 (ref)		
3 to 4	798	1.51 (1.27-1.80)	<0.001	798	1.38 (1.13-1.68)	0.001	
unknown	55	1.20 (0.68-2.13)	0.528	55	1.61 (0.83-3.15)	0.161	
Cancer Diagnosis							
non-cancer diagnosis	1280	1.0 (ref)	0.496	-	-	-	
cancer diagnosis	1348	1.06 (0.90-1.24)		-	-	-	
Procedure Type							
colon and small bowel	1687	1.0 (ref)		-	-	-	
rectal procedure	914	1.42 (1.20-1.68)	<0.001	-	-	-	
Surgical approach							
laparoscopic	1343			1342	1.0 (ref)		
open	930	2.72 (2.28-3.25)	< 0.001	930	3.08 (2.50-3.81)	<0.001	
stoma approach	232	0.67 (0.47-0.96)	0.028	232	0.75 (0.50-1.10)	0.142	
converted	113	2.75 (1.86-4.07)	< 0.001	113	2.72 (1.77-4.18)	<0.001	
unknown	10	0.78 (0.17-3.69)	0.753	10	2.24 (0.45-11.18)	0.324	
Surgical Complexity							
less complex procedures	1482	1.0 (ref)		-	-	-	
more complex procedures	1146	1.56 (1.32-1.84)	<0.001	-	-	-	
Complication(s) during primary stay							
no	1327	1.0 (ref)		1327	1.0 (ref)		
yes	1301	3.05 (2.56-3.62)	<0.001	1300	2.53 (2.09-3.04)	<0.001	
BMI class							
<18.5 (underweight)	57	1.64 (0.95-2.81)	0.074	57	1.35 (0.74-2.43)	0.328	
18.5-24.9 (normal weight)	789	1.0 (ref)		789	1.0 (ref)		
25.0-29.9 (overweight)	943	0.80 (0.66-0.98)	0.033	942	0.83 (0.67-1.04)	0.110	
≥30.0 (obese)	839	0.71 (0.58-0.88)	0.001	839	0.74 (0.59-0.94)	0.014	

Outcome = low overall compliance to ERAS care elements (<70%)

Acute Care Center						
Hospital #1	227	1.0 (ref)		277	1.0 (ref)	
Hospital #2	1089	2.33 (1.68-3.23)	<0.001	1088	4.56 (3.14-6.62)	<0.001
Hospital #3	459	0.81 (0.55-1.18)	0.263	459	1.10 (0.73-1.66)	0.653
Hospital #4	317	1.03 (0.70-1.54)	0.853	317	1.37 (0.89-2.12)	0.153
Hospital #5	277	1.18 (0.79-1.77)	0.414	277	2.01 (1.27-3.16)	0.003
Hospital #6	259	1.01 (0.66-1.52)	0.981	259	1.19 (0.76-1.86)	0.456
Diabetes						
no	2192	1.0 (ref)		-	-	-
yes	432	1.14 (0.91-1.41)	0.249	-	-	-
unknown	4	0.73 (0.08-7.08)	0.790	-	-	-

ASA, American Society of Anesthesiologists physical status classification; BMI, body mass index ; CI, confidence interval; ERAS, Enhanced Recovery After Surgery; LOS, length of hospital stay; N, number; OR, odds ratio;

*Nutrition risk was only calculated for pateints in the ERAS Group and defined as ≥ 2 on the Malnutrition Screening Tool (MST) or two "yes" answers on the Canadian Nutrition Screening Tool (CNST), patients in the ERAS Group whose nutrition screen status was unknown (N=908) were excluded from this analysis (-) dash indicates variables that were not included in the logistic regression models because they were not significant at univariable or multivariable level

Logistic Regression Full model (included all variables significant at univariable level) -2 LL = 2770.997, Negelkerke $R^2 = 25\%$, Hosmer & Lemeshow test P= 0.318, Classification = 73%; Fitted Model (retains variables significant at multivariable level) -2LL = 2775.011, Negelkerke $r^2 = 25\%$, Hosmer & Lemeshow test P=0.131, Classification = 73%
Chapter 8: Assessment of CT-defined muscle and adipose tissue features in relation to short term outcomes after elective surgery for colorectal cancer: a multicenter approach¹

8.1 Introduction

Computed tomography (CT)-defined features of skeletal muscle and adipose tissue have emerged as a new prognostic tools for postoperative complications, length of hospital stay (LOS), readmissions and mortality.¹⁻⁶ In colorectal cancer (CRC) surgery, 36 reports (N=7,666 patients) on CT-defined body composition features and post-surgical outcomes have appeared, focusing on skeletal muscle (SM, 17 studies) and visceral adipose tissue (VAT, 20 studies).^{1,3,7-30} Sarcopenia, visceral obesity (VO) and reduced muscle radiodensity (myosteatosis) have been associated with post-operative complications, extended LOS, and survival.¹³¹⁻²⁵ However, there is a lack of methodological consistency³¹⁻³⁴, and threshold values to define these features have been derived in from non-surgical, non-CRC patients. Threshold values associated with increased risk of mortality have been published^{5,25,35,36}, but were derived in unresectable cancers and their application to surgical settings is questionable³⁷ because features such as SM may have unique relationships with different outcomes.³⁸ Furthermore, effects of aging on body composition has largely been ignored in developing thresholds; SM declines after the age of 50, whereas fat mass increases with age, and declines into extreme old age.^{39,40} Use of standardized scores (z-scores) is an approach used to account for age and sex specific effects on body composition, but applying standardized scores requires a large study population.⁴¹ Lastly, sarcopenia, VO, and myosteatosis have been most often evaluated in isolation^{1,3,7-13,15}, and we suggest this is inadequate to describe a given individual. For example, in CRC, sarcopenic obesity associated with poorer postoperative outcomes, when compared to either feature

¹A version of this chapter has been published: Martin L, Hopkins J, Malietzis G, Jenkins JT, Sawyer MB, Brisebois R, MacLean A, Nelson G, Gramlich L, Baracos VE. Assessment of CT-defined muscle and adipose tissue features in relation to short term outcomes after elective surgery for colorectal cancer: a multicenter approach. Ann Surg Oncol, 2018; 25:2669-2680

alone.^{2,4,7,9} We propose a full assessment of all possible combinations of sarcopenia, VO and myosteatosis is needed.

Large data sets for CT-defined body composition features of resectable CRC patients are lacking, and heterogeneity of CT image analysis methodologies and differences defining surgical outcomes (e.g. complications) create difficulty aggregating existing data.³¹ As a result, the generalizability of results is difficult to determine. Development of a data set specific to CRC is necessary to characterize what is usual for CRC patients with respect to body composition features, and to evaluate these features in relation to postoperative outcomes.

We aimed to establish a large, pooled sample consisting of pre-operative CT-defined body composition in CRC patients eligible for elective surgery. Study cohorts with similar characteristics were identified, and sex and age norms for body composition features specific to CRC were developed. Thresholds for sarcopenia, myosteatosis, and VO were defined based on their association with LOS, and the combination of body composition features on LOS, major complications, and readmissions was evaluated. Post-operative complications are recognized to be a main driver of LOS and readmissions; however patients with altered body composition profiles may have extended LOS independent of postoperative complications.⁴²⁻⁴⁴ We adjusted for major complications in our analysis of LOS and readmissions.

8.2 Methods

8.2.1 Study design & participants

Our pooled analysis (see statistics for pooling criteria) includes data from Canadian and UK cohorts: The Canadian Cohort includes consecutive primary CRC patients who underwent elective surgical resection from August 2013 to March 2015⁴⁵ (Cohort 1) and January 2007 to December 2009 (Cohort 2) and in Alberta, Canada (population ~4 million). The UK Cohort

included consecutive patients with primary CRC who underwent surgery at St. Mark's Hospital, London, UK from January 2006 to December 2011.⁹ Included patients were adults (\geq 18 years) with primary CRC who underwent elective colorectal resection surgery, and had a preoperative CT image. Patients were excluded if they had recurrent disease confirmed before or during surgery, had emergency surgery or did not have a preoperative CT image available for analysis. The Canadian and UK cohorts had ethical approval from their respective institutional Research Ethics Boards.

8.2.2 Clinical data

For all study cohorts, demographic, surgical, and outcome data were prospectively collected over the perioperative period, and recorded in institutional surgical databases. Common baseline characteristics included: age, sex, weight, height, cancer site (colon vs. rectum, classified using the International Statistical Classification of Diseases and Related Health Problems *10th* Revision) and stage (American Joint Committee on Cancer (AJCC) or Union for International Cancer Control's (UICC)), American Society of Anesthesiologists (ASA) class, surgical approach (laparoscopic, open, or converted procedures). Definitions for short-term surgical outcomes were: 1) LOS: period between primary operation date and discharge from hospital (readmissions not included), 2) Major post-operative complications: any complication with Clavien-Dindo⁴⁶ Grade \geq 3, 3) Hospital readmission: hospital admission within 30 days of discharge for a primary CRC surgery, 4) Survival: mortality occurring within 30 days of surgery. *8.2.3 Computed tomography*

CT images acquired for cancer staging prior to surgery were used for quantitative assessment of SM and adipose tissue. The average time between CT image acquisition and surgery was 30 days. CT image analysis was done in Canada and UK using identical

methodology: SliceOmatic[©] (TomoVision, Magog, Quebec, Canada) with a single axial abdominal CT image landmarked at the 3rd lumbar vertebra (L3). For each cohort, a single trained observer, blinded to patient outcomes reviewed all images. Pre-determined Hounsfield Unit (HU) thresholds were: -29 HU to +150 HU for SM, -50 to -150 HU for VAT and -30 to -190 HU for subcutaneous adipose tissue (SAT). SM area (SMA), mean skeletal muscle radiodensity (SMR) and areas of VAT and SAT were reported. SMA, VAT, and SAT were normalized for height² and reported as: SM index (SMI, cm²/m²), VAT index (VATI, cm²/m²) and SAT index (SATI, cm²/m²). Inter-rater coefficients of variation were within the expected ranges: SMI (2%), SMR (4%), VATI (4-7%).^{47,48} Contrast agents can significantly alter SMR values^{49,50}, therefore only contrast enhanced images were used.

8.2.4 Statistics

Frequencies and summary statistics are reported. Variables were tested for normality using the Shapiro-Wilk test. Comparisons were assessed with parametric (Independent T-tests, ANOVA (Tukey post-hoc test), Chi-Square or Fischer's Exact Tests (post-hoc Bonferroni corrections)) or non-parametric (Mann-Whitney, Kruskal-Wallis Test) tests where appropriate.

Criteria for pooling cohorts was the absence of significant differences (using Z-tests) in age and sex-specific regression coefficients for SMI, SMR and VATI between linear regression models for individual cohorts.⁵¹ The pooled CRC cohort was stratified by sex and age (<50, 50-59, 60-69, 70-79, \geq 80 years) and Z-scores for each body composition feature were calculated using the pooled sex- and age- specific means and population standard deviations.

The primary outcome was postoperative LOS. LOS has a naturally skewed distribution and was treated as count data, hence we used a generalized linear model with a negative binomial distribution (NBM), appropriate for skewed distributions.⁵² Relationships between LOS and Z-scores for each body composition feature were evaluated. Threshold values were defined by examining the tissue-specific Z-score distributions, and determining the value that significantly associated with longer LOS using NBM. Z-score values significantly associated with longer LOS were defined for patients with low SMI (sarcopenia), low SMR (myosteatosis) and a high VATI (viscerally obese). Relationships between LOS and other relevant patient and surgical characteristics, including the geographic region (Canada, UK) were also evaluated. Results from NBMs are reported as incidence rate ratios (IRR with 95% CI), and as adjusted estimated marginal mean LOS. Logistic regression models evaluated the relationship of body composition features to major complications (none vs. ≥Grade 3) and hospital readmissions (none vs. readmitted) experienced with 30 days of discharge after primary surgery. Results are reported as odds ratios (OR with 95% CI).

For all regression analyses variables significant (P<0.1) at the univariable level were entered into multivariable analyses. The most parsimonious model was retained based on variables significant at the multivariable level and contributing to goodness of fit statistics (e.g. lowest Akaike Information Criteria, AIC). Results were considered significant at the P<0.05 level (two-tailed). Analysis was performed with IBM SPSS Statistics for Windows (v.23, Armonk, NY: IBM Corp).

8.3 Results

8.3.1 Patient characteristics

The pooled CRC cohort (Table 8-1) included 1,345 Canadian patients (Cohort 1 (N=384) & 2 (N=961)), and 755 patients from the UK. A majority had stage II+III cancers (76%), tumor in the colon (61%) and underwent laparoscopic surgeries (75%), with a low conversion rate of 4%.

Sex and age significantly associated with features of body composition. There was no significant difference in age and sex-specific regression coefficients for SMI, SMR and VATI in linear regression models between individual cohorts (Appendix VII Suppl Table 1), hence the cohorts were pooled. Pooled data was used for Z-score calculation, and is shown in Table 8-2. *8.3.2 Individual body composition features associate with LOS*

This analysis included N=1139 patients (Canadian Cohort 1 N=384; UK Cohort N=755) for whom LOS, postoperative complications, and readmission rates were available. Z-score distributions for SMI, SMR, VATI, and the univariable thresholds for z-scores that significantly associated with LOS, are presented (Figure 8-1).

Sarcopenia was defined as a z-score below -0.5 for SMI (Figure 8-1 A_a), myosteatosis was defined as a z-score below 0 for SMR (Figure 8-1 A_b) and VO was defined as a z-score above 0.5 for VATI (Figure 8-1 A_c).



Figure 8-1. Description of body composition features related to length of hospital stay

Panel 1A) Tissue-specific Z-Score distributions for: a) Skeletal Muscle Index (SMI), b) Skeletal Muscle Radiation Attenuation (SMR), c) Visceral Adipose Tissue Index (VATI). Z-scores were calculated using specific values for sex and age decade (<50, 50-59, 60-69, 70-79, ≥80) for each tissue feature, from the aggregated CRC data set. Zscore values that significantly associated with increased LOS are indicated by black arrows. The shaded areas in Panel A_{asc} denote the proportion of the distribution at risk for longer LOS. Tissue-specific threshold values, stratified by sex and age, are presented in the tables inset in Panel A_{a-c}. d) A proportional VENN diagram depicting overlap between different features of body composition (sarcopenia, myosteatosis, viscerally obese), representing 8 distinct body composition profiles. Panel 1B) L3 axial CT scans for two females of the same age. Patient 1 has none of the features sarcopenia, obesity, or myosteatosis. Patient 2 has all three features. Main panels: The L3 region includes: skeletal muscle, subcutaneous adipose tissue, visceral adipose tissue, and intramuscular adipose tissue (IMAT). Insets: A zoomed in view of the psoas and paraspinal muscles highlights fat infiltration to the muscle represented by different ranges of skeletal muscle attenuation and intramuscular adipose tissue area. Attenuation ranges were defined: normal attenuation (+30 to +150 HU), abnormal (low) attenuation (+1 to +29 HU and 0 to -29 HU). Patient 1, stage I colon cancer, has 65% of their psoas and paraspinal muscle area within the normal attenuation range, and minimal IMAT infiltration. Patient 2, stage IIA rectal cancer, has 49% of their skeletal muscle within the normal attenuation range, and extensive IMAT infiltration.

8.3.3 Multidimensional body composition profiles associate with LOS

Patients were classified as being affected by sarcopenia, myosteatosis, and/or VO; individual patients had none, 1, any 2, or all 3 features (Figure 8-1A_d). Figure 8-1B illustrates two female patients of the same sex and age, with none, and all 3 features, respectively.

In multivariable analysis (Table 8-3), surgical approach (P<0.001), major complications

(P<0.001), Canada vs UK (P<0.001), age (P<0.001) and 3 body composition profiles were

associated with LOS. These profiles were characterized by myosteatosis + sarcopenia (IRR 1.27

(95% CI 1.12-1.43), P<0.001), myosteatosis + VO (IRR 1.25 (1.10-1.42), P=0.001) and

myosteatosis + sarcopenia + VO (IRR 1.58 (1.29-1.93), P<0.001). Patients with myosteatosis

trended toward longer LOS (P=0.054). As expected, major postoperative complications predicted

longer LOS (IRR 2.42 (95% CI 2.18-2.68), P<0.001) however, the 3 body composition profiles

predicted longer LOS in patients with and without major complications (Table 8-4), or if they

were from Canada or the UK (Appendix VIII Suppl Table 2).

8.3.4 Post-surgical complications

The rate of 30-day hospital readmission was 9% (N=96; Canadian Cohort 1 N=42; UK Cohort N=54). Three body composition profiles predicted readmission in multivariable analysis (Table 8-5): VO only (P=0.018), myosteatosis + VO (P=0.005), and sarcopenia + myosteatosis + VO (P=0.038). Other variables predictive of readmission included open surgical approach (P=0.002), major complications (P<0.001), and stage III cancer (P=0.027). The rate of major complications was 15% (N=175; Canadian Cohort N=73; UK Cohort N=102). None of the body composition profiles predicted major complications (Appendix IV Suppl Table 3). Mortality within 30 days post-surgery occurred in too few patients (Canadian Cohort N=0, UK Cohort N=10 (0.9%) to make statistical inferences.

8.4 Discussion

Multiple prior studies in CRC have assessed single features (SM, SMR, and VAT) as stand-alone measures, or at most examined 2 features at the same time.^{7,9,12,14} While these studies identified a portion of patients at risk for poor outcomes, our analysis demonstrates without consideration of all three abnormalities, not all of the patients at risk are identified. We found body composition profiles characterized by abnormal skeletal muscle (myosteatosis + sarcopenia) negatively impacted on LOS, whereas body composition profiles characterized by VO predicted hospital readmission. Clearly the worst case body composition is the simultaneous presence of sarcopenia, myosteatosis and VO. A number of patients had sarcopenia, VO, or myosteatosis in isolation, which might reflect an early stage of the development toward a more deleterious body composition. The isolated presence of VO was significantly and independently associated with short-term hospital readmission and the isolated presence of myosteatosis showed a trend for increased LOS.

In prior work, we presented sex-specific threshold values for reduced SMI^{5,36} and SMR⁵, based on risk of mortality, not LOS. The current analysis was predicated on LOS as the primary outcome, for the reasons that mortality rates are low post-elective CRC surgery and thresholds

defined in relation to mortality may not be relevant for short-term postoperative outcomes.^{31,37} We recognize major complications to be a major driver of LOS and readmissions, and our analyses were adjusted accordingly. Importantly, body composition profiles at risk of longer LOS and readmission were independent of major complications, whether patients were from Canada or the UK, surgical approach, and age. As expected, major complications significantly extended LOS for all body composition profiles; however at risk body composition profiles extended LOS by an additional 4 (myosteatosis + sarcopenia or VO) to 9 days (myosteatosis + sarcopenia + VO).When in the longer term, cancer-related mortality data become available for these cohorts, it will be possible to evaluate in what manner mortality and pre-surgery body composition may be associated.

The radiological definition of body habitus is an emerging theme, with considerable recent proliferation in surgical oncology. However, this area is rife with differing approaches, and it is not possible to aggregate results or directly compare relevant outcomes.^{31,34} The need to standardize recording of surgical outcomes is an acknowledged issue⁵³, and the need to standardize body composition measures has become equally apparent^{31,34} in the recent CRC surgical literature relating postoperative complications to body composition.^{7-9,11-14,17-19,21,26-30,54} In our approach we sought to develop an understanding of body composition common to 2 regional patient populations. We selected 2 regions likely to be comparable with regard to body composition, and determined that regardless of patients being from Canada or UK they had the same postoperative risks conferred by body composition, even though the surgical practices differ between countries. Ethnicity has been described in the literature as factor related to various body composition features²⁵, however, data on ethnicity is not available at an individual level in our patient samples; we acknowledge that it should become part of future data collection efforts.

In CRC, CT body compositions analysis could be implemented as a pre-operative risk assessment to assist in treatment decision making. Body composition profiles identified here are pre-existing, we do not know their provenance or if they are reversible. Further clinical characterization of patient subsets showing abnormal body composition would be valuable to determine whether their surgical risk is associated with specific comorbidities⁵⁵, inflammation⁵⁶, or metabolic abnormalities.⁵⁷ Given that myosteatosis and VO are key risk factors, investigations for insulin resistance and metabolic syndrome might be of interest. Specific interventions remain to be investigated for their ability to support CRC patients with altered body compositions, and it is unknown whether and over what timescale these features could be modified. It is plausible that sarcopenia and myosteatosis have implications through reduced physical functioning and delayed return to activities of daily living.^{42,43} In the future it will be of interest to evaluate the physical functioning of patients with altered body composition profiles. Evidence-based multimodal prehabilitation programs have demonstrated some efficacy in contributing to improved cancer patient outcomes, including functional capacity and return to activities of daily living within ERAS programs.⁵⁸ It is unknown what effect such interventions would have on patients with altered body composition profiles, or would improve surgical outcomes.

This study contributes to advancing the understanding of CT data interpretation and suggests CT-defined features can help identify patients at risk for longer LOS and hospital readmission. Development of better risk stratification is essential to link patients to the appropriate therapeutic interventions.

8.5 References

- 1. Cakir H, Heus C, van der Ploeg TJ, Houdijk AP. Visceral obesity determined by CT scan and outcomes after colorectal surgery; a systematic review and meta-analysis. Int J Colorectal Dis 2015; 30:875-882.
- 2. Kazemi-Bajestani SM, Mazurak VC, Baracos V. Computed tomography-defined muscle and fat wasting are associated with cancer clinical outcomes. Semin Cell Dev Biol 2016; 54:2-10.
- 3. Malietzis G, Aziz O, Bagnall NM, Johns N, Fearon KC, Jenkins JT. The role of body composition evaluation by computerized tomography in determining colorectal cancer treatment outcomes: a systematic review. Eur J Surg Oncol 2015; 41:186-196.
- 4. Mei KL, Batsis JA, Mills JB, Holubar SD. Sarcopenia and sarcopenic obesity: do they predict inferior oncologic outcomes after gastrointestinal cancer surgery? Perioper Med (Lond) 2016; 5:30.
- 5. Martin L, Birdsell L, MacDonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol 2013; 31:1539-1547.
- 6. Yip C, Dinkel C, Mahajan A, Siddique M, Cook GJ, Goh V. Imaging body composition in cancer patients: visceral obesity, sarcopenia and sarcopenic obesity may impact on clinical outcome. Insights Imaging 2015; 6:489-497.
- 7. Boer BC, de Graaff F, Brusse-Keizer M, et al. Skeletal muscle mass and quality as risk factors for postoperative outcome after open colon resection for cancer. Int J Colorectal Dis 2016; 31:1117-1124.
- 8. Lodewick TM, van Nijnatten TJ, van Dam RM, et al. Are sarcopenia, obesity and sarcopenic obesity predictive of outcome in patients with colorectal liver metastases? HPB (Oxford) 2015; 17:438-446.
- 9. Malietzis G, Currie AC, Athanasiou T, et al. Influence of body composition profile on outcomes following colorectal cancer surgery. Br J Surg 2016; 103:572-580.
- 10. Moon HG, Ju YT, Jeong CY, et al. Visceral obesity may affect oncologic outcome in patients with colorectal cancer. Ann Surg Oncol 2008; 15:1918-1922.
- Ouchi A, Asano M, Aono K, Watanabe T, Oya S. Laparoscopic Colorectal Resection in Patients with Sarcopenia: A Retrospective Case-Control Study. J Laparoendosc Adv Surg Tech A 2016; 26:366-370.
- 12. Ozoya OO, Siegel EM, Srikumar T, Bloomer AM, DeRenzis A, Shibata DA-OhooX. Quantitative Assessment of Visceral Obesity and Postoperative Colon Cancer Outcomes. J Gastrointest Surg 2017; 21:534-542.

- 13. Pedziwiatr M, Pisarska M, Major P, et al. Laparoscopic colorectal cancer surgery combined with enhanced recovery after surgery protocol (ERAS) reduces the negative impact of sarcopenia on short-term outcomes. Eur J Surg Oncol 2016; 42:779-787.
- 14. Sabel MS, Terjimanian M, Conlon AS, et al. Analytic morphometric assessment of patients undergoing colectomy for colon cancer. J Surg Oncol 2013; 108:169-175.
- 15. van Vledder MG, Levolger S, Ayez N, Verhoef C, Tran TC, Ijzermans JN. Body composition and outcome in patients undergoing resection of colorectal liver metastases. Br J Surg 2012; 99:550-557.
- 16. Yu H, Joh YG, Son GM, Kim HS, Jo HJ, Kim HY. Distribution and Impact of the Visceral Fat Area in Patients With Colorectal Cancer. Ann Coloproctol 2016; 32:20-26.
- 17. Jones KI, Doleman B, Scott S, Lund JN, Williams JP. Simple psoas cross-sectional area measurement is a quick and easy method to assess sarcopenia and predicts major surgical complications. Colorectal Dis 2015; 17:O20-26.
- Margadant CC, Bruns ER, Sloothaak DA, et al. Lower muscle density is associated with major postoperative complications in older patients after surgery for colorectal cancer. Eur J Surg Oncol 2016; 42:1654-1659.
- 19. Cecchini S, Cavazzini E, Marchesi F, Sarli L, Roncoroni L. Computed tomography volumetric fat parameters versus body mass index for predicting short-term outcomes of colon surgery. World J Surg. 2011; 35:415-423.
- 20. Chen B, Zhang Y, Zhao S, et al. The impact of general/visceral obesity on completion of mesorectum and perioperative outcomes of laparoscopic TME for rectal cancer: A STARD-compliant article. Medicine 2016; 95:e4462.
- 21. Heus C, Cakir H, Lak A, Doodeman HJ, Houdijk AP. Visceral obesity, muscle mass and outcome in rectal cancer surgery after neo-adjuvant chemo-radiation. Int J Surg 2016; 29:159-164.
- 22. Huang DD, Wang SL, Zhuang CL, et al. Sarcopenia, as defined by low muscle mass, strength and physical performance, predicts complications after surgery for colorectal cancer. Colorectal Dis 2015; 17:O256-264.
- 23. Park BK, Park JW, Ryoo SB, Jeong SY, Park KJ, Park JG. Effect of Visceral Obesity on Surgical Outcomes of Patients Undergoing Laparoscopic Colorectal Surgery. World J Surg 2015; 39:2343-2353.
- 24. Scott SI, Farid S, Mann C, Jones R, Kang P, Evans J. Abdominal fat ratio a novel parameter for predicting conversion in laparoscopic colorectal surgery. Ann R Coll Surg Engl 2017; 99:46-50.

- 25. Cespedes Feliciano EM, Kroenke CH, Meyerhardt JA, et al. Association of Systemic Inflammation and Sarcopenia With Survival in Nonmetastatic Colorectal Cancer: Results From the C SCANS Study. JAMA Oncol 2017; 3:e172319.
- 26. Chen WZ, Chen XD, Ma LL, et al. Impact of Visceral Obesity and Sarcopenia on Short-Term Outcomes After Colorectal Cancer Surgery. Dig Dis Sci 2018; doi:10.1007/s10620-018-5019-2.
- 27. Hanaoka M, Yasuno M, Ishiguro M, et al. Morphologic change of the psoas muscle as a surrogate marker of sarcopenia and predictor of complications after colorectal cancer surgery. Int J Colorectal Dis 2017; 32:847-856.
- 28. Nakanishi R, Oki E, Sasaki S, et al. Sarcopenia is an independent predictor of complications after colorectal cancer surgery. Surg Today 2017; 48:151-157.
- 29. van der Kroft G, Bours D, Janssen-Heijnen DM, van Berlo D, Konsten D. Value of sarcopenia assessed by computed tomography for the prediction of postoperative morbidity following oncological colorectal resection: A comparison with the malnutrition screening tool. Clin Nutr 2018; 24:114-119.
- 30. Womer AL, Brady JT, Kalisz K, et al. Do psoas muscle area and volume correlate with postoperative complications in patients undergoing rectal cancer resection? Am J Surg 2018; 215:503-506.
- 31. Hopkins JJ, Skubleny D, Bigam DL, Baracos VE, Eurich DT, Sawyer MB. Barriers to the Interpretation of Body Composition in Colorectal Cancer: A Review of the Methodological Inconsistency and Complexity of the CT-Defined Body Habitus. Ann Surg Oncol 2018; 25:1381-1394.
- 32. Baracos VE. Psoas as a sentinel muscle for sarcopenia: a flawed premise. J Cachexia Sarcopenia Muscle 2017; 8:527-528.
- 33. Rutten IJG, Ubachs J, Kruitwagen R, Beets-Tan RGH, Olde Damink SWM, Van Gorp T. Psoas muscle area is not representative of total skeletal muscle area in the assessment of sarcopenia in ovarian cancer. J Cachexia Sarcopenia Muscle 2017; 8:630-638.
- Simonsen C, de Heer P, Bjerre ED, et al. Sarcopenia and Postoperative Complication Risk in Gastrointestinal Surgical Oncology: A Meta-analysis. Ann Surg, 2018; doi: 10.1097/SLA.0000000002679.
- 35. Doyle SL, Bennett AM, Donohoe CL, et al. Establishing computed tomography-defined visceral fat area thresholds for use in obesity-related cancer research. Nutr Res 2013; 33:171-179.

- 36. Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumors of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol 2008; 9:629-635.
- 37. Martin L. Diagnostic criteria for cancer cachexia: data versus dogma. Curr Opin Clin Nutr Metab Care 2016; 19:188-198.
- Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle 2017; 8:759-767.
- 39. Kuk JL, Saunders TJ, Davidson LE, Ross R. Age-related changes in total and regional fat distribution. Ageing Res Rev 2009; 8:339-348.
- 40. Rolland Y, Czerwinski S Fau Abellan Van Kan G, Abellan Van Kan G Fau Morley JE, et al. Sarcopenia: its assessment, etiology, pathogenesis, consequences and future perspectives. J Nutr Health Aging 2008; 12:433-450.
- Kuchnia AJ, Teigen LM, Cole AJ, et al. Phase Angle and Impedance Ratio: Reference Cut-Points From the United States National Health and Nutrition Examination Survey 1999-2004 From Bioimpedance Spectroscopy Data. J Parenter Enteral Nutr 2017; 41:1310-1315.
- 42. Kirk PS, Friedman JF, Cron DC, et al. One-year postoperative resource utilization in sarcopenic patients. J Surg Res 2015; 199:51-55.
- 43. Lieffers JR, Bathe OF, Fassbender K, Winget M, Baracos VE. Sarcopenia is associated with postoperative infection and delayed recovery from colorectal cancer resection surgery. Br J Cancer 2012; 107:931-936.
- 44. Du Y, Karvellas CJ, Baracos V, et al. Sarcopenia is a predictor of outcomes in very elderly patients undergoing emergency surgery. Surgery. 2014; 156:521-527.
- 45. Nelson G, Kiyang LN, Crumley ET, et al. Implementation of Enhanced Recovery After Surgery (ERAS) Across a Provincial Healthcare System: The ERAS Alberta Colorectal Surgery Experience. World J Surg. 2016; 40:1092-1103.
- 46. Dindo D, Demartines N, Clavien P-A. Classification of Surgical Complications. Ann Surg. 2004; 240:205-213.
- 47. Mourtzakis M, Prado CM, Lieffers JR, Reiman T, McCargar LJ, Baracos VE. A practical and precise approach to quantification of body composition in cancer patients using computed tomography images acquired during routine care. Appl Physiol Nutr Metab 2008; 33:997-1006.

- 48. Shuster A, Patlas M, Pinthus JH, Mourtzakis M. The clinical importance of visceral adiposity: a critical review of methods for visceral adipose tissue analysis. Br J Radiol. 2012; 85:1-10.
- 49. van der Werf A, Dekker IM, Meijerink MR, Wierdsma NJ, de van der Schueren MAE, Langius JAE. Skeletal muscle analyses: agreement between non-contrast and contrast CT scan measurements of skeletal muscle area and mean muscle attenuation. Clin Physiol Funct Imaging. 2017;38:366-372.
- 50. van Vugt JLA, Coebergh van den Braak RRJ, Schippers HJW, et al. Contrastenhancement influences skeletal muscle density, but not skeletal muscle mass, measurements on computed tomography. Clin Nutr. 2017; S0261-5614:30246-7.
- 51. Weaver BW, K.L. SPSS and SAS programs for comparing Pearson correlations and OLS regression coefficients. Behav Res Methods. 2013; 45:880-895.
- 52. Austin P. A Comparison of Statistical Modeling Strategies for Analyzing Length of Stay after CABG Surgery. Health Serv Outcomes Res Methodol. 2003; 3:107-133.
- 53. Martin RCG, Brennan MF, Jaques DP. Quality of Complication Reporting in the Surgical Literature. Ann Surg. 2002; 235:803-813.
- 54. Huang DD, Wang SL, Zhuang CL, et al. Sarcopenia, as defined by low muscle mass, strength and physical performance, predicts complications after surgery for colorectal cancer. Colorectal Dis. 2015; 17:O256-O264.
- 55. Xiao J, Caan BJ, Weltzien E, et al. Associations of pre-existing co-morbidities with skeletal muscle mass and radiodensity in patients with non-metastatic colorectal cancer. J Cachexia Sarcopenia Muscle 2018; doi: 10.1002/jcsm.12301.
- 56. Malietzis G, Johns N, Al-Hassi HO, et al. Low Muscularity and Myosteatosis Is Related to the Host Systemic Inflammatory Response in Patients Undergoing Surgery for Colorectal Cancer. Ann Surg. 2016; 263:320-325.
- 57. Lohsiriwat V, Pongsanguansuk W, Lertakyamanee N, Lohsiriwat D. Impact of metabolic syndrome on the short-term outcomes of colorectal cancer surgery. Dis Colon Rectum. 2010; 53:186-191.
- 58. Carli F, Gillis C, Scheede-Bergdahl C. Promoting a culture of prehabilitation for the surgical cancer patient. Acta Oncol. 2017; 56:128-133.

Variables	Pooled CRC Cohort
Sample Size (N)	2100
Data Collection Time Frame	2006-2015
Study Type	Retrospective, cohort
Sex (N, %)	
Female	830 (39)
Male	1270 (61)
Age (years; mean ±SD)	66.6 ± 11.9
BMI* (kg/m ² ; mean ±SD)	27.7 ±5.6
Tumor Site (N, %)	
Colon	1279 (61)
Rectum	821 (39)
Surgical Approach (N, %)	
Laparoscopic Surgery	1580 (75)
Open Surgery	413 (20)
Converted	83 (4)
Unknown	24 (1)
Cancer Stage [§] (N, %)	
Ι	385 (18)
II	713 (34)
III	887 (42)
IV	109 (5)
CT Analysis Software	SliceOmatic
HU range for muscle	-29 to 150
HU range for visceral adipose tissue	-150 to -50
HU range for subcutaneous adipose tissue	-190 to -30
Number of CT image observers	3
Outcomes Investigated	
Primary*	LOS
Secondary†	major complications (Grade ≥ 3)
	hospital readmission
	30-day mortality
LOS for Primary Admission	
Mean ±SD	9.42 ± 10.79
Median (IQR)	6.00 (6.00)
30-day major complications (Grade ≥3) (N, %)	175 (15)
30-day hospital readmission (N, %)	96 (9)
30-day Mortality	10 (1)

Table 8-1. Characteristics for the pooled colorectal surgical co	hort
--	------

BMI, body mass index; CRC, colorectal cancer; HU, Hounsfield Unit; IQR, interquartile range; L3, third lumbar vertebrae; LOS, length of hospital stay; N, number; SD, standard deviation *† available for N=1136 patients from Canadian Cohort 1 (N=384) and UK Cohort (N=755) | Complications graded according to the Clavien-Dindo classification

Body			A	ge Categori	es		D Voluos*	Total
Composition	Sex				T	1	I - v alues	TUTAL
Feature		<50	50-59	60-69	70-79	≥80		
Skeletal	Female	113.8	105.9	104.2	97.2	90.6	P<0.001	
Muscle Area	(mean, SD)	(21.4)	(17.1)	(17.7)	(15.2)	(15.2)		
(SMA, cm ²)	N	75	138	238	245	134		830
	Male	169.7	165.3	153.9	141.5	125.8	P<0.001	150.0
	(mean, SD)	(32.1)**	(29.9)**	$(24.7)^{**}$	(23.0)**	(20.5)**	1 \0.001	(28.5)**
	N	97	219	422	378	154		1270
	E L	12.0	41.0	10.5	20.6	26.0	D =0.001	20.7
	Female	43.0	41.0	40.5	38.0	36.9	P<0.001	$\frac{39.7}{(6.8)}$
Skeletal	(mean, SD)	(0.1)	(0.7)	(0.7)	(0.4)	(3.9)		(0.8)
Muscle Index	1	15	138	238	243	134		830
$(SMI, cm^2/m^2)$	Male	55.1	53.8	51.3	47.9	43.2	P<0.001	50.0
	(mean, SD)	(9.9)**	(9.2)**	(8.3)**	(8.1)**	(7.4)**		(9.1)**
	Ν	97	219	422	378	154		1270
Skalatal	Female	39.6	36.0	31.4	27.8	24.8	P<0.001	30.8
Muscle	(mean, SD)	(9.3)	(8.8)	(8.5)	(8.5)	(7.5)		(9.6)
Radiation	Ν	75	138	238	245	134		830
Attenuation	Male	42.0	37.6	32.9	29.7	28.1	P<0.001	32.9
(SMR, HU)	(mean, SD)	(7.8)	(8.8)	(8.6)*	(8.3)*	(8.7)**		(9.4)**
	N	97	219	422	378	154		1270
	Female	78.0	98.8	119.1	119.4	98.0	P<0.001	108.7
	(mean, SD)	(69.8)	(72.3)	(79.1)	(77.2)	(71.0)		(76.4)
Visceral	N	75	136	235	240	131		817
Adipose Tissue	Male	140.3	189.8	216.9	214.4	184.0	P<0.001	201.6
Area (VAT,	(mean, SD)	(93.1)**	(106.5)*	(114.9)**	(113.8)*	(95.7)**	1 0.001	$(111.4)^*$
cm ²)	())	()	*		*			*
	Ν	97	218	419	374	153		1261
	Female	29.6	38.3	46.5	47.1	40.0	P<0.001	42.7
V [*]	(mean, SD)	(26.7)	(28.3)	(30.5)	(30.4)	(29.7)		(30.1)
Visceral Adipose Tissue	N	75	136	235	240	131		817
Index (VATL	Male	45.7	61.6	71.9	72.5	63.3	P<0.001	67.3
cm^2/m^2)	(mean, SD)	(29.5)**	(34.0)**	(37.5)**	(38.5)**	(33.1)**	1 0.001	(36.9)**
,	N	97	218	419	374	153		1261
	Female	242.5	231.9	255.1	229.7	186.5	P<0.001	231.6
~ •	(mean, SD)	(139.3)	(118.6)	(130.5)	(111.0)	(93.1)	1 0.001	(120.2)
Subcutaneous	N	74	133	235	238	131		811
Aupose Hissue	Male	102.8	185.0	188.2	180.2	146.4	P <0.001	180.6
cm^2)	(mean SD)	$(107.8)^{*}$	(85 4)**	(102 5)**	(79.2)**	(58 1)**	1 <0.001	(89.8)**
	N	97	218	420	375	153		1263
	Famala	02.1	20.0	00.7	00.5	20.5	D<0.001	01.0
Subcutaneous	(mean SD)	(53.6)	(46.0)	(51.7)	(43.3)	(47.2)	1 ~0.001	(47.2)
Adipose Tissue	N	74	133	235	238	131		811
Index (SATI,	- ` M.1.	(2.4	(0.0	() [(1.0	50.1	D <0.001	(0.2
cm²/m²)	(mean SD)	(34.8)*	00.2 (27 0)**	02.3 (33 3)**	01.0	50.1 (10 /**	P<0.001	00.2 (29.5)**
	(incan, SD)	1 (37.0)	(41.7)	(33.3)	(20.7)		1	(47.5)

Table 8-2. Pre-surgical L3 CT-defined body composition features by sex and age for the pooled colorectal cancer (CRC) cohort (N=2100).

	Ν	97	218	420	375	153		1263
	Female (mean_SD)	318.1 (192.7)	329.1 (175.2)	374.2	347.6 (168.5)	284.5 (144.9)	P<0.001	339.4 (176.6)
Total Adipose Tissue Area (TAT, cm²)	N	74	133	235	236	131		809
	Male (mean, SD)	333.1 (185.6)	374.8 (174.5)*	403.0 (180.4)	393.5 (164.1)* *	330.4 (132.8)*	P<0.001	381.1 (171.7)
	Ν	97	218	419	374	153		1261**
Total Adinasa	Female (mean, SD)	120.8 (74.1)	127.4 (68.5)	146.1 (74.2)	137.4 (65.3)	116.7 (61.2)	P<0.001	133.4 (69.4)
i otal Adipose	Ν	74	133	235	236	131		809
Tissue Index	- '	, .	155	235	230	101		00)
Tissue Index (TATI, cm ² /m ²)	Male (mean, SD)	107.7 (59.4)	121.8 (55.9)	133.8 (58.7)*	133.1 (55.8)	113.4 (45.0)	P<0.001	127.0 (56.6)

CT, computed tomography; HU, Hounsfield Unit; L3, third lumbar vertebrae; N, number; SD, standard deviation P<0.05, P<0.001 (Independent T-test) for the comparison between males and females within each age category and overall

[†]One-way ANOVA for the comparison between age categories within each sex

Table 8-3. Regression models for variables associated with extended length of hospital stay (LOS) and 30-hospital readmission in patients with colorectal cancer undergoing elective surgery

	Negative Binomial Regression Model†					Logistic Regression Model‡						
		Outcome -	= Length	of hospi	ital stay (days)		Outcome = 30-day hospital readmission (yes vs. no)					no)
	Univariable			Multivariable			Univariable			Multivariable		
Variables	Ν	IRR	P-	Ν	IRR	P-	Ν	OR	P-	Ν	OR	P-Value
		(95% CI)	Value		(95% CI)	Value		(95% CI)	Value		(95% CI)	
Pre-surgical Body Composition Profiles												
no features												
No Sarcopenia, No myosteatosis, No Visceral Obesity	320	1.0 (ref)		319	1.0 (ref)		320	1.0 (ref)		316	1.0 (ref)	
1 feature												
Sarcopenic	140	1.05 (0.92-1.20)	0.472	139	1.13 (0.99-1.29)	0.070	140	0.71 (0.28-1.82)	0.474	139	0.87 (0.33-2.30)	0.786
Visceral Obesity	90	1.17 (1.00-1.35)	0.045	90	1.01 (0.86-1.18)	0.909	90	2.44 (1.14-5.24)	0.022	89	2.66 (1.18-6.00)	0.018
Myosteatosis	181	1.46 (1.23-1.72)	<0.00 1	181	1.13 (1.00-1.27)	0.054	181	1.43 (0.71-2.89)	0.317	180	1.44 (0.69-3.03)	0.333
2 features												
Sarcopenia + Visceral Obesity	5	1.37 (0.89-2.12)	0.152	5	1.15 (0.63-2.10)	0.657	-	-	-	-	-	-
Sarcopenia + Myosteatosis	187	1.53 (1.32-1.78)	<0.00 1	186	1.27 (1.12-1.43)	<0.001	186	1.09 (0.52-2.30)	0.816	186	1.18 (0.54-2.57)	0.677
Myosteatosis + Visceral Obesity	148	1.29 (1.09-1.52)	0.003	148	1.25 (1.10-1.42)	0.001	147	2.79 (1.46-5.33)	0.002	146	2.72 (1.36-5.46)	0.005
3 features												
Sarcopenia + Myosteatosis + Visceral Obesity	42	1.73 (1.34-2.24)	<0.00 1	42	1.58 (1.29-1.93)	<0.001	42	2.64 (0.99-7.04)	0.052	42	2.98 (1.06-5.46)	0.038
Cancer Site												

Colon	728	1.0 (ref)		-			727	1.0 (ref)				
Rectum	408	1.11 (1.01-1.22)	0.026	-	-	NS	407	1.50 (0.99-2.29)	0.059	-	-	NS
Surgical Approach												
Laparoscopic	648	1.0		640	1.0 (ref)		648	1.0 (ref)		635	1.0 (ref)	
Open	411	(ref) 1.80 (1.64, 1.06)	<0.00	397	1.85	<0.001	409	2.30	<0.001	390	2.08	0.003
Converted	77	(1.04-1.90) 1.74 (1.47-2.06)	<0.00 1	73	(1.70-2.01) 1.53 (1.31-1.79)	<0.001	77	(1.47-3.39) 2.19 (1.01-4.72	0.047	73	(1.29-3.54) 1.56 (0.67-3.68)	0.304
ASA Class (N, %)												
ASA 1	117	1.0 (ref)		-			117	1.0 (ref)				
ASA 2	794	1.32	<0.00	-	-	NS	794	1.54	0.327	-	-	NS
ASA 3+4	221	(1.14-1.55) 1.30 (1.09-1.55)	1 0.004	-	-	NS	219	(0.65-3.65) 2.82 (1 14-7 01)	0.025	-	-	NS
Major		(110) 1100)										
Complication within 30 days												
no	960	1.0 (ref)		941	1.0 (ref)		960	1.0 (ref)		934	1.0 (ref)	
yes	173	2.20 (1.96-2.46)	<0.00 1	169	2.42 (2.18-2.68)	<0.001	172	4.65 (2.97-7.26)	<0.001	164	4.67 (2.89-7.56)	<0.001
Cancer Stage*												
Ι	281	1.0 (ref)		-			280	1.0 (ref)		275	1.0 (ref)	
II	342	1.02 (0.91-1.16)	0.702	-	-	NS	341	1.05 (0.56-1.99)	0.875	330	1.15 (0.58-2.28)	0.69
III	398	1.05 (0.93-1.84)	0.415	-	-	NS	398	1.86 (1.05-3.28)	0.033	389	2.10 (1.14-3.87)	0.018
IV	109	1.00 (0.84-1.19)	1.00	-	-	NS	109	1.15 (0.49-2.74)	0.747	104	0.87 (0.33-2.30)	0.886
Geographic Region												
Canada	384	1.0 (ref)		381	1.0 (ref)		382	1.0 (ref)				
UK	752	1.26	<0.00	729	1.33	<0.001	752	0.63	0.030	-	-	NS

		(1.15-1.39)	1		(1.22-1.46)			(0.41-0.96)				
Sex												
Female	450						449	1.0 (ref)				
Male	686	1.20	<0.00	-	-	NS	685	1.22	0.382	-	-	NS
		(1.09-1.31)	1					(0.79 - 1.88)				
Age (years)	1136	1.01	0.001	1110	1.01	<0.001	1134	1.01	0.260	-	-	NS
		(1.01 - 1.01)			(1.01 - 1.01)			(0.99-1.03)				
Diabetes												
no	1031	1.0 (ref)		-			1029	1.0 (ref)				
yes	105	0.92	0.310	-	-	NS	105	0.88	0.744	-	-	NS
		(0.79 - 1.08)						(0.42 - 1.87)				

ASA, American Society of Anesthesiologists; CI, confidence interval; IRR, incident rate ratio; LOS, length of stay; N, number; OR, odds ratio Canadian cohort used American Joint Committee on Cancer (AJCC) version 6, UK cohort used Union for International Cancer Control's (UICC) version 5 †Model = Generalized linear model with negative binomial distribution, dependent variable = LOS; Best fit final model (Full Model (all variables significant at the univariable level) AIC = 6588.963, Fitted Model (removed non-significant variables from multivariable model) AIC = 6584.190). Variables not significant at univariable level = Cancer Stage, Diabetes; Variables non-significant at multivariable level = ASA Class (P=0.881), Cancer Site (P=0.681), Sex (P=0.110) ‡Model = Binary logistic regression model, dependent variable = 30-day hospital readmission (no. vs. yes; N=92 were readmitted); Best fit final model (Full Model (all variables significant at the univariable level) AIC = 391.737, Fitted Model (removed non-significant variables from multivariable model) AIC = 390.808). Variables not significant at univariable level = sex, age, diabetes; Variables non-significant at multivariable level = study cohort (P=0.655), ASA Class (P=0.161) Table 8-4. Estimated marginal mean LOS derived from the final adjusted NBM for each body composition profile demonstrating the independent effect of the three body composition profiles on LOS with or without major post-operative complications

Body composition profiles based on the absence or presence of sarcopenia, myosteatosis, or visceral obesity	Major Complication within 30 days post-surgery	Estimated Marginal Mean LOS (95% CI)	Mean Difference (days)	P-Value
no features				
No Sarcopenia,	yes	$16.44 (14.62-18.48)^{\circ}$	9.64	< 0.001
No Myosteatosis, No Visceral Obesity	no	6.80 (6.22-7.44) ^a		
1 feature				
Sarcopenic	yes	18.58 (16.05-21.52)	10.90	< 0.001
	no	7.69 (6.81-8.68)		
Visceral Obesity	yes	16.59 (14.03-19.62)	9.73	< 0.001
	no	6.86 (5.94-7.94)		
Myosteatosis	yes	18.53 (16.28-21.09)	10.86	< 0.001
	no	7.66 (6.86-8.56)		
2 features				
Sarcopenia +	yes	18.85 (10.24-34.70)	11.05	0.002
Visceral Obesity	no	7.80 (4.27-14.23)		
Sarcopenia +	yes	20.86 (18.21-23.89) ^d	12.23	<0.001
Myosteatosis*	no	8.63 (7.74-9.61) ^b		
Myosteatosis +	yes	20.57 (18.03-23.47) ^d	12.06	<0.001
Visceral Obesity*	no	8.51 (7.63-9.49) ^b		
3 features				
Sarcopenia+	yes	25.90 (21.12-31.78) ^d	15.19	<0.001
Myosteatosis + Visceral Obesity*	no	10.72 (8.86-12.97) ^b		

*profiles identified from negative binomial regression that are significantly (P<0.05) associated with extended LOS (compared to no features) independent of major complications, study cohort, surgical approach, and age ^{a,b} Significantly different from no features (No sarcopenia, No myosteatosis, No Visceral Obesity) with no major

complications

^{c,d} Significantly different from no features (No sarcopenia, No myosteatosis, No Visceral Obesity) with major complications

CHAPTER 9: Discussion and future directions

9.1 The status of diagnostic criteria

Cachexia has devastating effects on patients including weakness, fatigue, loss of independence, poor treatment tolerance, and death. Over the course of history, up until the present day, cachexia has been described by several signs and symptoms including involuntary weight loss (WL), wasting of skeletal muscle (SM) and adipose tissues, anorexia, and gastrointestinal symptoms.¹ However, the specific etiology has been difficult to define due to its complex and multifactorial origins related to the underlying disease and its treatment. Owing to the prominence of reduced food intake (FI) as well as the presence of inflammation; cachexia is also classified as a form of disease-associated malnutrition.²⁻⁴ There is no widely agreed upon definition or diagnostic criteria for cachexia or disease-associated malnutrition, which is consequently reflected within the International Statistical Classification of Diseases and Related Health Problems 10th revision (ICD-10), which sets the international standard for reporting disease and health conditions.⁵ The ICD-10 includes a core description of cachexia as a condition of involuntary WL >10% associated with chronic illness, characterized by atrophy of SM, due to inadequate dietary intake, malabsorption, or hypermetabolism, but lacks diagnostic criteria.^{5,6}

International research consortia including the Global Leadership on Malnutrition (GLIM) and the Society on Sarcopenia, Cachexia and Wasting Disorders,⁷⁻⁹ have focused research efforts on developing international consensus for a definition and diagnostic criteria for disease-associated malnutrition, cachexia, and cancer cachexia.⁷⁻⁹ These efforts are complimentary and have produced consensus definitions with overlapping core concepts, which are common to definitions outlined in the ICD-10, but that also include a minimum set of diagnostic criteria which require development and validation (Table 9-1).⁷⁻⁹

The overarching theme of this thesis has been to advance the development of diagnostic criteria for the diagnosis of cancer cachexia based on the International Consensus Framework for the Definition and Classification of Cancer Cachexia.⁹ Key requirements for diagnostic criteria include clinical availability, cost and time effectiveness, and clinical utility.^{3,8-10} The research presented throughout this thesis includes several advances related to the development of diagnostic criteria for cancer-associated WL, computed-tomography (CT)-defined SM and adipose tissue, FI, and inflammation.

9.1.1 Diagnostic criteria for cancer-associated weight loss

In the ICD-10, cachexia is diagnosed based on WL >10%.⁵ The National Cancer Institute Common Terminology Criteria for Adverse Events (NCI CTCAE) has a grading scheme (Grade 1-5) intended to capture the severity of an adverse event, along with recommendations for appropriate interventions.¹¹WL is included as an adverse event (Grade 1: 5-10%; Grade 2: 10-20%; Grade 3: >20%), but grades 4 and 5 WL are not defined, the etiology of WL is unaccounted for, and the impact of WL based on initial body weight is not considered. In Chapter 2, I presented and validated an updated classification for cancer-associated WL by combining %WL and BMI (Grades 0 to 4), which stratified patients according to overall survival (OS). An important finding from this analysis is that even small amounts of weight loss (i.e. 2.5-5.0%) associated with reduced OS, which may go undetected when using the classifications suggested in the ICD-10 and NCI CTCAE. The updated WL grades have been validated in three additional cohorts representing different care settings (palliative care, medical oncology) and cancer types (e.g. head and neck and lung cancers).¹²⁻¹⁴ Vagnildhaug et al.¹⁴, further demonstrated that higher WL grades (i.e. more severe) identified patients likely to progress to worsening stages of cancer cachexia; with worsening of physical function, food intake, and

appetite. The WL grades have also been included in the most recent European Society for Clinical Nutrition and Metabolism (ESPEN) guidelines on nutrition in cancer patients.¹⁵ Grading the severity of WL with consideration for initial body weight is an improvement over the bivariate classification of >10% used in the ICD-10, and contributes to refining the severity grades used in the NCI CTCAE. But questions remain, the WL grades were based on mortality as an outcome and will need to be validated against other outcomes that are important to patients including activities of daily living, quality of life, and tolerance to cancer treatment. Furthermore, we will have to consider how WL grades can be adopted into clinical practice. *9.1.2 Diagnostic criteria for CT-defined skeletal muscle*

Provisional diagnostic criteria for cancer cachexia include the identification of SM depletion, based on thresholds defined for several different assessment methods including cross-sectional imaging (CT or magnetic resonance imaging (MRI)), dual energy x-ray imaging (DEXA), anthropometry (mid upper-arm muscle area), and bioelectrical impedance analysis.⁹ There was no consensus for any one assessment method, but preference was given to SM quantified using computed tomography (CT). Sex-specific thresholds related to mortality were presented for CT-defined SM that were originally developed in obese patients with advanced cancer.¹⁶. The use of CT to quantify SM and adipose tissue is a new and developing area of assessment. Since these provisional criteria were presented our knowledge and research efforts have greatly expanded. There has been a proliferation of different methods (e.g. tissue-specific thresholds, software packages, tissue attenuation ranges, and CT image landmarks) used in the quantification of SM as well as adipose tissues using CT, which has contributed significant heterogeneity in the literature.^{17,18} Additional CT metrics have appeared including SM radiation attenuation, adipose tissue (visceral and subcutaneous) areas and attenuation, which appear to

have different associations to different clinical outcomes.¹⁹⁻²⁶ We are beginning to appreciate how patient characteristics such as age, sex, and total adiposity impact on features of body composition and need to be factored into our analyses.²⁵⁻²⁷ This new knowledge has presented a great challenge in attempting to identify data sets for aggregation in order to explore some of these concepts. In chapter 8, I present an advanced analysis using body composition metrics derived from CT image analysis in preoperative colorectal cancer patients. I created a large aggregated data set of pre-operative CT images to facilitate the comparison of body composition features by sex and age within a similar patient population. I defined thresholds for SM, SM attenuation, and visceral obesity that related to surgical outcomes. Lastly, I was able to explore the impact of a multidimensional body composition profile and demonstrate an increased risk of longer length of hospital stay (LOS) and 30-day hospital readmission. This analysis advanced our understanding of how age and sex affect features of body composition and that multidimensional assessments of body composition beyond SM mass are necessary to capture patients with the greatest risk of poor surgical outcomes. Adoption of CT-defined body composition into routine clinical practice remains to be determined. There are efforts underway in Edmonton, Alberta and Queensland, Australia to understand if using CT as an assessment of body composition is feasible in clinical practice. Once demonstrated to be feasible, it would then be possible to determine how CT-body composition could augment nutrition clinical practice in areas such as the identification of patients with altered body composition, tailoring of nutrition support (e.g. calculation of protein needs based on lean body mass (LBM)), and improved monitoring of effectiveness of nutrition interventions (e.g. gains in LBM).²⁸

9.1.3 Diagnostic criteria for food intake

There are many assessments of FI to explore as diagnostic criteria for the etiology of involuntary WL.⁹ Reduced FI is a key feature of cancer cachexia for which there are multiple possible etiologies, and identifying reductions in FI is recommended as an essential first step to guide nutrition care.¹⁵ Information about FI is collected on the majority of nutrition screening tools used in the oncology setting. However there are differences in measurement scales and time frames for how reductions in FI are captured. In Chapter 5, I demonstrated that despite differences in measurement scales there are points of alignment between tools that allows them to be combined into a single diagnostic criterion for FI. I further demonstrated in multivariable analysis adjusted for cancer type, cancer stage, functional status, age, and sex that reductions in food intake had the strongest overall association to WL, suggesting that reduction in FI play an important role in the etiology of WL. Few studies have provided a quantitative value for energy and protein intake corresponding to patient reported categories of food intake, however "severely reduced" FI appear to be associated with energy intakes <500 kcal per day which is grossly insufficient to support the maintenance of body weight.²⁹ This supports the conceptual validity of these patient-reported categories to identify patients with low intakes.

9.1.4 Diagnostic criteria for systemic inflammation

Alterations in metabolism are the most difficult criteria to define. They are driven by an individual patient's response to the presence of a tumor and cancer treatment, which induce inflammation, excess catabolism, and hypermetabolism to a variable extent. There is tremendous heterogeneity in tumor characteristics (e.g. grade, specific mutations, metabolic rate) and in how a patient will respond to both the tumor and its treatment, which makes the characterization of individual metabolic alterations incredibly difficult.³⁰ There is potentially a long list of tumor-derived factors and mechanisms contributing to SM and adipose tissue wasting.^{31,32} These factors

can be proteolytic (e.g. IL-6, growth/differentiation factor 15 (GDF15) and/or lipolytic (e.g. leukemia inhibitory factor (LIF), parathyroid hormone-related protein (PTHrP)) in their actions and can be produced directly by the tumor, or may arise from crosstalk between the tumor and host immune system.^{31,32} However, clinical evidence is limited; clinical measures for many of these factors are not available, the prevalence of these factors and the magnitude of their contribution to weight loss in humans is not well characterized. If we move out from the molecular etiologies and examine the overall metabolic processes such as the rates of protein synthesis, protein degradation, and energy expenditure there are clinical studies in patients with cancer cachexia suggesting increased rate of whole-body protein turnover and increased resting energy expenditure.³¹ However, these types of measurements are limited to the research setting as they are invasive, time consuming, and expensive and unlikely to be adapted into clinical practice. The evidence related to the identification of metabolic alterations is in a rudimentary state of development, and their measurement for the time being is limited to the research setting. Treatment response is a clinically relevant indicator of altered metabolism, whereby the presence of tumor is driving the underlying metabolic changes.^{9,33} However, it has not been collected as part of cancer cachexia research studies, but moving forward would seem an essential piece of information to understand responsiveness to nutrition interventions.

The most widely accepted clinical proxy for altered metabolism is C-reactive protein (CRP).⁹ CRP was not frequently part of the data collected in the aggregated clinical cachexia studies (e.g. <40% of patients in the aggregated data set had a measure of CRP). In Chapter 6, I demonstrated that CRP was significantly independently associated with WL in multivariable analyses when reductions in FI were also accounted for. These results indicate that CRP is, in part, contributing to the development of cancer-associated WL. However, it was clear that

neither CRP nor reductions in FI explained all of the WL in cancer patients. This was not surprising given the other factors likely to contribute to WL. As an example PTHrP, a tumorderived factor, was demonstrated to independently increase the risk of WL in a multivariable model adjusted for CRP, PS, cancer stage, albumin, and serum calcium. Although this study did not account for reductions in FI, it is preliminary evidence that other, and possibly tumorderived, factors are contributing to WL.^{34,35} At present what these other factors might be is largely unknown, and substantial advancement with regard to their identification and characterization will be required before they can be considered as diagnostic criteria. In the meantime, the addition of CRP to assessments of FI may help the clinician differentiate patients with different etiologies for their WL (e.g. predominantly reduced FI, or combination inflammation and reduced FI), which might also provide an initial indication about their ability to respond to nutrition interventions.^{8,36,37}

9.2 Strengths and limitations of a data driven approach

Health-related data is highly valued, as it can be used to provide information about disease characteristics, prevalence, risk factors, clinical practice, and outcomes of treatment.³⁸ However, availability of data sets large enough to fully examine these features is limited, and therefore aggregation of health-related data from multiple studies is necessary. Data aggregation for secondary analysis enhances our ability to compare patient populations and to answer research questions quickly and inexpensively.³⁹ Additionally, findings generated from secondary analyses may be more generalizable than findings from a single center.

I adopted a data-driven approach, a priori, to advance the development of diagnostic criteria for cancer cachexia. However, there were no large aggregated data sets specific to cancer cachexia available for analysis, and therefore a main objective of my thesis was to create large

aggregated data sets from clinical research studies conducted in cancer cachexia. In the process of aggregating data, it was evident that data related to candidate diagnostic criteria were not collected systematically: a heterogeneous mix of measures and assessments were used and criteria were collected with varying frequency across data sets (e.g. most frequent = WL; least frequent = metabolic alterations). The type of information gathered within these datasets highlights that our understanding pertaining to the different concepts of cancer cachexia are in various stages of progress. The balance of data included criteria related to weight loss and qualitative assessments of food intake, and few data sets included criteria related to skeletal muscle depletion, and metabolic alterations. The heterogeneity and lack of data pertaining to some of these concepts is particularly challenging. Data related to metabolic alterations were limited to a single clinical measure of inflammation, which was influenced by the prevailing ordering practices of a given country. For example, CRP (an acute phase protein produced in the liver) was common among European data sets, but not typical of Canadian data sets, which were more likely to have included neutrophils and lymphocytes (immune cells made in the bone marrow). Both of these measures are demonstrated to be strong prognostic indicators for overall survival^{40,41}, but they represent different aspects of the inflammatory process, and it is not clear if they can be compared as etiological criteria for WL. The relationship between neutrophils, lymphocytes and WL needs to be evaluated to determine if these measures could be used interchangeably as etiological diagnostic for WL.

A way forward for cachexia researchers would be to generate an agreed upon minimum data set that would facilitate data sharing and aggregation, similar to the approach adopted by the GLIM group.⁸ There are some excellent examples of systematic data collection using minimum data sets that could guide and inform this process. nutritionDay worldwide is a large scale,

international data collection effort with the goal to reduce disease-associated malnutrition among hospitalized individuals and nursing home residents.⁴² Nutrition Day is a 1-day data collection initiative involving 7,000 institutions, the same data is collected across participating sites using a standardized online data capture tool, and includes additional data capture elements for hospitalized oncology patients. Enhanced Recovery After Surgery (ERAS) protocols are also excellent examples of international systematic data collection initiatives in the surgical setting.⁴³ In addition to standardized data collection the ERAS Society recently developed a framework to facilitate standardization for reporting ERAS related studies.⁴⁴ Although the type of nutrition data collected within these different initiatives may not capture data relevant to the diagnosis of cancer cachexia, they provide a model and platform of how to accomplish standardized data collection. Additionally, there are numerous examples of how these initiatives used standardized data to demonstrate improved clinical outcomes, influence health systems, and change clinical practice.⁴⁵⁻⁵¹

There are however known limitations with aggregated data that include the lack of control over the types of patients originally included, which variables were collected, how variables were measured and recorded, completeness of the data collected, and limited longitudinal follow-up.³⁹ Other limitations can arise when changes to the standardized data fields are made, and while these are necessary to capture new developments and advancement, it can limit the evaluation of trends over time. In my work with Alberta Health Services' ERAS implementation program (AHS EIP), presented in Chapter 7, obtaining data that was not part of the medical record (e.g. weight loss history, postoperative nutritional supplement consumption, and mobility), was a challenge, and these fields had the highest rate of missing values. Some of these challenges can be mitigated through data linkages with other data sources where essential data

can be acquired. For example, a portion of the data presented in Chapter 8 was acquired from data linkages between our AHS EIP data and our provincial Picture Archiving Communication System (PACS). A subset of patients from AHS EIP were linked with CT images stored in PACS, and this data was used to identify CT-defined body composition features that predicted surgical outcomes. Development of minimum data sets for cachexia could be informed and aligned with these efforts and vice versa, our efforts to develop more robust diagnostic criteria for cancer cachexia may improve the type of nutritional data captured in these large standardized data collection initiatives.

Lastly, a main recommendation for the development of diagnostic criteria is that they relate to meaningful patient-centered outcomes. Reporting of patient centered outcomes also suffers from heterogeneity. In 2012, the International Consortium for Health Outcomes Measurement (ICHOM) Initiative was created to develop standard sets of patient-centered outcomes according to medical condition.⁵² The standard recommended set of outcome measures were defined based on the need to encompass overall disease control (e.g. overall survival (OS), cancer-specific survival (CSS), and recurrence-free survival (RFS), treatment complications (US National Cancer Institute Common Terminology Criteria for Adverse Events (CTCAE); Clavien-Dindo classification), and quality of life (European Organisation for Research and Treatment of Cancer (EORTC) Quality of Life C30 tool), in addition to condition specific assessments.⁵²⁻⁵⁶ Similar efforts are underway to standardize outcome measurement in anaesthesia and perioperative medicine trials.⁵⁷ These core outcomes include those related to surgical complications, organ failure and survival, cancer-specific and long term survival, patient-reported outcomes (e.g. QOL, satisfaction), health resource utilization (e.g. LOS, cost, fitness for discharge).⁵⁸

In our approach to the development of diagnostic criteria, we have considered these larger ongoing efforts of outcome standardization, and have evaluated diagnostic criteria in relation to outcomes that are of interest to medical and surgical oncologists to help support their validity and use. Identification of SM depletion using CT has gained particular attention in both medical and surgical oncology. The identification of SM depletion has been associated with outcomes related to disease control, treatment complications, and important patient reported outcomes (e.g. physical function, quality of life).^{22,26,59,60} The assessment of SM may change how oncologists dose chemotherapy, and new clinical trials are underway to test this paradigm.^{61,62}

9.3 Future development and implementation

9.3.1 Future development: acquisition of data relevant to cancer cachexia

Currently available data sets for the development of diagnostic criteria for cancer cachexia are limited with regard to the breadth and depth of data they contain. Additional data will be required to expand our understanding in specific areas, such as metabolic alterations. A clear distinction must be made between what kind of data are feasible to collect in the clinical versus research setting. Large scale initiatives such as nutritionDay, ERAS programs, and GLIM have provided valuable information related to the type of clinical and nutrition data that currently collected, as well as to what is feasible in large scale data collection efforts.^{2,3,63,64} Missing assessments include SM (adipose tissue), biological indicators of inflammation, and assessments of physical function and quality of life. The feasibility of acquiring this type of data in different practice settings needs to be determined. One such effort is currently underway, the International Nutrition Audit in FORegut TuMors (INFORM), to understand what types of data collection specific to nutrition care practices are feasible to collect as part of clinical practice, which will help generate a common minimum data set.⁶⁵

Large organizations such as the Southwest Oncology Group (SWOG) Cancer Research Network⁶⁶, or the Eastern Cooperative Oncology Group-American College of Radiology Imaging Network (ECOG-ACRIN)⁶⁷ design and conduct large multi-center clinical trials, and as a consequence collect large amounts of in-depth clinical, treatment, biospecimen, and outcome data. These networks are uniquely positioned to contribute to the understanding of alterations in metabolism. Linkages between clinical data and biological samples may allow for the exploration of underlying mechanisms of SM catabolism, tumor-derived factors, and the effects cancer treatment. For example, data available for retrospective analysis enables longitudinal assessments of SM changes using CT in response cancer treatments, which provides information about the impact of treatment regimens on SM. Knowledge about the types of treatments that affect SM could inform development of future clinical trials for interventions to support the maintenance of SM.

9.3.2 Future development: how can diagnostic criteria be used to advance nutrition care

Goals for nutrition care in cancer patients are to identify and treat nutrition and metabolic alterations to maintain SM and function, to limit interruptions in treatment, and improve quality of life.¹⁵ Nutrition interventions include dietetic counseling, oral nutrition supplements (ONS), and enteral or parenteral nutrition. Unfortunately, few high quality studies (e.g. randomized clinical trials) demonstrate effectiveness of these types of interventions in patients with cancer cachexia.^{68,69} In a meta-analysis by de van der Schueren et al.⁶⁸ patient selection criteria (e.g. included patients with weight loss of any etiology and near the end of life), lack of clinically relevant patient-centered outcomes, and low compliance to nutrition prescriptions have limited interpretations regarding the effectiveness of nutrition interventions. In a sub-set analysis patients consuming a high protein oral nutritional supplement (ONS) enriched with omega-3 fatty acids,
demonstrated improved body weight and SM while undergoing chemoradiation therapy.⁶⁸ The authors offered that omega-3 fatty acids may have contributed to the attenuation the inflammatory response and the high protein content contributed to maintenance of target protein intakes. Multimodal treatment approaches are recommended for the treatment of cancer cachexia, which includes symptom management, nutrition interventions, physical therapy, and psychosocial support, and these types of approaches have reported significant improvements in body weight, physical function, and quality of life.⁷⁰⁻⁷³ A randomized clinical trial investigating the effect of a multimodal intervention that includes nutrition counseling, a protein dense ONS enriched with omega-3 fatty acids, and an anti-inflammatory agent is currently underway.⁷⁴

Early identification of nutrition and metabolic alterations is essential to guide appropriate interventions. The diagnostic criteria developed and presented in this thesis identify patients with disparate clinical outcomes, and offer an initial clue about to characterize the etiology of WL. In Chapters 5 & 6, reductions in FI and CRP were identified as etiological criteria for WL, with reductions in FI having the strongest association. Nutrition interventions aimed at augmenting food intake with dietary counseling and ONS typically fall short of recommended energy intakes by 300-800 kcal/day, at this point in time it is not clear how to overcome this gap.⁶⁸ The use of artificial nutrition support (enteral or parenteral) in cancer patients is somewhat controversial, although it is recommended in cases where FI is severely compromised, the benefits and risks must be weighed carefully and commencement of artificial nutrition support is not recommended a prognosis of less than 2 months.¹⁵ Validated diagnostic criteria could be applied to identify patients for whom artificial nutrition support should not be initiated (i.e. patients with poor prognosis), but it is not clear how they could be used to identify patients who would benefit this approach. Therapeutic agents that enhance FI by targeting appetite centers in the brain are

197

promising treatment strategies. Agents such as ghrelin or ghrelin analogues (e.g. anamorelin)⁷⁵⁻⁷⁷ have been tested in clinical trials and demonstrated significant improvements in the LBM of advanced lung cancer patients. Other molecules of interest currently under investigation in animal models include melanocortin receptor antagonists (e.g. Melanocortin-4 Receptor Antagonists)⁷⁸, and antagonists to GDF15.⁷⁹ These types of therapies offered as part of multimodal interventions that include nutrition support could offer significant improvements in the treatment of cancer cachexia. Carefully designed clinical trials are required to test these classes of agents. Diagnostic criteria could be applied to help select patients likely to receive a benefit of the therapy (i.e. targeted etiology of WL and good prognosis).

Well-designed clinical trials are essential to test whether treatments for cancer cachexia are safe and effective. As these efforts continue to advance what can be done in the meantime to ensure that patients with cachexia are correctly identified, and that they receive nutrition and metabolic care?⁸⁰ The development of diagnostic criteria that are linked to patient-centered outcomes is progress towards more effective identification and treatment of patients with cancer cachexia. Advancement of clinical trial design for nutrition interventions are beginning to appear, where the effectiveness of nutrition care is being evaluated against the limiting of Grade \geq 3 treatment toxicities as a primary outcome.⁸¹

The assessments proposed within this thesis include WL, FI, CT-defined body composition, and CRP, which are acquired by different people involved in the care of patients with cancer. Alignment of these assessments in a common way would require leadership from the health system and across disciplines, headed by the oncologist physician or surgeon to foster interdisciplinary collaboration to ensure care providers are aware of what information is needed to identify a patient a risk of cancer cachexia and what steps should follow thereafter.⁸² ERAS

programs offer excellent examples of how to develop capacity within the health system and for care providers. ERAS programs require health systems and care providers to examine current care practices and how care is delivered to develop implementation strategies that are best suited to the local context. This implementation process contributes to the successful implementation of multimodal evidence based guidelines, and collection of data to evaluate implementation processes, evaluate change in clinical practice and improve outcomes.⁶⁴ It seems plausible that a similar type of process could be applied to oncology, where efforts to integrate supportive/palliative care (inclusive of nutrition) alongside cancer care pathways have proven effective.^{83,84} There are evidenced based care pathways for the diagnosis and treatment of different cancer types, but these pathways do not incorporate nutrition care alongside cancer care.^{85,86} I see an opportunity to align diagnostic criteria with nutrition care guidelines to create standardized nutrition care pathways for patients with cancer cachexia. Standardized nutrition care pathways could contribute the development of the nutrition intervention component that is essential for the multimodal treatment of cancer cachexia.⁸⁷⁻⁹¹ Standardized nutrition care pathways are risk stratified, and identify the need for increased levels of nutrition intervention based on risk of an outcome.⁹² Within this thesis I have identified candidate diagnostic criteria based on nutrition assessments that help identify patients at risk of poor outcomes, which could be used to develop an algorithm for nutrition care. This is an area that is open for development and future work will be directed toward this opportunity.

9.4 Conclusion

In conclusion the research presented in this thesis offers an initial step toward the development of diagnostic criteria for cancer cachexia. Identifying patients with increased risk of poor outcomes in different care settings can help guide care. Prospective validation of these

199

concepts is required. Health data are essential to the development of diagnostic criteria and the hope is the data collection efforts will continue and begin to incorporate additional parameters essential to the diagnosis of cancer cachexia.

9.4 References

- 1. Bennani-Baiti NW, D. What is cancer anorexia-cachexia syndrome? A historical perspective. R Coll Physicians Edinb. 2009; 39:257-262.
- 2. Cederholm T, Bosaeus I, Barazzoni R, et al. Diagnostic criteria for malnutrition An ESPEN Consensus Statement. Clin Nutr. 2015; 34:335-340.
- 3. Cederholm T, Jensen GL. To Create a Consensus on Malnutrition Diagnostic Criteria. JPEN J Parenter Enteral Nutr. 2017; 41:311-314.
- 4. Jensen GL, Mirtallo J, Compher C, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. JPEN J Parenter Enteral Nutr. 2010; 34:156-159.
- 5. World Health Organization. International Statistical Classification of Diseases and Related Health Problems 10th revision 2018, https://icd.who.int/browse10/2016/en, Accessed October 12, 2018.
- International Classification of Diseases, Tenth Revision, Clinical Modification (ICD-10-CM) Diagnosis Code R64. 2018; https://www.icd10data.com/ICD10CM/Codes/R00-R99/R50-R69/R64-/R64. Accessed October 12, 2018.
- 7. Evans WJ, Morley JE, Argiles J, et al. Cachexia: a new definition. Clin Nutr. 2008; 27:793-799.
- 8. Jensen GL, Cederholm T, Correia M, et al. GLIM Criteria for the Diagnosis of Malnutrition: A Consensus Report From the Global Clinical Nutrition Community. JPEN J Parenter Enteral Nutr. 2018; doi:10.1002/jpen.1440.
- 9. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.
- 10. Aggarwal R, Ringold S, Khanna D, et al. Distinctions between diagnostic and classification criteria? Arthritis Care Res. 2015; 67:891-897.
- 11. National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE) v.5.0. 2017;
 https://ctep.cancer.gov/protocoldevelopment/electronic_applications/docs/CTCAE_v5_Q uick_Reference_5x7.pdf. Accessed October 11, 2018.
- 12. Kubrak K. Prevalence and prognostic significance of malnutrition in patients with cancers of the head and neck. University of Alberta; 2018, unpublished manuscript.

- 13. Mytelka DS, Li L, Benoit K. Post-diagnosis weight loss as a prognostic factor in nonsmall cell lung cancer. J Cachexia Sarcopenia Muscle. 2018; 9:86-92.
- 14. Vagnildhaug OM, Blum D, Wilcock A, et al. The applicability of a weight loss grading system in cancer cachexia: a longitudinal analysis. J Cachexia Sarcopenia Muscle. 2017; 8:789-797.
- 15. Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.
- 16. Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008; 9:629-635.
- 17. Hopkins JJ, Skubleny D, Bigam DL, Baracos VE, Eurich DT, Sawyer MB. Barriers to the Interpretation of Body Composition in Colorectal Cancer: A Review of the Methodological Inconsistency and Complexity of the CT-Defined Body Habitus. Ann Surg Oncol. 2018; 25:1381-1394.
- 18. Boshier PR, Heneghan R, Markar SR, Baracos VE, Low DE. Assessment of body composition and sarcopenia in patients with esophageal cancer: a systematic review and meta-analysis. Dis Esophagus. 2018; doi: 10.1093/dote/doy047.
- 19. Ebadi M, Martin L, Ghosh S, et al. Subcutaneous adiposity is an independent predictor of mortality in cancer patients. Br J Cancer. 2017; 117:148-155.
- 20. Esfandiari N, Ghosh S, Prado CMM, Martin L, Mazurak V, Baracos VE. Age, Obesity, Sarcopenia, and Proximity to Death Explain Reduced Mean Muscle Attenuation in Patients with Advanced Cancer. J Frailty Aging. 2014; 3:3-8.
- 21. Fujiwara N, Nakagawa H, Kudo Y, et al. Sarcopenia, intramuscular fat deposition, and visceral adiposity independently predict the outcomes of hepatocellular carcinoma. J Hepatol. 2015; 63:131-140.
- 22. Kazemi-Bajestani SM, Mazurak VC, Baracos V. Computed tomography-defined muscle and fat wasting are associated with cancer clinical outcomes. Semin Cell Dev Biol. 2016; 54:2-10.
- 23. Malietzis G, Aziz O, Bagnall NM, Johns N, Fearon KC, Jenkins JT. The role of body composition evaluation by computerized tomography in determining colorectal cancer treatment outcomes: a systematic review. Eur J Surg Oncol. 2015; 41:186-196.
- 24. Malietzis G, Johns N, Al-Hassi HO, et al. Low Muscularity and Myosteatosis Is Related to the Host Systemic Inflammatory Response in Patients Undergoing Surgery for Colorectal Cancer. Ann Surg. 2016; 263:320-5.

- 25. Martin L, Birdsell L, MacDonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol. 2013; 31:1539-1547.
- 26. Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle. 2017; 8:759-767.
- 27. Earthman CP. Body Composition Tools for Assessment of Adult Malnutrition at the Bedside: A Tutorial on Research Considerations and Clinical Applications. JPEN J Parenter Enteral Nutr. 2015; 39:787-822.
- 28. Thibault R, Pichard C. The evaluation of body composition: a useful tool for clinical practice. Ann Nutr Metab. 2012; 60:6-16.
- 29. Guerdoux-Ninot E, Flori N, Janiszewski C, et al. Assessing dietary intake in accordance with guidelines: Useful correlations with an Ingesta Verbal/Visual Analogue Scale in medical oncology patients. Clinical Nutrition. 2018; doi:10.1016/j.clnu.2018.06.974.
- 30. Fearon K. The Sir David Cuthbertson Medal Lecture 1991: The mechanisms and treatment of weight loss in cancer. Proc Nutr Soc. 1992; 51:251-265.
- 31. Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.
- 32. Schiessel DL, Baracos VE. Barriers to cancer nutrition therapy: excess catabolism of muscle and adipose tissues induced by tumour products and chemotherapy. Proc Nutr Soc. 2018; doi:10.1017/s0029665118000186:1-9.
- 33. Friesen DE, Baracos VE, Tuszynski JA. Modeling the energetic cost of cancer as a result of altered energy metabolism: implications for cachexia. Theor Biol Med Model. 2015; 12:17.
- Hong N, Yoon HJ, Lee YH, et al. Serum PTHrP Predicts Weight Loss in Cancer Patients Independent of Hypercalcemia, Inflammation, and Tumor Burden. J Clin Endocrinol Metab. 2016; 101:1207-1214.
- 35. Lee JY, Hong N, Kim HR, et al. Effects of Serum Albumin, Calcium Levels, Cancer Stage and Performance Status on Weight Loss in Parathyroid Hormone-Related Peptide Positive or Negative Patients with Cancer. Endocrinol Metabol (Seoul, Korea). 2018; 33:97-104.
- 36. Arends J. Struggling with nutrition in patients with advanced cancer: nutrition and nourishment-focusing on metabolism and supportive care. Ann Oncol. 2018; 29:ii27-ii34.

- 37. Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.
- 38. Institute of Medicine (US) Committee on Health Research and the Privacy of Health Information. Beyond the HIPAA Privacy Rule: Enhancing Privacy, Improving Health Through Research, 2009; doi:10.17226/12458.
- 39. Schlomer BJ, Copp HL. Secondary Data Analysis of Large Data Sets in Urology: Successes and Errors to Avoid. J Urol. 2014; 191:587-596.
- 40. Douglas E, McMillan DC. Towards a simple objective framework for the investigation and treatment of cancer cachexia: the Glasgow Prognostic Score. Cancer Treat Rev. 2014; 40:685-691.
- 41. Guthrie GJ, Charles KA, Roxburgh CS, Horgan PG, McMillan DC, Clarke SJ. The systemic inflammation-based neutrophil-lymphocyte ratio: experience in patients with cancer. Crit Rev Oncol Hematol. 2013; 88:218-230.
- 42. nutritionDay Worldwide. 2018; https://www.nutritionday.org/. Accessed October 7, 2018.
- 43. ERAS Society. 2018; http://erassociety.org/. Accessed October 7, 2018.
- 44. Elias KM, Stone AB, McGinigle K, et al. The Reporting on ERAS Compliance, Outcomes, and Elements Research (RECOVER) Checklist: A Joint Statement by the ERAS((R)) and ERAS((R)) USA Societies. World J Surg. 2018; doi:10.1007/s00268-018-4753-0.
- 45. Schindler K, Pernicka E, Laviano A, et al. How nutritional risk is assessed and managed in European hospitals: a survey of 21,007 patients findings from the 2007-2008 cross-sectional nutritionDay survey. Clin Nutr. 2010; 29:552-559.
- 46. Schindler K, Themessl-Huber M, Hiesmayr M, et al. To eat or not to eat? Indicators for reduced food intake in 91,245 patients hospitalized on nutritionDays 2006-2014 in 56 countries worldwide: a descriptive analysis. Am J Clin Nutr. 2016; 104:1393-1402.
- 47. Streicher M, Themessl-Huber M, Schindler K, Sieber CC, Hiesmayr M, Volkert D. nutritionDay in Nursing Homes-The Association of Nutritional Intake and Nutritional Interventions With 6-Month Mortality in Malnourished Residents. J Am Med Dir Assoc. 2017; 18:162-168.
- 48. Navarro DA, Boaz M, Krause I, et al. Improved meal presentation increases food intake and decreases readmission rate in hospitalized patients. Clin Nutr. 2016; 35:1153-1158.
- 49. Lau CS, Chamberlain RS. Enhanced Recovery After Surgery Programs Improve Patient Outcomes and Recovery: A Meta-analysis. World J Surg. 2017; 41:899-913.

- 50. Lee L, Mata J Fau Ghitulescu GA, Ghitulescu Ga Fau Boutros M, et al. Costeffectiveness of Enhanced Recovery Versus Conventional Perioperative Management for Colorectal Surgery. Ann Surg. 2015; 262:1026-1033.
- 51. Visioni A, Shah R, Gabriel E, Attwood K, Kukar M, Nurkin S. Enhanced Recovery After Surgery for Noncolorectal Surgery?: A Systematic Review and Meta-analysis of Major Abdominal Surgery. Ann Surg. 2018; 267:57-65.
- 52. International Consortium for Health Outcomes Measurement. 2018; http://www.ichom.org/, accessed: October 11, 2018.
- 53. Ong WL, Schouwenburg MG, van Bommel ACM, et al. A Standard Set of Value-Based Patient-Centered Outcomes for Breast Cancer: The International Consortium for Health Outcomes Measurement (ICHOM) Initiative. JAMA Oncol. 2017; 3:677-685.
- 54. Zerillo JA, Schouwenburg MG, van Bommel ACM, et al. An International Collaborative Standardizing a Comprehensive Patient-Centered Outcomes Measurement Set for Colorectal Cancer. JAMA Oncol. 2017; 3:686-694.
- 55. Mak KS, van Bommel ACM, Stowell C, et al. Defining a standard set of patient-centred outcomes for lung cancer. Eur Resp J. 2016; 48:852-860.
- 56. Morgans AK, van Bommel AC, Stowell C, et al. Development of a Standardized Set of Patient-centered Outcomes for Advanced Prostate Cancer: An International Effort for a Unified Approach. Eur Urol. 2015; 68:891-898.
- 57. National Institute of Academic Anesthesia Health Services Research Center. Core Outcome Measures in Perioperative and Anaesthetic Care (COMPAC). 2018; https://www.niaa-hsrc.org.uk/HSRC-COMPAC accessed: October 11, 2018.
- 58. Myles PS, Grocott MP, Boney O, Moonesinghe SR, Group CO-S. Standardizing end points in perioperative trials: towards a core and extended outcome set. Br J Anaesth. 2016; 116:586-589.
- 59. Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapy-induced haematological toxicity in advanced non-small cell lung cancer. Lung Cancer. 2015; 90:85-91.
- 60. Baracos VE, Arribas L. Sarcopenic obesity: hidden muscle wasting and its impact for survival and complications of cancer therapy. Ann Oncol. 2018; 29:ii1-ii9.
- 61. Clinicaltrials.gov. Identifier: NCT03255434 LEAn Body Mass Normalization of OXaliplatin Based Chemotherapy (LEANOX). 2017; https://clinicaltrials.gov/ct2/show/NCT03255434, Accessed Octber 8, 2018.

- 62. Clinicaltrials.gov. Identifier: NCT01624051 A Study Comparing Chemotherapy Dosing Based on Either Standard Body Surface Area or Lean Body Mass in Patients With Advanced Lung Cancer. 2012; https://clinicaltrials.gov/ct2/show/NCT0162405, Accessed October 8, 2018.
- 63. Nicolo M, Compher CW, Still C, Huseini M, Dayton S, Jensen GL. Feasibility of accessing data in hospitalized patients to support diagnosis of malnutrition by the Academy-A.S.P.E.N. malnutrition consensus recommended clinical characteristics. JPEN J Parenter Enteral Nutr. 2014; 38:954-959.
- 64. Ljungqvist O, Scott M, Fearon KC. Enhanced Recovery After Surgery: A Review. JAMA Surg. 2017; 152:292-298.
- 65. Clinicaltrials.gov. Identifier: NCT02829489 International Nutrition Audit in FORegut TuMors (INFORM). 2016; https://clinicaltrials.gov/ct2/show/NCT02829489, Accessed October 8, 2018.
- 66. Southwest Oncology Group Clinical Research Network. 2018; https://www.swog.org/. Accessed October 8, 2018.
- 67. Eastern Cooperative Oncology Group-American College of Radiology Imaging Network. 2018; https://ecog-acrin.org/. Accessed October 8, 2018.
- 68. de van der Schueren MAE, Laviano A, Blanchard H, Jourdan M, Arends J, Baracos VE. Systematic review and meta-analysis of the evidence for oral nutritional intervention on nutritional and clinical outcomes during chemo(radio)therapy: current evidence and guidance for design of future trials. Ann Oncol. 2018; 29:1141-1153.
- 69. Chow R, Bruera E, Chiu L, et al. Enteral and parenteral nutrition in cancer patients: a systematic review and meta-analysis. Annals of palliative medicine. 2016; 5:30-41.
- 70. Gagnon B, Murphy J, Eades M, et al. A prospective evaluation of an interdisciplinary nutrition-rehabilitation program for patients with advanced cancer. Curr Oncol. 2013; 20:310-318.
- 71. Nasrah R, Kanbalian M, Van Der Borch C, Swinton N, Wing S, Jagoe RT. Defining the role of dietary intake in determining weight change in patients with cancer cachexia. Clin Nutr. 2018; 37:235-241.
- 72. Parmar MP, Swanson T, Jagoe RT. Weight changes correlate with alterations in subjective physical function in advanced cancer patients referred to a specialized nutrition and rehabilitation team. Support Care Cancer. 2013; 21:2049-2057.
- 73. Parmar MP, Vanderbyl BL, Kanbalian M, Windholz TY, Tran A-T, Jagoe RT. A multidisciplinary rehabilitation programme for cancer cachexia improves quality of life. BMJ Support Palliat Care. 2017; 7:441-449.

- 74. Clinicaltrials.gov. Identifier: NCT02330926 Multimodal Intervention for Cachexia in Advanced Cancer Patients Undergoing Chemotherapy (MENAC). 2015; https://clinicaltrials.gov/ct2/show/NCT02330926?term=MENAC&cond=Cancer+Cachex ia&rank=1. Accessed October 10, 2018.
- 75. Katakami N, Uchino J, Yokoyama T, et al. Anamorelin (ONO-7643) for the treatment of patients with non-small cell lung cancer and cachexia: Results from a randomized, double-blind, placebo-controlled, multicenter study of Japanese patients (ONO-7643-04). Cancer. 2018; 124:606-616.
- 76. Currow D, Temel JS, Abernethy A, Milanowski J, Friend J, Fearon KC. ROMANA 3: a phase 3 safety extension study of anamorelin in advanced non-small-cell lung cancer (NSCLC) patients with cachexia. Ann Oncol. 2017; 28:1949-1956.
- 77. Temel JS, Abernethy AP, Currow DC, et al. Anamorelin in patients with non-small-cell lung cancer and cachexia (ROMANA 1 and ROMANA 2): results from two randomised, double-blind, phase 3 trials. Lancet Oncol. 2016; 17:519-531.
- 78. Dallmann R, Weyermann P, Anklin C, et al. The orally active melanocortin-4 receptor antagonist BL-6020/979: a promising candidate for the treatment of cancer cachexia. J Cachexia Sarcopenia Muscle. 2011; 2:163-174.
- 79. Tsai VWW, Husaini Y, Sainsbury A, Brown DA, Breit SN. The MIC-1/GDF15-GFRAL Pathway in Energy Homeostasis: Implications for Obesity, Cachexia, and Other Associated Diseases. Cell Metab. 2018; 28:353-368.
- 80. Currow D, LeBlanc TW. It is time to rethink weight loss in cancer. Ann Oncol. 2018; 29:1090-1091.
- 81. Clinicaltrials.gov. Identifier: NCT01143285 Impact of Early and Active Nutritional and Dietary Management grade3 or More Toxicities Induced by Chemotherapy and Targeted Therapies Administered to Patients as First Intention for Non Surgical Metastatic Colorectal Cancer (CHIMIODIET). 2010; https://clinicaltrials.gov/ct2/show/NCT01143285, Accessed October 8, 2018.
- 82. Tappenden KA, Quatrara B, Parkhurst ML, Malone AM, Fanjiang G, Ziegler TR. Critical role of nutrition in improving quality of care: an interdisciplinary call to action to address adult hospital malnutrition. J Acad Nutr Diet. 2013; 113:1219-1237.
- 83. Girgis A, Durcinoska I, Koh E-S, et al. Development of Health Pathways to Standardize Cancer Care Pathways Informed by Patient-Reported Outcomes and Clinical Practice Guidelines. JCO Clinical Cancer Informatics. 2018; doi:10.1200/cci.18.00024:1-13.
- 84. Temel JS, Greer JA, Muzikansky A, et al. Early palliative care for patients with metastatic non-small-cell lung cancer. N Engl J Med. 2010; 363:733-742.

- 85. Fasola G, Menis J, Follador A, et al. Integrated care pathways in lung cancer: a quality improvement project. Int J Technol Assess Health Care. 2018; 34:3-9.
- Planchard D, Popat S, Kerr K, et al. Metastatic non-small cell lung cancer: ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann Oncol. 2018; 29:iv192-iv237.
- 87. Maddocks M, Hopkinson J, Conibear J, Reeves A, Shaw C, Fearon KC. Practical multimodal care for cancer cachexia. Curr Opin Support Palliat Care. 2016; 10:298-305.
- 88. Ramchandran K, Tribett E, Dietrich B, Von Roenn J. Integrating Palliative Care Into Oncology: A Way Forward. Cancer Control. 2015; 22:386-395.
- 89. Hui D, Bruera E. Integrating palliative care into the trajectory of cancer care. Nat Rev Clin Oncol. 2016; 13:159-171.
- 90. Sommerbakk R, Haugen DF, Tjora A, Kaasa S, Hjermstad MJ. Barriers to and facilitators for implementing quality improvements in palliative care results from a qualitative interview study in Norway. BMC Palliat Care. 2016; 15:61.
- 91. den Herder-van der Eerden M, Ewert B, Hodiamont F, Hesse M, Hasselaar J, Radbruch L. Towards accessible integrated palliative care: Perspectives of leaders from seven European countries on facilitators, barriers and recommendations for improvement. J Integr Care (Brighton). 2017; 25:222-232.
- 92. Keller HH, McCullough J, Davidson B, et al. The Integrated Nutrition Pathway for Acute Care (INPAC): Building consensus with a modified Delphi. Nutr J. 2015; 14:63.

Diagnostic Criteria	Cancer Cachexia ⁹	Cachexia ⁷	Malnutrition ⁸
Weight loss	>5% previous 6 months	>5% in previous 12 months or less	>5% previous 6 months
	>2% if BMI<20		>10% if longer than previous 6 months
BMI (kg/m ²)	<20	<20	<20 if <70 years
			<22 if >70 years
			<18.5 if <70 years (Asia)
			<20 if >70 years (Asia)
Skeletal muscle	Appendicular SMI consistent with sarcopenia (males <7.26 kg/m2; females <5.45 kg/m2) MUMA by anthropometry (men <32 cm2, women <18 cm2);31 CT lumbar SMI (men <55 cm2/m2; women <39 cm2/m2);33 whole body FFM BIA (men <14.6 kg/m2; women <11.4 kg/m2).	Low fat free mass	Reduced by validated body composition measuring techniques (DEXA, BIA, CT, MRI)
Muscle Strength	None	Decreased	None
Food Intake	Reduced food intake Anorexia	Anorexia	<pre><50% of estimated requirements for > 1 week, or any reduction for >2 weeks, or any chronic GI condition that adversely impacts food assimilation or absorption</pre>
Inflammation	Systemic inflammation	CRP IL-6	Acute disease – severe Chronic disease – mild to moderate
Fatigue	None	Fatigue	None
Other Biochemistry		Hemoglobin <120 g/L	

Table 9-1. A summary of proposed diagnostic criteria for cancer cachexia, cachexia, and malnutrition.

 Albumin <35 g/L</td>

 BIA, bioelectrical impedance analysis; BMI, body mass index; CRP, c-reactive protein; CT, computed tomography; DEXA, dual energy x-ray absorptiometry; FFM, fat free mass ; GI, gastrointestinal; IL, interleukin; MUMA, mid upper-arm muscle area; L3, third lumbar vertebrae; SMI, skeletal muscle index

BIBLIOGRAPHY

Aapro M, Arends J, Bozzetti F, et al. Early recognition of malnutrition and cachexia in the cancer patient: a position paper of a European School of Oncology Task Force. Ann Oncol. 2014; 25:1492-1499.

Aarts MA, Rotstein OD, Pearsall EA, et al. Postoperative ERAS Interventions Have the Greatest Impact on Optimal Recovery: Experience With Implementation of ERAS Across Multiple Hospitals. Ann Surg. 2018; 267: 992-997.

Aggarwal R, Ringold S, Khanna D, et al. Distinctions between diagnostic and classification criteria? Arthritis Care Res. 2015; 67:891-897.

Amano K, Maeda I, Morita T, et al. Eating-related distress and need for nutritional support of families of advanced cancer patients: a nationwide survey of bereaved family members. J Cachexia Sarcopenia Muscle. 2016; 7:527-534.

Antoun S, Bayar A, Ileana E, Laplanche A, Fizazi K, di PM, et al. High subcutaneous adipose tissue predicts the prognosis in metastatic castration-resistant prostate cancer patients in post chemotherapy setting. Eur J Cancer 2015; 51:2570-7.

Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017; 36:11-48.

Arends J, Baracos V, Bertz H, et al. ESPEN expert group recommendations for action against cancer-related malnutrition. Clin Nutr. 2017; 36:1187-1196.

Arends J. Struggling with nutrition in patients with advanced cancer: nutrition and nourishment-focusing on metabolism and supportive care. Ann Oncol. 2018; 29:ii27-ii34.

Aust S, Knogler T, Pils D, Obermayr E, Reinthaller A, Zahn L, et al. Skeletal muscle depletion and markers for cancer cachexia are strong prognostic factors in epithelial ovarian cancer. PLoS One 2015; 10:e0140403.

Austin P. A Comparison of Statistical Modeling Strategies for Analyzing Length of Stay after CABG Surgery. Health Serv Outcomes Res Methodol 2003; 3:107-133.

Clinical Oncology Society of Australia. Head and Neck Guideline Steering Committee. Evidence-based practice guidelines for the nutritional management of adult patients with head and neck cancer. Sydney: Cancer Council Australia. 2018; https://wiki.cancer.org.au/australiawiki/index.php?oldid=116710, Accessed July 19, 2018.

Baldwin C, Spiro A, Ahern R, Emery PW. Oral nutritional interventions in malnourished patients with cancer: a systematic review and meta-analysis. J Natl Cancer Inst. 2012; 104:371-385.

Balstad TR, Kaasa S, Solheim TS. Multimodal nutrition/anabolic therapy for wasting conditions. Curr Opin Clin Nutr Metab Care. 2014; 17:226-235.

Balstad TR, Solheim TS, Strasser F, Kaasa S, Bye A. Dietary treatment of weight loss in patients with advanced cancer and cachexia: a systematic literature review. Crit Rev Oncol Hematol. 2014; 91:210-221.

Baracos VE, Arribas L. Sarcopenic obesity: hidden muscle wasting and its impact for survival and complications of cancer therapy. Ann Oncol. 2018; 29:ii1-ii9.

Baracos VE, Martin L, Korc M, Guttridge DC, Fearon KCH. Cancer-associated cachexia. Nat Rev Dis Primers. 2018; 4:17105.

Baracos VE. Psoas as a sentinel muscle for sarcopenia: a flawed premise. J Cachexia Sarcopenia Muscle 2017; 8:527-528.

Barajas Galindo DE, Vidal-Casariego A, Calleja-Fernandez A, et al. Appetite disorders in cancer patients: Impact on nutritional status and quality of life. Appetite. 2017; 114:23-27.

Barberan-Garcia A, Ubre M, Roca J, et al. Personalised Prehabilitation in High-risk Patients Undergoing Elective Major Abdominal Surgery: A Randomized Blinded Controlled Trial. Ann Surg. 2018; 267:50-56.

Bauer MS, Damschroder L, Hagedorn H, Smith J, Kilbourne AM. An introduction to implementation science for the non-specialist. BMC Psychol. 2015; 3:32.

Bennani-Baiti NW, D. What is cancer anorexia-cachexia syndrome? A historical perspective. R Coll Physicians Edinb. 2009; 39:257-262.

Blainey S: Consultation and clinical history-taking skills, in Ranson M, Abbott H, Braithwaite W (eds): Clinical examination skills for health professionals, M&K Update Ltd, 2014, pp 1-19

Blauwhoff-Buskermolen S, Langius JAE, Becker A, Verheul HMW, de van der Schueren MAE. The influence of different muscle mass measurements on the diagnosis of cancer cachexia. J Cachexia Sarcopenia Muscle. 2017; 8:615-622.

Blum D, Omlin A, Baracos VE, et al. Cancer cachexia: a systematic literature review of items and domains associated with involuntary weight loss in cancer. Crit Rev Oncol Hematol. 2011; 80:114-144.

Blum D, Stene GB, Solheim TS, et al. Validation of the Consensus-Definition for Cancer Cachexia and evaluation of a classification model--a study based on data from an international multicentre project (EPCRC-CSA). Ann Oncol. 2014; 25:1635-1642.

Boer BC, de Graaff F, Brusse-Keizer M, et al. Skeletal muscle mass and quality as risk factors for postoperative outcome after open colon resection for cancer. Int J Colorectal Dis 2016; 31:1117-1124.

Boshier PR, Heneghan R, Markar SR, Baracos VE, Low DE. Assessment of body composition and sarcopenia in patients with esophageal cancer: a systematic review and meta-analysis. Dis Esophagus. 2018; doi:10.1093/dote/doy047.

Bozzetti F, Cotogni P, Lo VS, Pironi L, Giardiello D, Mariani L. Development and validation of a nomogram to predict survival in incurable cachectic cancer patients on home parenteral nutrition. Ann Oncol 2015; 26:2335-40.

Bozzetti F, Mariani L, Lo VS, et al. The nutritional risk in oncology: a study of 1,453 cancer outpatients. Support Care Cancer, 2012; 20:1919-1928.

Burden ST, Hill J, Shaffer JL, Todd C. Nutritional status of preoperative colorectal cancer patients. J Hum Nutr Diet. 2010; 23:402-407.

Burfeind KG, Michaelis KA, Marks DL. The central role of hypothalamic inflammation in the acute illness response and cachexia. Semin Cell Dev Biol. 2016; 54:42-52.

Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle 2017; 8:759-767.

Bye A, Wesseltoft-Rao N, Iversen PO, et al. Alterations in inflammatory biomarkers and energy intake in cancer cachexia: a prospective study in patients with inoperable pancreatic cancer. Med Oncol. 2016; 33:54.

Caccialanza R, Pedrazzoli P, Cereda E, et al. Nutritional Support in Cancer Patients: A Position Paper from the Italian Society of Medical Oncology (AIOM) and the Italian Society of Artificial Nutrition and Metabolism (SINPE). J Cancer. 2016; 7:131-135.

Cakir H, Heus C, van der Ploeg TJ, Houdijk AP. Visceral obesity determined by CT scan and outcomes after colorectal surgery; a systematic review and meta-analysis. Int J Colorectal Dis 2015; 30:875-882.

Statistics Canada. Body mass index, overweight or obese, self-reported, adult, by age group and sex. 2016; http://www.statcan.gc.ca/tables-tableaux/sum-som/l01/cst01/health81b-eng.htm, Accessed October 11, 2018.

Carli F, Gillis C, Scheede-Bergdahl C. Promoting a culture of prehabilitation for the surgical cancer patient. Acta Oncol. 2017; 56:128-133.

Cecchini S, Cavazzini E, Marchesi F, Sarli L, Roncoroni L. Computed tomography volumetric fat parameters versus body mass index for predicting short-term outcomes of colon surgery. World J Surg. 2011; 35:415-423.

Cederholm T, Bosaeus I, Barazzoni R, et al. Diagnostic criteria for malnutrition - An ESPEN Consensus Statement. Clin Nutr. 2015; 34:335-340.

Cederholm T, Jensen GL. To Create a Consensus on Malnutrition Diagnostic Criteria. JPEN J Parenter Enteral Nutr. 2017; 41:311-314.

Center for Disease Control and Prevention. Obesity and Overweight. 2016; http://www.cdc.gov/nchs/fastats/obesity-overweight.htm, Accessed October 11, 2018.

Cespedes Feliciano EM, Kroenke CH, Meyerhardt JA, et al. Association of Systemic Inflammation and Sarcopenia With Survival in Nonmetastatic Colorectal Cancer: Results From the C SCANS Study. JAMA Oncol 2017; 3:e172319.

Chan LN, Compher C, DiBaise JK, et al. American Society for Parenteral and Enteral Nutrition research agenda. JPEN J Parenter Enteral Nutr. 2014; 38:13-18.

Chasen MR, Feldstain A, Gravelle D, et al. An interprofessional palliative care oncology rehabilitation program: effects on function and predictors of program completion. Curr Oncol, 2013; 20:301-309.

Chen B, Zhang Y, Zhao S, et al. The impact of general/visceral obesity on completion of mesorectum and perioperative outcomes of laparoscopic TME for rectal cancer: A STARD-compliant article. Medicine 2016; 95:e4462.

Chen WZ, Chen XD, Ma LL, et al. Impact of Visceral Obesity and Sarcopenia on Short-Term Outcomes After Colorectal Cancer Surgery. Dig Dis Sci 2018; doi:10.1007/s10620-018-5019-2.

Choi Y, Oh DY, Kim TY, et al. Skeletal muscle depletion predicts the prognosis of patients with advanced pancreatic cancer undergoing palliative chemotherapy, independent of body mass index. PLoS One 2015; 10:e0139749.

Chow CC, Hall KD. Short and long-term energy intake patterns and their implications for human body weight regulation. Physiol Behav. 2014; 134:60-65.

Chow R, Bruera E, Chiu L, et al. Enteral and parenteral nutrition in cancer patients: a systematic review and meta-analysis. Ann Palliat Med. 2016; 5:30-41.

Clinicaltrials.gov. Identifier: NCT01143285 Impact of Early and Active Nutritional and Dietary Management grade3 or More Toxicities Induced by Chemotherapy and Targeted Therapies Administered to Patients as First Intention for Non Surgical Metastatic Colorectal Cancer (CHIMIODIET). 2010; https://clinicaltrials.gov/ct2/show/NCT01143285, Accessed October 8, 2018.

Clinicaltrials.gov. Identifier: NCT01624051 A Study Comparing Chemotherapy Dosing Based on Either Standard Body Surface Area or Lean Body Mass in Patients With Advanced Lung Cancer. 2012; https://clinicaltrials.gov/ct2/show/NCT0162405, Accessed October 8, 2018.

Clinicaltrials.gov. Identifier: NCT02330926 Multimodal Intervention for Cachexia in Advanced Cancer Patients Undergoing Chemotherapy (MENAC). 2015;

https://clinicaltrials.gov/ct2/show/NCT02330926?term=MENAC&cond=Cancer+Cachexia&ran k=1, Accessed October 10, 2018.

Clinicaltrials.gov. Identifier: NCT02829489 International Nutrition Audit in FORegut TuMors (INFORM). 2016; https://clinicaltrials.gov/ct2/show/NCT02829489, Accessed October 8, 2018.

Clinicaltrials.gov. Identifier: NCT03255434 LEAn Body Mass Normalization of OXaliplatin Based Chemotherapy (LEANOX). 2017; https://clinicaltrials.gov/ct2/show/NCT03255434, Accessed October 8, 2018.

Cooper AB, Slack R, Fogelman D, et al. Characterization of anthropometric changes that occur during neoadjuvant therapy for potentially resectable pancreatic cancer. Ann Surg Oncol 2015; 22:2416-23.

Cooper C, Burden ST, Cheng H, Molassiotis A. Understanding and managing cancer-related weight loss and anorexia: insights from a systematic review of qualitative research. J Cachexia Sarcopenia Muscle. 2015; 6:99-111.

Correia MI, Hegazi RA, Higashiguchi T, et al. Evidence-based recommendations for addressing malnutrition in health care: an updated strategy from the feedM.E. Global Study Group. J Am Med DirAssoc. 2014; 15:544-550.

Currow D, LeBlanc TW. It is time to rethink weight loss in cancer. Ann Oncol. 2018; 29:1090-1091.

Currow D, Temel JS, Abernethy A, Milanowski J, Friend J, Fearon KC. ROMANA 3: a phase 3 safety extension study of anamorelin in advanced non-small-cell lung cancer (NSCLC) patients with cachexia. Ann Oncol. 2017; 28:1949-1956.

Dallmann R, Weyermann P, Anklin C, et al. The orally active melanocortin-4 receptor antagonist BL-6020/979: a promising candidate for the treatment of cancer cachexia. J Cachexia Sarcopenia Muscle. 2011; 2:163-174.

Darnall BD, Suarez EC. Sex and gender in psychoneuroimmunology research: Past, present and future. Brain Behav Immun. 2009; 23:595-604.

De PG, Silvestris F. Obesity as a major risk factor for cancer. J Obes. 2013; 2013:291546.

de van der Schueren MAE, Laviano A, Blanchard H, Jourdan M, Arends J, Baracos VE. Systematic review and meta-analysis of the evidence for oral nutritional intervention on nutritional and clinical outcomes during chemo(radio)therapy: current evidence and guidance for design of future trials. Ann Oncol. 2018; 29:1141-1153.

Deans DAC, Tan BH, Wigmore SJ, et al. The influence of systemic inflammation, dietary intake and stage of disease on rate of weight loss in patients with gastro-oesophageal cancer. Br J Cancer, 2009; 100:63-69.

Dechaphunkul T, Martin L, Alberda C, Olson K, Baracos V, Gramlich L. Malnutrition assessment in patients with cancers of the head and neck: a call to action and consensus. Crit Rev Oncol Hematol 2013; 88:459-76.

den Herder-van der Eerden M, Ewert B, Hodiamont F, Hesse M, Hasselaar J, Radbruch L. Towards accessible integrated palliative care: Perspectives of leaders from seven European countries on facilitators, barriers and recommendations for improvement. J Integr Care (Brighton). 2017; 25:222-232.

Dewys WD, Begg C, Lavin PT, et al. Prognostic effect of weight loss prior to chemotherapy in cancer patients. Eastern Cooperative Oncology Group. Am J Med, 1980; 69:491-497.

Dindo D, Demartines N, Clavien PA. Classification of Surgical Complications. Ann Surg 2004; 240:205-213.

Douglas E, McMillan DC. Towards a simple objective framework for the investigation and treatment of cancer cachexia: the Glasgow Prognostic Score. Cancer Treat Rev. 2014; 40:685-691.

Doyle SL, Bennett AM, Donohoe CL, et al. Establishing computed tomography-defined visceral fat area thresholds for use in obesity-related cancer research. Nutr Res 2013; 33:171-179.

Du Y, Karvellas CJ, Baracos V, et al. Sarcopenia is a predictor of outcomes in very elderly patients undergoing emergency surgery. Surgery. 2014; 156:521-527.

Earthman CP. Body Composition Tools for Assessment of Adult Malnutrition at the Bedside: A Tutorial on Research Considerations and Clinical Applications. JPEN J Parenter Enteral Nutr. 2015; 39:787-822.

Eastern Cooperative Oncology Group-American College of Radiology Imaging Network. 2018; https://ecog-acrin.org/. Accessed October 8, 2018.

Ebadi M, Martin L, Ghosh S, et al. Subcutaneous adiposity is an independent predictor of mortality in cancer patients. Br J Cancer. 2017; 117:148-155.

Elias KM, Stone AB, McGinigle K, et al. The Reporting on ERAS Compliance, Outcomes, and Elements Research (RECOVER) Checklist: A Joint Statement by the ERAS((R)) and ERAS((R)) USA Societies. World J Surg. 2018; doi:10.1007/s00268-018-4753-0.

ERAS Society. 2018; http://erassociety.org/. Accessed October 7, 2018.

ERAS Society. ERAS Implementation Program (EIP). 2018. https://www.encare.net/healthcare-professionals/products-andservices/eras-implementation-program-eip. Accessed March 5, 2018.

Esfandiari N, Ghosh S, Prado CMM, Martin L, Mazurak V, Baracos VE. Age, Obesity, Sarcopenia, and Proximity to Death Explain Reduced Mean Muscle Attenuation in Patients with Advanced Cancer. J Frailty Aging. 2014; 3:3-8.

Evans WJ, Morley JE, Argiles J, et al. Cachexia: a new definition. Clin Nutr. 2008; 27:793-799.

Farkas J, von HS, Kalantar-Zadeh K, Morley JE, Anker SD, Lainscak M. Cachexia as a major public health problem: frequent, costly, and deadly. J Cachexia Sarcopenia Muscle. 2013; 4:173-178.

Fasola G, Menis J, Follador A, et al. Integrated care pathways in lung cancer: a quality improvement project. Int J Technol Assess Health Care. 2018; 34:3-9.

Fearon K, Arends J, Baracos V. Understanding the mechanisms and treatment options in cancer cachexia. Nat Rev Clin Oncol, 2013; 10:90-99.

Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus framework. Lancet Oncol. 2011; 12:489-495.

Fearon K. The Sir David Cuthbertson Medal Lecture 1991: The mechanisms and treatment of weight loss in cancer. Proc Nutr Soc. 1992; 51:251-265.

Fearon KC, Barber MD, Moses AG, et al. Double-blind, placebo-controlled, randomized study of eicosapentaenoic acid diester in patients with cancer cachexia. J Clin Oncol, 2006; 24:3401-3407.

Fearon KC, Von Meyenfeldt MF, Moses AG, et al. Effect of a protein and energy dense N-3 fatty acid enriched oral supplement on loss of weight and lean tissue in cancer cachexia: a randomised double blind trial. Gut, 2003; 52:1479-1486.

Fearon KC, Voss AC, Hustead DS. Definition of cancer cachexia: effect of weight loss, reduced food intake, and systemic inflammation on functional status and prognosis. Am J Clin Nutr 2006; 83:1345-50.

Ferguson M, Capra S, Bauer J, Banks M. Development of a valid and reliable malnutrition screening tool for adult acute hospital patients. Nutrition. 1999; 15:458-464.

Flegal KM, Kit BK, Orpana H, et al. Association of all-cause mortality with overweight and obesity using standard body mass index categories: a systematic review and meta-analysis. JAMA, 2013; 309:71-82.

French SA, Epstein LH, Jeffery RW, Blundell JE, Wardle J. Eating behavior dimensions: associations with energy intake and body weight: a review. Appetite. 2012; 59:541-549.

French Speaking Society of Clinical Nutrition and Metabolism. Clinical nutrition guidelines of the French Speaking Society of Clinical Nutrition and Metabolism (SFNEP): Summary of

recommendations for adults undergoing non-surgical anticancer treatment. Dig Liver Dis. 2014; 46:667-674.

Friesen DE, Baracos VE, Tuszynski JA. Modeling the energetic cost of cancer as a result of altered energy metabolism: implications for cachexia. Theor Biol Med Model. 2015; 12:17.

Fujiwara N, Nakagawa H, Kudo Y, et al. Sarcopenia, intramuscular fat deposition, and visceral adiposity independently predict the outcomes of hepatocellular carcinoma. J Hepatol. 2015; 63:131-140.

Fukushima H, Nakanishi Y, Kataoka M, Tobisu K, Koga F. Prognostic significance of sarcopenia in patients with metastatic renal cell carcinoma. J Urol 2016; 195:26-32.

Gagnon B, Abrahamowicz M, Xiao Y, et al. Flexible modeling improves assessment of prognostic value of C-reactive protein in advanced non-small cell lung cancer. Br J Cancer. 2010; 102:1113-1122.

Gagnon B, Agulnik JS, Gioulbasanis I, et al. Montreal prognostic score: estimating survival of patients with non-small cell lung cancer using clinical biomarkers. Br J Cancer, 2013; 109:2066-2071.

Gagnon B, Murphy J, Eades M, et al. A prospective evaluation of an interdisciplinary nutrition-rehabilitation program for patients with advanced cancer. Curr Oncol. 2013; 20:310-318.

Garcia JM, Boccia RV, Graham CD, et al. Anamorelin for patients with cancer cachexia: an integrated analysis of two phase 2, randomised, placebo-controlled, double-blind trials. Lancet Oncol 2015; 16:108-16.

Gibson DJ, Burden ST, Strauss BJ, Todd C, Lal S. The role of computed tomography in evaluating body composition and the influence of reduced muscle mass on clinical outcome in abdominal malignancy: a systematic review. Eur J Clin Nutr 2015; 69:1079-86.

Gillis C, Nguyen TH, Liberman AS, Carli F. Nutrition adequacy in enhanced recovery after surgery: a single academic center experience. Nutr Clin Pract. 2015; 30:414-419.

Gillis CG, Marlett N, MacKean, G, et al. Patients as partners in Enhanced Recovery After Surgery: A qualitative patient-led study. BMJ Open. 2017; 7:10.

Gioulbasanis I, Georgoulias P, Vlachostergios PJ, et al. Mini Nutritional Assessment (MNA) and biochemical markers of cachexia in metastatic lung cancer patients: interrelations and associations with prognosis. Lung Cancer, 2011; 74:516-520.

Gioulbasanis I, Martin L, Baracos VE, Thezenas S, Koinis F, Senesse P. Nutritional assessment in overweight and obese patients with metastatic cancer: does it make sense? Ann Oncol. 2015; 26:217-221.

Girgis A, Durcinoska I, Koh E-S, et al. Development of Health Pathways to Standardize Cancer Care Pathways Informed by Patient-Reported Outcomes and Clinical Practice Guidelines. JCO Clinical Cancer Informatics. 2018; doi:10.1200/cci.18.00024:1-13.

Glare P, Jongs W, Zafiropoulos B. Establishing a cancer nutrition rehabilitation program (CNRP) for ambulatory patients attending an Australian cancer center. Support Care Cancer. 2011; 19:445-454.

Gramlich LM, Sheppard CE, Wasylak T, et al. Implementation of Enhanced Recovery After Surgery: a strategy to transform surgical care across a health system. Implement Sci. 2017; 12:67.

Grant MC, Yang D, Wu CL, Makary MA, Wick EC. Impact of Enhanced Recovery After Surgery and Fast Track Surgery Pathways on Healthcare-associated Infections: Results From a Systematic Review and Meta-analysis. Ann Surg. 2017; 265:68-79.

Greco MC, G; Beretta,L; Gemma,M; Pecorelli,N; Braga M. Enhanced Recovery Program in Colorectal Surgery: A Meta-analysis of Randomized Controlled Trials. World J Surg 2014; 38:1531-1541.

ERAS Compliance Group. The Impact of Enhanced Recovery Protocol Compliance on Elective Colorectal Cancer Resection: Results From an International Registry. Ann Surg. 2015; 261:1153-1159.

Guerdoux-Ninot E, Flori N, Janiszewski C, et al. Assessing dietary intake in accordance with guidelines: Useful correlations with an Ingesta Verbal/Visual Analogue Scale in medical oncology patients. Clin Nutr. 2018; doi:10.1016/j.clnu.2018.06.974.

Guerdoux-Ninot E, Kilgour RD, Janiszewski C, et al. Meal context and food preferences in cancer patients: results from a French self-report survey. Springer Plus. 2016; 5:810.

Gustafsson UO, Hausel J, Thorell A, et al. Adherence to the enhanced recovery after surgery protocol and outcomes after colorectal cancer surgery. Arch Surg. 2011; 146:571-577.

Gustafsson UO, Ljungqvist O. Perioperative nutritional management in digestive tract surgery. Curr Opin Clin Nutr Metab Care. 2011; 14:504-509.

Gustafsson UO, Scott MJ, Schwenk W, et al. Guidelines for perioperative care in elective colonic surgery: Enhanced Recovery After Surgery (ERAS(R)) Society recommendations. Clin Nutr. 2012; 31:783-800.

Guthrie GJ, Charles KA, Roxburgh CS, Horgan PG, McMillan DC, Clarke SJ. The systemic inflammation-based neutrophil-lymphocyte ratio: experience in patients with cancer. Crit Rev Oncol Hematol 2013; 88:218-30.

Hanaoka M, Yasuno M, Ishiguro M, et al. Morphologic change of the psoas muscle as a

surrogate marker of sarcopenia and predictor of complications after colorectal cancer surgery. Int J Colorectal Dis 2017; 32:847-856.

Harada K, Ida S, Baba Y, et al. Prognostic and clinical impact of sarcopenia in esophageal squamous cell carcinoma. Dis Esophagus 2016; 29:627-633.

Harrell FE, Jr., Lee KL, Mark DB. Multivariable prognostic models: issues in developing models, evaluating assumptions and adequacy, and measuring and reducing errors. Stat Med, 1996; 15:361-387.

Haverkort EB, de Haan RJ, Binnekade JM, et al. Self-reporting of height and weight: valid and reliable identification of malnutrition in preoperative patients. Am J Surg, 2012; 203:700-707.

Hetzler KL, Hardee JP, Puppa MJ, et al. Sex Differences in the Relationship of IL-6 Signaling to Cancer Cachexia Progression. Biochim Biophys Acta. 2015; 1852:816-825.

Heus C, Cakir H, Lak A, Doodeman HJ, Houdijk AP. Visceral obesity, muscle mass and outcome in rectal cancer surgery after neo-adjuvant chemo-radiation. Int J Surg 2016; 29:159-164.

Hills AP, Mokhtar N, Byrne NM. Assessment of physical activity and energy expenditure: an overview of objective measures. Front Nutr. 2014; 1:doi: 10.3389/fnut.2014.00005.

Hong N, Yoon HJ, Lee YH, et al. Serum PTHrP Predicts Weight Loss in Cancer Patients Independent of Hypercalcemia, Inflammation, and Tumor Burden. J Clin Endocrinol Metab. 2016; 101:1207-1214.

Hopkins JJ, Skubleny D, Bigam DL, Baracos VE, Eurich DT, Sawyer MB. Barriers to the Interpretation of Body Composition in Colorectal Cancer: A Review of the Methodological Inconsistency and Complexity of the CT-Defined Body Habitus. Ann Surg Oncol 2018; 25:1381-1394.

Huang DD, Wang SL, Zhuang CL, et al. Sarcopenia, as defined by low muscle mass, strength and physical performance, predicts complications after surgery for colorectal cancer. Colorectal Dis 2015; 17:O256-264.

Hubbard JM, Cohen HJ, Muss HB. Incorporating biomarkers into cancer and aging research. J Clin Oncol 2014; 32:2611-6.

Hui D, Bruera E. Integrating palliative care into the trajectory of cancer care. Nat Rev Clin Oncol. 2016; 13:159-171.

Institute of Medicine (US) Committee on Health Research and the Privacy of Health Information. Beyond the HIPAA Privacy Rule: Enhancing Privacy, Improving Health Through Research. 2009; doi:10.17226/12458. International Classification of Diseases, Tenth Revision, Clinical Modification (ICD-10-CM) Diagnosis Code R64. 2018; https://www.icd10data.com/ICD10CM/Codes/R00-R99/R50-R69/R64-/R64. Accessed October 12, 2018.

International Consortium for Health Outcomes Measurement. 2018; http://www.ichom.org/, accessed: October 11, 2018.

International Expert Committee report on the role of the A1C assay in the diagnosis of diabetes. Diabetes Care 2009; 32:1327-34.

Isenring E, Zabel R, Bannister M, et al. Updated evidence-based practice guidelines for the nutritional management of patients receiving radiation therapy and/or chemotherapy. Nutrition & Dietetics. 2013; 70:312-324.

Jager-Wittenaar H, Dijkstra PU, Dijkstra G, et al. High prevalence of cachexia in newly diagnosed head and neck cancer patients: An exploratory study. Nutrition. 2017; 35:114-118.

Jager-Wittenaar H, Ottery FD. Assessing nutritional status in cancer: role of the Patient-Generated Subjective Global Assessment. Curr Opin Clin Nutr Metab Care. 2017; 20:322-329.

Jensen GL, Cederholm T, Correia M, et al. GLIM Criteria for the Diagnosis of Malnutrition: A Consensus Report From the Global Clinical Nutrition Community. JPEN J Parenter Enteral Nutr. 2018; doi:10.1002/jpen.1440.

Jensen GL, Hsiao PY, Wheeler D. Adult nutrition assessment tutorial. JPEN J Parenter Enteral Nutr, 2012; 36:267-274.

Jensen GL, Mirtallo J, Compher C, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. JPEN J Parenter Enteral Nutr. 2010; 34:156-159.

Jie B, Jiang ZM, Nolan MT, Zhu SN, Yu K, Kondrup J. Impact of preoperative nutritional support on clinical outcome in abdominal surgical patients at nutritional risk. Nutrition. 2012; 28:1022-1027.

Johns N, Tan BH, MacMillan M, et al. Genetic basis of interindividual susceptibility to cancer cachexia: selection of potential candidate gene polymorphisms for association studies. J Genet 2014; 93:893-916.

Jones KI, Doleman B, Scott S, Lund JN, Williams JP. Simple psoas cross-sectional area measurement is a quick and easy method to assess sarcopenia and predicts major surgical complications. Colorectal Dis 2015; 17:O20-26.

Kalantar-Zadeh K, Horwich TB, Oreopoulos A, et al. Risk factor paradox in wasting diseases. Curr Opin Clin Nutr Metab Care, 2007; 10:433-442.

Kapoor N, Naufahu J, Tewfik S, Bhatnagar S, Garg R, Tewfik I. A Prospective Randomized Controlled Trial to Study the Impact of a Nutrition-Sensitive Intervention on Adult Women With Cancer Cachexia Undergoing Palliative Care in India. Integr Cancer Ther. 2017; 16:74-84.

Katakami N, Uchino J, Yokoyama T, et al. Anamorelin (ONO-7643) for the treatment of patients with non-small cell lung cancer and cachexia: Results from a randomized, double-blind, placebo-controlled, multicenter study of Japanese patients (ONO-7643-04). Cancer. 2018; 124:606-616.

Kazemi-Bajestani SM, Mazurak VC, Baracos V. Computed tomography-defined muscle and fat wasting are associated with cancer clinical outcomes. Semin Cell Dev Biol 2016; 54:2-10.

Keller HH, McCullough J, Davidson B, et al. The Integrated Nutrition Pathway for Acute Care (INPAC): Building consensus with a modified Delphi. Nutr J. 2015; 14:63.

Kimura M, Naito T, Kenmotsu H, Taira T, Wakuda K, Oyakawa T, et al. Prognostic impact of cancer cachexia in patients with advanced non-small cell lung cancer. Support Care Cancer 2015; 23:1699-708.

Kirk PS, Friedman JF, Cron DC, et al. One-year postoperative resource utilization in sarcopenic patients. J Surg Res 2015; 199:51-55.

Kubrak C, Olson K, Jha N, et al. Clinical determinants of weight loss in patients receiving radiation and chemoirradiation for head and neck cancer: a prospective longitudinal view. Head Neck, 2013; 35:695-703.

Kubrak K. Prevalence and prognostic significance of malnutrition in patients with cancers of the head and neck. University of Alberta; 2018, unpublished manuscript.

Kuchnia AJ, Teigen LM, Cole AJ, et al. Phase Angle and Impedance Ratio: Reference Cut-Points From the United States National Health and Nutrition Examination Survey 1999-2004 From Bioimpedance Spectroscopy Data. J Parenter Enteral Nutr 2017; 41:1310-1315.

Kuk JL, Saunders TJ, Davidson LE, Ross R. Age-related changes in total and regional fat distribution. Ageing Res Rev 2009; 8:339-348.

Laird BJ, Kaasa S, McMillan DC, et al. Prognostic factors in patients with advanced cancer: a comparison of clinicopathological factors and the development of an inflammation-based prognostic system. Clin Cancer Res. 2013; 19:5456-5464.

Laird BJ, McMillan DC, Fayers P, et al. The systemic inflammatory response and its relationship to pain and other symptoms in advanced cancer. Oncologist. 2013; 18:1050-1055.

Lam YY, Ravussin E. Analysis of energy metabolism in humans: A review of methodologies. Mol Metab. 2016; 5:1057-1071.

Laporte M, Keller HH, Payette H, et al. Validity and reliability of the new Canadian Nutrition Screening Tool in the 'real-world' hospital setting. Eur J Clin Nutr. 2015; 69:558-564.

Laporte, M. Comparison of nutrition screening tools for hospital use. Canadian Malnutrition Task Force. 2013, http://nutritioncareincanada.ca/hospital-use.pdf. Accessed June 27, 2018.

Lau CS, Chamberlain RS. Enhanced Recovery After Surgery Programs Improve Patient Outcomes and Recovery: A Meta-analysis. World J Surg. 2017; 41:899-913.

Lawrence DP, Kupelnick B, Miller K, Devine D, Lau J. Evidence Report on the Occurrence, Assessment, and Treatment of Fatigue in Cancer Patients. JNCI Monographs. 2004; 2004:40-50.

LeBlanc TW, Abernethy AP. Patient-reported outcomes in cancer care - hearing the patient voice at greater volume. Nat Rev Clin Oncol. 2017; 14:763-772.

LeBlanc TW, Nipp RD, Rushing CN, Samsa GP, Locke SC, Kamal AH, et al. Correlation between the international consensus definition of the Cancer Anorexia-Cachexia Syndrome (CACS) and patient-centered outcomes in advanced non-small cell lung cancer. J Pain Symptom Manage 2015; 49:680-9.

LeBlanc TW, Samsa GP, Wolf SP, Locke SC, Cella DF, Abernethy AP. Validation and realworld assessment of the Functional Assessment of Anorexia-Cachexia Therapy (FAACT) scale in patients with advanced non-small cell lung cancer and the cancer anorexia-cachexia syndrome (CACS). Support Care Cancer 2015; 23:2341-7.

Lee J, Taneja V, Vassallo R. Cigarette Smoking and Inflammation: Cellular and Molecular Mechanisms. J Dent Res. 2012; 91:142-149.

Lee JY, Hong N, Kim HR, et al. Effects of Serum Albumin, Calcium Levels, Cancer Stage and Performance Status on Weight Loss in Parathyroid Hormone-Related Peptide Positive or Negative Patients with Cancer. Endocrinol Metab (Seoul, Korea). 2018; 33:97-104.

Lee L, Mata J, Ghitulescu GA, Boutros M, et al. Cost-effectiveness of Enhanced Recovery Versus Conventional Perioperative Management for Colorectal Surgery. Ann Surg. 2015; 262:1026-1033.

Levolger S, van Vugt JL, de Bruin RW, JN IJ. Systematic review of sarcopenia in patients operated on for gastrointestinal and hepatopancreatobiliary malignancies. Br J Surg. 2015; 102:1448-1458.

Li L, Kalantar-Zadeh K. Obesity that makes kidney cancer more likely but helps fight it more strongly. J Natl Cancer Inst, 2013; 105:1848-1849.

Lieffers JR, Bathe OF, Fassbender K, Winget M, Baracos VE. Sarcopenia is associated with postoperative infection and delayed recovery from colorectal cancer resection surgery. Br J

Cancer 2012; 107:931-936.

Lieffers JR, Mourtzakis M, Hall KD, McCargar LJ, Prado CM, Baracos VE. A viscerally driven cachexia syndrome in patients with advanced colorectal cancer: contributions of organ and tumor mass to whole-body energy demands. Am J Clin Nutr. 2009; 89:1173-1179.

Ljungqvist O, Scott M, Fearon KC. Enhanced Recovery After Surgery: A Review. JAMA Surg. 2017; 152:292-298.

Ljungqvist O. ERAS - Enhanced Recovery After Surgery: moving evidence-based perioperative care to practice. JPEN J Parenter Enteral Nutr. 2014; 38:559-566.

Lodewick TM, van Nijnatten TJ, van Dam RM, et al. Are sarcopenia, obesity and sarcopenic obesity predictive of outcome in patients with colorectal liver metastases? HPB (Oxford) 2015; 17:438-446.

Lohsiriwat V, Pongsanguansuk W, Lertakyamanee N, Lohsiriwat D. Impact of metabolic syndrome on the short-term outcomes of colorectal cancer surgery. Dis Colon Rectum 2010; 53:186-191.

Ma C, Bandukwala S, Burman D, et al. Interconversion of three measures of performance status: an empirical analysis. Eur J Cancer, 2010; 46:3175-3183.

MacDonald N. Terminology in cancer cachexia: importance and status. Curr Opin Clin Nutr Metab Care. 2012; 15:220-225.

Maddocks M, Hopkinson J, Conibear J, Reeves A, Shaw C, Fearon KC. Practical multimodal care for cancer cachexia. Curr Opin Support Palliat Care. 2016; 10:298-305.

Mak KS, van Bommel ACM, Stowell C, et al. Defining a standard set of patient-centred outcomes for lung cancer. Eur Resp J. 2016; 48:852-860.

Malietzis G, Aziz O, Bagnall NM, Johns N, Fearon KC, Jenkins JT. The role of body composition evaluation by computerized tomography in determining colorectal cancer treatment outcomes: a systematic review. Eur J Surg Oncol 2015; 41:186-196.

Malietzis G, Currie AC, Athanasiou T, et al. Influence of body composition profile on outcomes following colorectal cancer surgery. Br J Surg 2016; 103:572-580.

Malietzis G, Johns N, Al-Hassi HO, et al. Low Muscularity and Myosteatosis Is Related to the Host Systemic Inflammatory Response in Patients Undergoing Surgery for Colorectal Cancer. Ann Surg. 2016; 263:320-5.

Margadant CC, Bruns ER, Sloothaak DA, et al. Lower muscle density is associated with major postoperative complications in older patients after surgery for colorectal cancer. Eur J Surg Oncol 2016; 42:1654-1659.

Martin L, Birdsell L, MacDonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol 2013; 31:1539-1547.

Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancerassociated weight loss. J Clin Oncol. 2015; 33:90-99.

Martin L, Watanabe S, Fainsinger R, et al. Prognostic factors in patients with advanced cancer: use of the patient-generated subjective global assessment in survival prediction. J Clin Oncol. 2010; 28:4376-4383.

Martin L. Diagnostic criteria for cancer cachexia: data versus dogma. Curr Opin Clin Nutr Metab Care 2016; 19:188-198.

Martin L, C. How much does reduced food intake contribute to cancer-associated weight loss? Curr Opin Support Palliat Care. 2018; 12.

Martin L, Laird B, Gagnon B, Chasen M, Baracos VE, Senesse P. Reduced food intake as an explanatory variable for cancer-associated weight loss (WL): Different clinical assessments of food intake provide equivalent results. University of Alberta; 2018, unpublished manuscript.

Martin RCG, Brennan MF, Jaques DP. Quality of Complication Reporting in the Surgical Literature. Ann Surg 2002; 235:803-813.

McMillan DC. The systemic inflammation-based Glasgow Prognostic Score: a decade of experience in patients with cancer. Cancer Treat Rev 2013; 39:534-40.

Mei KL, Batsis JA, Mills JB, Holubar SD. Sarcopenia and sarcopenic obesity: do they predict inferior oncologic outcomes after gastrointestinal cancer surgery? Perioper Med (Lond) 2016; 5:30.

Messenger DE, Curtis NJ, Jones A, Jones EL, Smart NJ, Francis NK. Factors predicting outcome from enhanced recovery programmes in laparoscopic colorectal surgery: a systematic review. Surg Endosc. 2016; 31:2050-2071.

Mislang AR, Di Donato S, Hubbard J, et al. Nutritional management of older adults with gastrointestinal cancers: An International Society of Geriatric Oncology (SIOG) review paper. J Geriatr Oncol. 2018; 9:382-392.

Miyamoto Y, Baba Y, Sakamoto Y, et al. Negative impact of skeletal muscle loss after systemic chemotherapy in patients with unresectable colorectal cancer. PLoS One 2015; 10:e0129742.

Mohan A, Poulose R, Kulshreshtha I, et al. High prevalence of malnutrition and deranged relationship between energy demands and food intake in advanced non-small cell lung cancer. Eur J Cancer Care (Engl). 2017; 26: doi: 10.1111/ecc.12503.

Moon HG, Ju YT, Jeong CY, et al. Visceral obesity may affect oncologic outcome in patients with colorectal cancer. Ann Surg Oncol 2008; 15:1918-1922.

Morgans AK, van Bommel AC, Stowell C, et al. Development of a Standardized Set of Patientcentered Outcomes for Advanced Prostate Cancer: An International Effort for a Unified Approach. Eur Urol. 2015; 68:891-898.

Mourtzakis M, Prado CM, Lieffers JR, Reiman T, McCargar LJ, Baracos VE. A practical and precise approach to quantification of body composition in cancer patients using computed tomography images acquired during routine care. Appl Physiol Nutr Metab 2008; 33:997-1006.

Muscaritoli M, Lucia S, Farcomeni A, et al. Prevalence of malnutrition in patients at first medical oncology visit: the PreMiO study. Oncotarget. 2017; 8:79884-79896.

Myles PS, Grocott MP, Boney O, et al. Standardizing end points in perioperative trials: towards a core and extended outcome set. Br J Anaesth. 2016; 116:586-589.

Mytelka DS, Li L, Benoit K. Post-diagnosis weight loss as a prognostic factor in non-small cell lung cancer. J Cachexia Sarcopenia Muscle. 2018; 9:86-92.

Nakanishi R, Oki E, Sasaki S, et al. Sarcopenia is an independent predictor of complications after colorectal cancer surgery. Surg Today 2017; 48:151-157.

Nasrah R, Kanbalian M, Van Der Borch C, Swinton N, Wing S, Jagoe RT. Defining the role of dietary intake in determining weight change in patients with cancer cachexia. Clin Nutr. 2018; 37:235-241.

National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE) v.4.03. 2009; https://www.eortc.be/services/doc/ctc/CTCAE_4.03_2010-06-14_QuickReference_5x7.pdf

National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE) v.5.0. 2017.

https://ctep.cancer.gov/protocolDevelopment/electronic_applications/docs/CTCAE_v5_Quick_ Reference_8.5x11.pdf, Accessed: October 11, 2018.

National Institute of Academic Anesthesia Health Services Research Center. Core Outcome Measures in Perioperative and Anaesthetic Care (COMPAC). 2018; https://www.niaa-hsrc.org.uk/HSRC-COMPAC accessed: October 11, 2018.

Navarro DA, Boaz M, Krause I, et al. Improved meal presentation increases food intake and decreases readmission rate in hospitalized patients. Clin Nutr. 2016; 35:1153-1158.

Nelson G, Kiyang LN, Chuck A, Thanh NX, Gramlich LM. Cost impact analysis of Enhanced Recovery After Surgery program implementation in Alberta colon cancer patients. Curr Oncol.

2016; 23:e221-227.

Nelson G, Kiyang LN, Crumley ET, et al. Implementation of Enhanced Recovery After Surgery (ERAS) Across a Provincial Healthcare System: The ERAS Alberta Colorectal Surgery Experience. World J Surg 2016; 40:1092-1103.

Nguyen D, El-Serag H. The Epidemiology of Obesity. Gastroenterol Clin North Am, 2010; 39:1-7.

Nguyen TY, Batterham MJ, Edwards C. Comparison of resting energy expenditure between cancer subjects and healthy controls: a meta-analysis. Nutr Cancer. 2016; 68:374-387.

Nicolo M, Compher CW, Still C, Huseini M, Dayton S, Jensen GL. Feasibility of accessing data in hospitalized patients to support diagnosis of malnutrition by the Academy-A.S.P.E.N. malnutrition consensus recommended clinical characteristics. JPEN J Parenter Enteral Nutr. 2014; 38:954-959.

nutritionDay Worldwide. 2018; https://www.nutritionday.org/. Accessed October 7, 2018.

Ong WL, Schouwenburg MG, van Bommel ACM, et al. A Standard Set of Value-Based Patient-Centered Outcomes for Breast Cancer: The International Consortium for Health Outcomes Measurement (ICHOM) Initiative. JAMA Oncol. 2017; 3:677-685.

Ottery FD. Definition of standardized nutritional assessment and interventional pathways in oncology. Nutrition. 1996; 12:S15-19.

Ouchi A, Asano M, Aono K, Watanabe T, Oya S. Laparoscopic Colorectal Resection in Patients with Sarcopenia: A Retrospective Case-Control Study. J Laparoendosc Adv Surg Tech A 2016; 26:366-370.

Ozoya OO, Siegel EM, Srikumar T, Bloomer AM, DeRenzis A, Shibata DA-OhooX. Quantitative Assessment of Visceral Obesity and Postoperative Colon Cancer Outcomes. J Gastrointest Surg 2017; 21:534-542.

Park BK, Park JW, Ryoo SB, Jeong SY, Park KJ, Park JG. Effect of Visceral Obesity on Surgical Outcomes of Patients Undergoing Laparoscopic Colorectal Surgery. World J Surg 2015; 39:2343-2353.

Parmar MP, Swanson T, Jagoe RT. Weight changes correlate with alterations in subjective physical function in advanced cancer patients referred to a specialized nutrition and rehabilitation team. Support Care Cancer. 2013; 21:2049-2057.

Parmar MP, Vanderbyl BL, Kanbalian M, Windholz TY, Tran A-T, Jagoe RT. A multidisciplinary rehabilitation programme for cancer cachexia improves quality of life. BMJ Support Palliat Care. 2017; 7:441-449.

Pedziwiatr M, Pisarska M, Major P, et al. Laparoscopic colorectal cancer surgery combined with enhanced recovery after surgery protocol (ERAS) reduces the negative impact of sarcopenia on short-term outcomes. Eur J Surg Oncol 2016; 42:779-787.

Pencina MJ, D'Agostino RB. Overall C as a measure of discrimination in survival analysis: model specific population value and confidence interval estimation. Stat Med, 2004; 23:2109-2123.

Perry GS, Byers TE, Mokdad AH, et al. The validity of self-reports of past body weights by U.S. adults. Epidemiology, 1995; 6:61-66.

Planchard D, Popat S, Kerr K, et al. Metastatic non-small cell lung cancer: ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann Oncol. 2018; 29:iv192-iv237.

Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008; 9:629-635.

Psutka SP, Boorjian SA, Moynagh MR, et al. Decreased skeletal muscle mass is associated with an increased risk of mortality after radical nephrectomy for localized renal cell cancer. J Urol 2016; 195:270-6.

Purcell SA, Elliott SA, Ryan AM, Sawyer MB, Prado CM. Accuracy of a portable indirect calorimeter for measuring resting energy expenditure in individuals with cancer. JPEN J Parenter Enteral Nutr. 2018; doi.10.1002/jpen.1310.

Ramchandran K, Tribett E, Dietrich B, Von Roenn J. Integrating Palliative Care Into Oncology: A Way Forward. Cancer Control. 2015; 22:386-395.

Reisinger KW, Derikx JP, van Vugt JL, et al. Sarcopenia is associated with an increased inflammatory response to surgery in colorectal cancer. Clin Nutr 2016; 35:924-7.

Rolland Y, Czerwinski S, Abellan Van Kan, et al. Sarcopenia: its assessment, etiology, pathogenesis, consequences and future perspectives. J Nutr Health Aging 2008; 12:433-450.

Rollins KE, Tewari N, Ackner A, et al. The impact of sarcopenia and myosteatosis on outcomes of unresectable pancreatic cancer or distal cholangiocarcinoma. Clin Nutr 2016;35:1103-9.

Rubenstein LZ, Harker JO, Salva A, Guigoz Y, Vellas B. Screening for undernutrition in geriatric practice: developing the short-form mini-nutritional assessment (MNA-SF). J Gerontol A Bio Sci Med Sci. 2001; 56:M366-372.

Rutten IJG, Ubachs J, Kruitwagen R, Beets-Tan RGH, Olde Damink SWM, Van Gorp T. Psoas muscle area is not representative of total skeletal muscle area in the assessment of sarcopenia in ovarian cancer. J Cachexia Sarcopenia Muscle 2017; 8:630-638.

Sabel MS, Terjimanian M, Conlon AS, et al. Analytic morphometric assessment of patients undergoing colectomy for colon cancer. J Surg Oncol 2013; 108:169-175.

Salsman JM, Beaumont JL, Wortman K, Yan Y, Friend J, Cella D. Brief versions of the FACIT-fatigue and FAACT subscales for patients with non-small cell lung cancer cachexia. Support Care Cancer 2015; 23:1355-64.

Sandrucci S, Beets G, Braga M, Dejong K, Demartines N. Perioperative nutrition and enhanced recovery after surgery in gastrointestinal cancer patients. A position paper by the ESSO Task Force in collaboration with the ERAS Society (ERAS Coalition). European Journal of Surgical Oncology. 2018;10.1016/j.ejso.2017.12.010.

Schiessel DL, Baracos VE. Barriers to cancer nutrition therapy: excess catabolism of muscle and adipose tissues induced by tumour products and chemotherapy. Proc Nutr Soc. 2018; doi:10.1017/s0029665118000186:1-9.

Schindler K, Pernicka E, Laviano A, et al. How nutritional risk is assessed and managed in European hospitals: a survey of 21,007 patients findings from the 2007-2008 cross-sectional nutritionDay survey. Clin Nutr. 2010; 29:552-559.

Schindler K, Themessl-Huber M, Hiesmayr M, et al. To eat or not to eat? Indicators for reduced food intake in 91,245 patients hospitalized on nutritionDays 2006-2014 in 56 countries worldwide: a descriptive analysis. Am J Clin Nutr. 2016; 104:1393-1402.

Schlomer BJ, Copp HL. Secondary Data Analysis of Large Data Sets in Urology: Successes and Errors to Avoid. J Urol. 2014; 191:587-596.

Scott SI, Farid S, Mann C, Jones R, Kang P, Evans J. Abdominal fat ratio - a novel parameter for predicting conversion in laparoscopic colorectal surgery. Ann R Coll Surg Engl 2017; 99:46-50.

Senesse P, Isambert A, Janiszewski C, et al. Management of cancer cachexia and guidelines implementation in a comprehensive cancer center: a physician-led cancer nutrition program adapted to the practices of a country. Journal of pain and symptom management. 2017; doi:10.1016/j.jpainsymman.2017.01.010.

Shuster A, Patlas M, Pinthus JH, Mourtzakis M. The clinical importance of visceral adiposity: a critical review of methods for visceral adipose tissue analysis. Br J Radiol 2012; 85:1-10.

Simmons CP, Koinis F, Fallon MT, et al. Prognosis in advanced lung cancer--A prospective study examining key clinicopathological factors. Lung Cancer. 2015; 88:304-309.

Simonsen C, de Heer P, Bjerre ED, et al. Sarcopenia and Postoperative Complication Risk in Gastrointestinal Surgical Oncology: A Meta-analysis. Ann Surg, 2018; doi: 10.1097/SLA.00000000002679.

Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapyinduced haematological toxicity in advanced non-small cell lung cancer. Lung Cancer. 2015; 90:85-91.

Solheim TS, Blum D, Fayers PM, et al. Weight loss, appetite loss and food intake in cancer patients with cancer cachexia: three peas in a pod? - analysis from a multicenter cross sectional study. Acta Oncol. 2014; 53:539-546.

Sommerbakk R, Haugen DF, Tjora A, Kaasa S, Hjermstad MJ. Barriers to and facilitators for implementing quality improvements in palliative care - results from a qualitative interview study in Norway. BMC Palliat Care. 2016; 15:61.

Southwest Oncology Group Clinical Research Network. 2018; https://www.swog.org/. Accessed October 8, 2018.

Spiro A, Baldwin C, Patterson A, et al. The views and practice of oncologists towards nutritional support in patients receiving chemotherapy. Br J Cancer, 2006; 95:431-434.

Steenhagen E. Enhanced Recovery After Surgery: It's time to change practice! Nutr in Clin Pract. 2016; 31:18-29.

Strasser F, Lutz TA, Maeder MT, et al. Safety, tolerability and pharmacokinetics of intravenous ghrelin for cancer-related anorexia/cachexia: a randomised, placebo-controlled, double-blind, double-crossover study. Br J Cancer, 2008; 98:300-308.

Streicher M, Themessl-Huber M, Schindler K, Sieber CC, Hiesmayr M, Volkert D. nutritionDay in Nursing Homes-The Association of Nutritional Intake and Nutritional Interventions With 6-Month Mortality in Malnourished Residents. J Am Med Dir Assoc. 2017; 18:162-168.

Tamandl D, Paireder M, Asari R, Baltzer PA, Schoppmann SF, Ba-Ssalamah A. Markers of sarcopenia quantified by computed tomography predict adverse long-term outcome in patients with resected oesophageal or gastro-oesophageal junction cancer. Eur Radiol 2016; 26:1359-67.

Tao W, Lagergren J. Clinical management of obese patients with cancer. Nat Rev Clin Oncol, 2013; 10:519-533.

Tappenden KA, Quatrara B, Parkhurst ML, Malone AM, Fanjiang G, Ziegler TR. Critical role of nutrition in improving quality of care: an interdisciplinary call to action to address adult hospital malnutrition. J Acad Nutr Diet. 2013; 113:1219-1237.

Tegels JJ, van Vugt JL, Reisinger KW, Hulsewe KW, Hoofwijk AG, Derikx JP, et al. Sarcopenia is highly prevalent in patients undergoing surgery for gastric cancer but not associated with worse outcomes. J Surg Oncol 2015; 112:403-7.

Temel JS, Abernethy AP, Currow DC, et al. Anamorelin in patients with non-small-cell lung

cancer and cachexia (ROMANA 1 and ROMANA 2): results from two randomized, doubleblind, phase 3 trials. Lancet Oncol. 2016; 17:519-531.

Temel JS, Greer JA, Muzikansky A, et al. Early palliative care for patients with metastatic nonsmall-cell lung cancer. N Engl J Med. 2010; 363:733-742.

Thibault R, Goujon N, Le Gallic E, et al. Use of 10-point analogue scales to estimate dietary intake: a prospective study in patients nutritionally at-risk. Clin Nutr. 2009; 28:134-140.

Thibault R, Pichard C. The evaluation of body composition: a useful tool for clinical practice. Ann Nutr Metab. 2012; 60:6-16.

Thompson KL, Elliott L, Fuchs-Tarlovsky V, Levin RM, Voss AC, Piemonte T. Oncology Evidence-Based Nutrition Practice Guideline for Adults. J Acad Nutr Diet. 2017; 117:297-310 e247.

Thoresen L, Fjeldstad I, Krogstad K, et al. Nutritional status of patients with advanced cancer: the value of using the subjective global assessment of nutritional status as a screening tool. Palliat Med, 2002; 16:33-42.

Thoresen L, Frykholm G, Lydersen S, et al. Nutritional status, cachexia and survival in patients with advanced colorectal carcinoma. Different assessment criteria for nutritional status provide unequal results. Clin Nutr. 2013; 32:65-72.

Thoresen L, Frykholm G, Lydersen S, et al. The association of nutritional assessment criteria with health-related quality of life in patients with advanced colorectal carcinoma. Eur J Cancer Care (Engl), 2012; 21:505-516.

Tsai VWW, Husaini Y, Sainsbury A, Brown DA, Breit SN. The MIC-1/GDF15-GFRAL Pathway in Energy Homeostasis: Implications for Obesity, Cachexia, and Other Associated Diseases. Cell Metab. 2018; 28:353-368.

Uster A, Ruehlin M, Mey S, et al. Effects of nutrition and physical exercise intervention in palliative cancer patients: A randomized controlled trial. Clin Nutr. 2018; 37:1202-1209.

Vagnildhaug OM, Balstad TR, Almberg SS, et al. A cross-sectional study examining the prevalence of cachexia and areas of unmet need in patients with cancer. Support Care Cancer. 2018; 26:1871-1880.

Vagnildhaug OM, Blum D, Wilcock A, et al. The applicability of a weight loss grading system in cancer cachexia: a longitudinal analysis. J Cachexia Sarcopenia Muscle. 2017; 8:789-797.

van der Kroft G, Bours D, Janssen-Heijnen DM, van Berlo D, Konsten D. Value of sarcopenia assessed by computed tomography for the prediction of postoperative morbidity following oncological colorectal resection: A comparison with the malnutrition screening tool. Clin Nutr 2018; 24:114-119.

van der Werf A, Arthey K, Hiesmayr M, et al. The determinants of reduced dietary intake in hospitalised colorectal cancer patients. Support Care Cancer. 2018; 26:2039-2047.

van der Werf A, Dekker IM, Meijerink MR, Wierdsma NJ, de van der Schueren MAE, Langius JAE. Skeletal muscle analyses: agreement between non-contrast and contrast CT scan measurements of skeletal muscle area and mean muscle attenuation. Clin Physiol Funct Imaging. 2017;38:366-372.

van Vledder MG, Levolger S, Ayez N, Verhoef C, Tran TC, Ijzermans JN. Body composition and outcome in patients undergoing resection of colorectal liver metastases. Br J Surg 2012; 99:550-557.

van Vugt JL, Braam HJ, van Oudheusden TR, et al. Skeletal muscle depletion is associated with severe postoperative complications in patients undergoing cytoreductive surgery with hyperthermic intraperitoneal chemotherapy for peritoneal carcinomatosis of colorectal cancer. Ann Surg Oncol 2015; 22:3625-31.

van Vugt JL, Levolger S, de Bruin RW, van RJ, Metselaar HJ, Ijzermans JN. Systematic review and meta-analysis of the impact of computed tomography assessed skeletal muscle mass on outcome in patients awaiting or undergoing liver transplantation. Am J Transplant. 2016; 16:2277-2292.

van Vugt JLA, Coebergh van den Braak RRJ, Schippers HJW, et al. Contrast-enhancement influences skeletal muscle density, but not skeletal muscle mass, measurements on computed tomography. Clin Nutr 2017; S0261-5614:30246-7.

Vazeille C, Jouinot A, Durand JP, et al. Relation between hypermetabolism, cachexia, and survival in cancer patients: a prospective study in 390 cancer patients before initiation of anticancer therapy. Am J Clin Nutr. 2017; 105:1139-1147.

Vellas B, Villars H, Abellan G, et al. Overview of the MNA - Its history and challenges. J Nutr Health Aging. 2006; 10:456-635.

Vigano AAL, Morais JA, Ciutto L, et al. Use of routinely available clinical, nutritional, and functional criteria to classify cachexia in advanced cancer patients. Clin Nutr. 2017; 36:1378-1390.

Visioni A, Shah R, Gabriel E, Attwood K, Kukar M, Nurkin S. Enhanced Recovery After Surgery for Noncolorectal Surgery?: A Systematic Review and Meta-analysis of Major Abdominal Surgery. Ann Surg. 2018; 267:57-65.

Wallengren O, Bosaeus I, Lundholm K. Dietary energy density, inflammation and energy balance in palliative care cancer patients. Clin Nutr, 2013; 32:88-92.

Wallengren O, Iresjo BM, Lundholm K, Bosaeus I. Loss of muscle mass in the end of life in

patients with advanced cancer. Support Care Cancer 2015; 23:79-86.

Wallengren O, Lundholm K, Bosaeus I. Diagnostic criteria of cancer cachexia: relation to quality of life, exercise capacity and survival in unselected palliative care patients. Support Care Cancer. 2013; 21:1569-1577.

Weaver BW, K.L. SPSS and SAS programs for comparing Pearson correlations and OLS regression coefficients. Behav Res Methods 2013; 45:880-895.

Weimann A, Braga M, Carli F, et al. ESPEN guideline: Clinical nutrition in surgery. Clin Nutr. 2017; 36:623-650.

Wesseltoft-Rao N, Hjermstad MJ, Ikdahl T, et al. Comparing two classifications of cancer cachexia and their association with survival in patients with unresected pancreatic cancer. Nutr Cancer. 2015; 67:472-480.

Wigmore SJ, Barber MD, Ross JA, Tisdale MJ, Fearon KC. Effect of oral eicosapentaenoic acid on weight loss in patients with pancreatic cancer. Nutr Cancer. 2000; 36:177-184.

Williams JD, Wischmeyer PE. Assessment of perioperative nutrition practices and attitudes-A national survey of colorectal and GI surgical oncology programs. Am J Surg. 2017; 213:1010-1018.

Wischmeyer PE, Carli F, Evans DC, et al. American Society for Enhanced Recovery and Perioperative Quality Initiative Joint Consensus Statement on Nutrition Screening and Therapy Within a Surgical Enhanced Recovery Pathway. Anesth Analg. 2018;126(6):1883-1895.

Wolin KY, Carson K, Colditz GA. Obesity and cancer. Oncologist. 2010; 15:556-565.

Womer AL, Brady JT, Kalisz K, et al. Do psoas muscle area and volume correlate with postoperative complications in patients undergoing rectal cancer resection? Am J Surg 2018; 215:503-506.

World Health Organization. Annex 2: Recommendations for the preparation, characterization and establishment of international and other biological reference standards (revised 2004). WHO Technical Report Series. 2006; 932:73-131.

World Health Organization. International Statistical Classification of Diseases and Related Health Problems 10th revision. 2018; https://icd.who.int/browse10/2016/en, Accessed October 12, 2018.

Xiao J, Caan BJ, Weltzien E, et al. Associations of pre-existing co-morbidities with skeletal muscle mass and radiodensity in patients with non-metastatic colorectal cancer. J Cachexia Sarcopenia Muscle 2018: doi:10.1002/jcsm.12301

Yip C, Dinkel C, Mahajan A, Siddique M, Cook GJ, Goh V. Imaging body composition in
cancer patients: visceral obesity, sarcopenia and sarcopenic obesity may impact on clinical outcome. Insights Imaging 2015; 6:489-497.

Yu H, Joh YG, Son GM, Kim HS, Jo HJ, Kim HY. Distribution and Impact of the Visceral Fat Area in Patients With Colorectal Cancer. Ann Coloproctol 2016; 32:20-26.

Zerillo JA, Schouwenburg MG, van Bommel ACM, et al. An International Collaborative Standardizing a Comprehensive Patient-Centered Outcomes Measurement Set for Colorectal Cancer. JAMA Oncol. 2017; 3:686-694.

Appendix I

Chapter 2 Supplementary Table 1. Predictive value of BMI and WL as continuous variables in a multivariable Cox proportional hazard model applied to advanced cancer stages only (stages III+IV).

	B	SE	Wald	df	P-value	HR	95.0% C	I for HR
							Lower	Upper
BMI	010	.003	11.885	1	.001	.990	.984	.996
%WL	015	.002	64.564	1	.000	.985	.981	.989
Sex								
Male								
Female	044	.030	2.086	1	.149	.957	.902	1.016
Age	.006	.001	21.687	1	.000	1.006	1.003	1.008
Cancer Diagnosis			545.429	11	.000			
colorectal								
breast	.361	.089	16.582	1	.000	1.434	1.206	1.706
gastro-esophageal	.637	.057	124.677	1	.000	1.891	1.691	2.115
genitourinary	.569	.077	54.269	1	.000	1.766	1.518	2.055
Hean & neck	428	.066	41.882	1	.000	.652	.573	.742
other	.365	.080	21.091	1	.000	1.441	1.233	1.684
other GI	.718	.085	72.112	1	.000	2.050	1.737	2.419
pancreas	.765	.067	132.426	1	.000	2.150	1.887	2.449
respiratory	.637	.046	194.634	1	.000	1.890	1.729	2.067
unknown primary	.663	.123	29.021	1	.000	1.940	1.525	2.470
hematological	.062	.111	.311	1	.577	1.064	.856	1.322
liver/intrahepatic bile ducts	.987	.132	55.670	1	.000	2.683	2.070	3.476
Performance Status			798.326	4	.000			
0								
1	.287	.050	32.292	1	.000	1.332	1.207	1.471
2	.730	.055	177.605	1	.000	2.076	1.864	2.311
3	1.149	.055	437.338	1	.000	3.155	2.833	3.514
4	1.727	.093	344.410	1	.000	5.622	4.685	6.747

B, beta coefficient; BMI, body mass index; df, degrees of freedom; GI, gastrointestinal; HR, hazard ratio; SE, standard error; %WL, percent weight loss.

Appendix II

	В	SE	Wald	df	P-value	HR	95.0% C	I for HR
							Lower	Upper
BMI adjusted WL Grades			144.225	4	.000			
0								
1	.118	.062	3.670	1	.055	1.125	.997	1.270
2	.204	.057	12.621	1	.000	1.226	1.096	1.372
3	.323	.051	39.507	1	.000	1.381	1.249	1.527
4	.541	.052	108.573	1	.000	1.717	1.551	1.901
Sex								
Male						1.000		
Female	058	.030	3.607	1	.058	.944	.890	1.002
Age	.006	.001	22.239	1	.000	1.006	1.003	1.008
Cancer Diagnosis			544.858	11	.000			
colorectal								
breast	.373	.089	17.697	1	.000	1.452	1.220	1.727
gastro-esophageal	.641	.057	126.234	1	.000	1.899	1.698	2.123
genitourinary	.571	.077	54.473	1	.000	1.769	1.520	2.059
head & neck	410	.066	38.387	1	.000	.664	.583	.756
other	.363	.080	20.737	1	.000	1.437	1.230	1.680
other GI	.713	.085	71.159	1	.000	2.040	1.728	2.407
pancreas	.774	.066	135.918	1	.000	2.168	1.904	2.469
respiratory	.646	.046	200.856	1	.000	1.908	1.745	2.086
unknown primary	.690	.123	31.360	1	.000	1.994	1.566	2.538
hematological	.073	.111	.433	1	.510	1.076	.866	1.337
Liver/intrahepatic bile ducts	.994	.132	56.440	1	.000	2.701	2.084	3.500
Performance Status			780.587	4	.000			
0								
1	.269	.051	28.344	1	.000	1.309	1.186	1.446
2	.706	.055	166.165	1	.000	2.025	1.819	2.254
3	1.126	.055	421.047	1	.000	3.085	2.770	3.435
4	1.694	.093	330.672	1	.000	5.443	4.535	6.534

Chapter 2 Supplementary Table 2. Predictive value of BMI adjusted WL Grades in a multivariable Cox proportional hazard model applied to advanced stages only (stage III + IV)

B, beta coefficient; BMI, body mass index; df, degrees of freedom; GI, gastrointestinal; HR, hazard ratio; SE, standard error; %WL, percent weight loss.

Appendix III

Chapter 5 Supplementary Table 1. Univariable multinomial logistic regression analysis for the association of selected variables to cancer-associated weight loss for three different models based on food intake assessments from three data sets: PG-SGA, I-VVAS, and the MNA

PG-SGA MLR	k model					ι	J nivariab	le MLR	Analysis				
		W	L 2.5-5.9	9%	W	L 6.0-10.	9%	W	L 11.0-14.	9%		WL ≥15.0	0%
	Ν	N=	1139 (28	.8%)	N=	1347 (21.	9%)	N	= 816 (13.2	%)	N	= 1082 (17	.6%)
Variables in model	(%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P-Value
Current Food Intake													
severely reduced*	1284 (20.8)	1.19 (0.13)	3.28 (2.53- 4.24)	< 0.001	2.08 (0.12)	8.02 (6.30- 10.21)	< 0.001	2.86 (0.14)	17.41 (13.14- 23.07)	<0.001	3.39 (0.14)	29.79 (22.69- 39.13)	< 0.001
normal food, reduced amount	2765 (44.9)	0.83 (0.08)	2.29 (1.95- 2.69)	<0.001	1.39 (0.08)	4.02 (3.41- 4.74)	<0.001	1.86 (0.11)	6.43 (5.15- 8.02)	<0.001	2.10 (0.11)	8.19 (6.57- 10.21)	<0.001
normal food, normal amount	2112 (34.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis*													
respiratory	1913 (31.1)	0.53 (0.10)	1.70 (1.40- 2.08)	<0.001	0.80 (0.11)	2.23 (1.81- 2.74)	<0.001	1.14 (0.14)	3.12 (2.38- 4.08)	<0.001	1.14 (0.13)	3.12 (2.42- 4.03)	<0.001
other	712 (11.6)	0.24 (0.14)	1.27 (0.97- 1.67)	0.078	0.72 (0.13)	2.06 (1.59- 2.67)	<0.001	0.91 (0.17)	2.49 (1.78- 3.49)	<0.001	1.19 (0.16)	3.27 (2.41- 4.43)	<0.001
genitourinary	386 (6.3)	0.61 (0.18)	1.83 (1.28- 2.62)	0.001	1.20 (0.17)	3.30 (2.36- 4.63)	<0.001	1.39 (0.21)	4.02 (2.66- 6.08)	< 0.001	1.88 (0.19)	6.58 (4.58- 9.46)	<0.001
upper GI	950 (15.4)	0.92 (0.15)	2.52 (1.89- 3.36)	< 0.001	1.65 (0.14)	5.20 (3.96- 6.84)	< 0.001	2.23 (0.16)	9.27 (6.72- 12.79)	< 0.001	2.65 (0.15)	14.10 (10.47- 19.00)	< 0.001

lower GI	1009 (16.4)	0.86 (0.13)	2.37 (1.86- 3.03)	< 0.001	1.35 (0.12)	3.84 (3.02- 4.90)	< 0.001	1.59 (0.16)	4.90 (3.61- 6.66)	< 0.001	1.49 (0.15)	4.42 (3.29- 5.94)	<0.001
head & neck	1191 (19.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	1483 (24.1)	0.81 (0.12)	2.24 (1.77- 2.84)	< 0.001	1.68 (0.12)	5.39 (4.28- 6.78)	< 0.001	2.38 (0.14)	10.84 (8.17- 14.37)	< 0.001	3.28 (0.16)	26.60 (19.42- 36.44)	<0.001
PS 2	1181 (19.2)	0.73 (0.12)	2.07 (1.64- 2.62)	< 0.001	1.55 (0.12)	4.69 (3.72- 5.90)	<0.001	1.95 (0.15)	7.00 (5.23- 9.39)	< 0.001	2.75 (0.16)	15.63 (11.32- 21.57)	<0.001
PS 1	2110 (34.2)	0.60 (0.09)	1.82 (1.52- 2.19)	< 0.001	1.04 (0.10)	2.84 (2.34- 3.45)	< 0.001	1.22 (0.14)	3.40 (2.61- 4.43)	<0.001	1.86 (0.15)	6.43 (4.75- 8.71)	<0.001
PS 0	1387 (22.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	4580 (74.3)	0.47 (0.11)	1.60 (1.29- 1.98)	<0.001	0.89 (0.11)	2.44 (1.95- 3.06)	<0.001	1.06 (0.14)	2.89 (2.18- 3.84)	<0.001	1.51 (0.15)	4.54 (3.39- 6.08)	<0.001
stage 3	856 (13.9)	0.47 (0.14)	1.60 (1.23- 2.10)	0.001	0.64 (0.14)	1.90 (1.44- 2.51)	<0.001	0.73 (0.18)	2.08 (1.47- 2.95)	<0.001	0.82 (0.18)	2.27 (1.59- 3.25)	<0.001
stage 1-2	725 (11.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
female	2558 (41.5)	-0.02 (0.08)	0.98 (0.84- 1.14)	0.788	0.12 (0.07)	1.12 (0.97- 1.30)	0.110	0.03 (0.09)	1.03 (0.87- 1.22)	0.720	0.02 (0.08)	1.02 (0.88- 1.19)	0.787
male	3603 (58.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	6161 (100)	0.002 (0.00)	1.00 (0.99- 1.00)	0.554	- 0.004 (0.00)	1.00 (0.99- 1.00)	0.223	0.001 (0.00)	1.00 (0.99- 1.01)	0.757	-0.01 (0.00)	0.99 (0.98- 1.00)	0.004
BMI Categories (kg/m2)													

<20.0	935 (15.2)	0.50 (0.14)	1.65 (1.24- 2.19)	0.001	1.10 (0.13)	3.00 (2.31- 3.90)	<0.001	1.64 (0.15)	5.14 (3.84- 6.88)	<0.001	2.56 (0.14)	12.97 (9.85- 17.08)	<0.001
20.0 to 21.9	863 (14.0)	0.26 (0.13)	1.29 (1.00- 1.68)	0.054	0.92 (0.12)	2.52 (1.99- 3.20)	< 0.001	1.16 (0.14)	3.20 (2.41- 4.25)	<0.001	1.62 (0.14)	5.05 (3.83- 6.66)	<0.001
22.0 to 24.9	1540 (25.0)	0.45 (0.10)	1.56 (1.28- 1.91)	< 0.001	0.74 (0.10)	2.10 (1.72- 2.57)	< 0.001	0.99 (0.12)	2.69 (2.11- 3.43)	<0.001	1.25 (0.13)	3.48 (2.72- 4.46)	<0.001
25.0 to 27.9	1289 (20.9)	0.26 (0.10)	1.30 (1.06- 1.59)	0.011	0.43 (0.10)	1.54 (1.25- 1.89)	< 0.001	0.39 (0.13)	1.47 (1.13- 1.92)	0.004	0.69 (0.13)	1.99 (1.53- 2.59)	<0.001
≥28.0	1534		1.0			1.0			1.0 (ref)			1.0	
<i>I</i> -VVAS MLR model	(24.9)		(IeI)			Univ	ariable M	LR Ana	lysis			(lel)	<u> </u>
		W	L 2.5-5.9	9%	W	L 6.0-10.	9%	W	L 11.0-14.9	9%		WL ≥15.	0%
	Ν	N=	655 (20.	6%)	N=	= 818 (25.7	7%)	N	= 376 (11.8	%)	1	N= 409 (12	.8%)
variables in model	(%)	β (SE)	OR (95% CD	P- Value	β (SE)	OR (95% CD	P- Value	β (SE)	OR (95% CD	P- Value	β (SE)	OR (95% CD	P-Value
Current Food Intake												01)	
severely reduced intake (scores 1-4)	736 (23.1)	1.02 (0.16)	2.76 (2.01- 3.79)	<0.001	1.76 (0.15)	5.83 (4.35- 7.80)	<0.001	2.58 (0.19)	13.23 (9.18- 19.07)	<0.001	3.33 (0.20)	28.04 (18.95- 41.48)	<0.001
moderately reduced intake (scores 5-9)	1256 (39.4)	0.90 (0.11)	2.46 (1.97- 3.08)	<0.001	1.28 (0.11)	3.61 (2.90- 4.49)	<0.001	1.71 (0.16)	5.53 (4.03- 7.58)	<0.001	2.11 (0.18)	8.24 (5.76- 11.78)	<0.001
normal (score 10)	1193		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis						(101)						(101)	
U	262	0.32	1.06		0.30	0.88		0.83	0.99		0.84	0.68	

other	514 (16.1)	-0.04 (0.17)	0.74 (0.49- 1.11)	0.151	0.01 (0.16)	0.66 (0.46- 0.95)	0.026	0.57 (0.23)	0.76 (0.48- 1.22)	0.259	0.39 (0.24)	0.44 (0.28- 0.69)	<0.001
genitourinary	478 (15.0)	0.45 (0.17)	1.21 (0.81- 1.82)	0.354	0.25 (0.17)	0.84 (0.58- 1.23)	0.379	0.51 (0.24)	0.72 (0.43- 1.18)	0.190	0.76 (0.24)	0.63 (0.40- 0.99)	0.046
upper GI	519 (16.3)	0.95 (0.19)	1.99 (1.29- 3.07)	0.002	0.88 (0.18)	1.58 (1.06- 2.36)	0.025	1.69 (0.23)	2.33 (1.44- 3.77)	0.001	2.05 (0.23)	2.28 (1.47- 3.53)	< 0.001
lower GI	433 (13.6)	0.16 (0.18)	0.91 (0.59- 1.38)	0.647	0.13 (0.17)	0.75 (0.51- 1.10)	0.142	0.77 (0.23)	0.93 (0.58- 1.51)	0.780	0.72 (0.24)	0.60 (0.38- 0.95)	0.031
head & neck	346 (10.9)	0.26 (0.20)	0.77 (0.52- 1.15)	0.197	0.42 (0.18)	0.65 (0.46- 0.94)	0.021	0.84 (0.26)	0.43 (0.26- 0.71)	0.001	1.22 (0.25)	0.29 (0.18- 0.48)	< 0.001
breast	533 (16.7)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	583 (18.3)	0.58 (0.17)	1.78 (1.29- 2.47)	0.001	1.07 (0.17)	2.93 (2.11- 4.07)	< 0.001	1.89 (0.25)	6.65 (4.05- 10.92)	< 0.001	2.22 (0.24)	9.19 (5.72- 14.78)	<0.001
PS 2	994 (31.2)	0.54 (0.15)	1.71 (1.29- 2.28)	<0.001	1.22 (0.15)	3.39 (2.54- 4.53)	<0.001	2.01 (0.23)	7.49 (4.73- 11.87)	<0.001	2.08 (0.23)	8.02 (5.11- 12.59)	<0.001
PS 1	1055 (33.1)	0.32 (0.14)	1.37 (1.05- 1.80)	0.021	0.85 (0.14)	2.33 (1.77- 3.08)	<0.001	1.39 (0.23)	4.01 (2.53- 6.35)	<0.001	1.17 (0.24)	3.22 (2.03- 5.10)	< 0.001
PS 0	554 (17.4)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	2635 (82.7)	0.37 (0.17)	3.36 (1.96- 5.77)	< 0.001	0.75 (0.18)	2.71 (1.62- 4.54)	<0.001	1.00 (0.26)	2.12 (1.49- 3.01)	< 0.001	1.21 (0.28)	1.45 (1.03- 2.04)	0.032
stage 3	299 (9.4)	0.39 (0.23)	3.11 (1.63- 5.93)	0.001	0.59 (0.23)	2.62 (1.40- 4.92)	0.003	0.96 (0.32)	1.81 (1.15- 2.86)	0.011	1.13 (0.33)	1.48 (0.94- 2.33)	0.090
stage 1-2	251 (7.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	

Sex													
female	1553 (48.8)	-0.19 (0.10)	0.83 (0.68- 1.01)	0.064	-0.12 (0.10)	0.89 (0.74- 1.08)	0.229	-0.32 (0.12)	0.73 (0.57- 0.93)	0.010	-0.55 (0.12)	0.58 (0.46- 0.73)	< 0.001
male	1632 (51.2)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	3185 (100)	0.01 (0.00)	1.01 (1.00- 1.02)	0.072	0.01 (0.00)	1.01 (1.00- 1.01)	0.186	0.004 (0.01)	1.00 (0.99- 1.01)	0.373	- 0.002 (0.01)	1.00 (0.99- 1.01)	0.662
BMI Categories (kg/m2)						, í			, , , , , , , , , , , , , , , , , , ,				
<20.0	707 (22.2)	0.70 (0.18)	2.0 (1.4- 2.9)	<0.001	1.15 (0.17)	3.2 (2.3- 4.5)	<0.001	1.80 (0.23)	6.0 (3.8- 9.5)	<0.001	2.81 (0.25)	16.6 (10.1- 27.1)	<0.001
20.0 to 21.9	551 (17.3)	0.30 (0.18)	1.3 (1.0- 1.9)	0.095	0.80 (0.17)	2.2 (1.6- 3.1)	<0.001	1.18 (0.23)	3.2 (2.1- 5.1)	<0.001	1.33 (0.27)	3.8 (2.2- 6.4)	< 0.001
22.0 to 24.9	890 (27.9)	0.49 (0.15)	1.6 (1.2- 2.2)	0.001	0.55 (0.15)	1.7 (1.3- 2.3)	<0.001	0.67 (0.22)	2.0 (1.3- 3.0)	0.003	1.10 (0.25)	3.0 (1.8- 4.9)	<0.001
25.0 to 27.9	571 (17.9)	-0.04 (0.17)	1.0 (0.7- 1.3)	0.793	0.28 (0.16)	1.3 (1.0- 1.8)	0.087	0.62 (0.23)	1.9 (1.2- 2.9)	0.007	0.26 (0.29)	1.3 (0.7- 2.3)	0.374
≥28.0	467 (14.7)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
MNA MLR model						Univa	ariable M	LR Anal	ysis				
		V	VL 2.5-5.	9%	V	VL 6.0-10	.9%	V	VL 11.0-14	.9%		WL ≥15	.0%
Variables in	Ν	N=	= 534 (22.	.7%)	N	= 537 (22.	.8%)	ľ	N= 211 (9.0	%)		N= 211 (9	9.0%)
model	(%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P-Value
Current Food Intake													
severely decreased	267 (11.4)	1.12 (0.27)	3.05 (1.78- 5.22)	< 0.001	2.51 (0.24)	12.33 (7.64- 19.91)	<0.001	3.09 (0.28)	21.96 (12.64- 38.15)	<0.001	4.41 (0.29)	82.31 (46.51- 145.65	<0.001

)	
moderately decreased	1000 (42.4)	1.46 (0.12)	4.31 (3.39- 4.58)	< 0.001	2.26 (0.13)	9.54 (7.35- 12.38)	< 0.001	2.15 (0.19)	8.57 (5.90- 12.43)	< 0.001	2.48 (0.24)	11.93 (7.52- 18.91)	< 0.001
no decrease	1090 (46.2)		1.0 (ref)										
Cancer Diagnosis													
respiratory	364 (15.4)	0.97 (0.18)	2.64 (1.85- 3.75)	< 0.001	1.33 (0.20)	3.78 (2.55- 5.59)	< 0.001	1.43 (0.34)	4.20 (2.15- 8.19)	< 0.001	1.46 (0.65)	4.32 (2.18- 8.59)	< 0.001
other	292 (12.4)	0.63 (0.20)	1.88 (1.28- 2.76)	0.001	0.92 (0.22)	2.52 (1.64- 3.86)	<0.001	1.55 (0.34)	4.69 (2.41- 9.13)	< 0.001	1.65 (0.35)	5.21 (2.65- 10.25)	<0.001
genitourinary	433 (18.4)	0.78 (0.17)	2.18 (1.56- 3.06)	<0.001	1.20 (1.19)	3.32 (2.28- 4.83)	<0.001	1.27 (0.33)	3.55 91.85- 6.81)	<0.001	1.30 (0.34)	3.67 (1.88- 7.17)	< 0.001
upper GI	384 (16.3)	1.64 (0.22)	5.15 (3.38- 7.85)	<0.001	2.47 (0.22)	11.83 (7.68- 18.24)	<0.001	3.20 (0.32)	24.63 1305- 43.50)	< 0.001	3.45 (0.33)	31.42 (16.54- 59.71)	<0.001
lower GI	403 (17.1)	1.19 (0.19)	3.29 (2.28- 4.74)	<0.001	1.86 (0.20)	6.44 (4.36- 9.51)	<0.001	2.36 (0.32)	10.58 (5.70- 19.65)	<0.001	2.21 (0.33)	9.11 (4.77- 17.42)	< 0.001
breast	481 (20.4)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
MNA Section C: Mobility													
bed or chair bound	225 (9.5)	0.58 (0.23)	1.78 (1.13- 2.80)	0.012	1.02 (0.22)	2.78 (1.82- 4.24)	< 0.001	1.54 (0.25)	4.67 (2.87- 7.60)	<0.001	2.13 (0.23)	8.41 (5.34- 13.24)	<0.001
able to get out of bed/chair but does not go out	417 (17.7)	0.58 (0.15)	1.79 (1.32- 2.43)	<0.001	0.93 (0.15)	2.54 (1.90- 3.39)	<0.001	0.87 (020)	2.38 (1.60- 3.55)	<0.001	1.11 (0.20)	3.03 (2.03- 4.51)	< 0.001
goes out	1715 (72.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	1420 (60.2)	0.95 (0.14)	2.58 (1.97-	< 0.001	1.68 (0.16)	5.37 (3.93-	< 0.001	1.85 (0.25)	6.34 (3.89-	< 0.001	2.04 (0.27)	7.70 (4.57-	< 0.001

			3.38)			7.33)			10.33)			12.96)	
stage 3	423 (17.9)	1.00 (0.17)	2.71 91.95- 3.77)	< 0.001	1.15 (0.20)	3.15 (2.15- 4.63)	< 0.001	1.30 (0.30)	3.68 (2.05- 6.600	< 0.001	1.40 (0.32)	4.07 (2.19- 7.56)	<0.001
stage 1-2	514 (21.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
female	1207 (51.2)	-0.44 (0.11)	0.64 (0.52- 0.80)	< 0.001	-0.51 (011)	0.60 (0.48- 0.75)	< 0.001	-0.45 (0.15)	0.64 (0.47- 0.87)	0.004	-0.41 (0.15)	0.67 (0.49- 0.90)	0.008
male	1150 (48.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	2357 (100)	0.01 (0.00)	1.01 (1.00- 1.02)	0.005	0.01 (0.00)	1.01 (1.00- 1.02)	0.003	0.03 (0.01)	1.03 (1.02- 1.04)	< 0.001	0.03 (0.01)	1.03 (1.02- 1.04)	<0.001
BMI Categories (kg/m2)													
<20.0	278 (11.8)	0.83 (0.25)	2.29 (1.42- 3.72)	0.001	1.61 (0.24)	5.02 (3.17- 7.96)	<0.001	2.78 (0.37)	16.17 (7.83- 33.39)	<0.001	4.17 (0.43)	64.43 (28.00- 148.31)	<0.001
20.0 to 21.9	337 (14.3)	0.06 (0.21)	1.06 (0.70- 1.61)	0.784	1.03 (0.20)	2.81 (1.91- 4.12)	< 0.001	1.98 (0.35)	7.22 (3.64- 14.30)	< 0.001	2.72 (0.42)	15.16 (6.65- 34.54)	<0.001
22.0 to 24.9	792 (33.6)	0.82 (0.15)	2.28 (1.69- 3.06)	< 0.001	0.85 (0.17)	2.34 (1.69- 3.24)	< 0.001	1.71 (0.32)	5.51 (2.92- 10.41)	< 0.001	1.75 (0.42)	5.77 (2.55- 13.07)	<0.001
25.0 to 27.9	528 (22.4)	0.23 (0.17)	1.26 (0.90- 1.75)	0.173	0.65 (0.18)	1.91 (1.36- 2.70)	< 0.001	1.51 (0.33)	4.53 (2.35- 8.74)	< 0.001	1.36 (0.44)	3.88 (1.65- 9.16)	0.002
≥28.0	422 (17.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)				

β, beta coefficient; BMI, body mass index; *I*-VVAS, *Ingesta*-verbal/visual analogue scale GI, gastrointestinal; MNA, Mini-Nutrition Assessment; MLR, multinomial logistic regression; N, number; OR, odds ratio; PGSGA, Patient-generated subjective global assessment; PS, performance status; WL, weight loss *Severely reduced includes the following PG-SGA categories (little solid food, only liquids/oral nutritional supplements, very little of anything)

Appendix IV.

Chapter 5 Supplementary Table 2. Multivariable multinomial logistic regression (MLR) analysis for the association of selected variables to cancer-associated weight loss for three different models based on food intake assessments from three data sets: PG-SGA, *I*-VVAS, and the MNA

PG-SGA MLR model ¹						Mu	ltivariable	e MLR Ai	nalysis				
		V	VL 2.5-5.99	/0	V	VL 6.0-10.9)%	W	L 11.0-14.9)%	l l	VL >=15.0%	/o
Variables in	N (0/)	N=	1139 (28.8	%)	N=	= 1347 (21.9	9%)	N=	= 816 (13.2)	%)	N=	= 1082 (17.6	%)
model	IN (70)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value
Current Food Intake													
severely reduced*	1284 (20.8)	1.09 (0.14)	2.98 (2.26- 3.92)	< 0.001	1.79 (0.13)	5.98 (4.59- 7.78)	<0.001	2.37 (0.16)	10.70 (7.85- 14.58)	<0.001	2.73 (0.16)	15.27 (11.25- 20.73)	<0.001
normal food, reduced amount	2765 (44.9)	0.66 (0.09)	1.94 (1.63- 2.30)	<0.001	1.08 (0.09)	2.93 (2.46- 3.50)	< 0.001	1.45 (0.12)	4.25 (3.36- 5.38)	< 0.001	1.61 (0.12)	5.02 (3.95- 6.38)	<0.001
normal food, normal amount	2112 (34.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis*													
respiratory	1913 (31.1)	0.43 (0.11)	1.53 (1.23- 1.90)	<0.001	0.59 (0.12)	1.80 (1.43- 2.27)	<0.001	0.89 (0.15)	2.45 (1.81- 3.31)	<0.001	0.94 (0.15)	2.56 (1.89- 3.46)	<0.001
other	712 (11.6)	-0.05 (0.15)	0.95 (0.70- 1.29)	0.753	0.16 (0.15)	1.17 (0.87- 1.59)	0.299	0.29 (0.20)	1.34 90.91- 1.96)	0.136	0.49 (0.19)	1.63 (1.13- 2.35)	0.009
genitourinary	386 (6.3)	0.30 (0.20)	1.35 (0.92- 1.99)	0.124	0.62 (0.19)	1.85 (1.28- 2.69)	0.001	0.67 (0.23)	1.96 (1.24- 3.10)	0.004	1.13 (0.22)	3.09 (2.02- 4.73)	<0.001
upper GI	950 (15.4)	0.55 (0.16)	1.73 (1.27- 2.34)	<0.001	1.02 (0.15)	2.77 (2.05- 3.73)	< 0.001	1.51 (0.18)	4.54 (3.19- 6.46)	< 0.001	1.88 (0.17)	6.57 (4.67- 9.26)	<0.001

lower GI	1009 (16.4)	0.76 (0.13)	2.13 (1.65- 2.76)	< 0.001	1.27 (0.14)	3.56 (2.73- 4.64)	<0.001	1.59 (0.17)	4.92 93.51- 6.89)	< 0.001	1.66 (0.17)	5.25 (3.73- 7.39)	<0.001
head & neck	1191 (19.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	1483 (24.1)	0.38 (0.14)	1.47 (1.12- 1.92)	0.005	0.89 (0.14)	2.44 (1.87- 3.18)	< 0.001	1.38 (0.16)	3.97 (2.88- 5.49)	< 0.001	2.00 (0.18)	7.36 (5.16- 10.51)	<0.001
PS 2	1181 (19.2)	0.44 (0.13)	1.55 (1.20- 2.01)	0.001	0.97 (0.13)	2.63 (2.03- 3.40)	<0.001	1.23 (0.17)	3.42 (2.48- 4.73)	<0.001	1.82 (0.18)	6.15 (4.31- 8.79)	<0.001
PS 1	2110 (34.2)	0.39 (0.10)	1.48 (1.22- 1.79)	<0.001	0.64 (0.11)	1.90 (1.54- 2.35)	< 0.001	0.73 (014)	2.07 (1.56- 2.75)	< 0.001	1.23 (0.17)	3.43 (2.47- 4.75)	<0.001
PS 0	1387 (22.5)					1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	4580 (74.3)	0.21 (0.12)	1.23 (0.97- 1.56)	0.086	0.40 (0.13)	1.50 (1.16- 1.93)	0.002	0.33 (0.17)	1.39 (1.00- 1.93)	0.047	0.49 (0.18)	1.63 (1.15- 2.30)	0.006
stage 3	856 (13.9)	0.30 (0.14)	1.35 (1.02- 1.80)	0.036	0.41 (0.16)	1.50 (1.11- 2.04)	0.009	0.45 (0.19)	1.57 (1.08- 2.30)	0.019	0.53 (0.21)	1.71 (1.14- 2.56)	0.009
stage 1-2	725 (11.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
female	2558 (41.5)	-0.16 (0.08)	0.85 (0.72- 1.00)	0.056	-0.14 (0.08)	0.87 (0.74- 1.03)	0.101	-0.29 (0.10)	0.75 (0.62- 0.90)	0.003	-0.39 (0.09)	0.68 (0.56- 0.81)	<0.001
male	3603 (58.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	6161 (100)	-0.01 (0.00)	0.99 (0.99- 1.00)	0.016	-0.00 (0.00)	0.99 (0.98- 0.99)	<0.001	-0.01 (0.00)	0.99 (0.98- 1.00)	0.005	-0.02 (0.00)	0.98 (0.97- 0.98)	<0.001
BMI Categories (kg/m2)													

<20.0	935 (15.2)	0.47 (0.15)	1.61 (120- 2.15)	0.001	1.02 (0.14)	2.76 (2.09- 3.66)	<0.001	1.53 (0.16)	4.62 (3.36- 6.34)	<0.001	2.41 (0.16)	11.10 (8.14- 15.12)	< 0.001
20.0 to 21.9	863 (14.0)	0.26 (0.14)	1.29 (0.99- 1.69)	0.063	0.92 (0.13)	2.52 (1.95- 3.26)	<0.001	1.15 (0.16)	3.15 (2.31- 4.28)	<0.001	1.58 (0.16)	4.87 (3.57- 6.64)	< 0.001
22.0 to 24.9	1540 (25.0)	0.41 (0.11)	1.50 (1.22- 1.85)	< 0.001	0.68 (0.11)	1.97 (1.59- 2.44)	<0.001	0.89 (0.13)	2.44 (1.88- 3.18)	<0.001	1.13 (0.14)	3.10 (2.36- 4.08)	< 0.001
25.0 to 27.9	1289 (20.9)	0.26 (0.11)	1.30 (1.06- 1.60)	0.014	0.43 (0.11)	1.53 (1.23- 1.91)	<0.001	0.36 (0.14)	1.43 (1.08- 1.90)	0.013	0.65 (0.15)	1.92 (1.43- 2.56)	< 0.001
≥28.0	1534 (24.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
<i>I</i> -VVAS MLR model ²						Mu	tivariable	e MLR Ar	nalysis				
		V	WL 2.5-5.9% WL 6.0-10.9% N= 655 (20.6%) N= 818 (25.7%)						L 11.0-14.9	%	\ \	VL >=15.0%	/ ₀
X 7	NI (0/)	N=	= 655 (20.69	%)	N	= 818 (25.7	%)	N=	= 376 (11.89	%)	N	= 409 (12.8%	%)
model	IN (70)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value
Current Food Intake													
severely reduced intake (scores 1-4)	736 (23.1)	0.86 (0.18)	2.37 (1.68- 3.35)	< 0.001	1.53 (0.16)	4.62 (3.36- 6.37)	<0.001	2.13 (0.21)	8.41 (5.63- 12.58)	<0.001	2.66 (0.22)	14.36 (9.28- 22.23)	< 0.001
moderately reduced intake (scores 5-9)	1256 (39.4)	0.82 (0.12)	2.28 (1.79- 2.90)	< 0.001	1.13 (0.12)	3.10 (2.45- 3.92)	<0.001	1.45 (0.17)	4.25 (3.04- 5.94)	<0.001	1.73 (0.20)	5.62 (3.83- 8.26)	< 0.001
normal (score 10)	1193 (37.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis													
respiratory	363 (11.4)	0.25 (0.21)	1.29 (0.85- 1.96)	0.235	0.21 (0.21)	1.24 (0.83- 1.85)	0.296	0.65 (0.28)	1.91 (1.10- 3.31)	0.021	0.48 (0.30)	1.62 (0.90- 2.91)	0.106
other	514	-0.04	0.97 (0.67-	0.851	0.01 (0.18)	1.01 (0.72-	0.937	0.55 (0.25)	1.74 (1.07-	0.027	0.26 (0.27)	1.29 (0.76-	0.345

	470	0.42	1.53		0.00	1.24		0.20	1.46		0.52	1.70	
genitourinary	478 (15.0)	(0.43) (0.20)	(1.03- 2.27)	0.034	0.22 (0.20)	(0.85- 1.83)	0.266	0.38 (0.28)	(0.84- 2.52)	0.179	0.53 (0.29)	(0.97- 2.99)	0.065
upper GI	519 (16.3)	0.85 (0.22)	2.35 (1.54- 3.58)	<0.001	0.74 (0.21)	2.10 (1.39- 3.16)	< 0.001	1.51 (0.27)	4.52 (2.64- 7.73)	< 0.001	1.87 (0.28)	6.51 (3.73- 11.35)	<0.001
lower GI	433 (13.6)	0.28 (0.20)	1.33 (0.89- 1.98)	0.164	0.37 (0.20)	1.45 (0.98- 2.13)	0.061	1.11 (0.27)	3.04 (1.79- 5.15)	<0.001	1.07 (0.29)	2.90 (1.65- 5.12)	<0.001
head & neck	346 (10.9)	0.27 (0.24)	1.31 (0.82- 2.07)	0.257	0.45 (0.22)	1.57 (1.01- 2.42)	0.043	0.84 (0.30)	2.32 (1.28- 4.22)	0.006	1.07 (0.31)	2.92 (1.58- 5.37)	0.001
breast	533 (16.7)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	583 (18.3)	0.18 (0.20)	1.20 (0.81- 1.77)	0.358	0.45 (0.20)	1.56 (1.05- 2.32)	0.026	1.28 (0.29)	3.59 (2.02- 6.38)	<0.001	1.47 (0.30)	4.33 (2.42- 7.77)	<0.001
PS 2	994 (31.2)	0.21 (0.17)	1.23 (0.89- 1.70)	0.215	0.68 (0.17)	1.98 (1.42- 2.75)	<0.001	1.35 (0.26)	3.85 (2.32- 6.39)	<0.001	1.23 (0.26)	3.42 (2.03- 5.74)	<0.001
PS 1	1055 (33.1)	0.16 (0.15)	1.17 (0.88- 1.56)	0.282	0.59 (0.15)	1.80 (1.34- 2.43)	<0.001	1.09 (0.25)	2.97 (1.83- 4.83)	<0.001	0.77 (0.26)	2.16 91.30- 3.58)	0.003
PS 0	554 (17.4)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	2635 (82.7)	0.14 (0.18)	1.15 (0.80- 1.65)	0.459	0.36 (0.19)	1.43 (0.98- 2.08)	0.065	0.39 (0.28)	1.48 (0.85- 2.58)	0.164	0.46 (0.31)	1.58 (0.86- 2.91)	0.141
stage 3	299 (9.4)	0.23 (0.24)	1.26 (0.78- 2.03)	0.348	0.32 (0.25)	1.38 (0.84- 2.25)	0.205	0.47 (0.35)	1.59 (0.81- 3.14)	0.179	0.54 (0.37)	1.72 (0.83- 3.58)	0.146
stage 1-2	251 (7.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
female	1553 (48.8)	-0.15 (0.13)	0.86 (0.67-	0.234	-0.18 (0.12)	0.84 (0.66-	0.150	-0.38 (0.15)	0.68 (0.51-	0.013	-0.71 (0.16)	0.49 (0.36-	< 0.001

			1.10)			1.07)			0.92)			0.67)	
male	1632 (51.2)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	3185 (100)	0.00 (0.00)	1.00 (0.99- 1.01)	0.546	0.00 (0.00)	1.00 (0.99- 1.01)	0.838	0.00 (0.01)	1.00 (0.98- 1.01)	0.386	-0.01 (0.01)	0.99 (0.98- 1.00)	0.095
BMI Categories (kg/m2)													
<20.0	707 (22.2)	0.64 (0.19)	1.90 (1.31)	0.001	0.98 (0.18)	2.66 (1.85)	<0.001	1.56 (0.25)	4.76 (2.94- 7.72)	<0.001	2.57 (0.27)	13.12 (7.70- 22.37)	<0.001
20.0 to 21.9	551 (17.3)	0.21 (0.18)	1.23 (0.86- 1.77)	0.249	0.67 (0.18)	1.94 (1.38- 2.75)	< 0.001	1.01 (0.25)	2.73 (1.69- 4.43)	<0.001	1.13 (0.29)	3.10 (1.77- 5.43)	< 0.001
22.0 to 24.9	890 (27.9)	0.47 (0.15)	1.60 (1.18- 2.17)	0.002	0.51 (0.16)	1.67 (1.22- 2.28)	0.001	0.60 (0.23)	1.83 (1.16- 2.90)	0.010	1.01 (0.27)	2.76 (1.63- 4.67)	< 0.001
25.0 to 27.9	571 (17.9)	-0.13 (0.17)	0.88 (0.63- 1.23)	0.458	0.20 (0.17)	1.22 (0.87- 1.71)	0.243	0.51 (0.25)	1.67 (1.03- 2.70)	0.037	0.10 (0.31)	1.11 (0.61- 2.02)	0.737
≥28.0	467 (14.7)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
MNA MLR mo	del ³					Mu	ltivariable	e MLR Ai	nalysis				
		V	VL 2.5-5.9%	6	W	VL 6.0-10.9)%	W	L 11.0-14.9)%	\ \	VL >=15.0%	6
Variables in		N=	= 534 (22.79	%)	N	= 537 (22.8	%)	N	= 211 (9.09	(a)	N	= 211 (9.0%	6)
model	N (%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value
Current Food Intake													
severely decreased	267 (11.4)	0.77 (0.29)	2.17 (1.24- 3.80)	0.007	1.93 (0.26)	6.87 (4.12- 11.46)	<0.001	2.28 (0.31)	9.73 (5.33- 17.75)	<0.001	3.29 (0.32)	26.86 (14.29- 50.49)	<0.001
moderately decreased	1000 (42.4)	1.32	3.74 (2.89-	<0001	2.00 (0.14)	7.38 (5.57-	< 0.001	1.787 (0.20)	5.94 (3.99-	< 0.001	1.95 (0.25)	7.05 (4.29-	< 0.001

no decrease	1090 (46.2)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis													
respiratory	364 (15.4)	0.48 (022)	1.62 (1.05- 2.50)	0.029	0.63 (0.25)	1.87 (1.16- 3.04)	0.011	0.75 (0.38)	2.12 (1.00- 4.48)	0.051	0.74 (0.42)	2.10 (0.92- 4.82)	0.079
other	292 (12.4)	0.29 (0.23)	1.34 (0.85- 2.10)	0.204	0.43 (0.26)	1.54 (0.92- 2.56)	0.100	0.96 (0.38)	2.62 (1.24- 5.53)	0.012	0.85 (0.42)	2.35 (1.03- 5.35)	0.042
genitourinary	433 (18.4)	0.42 (0.20)	1.51 (1.03- 2.23)	0.037	0.66 (0.23)	1.93 (1.24- 3.01)	0.004	0.70 (0.36)	2.02 (0.99- 4.13)	0.053	0.67 (0.40)	1.96 (0.89- 4.31)	0.092
upper GI	384 (16.3)	1.08 (0.24)	2.94 (1.83- 4.73)	<0.001	1.60 (0.26)	4.95 (2.99- 8.20)	<0.001	2.42 (0.36)	1124 (5.53- 22.86)	<0.001	2.55 (0.39)	12.75 (5.90- 27.53)	<0.001
lower GI	403 (17.1)	0.88 (0.22)	2.40 (1.57- 3.66)	<0.001	1.54 (0.24)	4.67 (2.94- 7.42)	<0.001	2.03 (0.35)	7.62 (3.83- 15.15)	<0.001	1.93 (0.39)	6.87 (3.19- 14.82)	<0.001
breast	481 (20.4)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
MNA Section C: Mobility													
bed or chair bound	225 (9.5)	0.09 (0.25)	1.09 (0.66- 1.79)	0.734	0.30 (0.25)	1.34 (0.83- 2.18)	0.232	0.25 (0.24)	2.13 (1.20- 3.77)	0.010	1.08 (0.30)	2.94 (1.64- 5.27)	<0.001
able to get out of bed/chair but does not go out	417 (17.7)	0.23 (0.17)	1.26 (0.90- 1.77)	0.179	0.40 (0.18)	1.49 (1.06- 2.11)	0.023	0.76 (0.29)	1.28 (0.81- 2.03)	0.287	0.37 (0.25)	1.44 (0.88- 2.36)	0.145
goes out	1715 (72.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	1420 (60.2)	0.48 (0.16)	1.62 (1.19- 2.21)	0.002	1.05 (0.18)	2.86 (1.99- 4.10)	< 0.001	1.02 (0.27)	2.77 (1.62- 4.74)	< 0.001	1.10 (0.32)	3.01 (1.61- 5.61)	0.001
stage 3	423 (17.9)	0.69 (0.18)	2.00 (1.40- 2.87)	< 0.001	0.75 (0.22)	2.11 (1.37- 3.25)	0.001	0.77 (0.33)	2.16 (1.14- 4.08)	0.018	0.92 (0.38)	2.51 (1.20- 5.23)	0.014

stage 1-2	514 (21.8)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
female	1207 (51.2)	-0.14 (0.14)	0.87 (0.67- 1.14)	0.315	-0.21 (0.14)	0.81 (0.61- 1.07)	0.141	-0.13 (0.19)	0.88 (0.61- 1.26)	0.482	-0.26 (0.200	0.77 (0.52- 1.14)	0.189
male	1150 (48.8)												
Age (years)	2357 (100)	0.01 (0.05)	1.01 (0.92- 1.12)	0.782	-0.04 (0.05)	0.96 (0.87- 1.07)	0.469	0.126 (0.07)	1.13 (0.98- 1.31)	0.091	0.16 (0.08)	1.17 (1.00- 1.37)	0.050
BMI Categories (kg/m2)													
<20.0	278 (11.8)	0.66 (0.26)	1.93 (1.15- 3.23)	0.012	1.26 (0.27)	3.51 (2.08- 5.93)	<0.001	2.45 (0.40)	11.59 (5.31- 25.32)	<0.001	3.77 (0.47)	43.47 (17.46- 108.24)	<0.001
20.0 to 21.9	337 (14.3)	-0.07 (0.23)	0.93 (0.60- 1.46)	0.76	0.78 (0.23)	2.19 (1.40- 3.40)	0.001	1.81 (0.37)	6.11 (2.94- 12.69)	<0.001	2.59 (0.46)	13.33 (5.45- 32.57)	<0.001
22.0 to 24.9	792 (33.6)	0.70 (0.16)	2.014 (1.47- 2.77)	<0.001	0.63 (0.19)	1.88 (1.30- 2.72)	0.001	1.56 (0.34)	4.74 (2.42- 9.25)	<0.001	1.64 (0.44)	5.14 (2.15- 12.28)	<0.001
25.0 to 27.9	528 (22.4)	0.20 (0.180	1.22 (0.86- 1.73)	0.266	0.59 (0.20)	1.80 (1.22- 2.67)	0.003	1.49 (0.35)	4.46 (2.23- 8.91)	< 0.001	1.38 (0.47)	3.99 (1.60- 9.97)	<0.001
≥28.0	422 (17.9)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	

β, beta coefficient; BMI, body mass index; *I*-VVAS, *Ingesta*-verbal/visual analogue scale GI, gastrointestinal; MNA, Mini-Nutrition Assessment; MLR, multinomial logistic regression; N, number; OR, odds ratio; PGSGA, Patient-generated subjective global assessment; PS, performance status; WL, weight loss *Severely reduced includes the following PG-SGA categories (little solid food, only liquids/oral nutritional supplements, very little of anything) ¹PG-SGA MLR Model - Reference weight stable (+/- 2.4%) N=1777 (28.8%); Intercept only model: -2 log likelihood (LL)=18682.909, AIC 18690.909; Final Model: -2LL=16403.448, AIC=16555.448, χ 2=2279.461 (df=72), P<0.001; Pseudo R2 (Nagelkerke) = 0.323; Pearson goodness-of-fit P=0.063 ²*I*-VVAS MLR Model - Reference weight stable (+/-2.4%) N=928 (21.9%); Intercept only model: -2 LL=9671.159, AIC=9679.159; Final Model: -2LL=8658.277, AIC=8818.277, χ 2=1012.883 (df=76), P<0.001; Pseudo r2 (Nagelkerke)=0.285; Pearson goodness-of-fit P=0.792 ³MNA MLR Model - Reference weight stable (±2.4%) N=864 (36.7%); Intercept only model: -2 log likelihood (LL)= 6755.457, AIC 6763.457; Final Model: -2LL=5533.028, AIC=5677.028, χ 2=1222.429 (df=68), P<0.001; Pseudo R2 (Nagelkerke)=0.427, Pearson goodness-of-fit P=0.198

Appendix V.

Chapter 6 Supplementary Table 1. Univariable multinomial logistic regression (MLR) analysis for the association of selected variables to cancer-associated weight loss.

		WL 2.5-5.9%			WL 6.0-10.9%		W	L 11.0-14.9)%	V	VL >=15.0%	6	
		N=	= 711 (20.79	%)	N=	= 822 (23.9	9%)	N=	= 390 (11.3	%)	N	= 501 (14.69	%)
Variables in model	N (%)	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value	β (SE)	OR (95% CI)	P- Value
CRP Categories													
≥43.0 mg/L	1032 (30.0)	0.41 (0.13)	1.51 (1.18- 1.93)	0.001	0.89 (0.12)	2.45 (1.94- 3.08)	<0.001	1.12 (0.15)	3.06 (2.30- 4.08)	< 0.001	1.50 (0.14)	4.46 (3.40- 5.86)	<0.001
10.0-42.9 mg/L	1036 (30.1)	0.28 (0.11)	1.32 (1.06- 1.66)	0.014	0.42 (0.11)	1.52 (1.22- 1.90)	<0.001	0.51 (0.15)	1.67 (1.25- 2.23)	0.001	0.80 (0.14)	2.23 (1.69- 2.94)	<0.001
<10.0 mg/L	1375 (39.9)		1.0 (ref)									1.0 (ref)	
Current Food Intake													
severely reduced intake (scores 1-4)	686 (19.9)	0.99 (0.16)	2.69 (1.96- 3.69)	< 0.001	1.66 (0.15)	5.26 (3.91- 7.07)	<0.001	2.37 (0.18)	10.73 (7.53- 15.29)	< 0.001	2.88 (0.17)	17.77 (12.64- 24.98)	<0.001
moderately reduced intake (scores 5-9)	1380 (40.1)	0.91 (0.11)	2.49 (2.02)	<0.001	1.28 (0.11)	3.61 (2.93- 4.46)	<0.001	1.56 (0.15)	4.77 (3.56- 6.68)	<0.001	1.82 (0.15)	6.20 (4.65- 8.27)	< 0.001
normal (score 10)	1377 (40.0)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Diagnosis													
respiratory	786 (22.8)	0.29 (0.17)	1.34 (0.97- 1.86)	0.080	-0.09 (0.16)	0.92 (0.67- 1.24)	0.573	0.33 (0.22)	1.39 (0.90- 2.14)	0.139	0.32 (0.23)	1.38 (0.89- 2.15)	0.155
other	353 (10.3)	0.02 (0.20)	1.02 (0.69- 1.52)	0.906	-0.24 (0.19)	0.78 (0.54- 1.14)	0.198	0.22 (0.26)	1.25 (0.75- 2.08)	0.390	0.35 (0.26)	1.43 (0.86- 2.37)	0.171
genitourinary	607	0.40	1.50	0.023	0.09	1.09	0.585	0.45	1.58	0.051	0.79	2.20	0.001

	(17.6)	(0.18)	(1.06-		(0.17)	(0.79-		(0.23)	(1.00-		(0.23)	(1.41-	
			2.12)			1.52)			2.49)			3.45)	
upper GI	685 (19.9)	0.60 (0.18)	1.82 (1.27- 2.60)	0.001	0.36 (0.17)	1.43 (1.03- 1.99)	0.033	1.03 (0.23)	2.81 (1.80- 4.37)	<0.001	1.63 (0.22)	5.09 (3.31- 7.82)	<0.001
lower GI	276 (8.0)	0.21 (0.22)	1.23 (0.79- 1.91)	0.357	0.14 (0.20)	1.15 (0.77- 1.71)	0.493	0.69 (0.27)	1.99 (1.18- 3.37)	0.010	0.66 (0.28)	1.94 (1.13- 3.33)	0.016
head & neck	307 (8.9)	0.32 (0.22)	1.23 (0.79- 1.91)	0.153	0.37 (0.20)	1.45 (0.98- 2.16)	0.062	0.62 (0.28)	1.87 (1.08- 3.22)	0.024	1.33 (0.25)	3.77 (3.29- 6.19)	<0.001
breast	429 (12.5)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
ECOG PS													
PS 3-4	570 (16.6)	0.42 (0.16)	1.53 (1.11- 2.10)	0.009	0.99 (0.16)	2.68 (1.96- 3.66)	< 0.001	1.24 (0.21)	3.46 (2.31- 5.17)	< 0.001	1.89 (0.20)	6.65 (4.47- 9.89)	<0.001
PS 2	1058 (30.7)	0.29 (0.14)	1.33 (1.02- 1.74)	0.034	0.99 (0.13)	2.68 (2.06- 3.49)	<0.001	1.21 (0.18)	3.36 (2.37- 4.77)	<0.001	1.72 (0.18)	5.58 (3.89- 8.00)	<0.001
PS 1	1081 (31.4)	0.35 (0.13)	1.42 (1.11- 1.82)	0.005	0.64 (0.13)	1.90 (1.47- 2.46)	<0.001	0.69 (0.18)	1.99 (1.39- 2.84)	<0.001	1.08 (0.19)	2.94 (2.03- 4.25)	<0.001
PS 0	734 (21.3)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Cancer Stage													
stage 4	2698 (78.)	0.26 (0.17)	1.29 (0.92- 1.82)	0.139	0.45 (0.17)	1.57 (1.12- 2.20)	0.010	0.58 (0.24)	1.78 (1.12- 2.84)	0.014	0.97 (0.25)	2.65 (1.63- 4.29)	<0.001
stage 3	482 (14.0)	0.13 (0.21)	1.14 (0.76- 1.71)	0.534	0.21- 0.21)	1.23 (0.82- 1.84)	0.323	0.32 (0.28)	1.38 (0.80- 2.38)	0.253	0.79 (0.28)	2.20 (1.27- 3.81)	0.005
stage 1-2	263 (7.6)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Sex													
male	1873 (54.4)	0.18 (0.10)	1.19 (0.99- 1.45)	0.071	0.06 (0.09)	1.06 (0.88- 1.28)	0.521	0.30 (0.12)	1.35 (1.07- 1.71)	0.012	0.52 (0.11)	1.68 (1.35- 2.09)	< 0.001

female	1570 (45.6)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	
Age (years)	3443 (100)	-0.001 (0.004)	0.999 (0.992- 1.01)	0.900	- 0.002 (0.00 4)	0.998 (0.991- 1.01)	0.654	-0.001 (0.005)	0.999 (0.990- 1.01)	0.846	-0.007 (0.004)	0.993 (0.985- 1.002)	0.122
BMI Categories (kg/m2)													
<20.0	655 (19.0)	0.65 (0.17)	1.92 (1.37- 2.70)		1.16 (0.17)	3.18 (2.29- 4.43)	<0.001	2.04 (0.23)	7.67 (4.87- 12.09)	<0.001	2.35 (0.19)	10.45 (7.16- 15.25)	<0.001
20.0 to 21.9	566 (16.4)	0.43 (0.17)	1.54 (1.11- 2.13)		1.05 (0.16)	2.86 (2.10- 3.90)	<0.001	1.60 (0.23)	4.94 (3.13- 7.78)	<0.001	1.06 (0.21)	2.90 (1.92- 4.37)	<0.001
22.0 to 24.9	953 (27.7)	0.50 (0.14)	1.64 (1.25- 2.15)		0.69 (0.14)	1.99 (1.51- 2.62)	<0.001	1.29 (0.22)	3.63 (2.38- 5.53)	<0.001	088 (0.19)	2.40 (1.67- 3.47)	<0.001
25.0 to 27.9	646 (18.8)	0.18 (0.15)	1.20 (0.89- 1.60)	0.229	0.41 (0.15)	1.50 (1.12- 2.01)	0.007	0.86 (0.23)	2.36 (1.50- 3.71)	<0.001	0.11 (0.22)	1.11 (0.72- 1.70)	0.630
≥28.0	643 (18.1)		1.0 (ref)			1.0 (ref)			1.0 (ref)			1.0 (ref)	

β, beta coefficient; BMI, body mass index; CRP, C-reactive protein; MLR, multinomial logistic regression; N, number; OR, odds ratio; PS, performance status; SE, standard error; WL, weight loss

Appendix VI

Chapter 7 Supplementary Table 1. Logistic regression models of the factors associated with extended LOS (model 1), postoperative complications (model 2) for patients in the ERAS implementation group (N=2628)

			 	Model 1]	Model 2	2	
		Outc	ome = Ex	tended	LOS (> 5 days)		Outcome = a	ny comp	lication	during primary s	stay
Factors	Total	Univariab	ole		Final Multivariab	ole	Univariabl	e		Final Multi	ivariable
	N	OR (95% CI)	P- Value	N	OR (95% CI)	P- Value	OR (95% CI)	P- Value	N	OR (95% CI)	P- Value
Nutrition Screen*											
not at nutritional risk	2317	1.0 (ref)		2298	1.0 (ref)		1.0 (ref)		-	-	-
at nutritional risk	311	1.73 (1.36-2.21)	<0.001	302	1.40 (1.00-1.96)	0.052	1.34 (1.06-1.70)	0.016	-	-	-
Sex											
female	1137	1.0 (ref)		1124	1.0 (ref)		1.0 (ref)		1136	1.0 (ref)	
male	1490	1.33 (1.14-1.56)	<0.001	1476	1.29 (1.04-1.60)	0.022	1.27 (1.09-1.48)	0.002	1490	1.26 (1.07-1.49)	0.006
Age (years)	2627	1.02 (1.01-1.02) <0.001		2600	1.02 (1.01-1.03)	<0.001	1.01 (1.01-1.02)	0.000		1.01 (1.01-1.02)	<0.001
ASA Class											
1 to 2	1774	1.0 (ref)		1766	1.0 (ref)		1.0 (ref)		1773	1.0 (ref)	
3 to 4	798	1.76 (1.48-2.08)	<0.001	779	1.37 (1.08-1.74)	0.010	1.50 (1.27-1.78)	0.000	798	1.21 (1.01-1.45)	0.044
unknown	55	1.04 (0.61-1.79)	0.88	55	0.91 (0.41-2.02)	0.82	0.83 (0.48-1.42)	0.493	55	1.00 (0.54-1.84)	0.992
Cancer Diagnosis											
non-cancer diagnosis	1280	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
cancer diagnosis	1348	1.49 (1.27-1.73)	<0.001	-	-	-	1.21 (1.04-1.41)	0.014	-	-	-
Procedure Type											
colon and small bowel procedure	1687	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
rectal procedure	914	1.98 (1.69-2.33)	<0.001	-	-	-	1.37 (1.17-1.61)	0.000	-	-	-
Surgical approach											

253

laparoscopic	1343	1.0 (ref)		1332	1.0 (ref)		1.0 (ref)		1342	1.0 (ref)	
open	930	4.31 (3.60-5.16)	<0.001	915	3.13 (2.44-4.03)	<0.001	2.47 (2.08-2.93)	0.000	929	2.04 (1.68-2.48)	<0.001
stoma approach	232	0.83 (0.61-1.12)	0.220	231	1.00 (0.68-1.49)	0.981	0.93 (0.70-1.24)	0.621	232	1.08 (0.80-1.47)	0.609
converted	113	4.03 (2.66-6.10)	<0.001	112	2.83 (1.69-4.73)	<0.001	2.35 (1.58-3.48)	0.000	113	1.81 (1.19-2.75)	0.005
unknown	10	0.46 (0.10-2.16)	0.324	10	0.35 (0.05-2.34)	0.279	0.61 (0.16-2.40)	0.487	10	0.60 (0.15-2.41)	0.472
Surgical											
less complex procedures	1482	1.0 (ref)		1466	1.0 (ref)		1.0 (ref)		1481	1.0 (ref)	
more complex procedures	1146	2.11 (1.80-2.47)	<0.001	1134	1.72 (1.37-2.16)	<0.001	1.49 (1.28-1.75)	0.000	1145	1.18 (0.99-1.41)	0.059
Compliance to ERAS Protocol											
>70% (good compliance)	1796	1.0 (ref)		1788	1.0 (ref)		1.0 (ref)		1794	1.0 (ref)	
<70% (low	832	3.42 (2.87-4.08)	<0.001	812	2.57 (2.01-3.29)	<0.001	3.05 (2.56-3.62)	0.000	832	2.69 (2.23-3.24)	<0.001
compliance)											
during primary											
stay											
no	1327	1.0 (ref)		1326	1.0 (ref)		-	-	-	-	-
yes	1301	16.58 (13.67- 20.12)	<0.001	1274	17.00 (13.60- 21.26)	<0.001	-	-	-	-	-
Acute Care Center											
Hospital Site #1	227	1.0 (ref)		223	1.0 (ref)				227	1.0 (ref)	
Hospital Site #2	1089	0.62 (0.47-0.83)	0.001	1080	0.54 (0.35-0.82)	0.004	0.90 90.68-1.20)	0.483	1088	0.92 (0.67-1.27)	0.609
Hospital Site #3	459	1.19 (0.86-1.65)	0.290	456	2.10 (1.34-3.28)	0.001	0.90 (0.66-1.24)	0.528	459	1.10 (0.78-1.55)	0.580
Hospital Site #4	317	0.59 (0.42-0.83)	0.003	311	0.83 (0.52-1.34)	0.443	0.50 (0.35-0.71)	< 0.00	316	0.49 (0.34-0.71)	<0.001
Hospital Site #5	277	0.76 (0.53-1.09)	0.132	273	1.39 (0.84-2.30)	0.197	0.55 (0.38-0.78)	0.001	277	0.62 (0.41-0.91)	0.016
Hospital Site #6	259	0.96 (0.67-1.37)	0.814	257	1.02 (0.62-1.67)	0.948	1.01 (0.70-1.44)	0.965	259	1.09 (0.75-1.60)	0.642
Diabetes											
no	2192	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
yes	432	1.43 (1.16-1.77)	0.001	-	-	-	1.34 (1.09-1.65)	0.006	-	-	-

unknown	4	0.56 (0.05-6.18)	0.636	-	-	-	1.07 (0.15-7.60)	0.946	-	-	-
BMI class											
<18.5	57	0.98 (0.57-1.68)	0.935	-	-	-	1.31 (0.76-2.25)	0.329	-	-	-
(underweight)	700	10(0)					10(0				
18.5-24.9 (normal weight)	/89	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
25.0-29.9	943	1.03 (0.85-1.25)	0.744	-	-	-	1.03 (0.86-1.25)	0.729	-	-	-
(overweight)											
≥30.0 (obese)	839	0.95 (0.78-1.15)	0.589	-	-	-	0.96 (0.79-1.16)	0.640	-	-	-
Alcohol use [†]											
No	2378	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
Yes	219	0.85 (0.64-1.12)	0.237	-	-	-	0.83 (0.63-1.10)	0.191	-	-	-
Stopped because of	5	1.57 (0.26-9.4)	0.623	-	-	-	1.51 (0.25-9.05)	0.652	-	-	-
surgery	26	1 22 (0 5(2 (5)	0 (1(1 19 (0 54 2 55)	0.694			
Unknown	26	1.22 (0.56-2.65)	0.616	-	-	-	1.18 (0.54-2.55)	0.684	-	-	-
Smoker											
no	2160	1.0 (ref)		-	-	-	1.0 (ref)		-	-	-
yes	455	1.09 (0.89-1.34)	0.394	-	-	-	1.04 (0.85-1.280	0.678	-	-	-
unknown	13	2.43 (0.75-7.90)	0.141	-	-	-	1.65 (0.54-5.05)	0.382	-	-	-

ASA, American Society of Anesthesiologists physical status classification; BMI, body mass index; CI, confidence interval; ERAS, Enhanced Recovery After Surgery; LOS, length of hospital stay; N, number; OR, odds ratio

*Nutrition risk was only calculated for the ERAS GRoup and defined as ≥ 2 on the Malnutrition Screening Tool (MST) or two "yes" answers on the Canadian Nutrition Screening Tool (CNST), patients in the ERAS implementation group whose nutrition screen status was unknown (N=908) were excluded from this analysis

(-) dash indicates variables that were not included in the logistic regression models because they were not significant at univariable or multivariable level

Model 1: Full model (included all variables significant at univariable level) -2 LL = 2196.376, Negelkerke R^2 = 56%, Hosmer &

Lemeshow test P= 0.717, Classification = 82%; Fitted Model (retains variables significant at multivariable level) -2LL =2199.019,

Negelkerke $r^2 = 56\%$, Hosmer & Lemeshow test P=0.182, Classification = 82%

Model 2: Full model (included all variables significant at univariable level) -2 LL = 3311.719 Negelkerke R² = 16%, Hosmer & Lemeshow test P= 0.564, Classification = 65%; Fitted Model (retains variables significant at multivariable level) -2LL = 3316.942, Negelkerke r² = 15%, Hosmer & Lemeshow test P=0.584, Classification = 65%

Appendix VII

Chapter 8. Supplementary Table 1. Simple linear regression models for the effect of sex and age on features of body composition for each study cohort and the aggregated colorectal cancer data

Linear Regression Models	В	SE <i>B</i>	β	t	P-Value
Skeletal Muscle Index (cm ² /m ²)					
Canada Cohort 1 (N=384)					
Intercept	47.36	2.66		17.82	0.000
Age (years)	-0.28 ^a	0.04	-0.31	-7.91	0.000
Sex (male)	11.49 ^b	0.81	0.56	14.20	0.000
R^2	0.40				
Canada Cohort 2 (N=961)					
Intercept	43.02	1.67		25.83	0.000
Age (years)	-0.21 ^a	0.02	-0.25	-9.88	0.000
Sex (male)	10.95 ^b	0.51	0.55	21.40	0.000
R^2	0.37				
UK Cohort 3 (N=755)					
Intercept	44.68	1.85		24.20	0.000
Age (years)	-0.22 ^a	0.02	-0.29	-9.39	0.000
Sex (male)	8.73 ^b	0.57	0.46	15.19	0.000
R ²	0.30				
Pooled CRC Cohort (N=2100)					
Intercept	44.64	1.13		39.34	0.000
Age (years)	-0.23	0.01	-0.28	-15.79	0.000
Sex (male)	10.24	0.35	0.52	29.24	0.000
R^2	0.35				
Skeletal Muscle Radiodensity (HU)					
Canada Cohort 1 (N=384)					
Intercept	57.85	2.97		19.51	0.000
Age (years)	-0.38 ^a	0.04	-0.44	-9.62	0.000
Sex (male)	1.62 ^b	0.90	0.08	1.79	0.074
R^2	0.19				
Canada Cohort 2 (N=961)					
Intercept	52.87	1.78		29.71	0.000
Age (years)	-0.34 ^a	0.02	-0.44	-15.24	0.000
Sex (male)	1.45 ^b	0.55	0.08	2.66	0.008
R^2	0.20				
UK Cohort 3 (N=755)					
Intercept	50.79	1.91		26.61	0.000
Age (years)	-0.37 ^a	0.02	-0.49	-15.63	0.000
Sex (male)	2.59 ^b	0.59	0.14	4.36	0.000
\mathbf{R}^2	0.26				
Pooled CRC Cohort (N=2100)					

Intercept	53.28	1.22		43.77	0.000
Age (years)	-0.36	0.02	-0.46	-23.61	0.000
Sex (male)	1.91	0.38	0.10	5.10	0.000
R^2	0.22				
Visceral Adipose Tissue Index (cm ² /m ²)					
Canada Cohort 1 (N=384)					
Intercept	-3.72	12.53		-0.30	0.767
Age (years)	0.30 ^a	0.17	0.09	1.83	0.068
Sex (male)	29.76 ^b	3.82	0.37	7.80	0.000
R^2	0.14				
Canada Cohort 2 (N=961)					
Intercept	-12.05	7.53		-1.60	0.110
Age (years)	0.49^{a}	0.10	0.15	5.08	0.000
Sex (male)	24.90 ^c	2.31	0.33	10.77	0.000
R^2	0.12				
UK Cohort 3 (N=755)					
Intercept	-1.77	7.30		-0.24	0.808
Age (years)	0.26 ^a	0.09	0.10	2.86	0.004
Sex (male)	21.84 ^d	2.29	0.33	9.53	0.000
R^2	0.12				
Pooled CRC Cohort (N=2100)					
Intercept	-5.68	4.95		-1.15	0.252
Age (years)	0.35	0.06	0.12	5.64	0.000
Sex (male)	24.69	1.53	0.33	16.10	0.000
R^2	0.12				

B = unstandardized beta, slope of the line between predictor and dependent variable

SE = standard error for B

 β = standardized beta, values range between 0 to 1 or -1 to 0, indicates the direction and strength of relationship between predictor and dependent variable

 $R^2 = effect size$

^{a,b} Skeletal muscle index = no difference between study cohort correlation coefficients for age^a (P>0.05) or sex^b (P>0.05), based on Z-test⁵¹

^{a,b}Skeletal muscle attenuation = no difference between study cohort correlation coefficients age^a (P>0.05) or sex^b (P>0.05), based on Z-test⁵¹

Appendix VIII

Chapter 8 Supplementary Table 2. Estimated marginal mean LOS derived from the final adjusted negative binomial regression model for each body composition profile demonstrating the independent effect of the body composition profiles on LOS with or without major post-operative complications.

Body composition profiles based on the absence or presence of sarcopenia, myosteatosis, or visceral obesity	Geographic Region	Estimated Marginal Mean LOS (95% CI)	Mean Difference (days)	P-Value
no features				
Non-sarcopenic, Non-myosteatotic, Non-viscerally obese	UK	12.21 (11.05-13.49) ^a	3.05	< 0.001
	Canada	9.16 (8.26-10.15) ^c		
1 feature				
Sarcopenic	UK	13.80 (12.14-15.69)	3.45	< 0.001
	Canada	10.35 (9.03-11.86)		
Viscerally Obese	UK	12.32 (10.51-14.45)	3.08	< 0.001
	Canada	9.24 (7.96-10.74)		
Myosteatosis	UK	13.76 (12.36-15.32)	3.44	< 0.001
	Canada	10.32 (9.09-11.72)		
2 features				
Sarcopenia + Viscerally Obese	UK	14.00 (7.62-25.71)	3.50	0.005
	Canada	10.50 (5.75-19.18)		
Sarcopenia + Myosteatosis*	UK	15.49 (13.89-17.27) ^b	3.87	< 0.001
	Canada	11.62 (10.20-13.23) ^d		
Myosteatosis + Viscerally Obese*	UK	15.28 (13.59-17.17) ^b	3.82	< 0.001
	Canada	11.46 (10.16-12.92) ^d		
3 features				
Sarcopenia+ Myosteatosis + Viscerally Obese*	UK	19.24 (15.85-23.36) ^b	4.81	< 0.001
	Canada	$14.\overline{43} (11.84-17.59)^{d}$		

*profiles identified from negative binomial regression that are significantly (P<0.05) associated with extended LOS (compared to no features) independent of major complications, study cohort, surgical approach, and age ^{a,b} Significantly different from no features (Non-sarcopenic, Non-myosteatotic, Non-viscerally obese) from UK

^{c,d} Significantly different from no features (Non-sarcopenic, Non-myosteatotic, Non-viscerally obese) from Canada

Appendix IX

Chapter 8. Supplementary Table 3. Factors predictive of major complications (Clavien Dindo Grade \geq 3) in colorectal cancer patients undergoing elective surgery.

			Univariable			Multivariable	
Variable	Ν	Beta	OR (95% CI)	P-Value	Beta	OR (95% CI)	P-Value
Pre-surgical Body Composition Profiles Sarcopenia+ Myosteatosis + Viscerally Obese	42	0.022	1.02 (0.41-2.57)	0.963			
Sarcopenia +	5	-19.389	-	0.999			
Viscerally Obese Myosteatosis + Viscerally Obese	148	0.121	1.13 (0.65-1.95)	0.664			
Viscerally Obese	90	-0.383	0.68 (0.32-1.45)	0.321			
Sarcopenia + Myosteatosis	189	-0.213	0.81 (0.47-1.39)	0.443			
Sarcopenia	140	-0.465	0.63 (0.33-1.21)	0.162			
Myosteatosis	181	0.35	1.42 (0.87-2.31)	0.160			
None	321						
Sample							
Sample 1	384						
Sample 2	755	-0.046	0.955 (0.669-1.362)	0.798			
Sex							
Female	451						
Male	688	0.972	2.644 (1.7711-3.946)	< 0.001	0.881	2.414 (1.610-3.622)	< 0.001
Cancer Site							
Colon	731						
Rectum	408	0.803	2.233 (1.588-3.141)	< 0.001	0.696	2.005 (1.418-2.836)	< 0.001

Surgical Approach					
Laparoscopic	649				
Open	413	0.013	1.013 (0.706-1.454)	0.943	
Converted	77	0.272	1.312 (0.694-2.482)	0.404	
Age (years)	1139	-0.003	0.997 (0.983-1.011)	0.637	
ASA Class (N, %)					
ASA 1	118				
ASA 2	796	-0.102	0.903 (0.519-1.570)	0.717	
ASA 3+4	221	0.042	1.043 (0.554-1.964)	0.897	
AJCC Stage					
Ι	283				
II	343	-0.126	0.881 (0.568-1.368)	0.574	
III	398	-0.345	0.708 (0.456-1.100)	0.125	
IV	109	-0.521	0.594 (0.295-1.1.95)	0.144	
Diabetes					
no	1028				
yes	105	-0.008	0.992 (0.550-1.791)	0.979	

ASA, American Society of Anesthesiologists; CI, confidence interval; IRR, incident rate ratio; LOS, length of stay; N, number; NS, non-significant OR, odds ratio

*Canadian cohort used American Joint Committee on Cancer (AJCC) version 6, UK cohort used Union for International Cancer Control's (UICC) version 5 +Model = binary logistic regression, dependent variable = major complications (Grade \geq 3)